Table of Contents

Original Sheet No. 0	13
First Revised Sheet No. 1	14
Original Sheet No. 1A	15
Original Sheet No. 1B	16
First Revised Sheet No. 2	17
First Revised Sheet No. 3	18
Second Revised Sheet No. 4	19
Sheet Nos. 5 - 9	20
First Revised Sheet No. 10	21
Sheet Nos. 11 - 14	22
Second Revised Sheet No. 15	23
Sheet Nos. 16 - 19	24
Third Revised Sheet No. 20	25
First Revised Sheet No. 21	26
Tenth Revised Sheet No. 22	27
Ninth Revised Sheet No. 22	28
Second Revised Sheet No. 23	29
Eighth Revised Sheet No. 24	30
Tenth Revised Sheet No. 25	31
Ninth Revised Sheet No. 25	32
Second Revised Sheet No. 26	33
Ninth Revised Sheet No. 27	34
Eighth Revised Sheet No. 27	35
Ninth Revised Sheet No. 28	36
Eighth Revised Sheet No. 28	37
Ninth Revised Sheet No. 29	38
Eighth Revised Sheet No. 29	39
First Revised Sheet No. 30	40
Second Revised Sheet No. 30	41
Ninth Revised Sheet No. 31	42
Eighth Revised Sheet No. 31	43
First Revised Sheet No. 32	44
Second Revised Sheet No. 32	45
Ninth Revised Sheet No. 33	46
Eighth Revised Sheet No. 33	47
Eighth Revised Sheet No. 34	48
Eighth Revised Sheet No. 35	49
Seventh Revised Sheet No. 36	50
Seventh Revised Sheet No. 37	51
Fifth Revised Sheet No. 37A	52
Third Revised Sheet No. 38	53
First Revised Sheet No. 38A	54
Ninth Revised Sheet No. 39	55
Eighth Revised Sheet No. 39	56
Fourth Revised Sheet No. 40	57

Original Sheet No. 41	58
Eleventh Revised Sheet No. 42	59
Second Revised Sheet No. 42A	60
Sheet Nos. 43 - 49	61
First Revised Sheet No. 50	62
First Revised Sheet No. 51	63
First Revised Sheet No. 52	64
Second Revised Sheet No. 53	65
First Revised Sheet No. 54	66
First Revised Sheet No. 55	67
First Revised Sheet No. 56	68
First Revised Sheet No. 57	69
First Revised Sheet No. 58	70
Second Revised Sheet No. 59	71
First Revised Sheet No. 60	72
First Revised Sheet No. 60A	73
First Revised Sheet No. 60B	74
First Revised Sheet No. 61	75
First Revised Sheet No. 61A	76
First Revised Sheet No. 61B	77
First Revised Sheet No. 62	78
First Revised Sheet No. 62A	79
First Revised Sheet No. 62B	80
First Revised Sheet No. 63	81
First Revised Sheet No. 63A	82
First Revised Sheet No. 63B	83
First Revised Sheet No. 64	84
First Revised Sheet No. 64A	85
First Revised Sheet No. 64B	86
First Revised Sheet No. 64C	87
First Revised Sheet No. 65	88
First Revised Sheet No. 65A	89
First Revised Sheet No. 65B	90
First Revised Sheet No. 66	91
First Revised Sheet No. 66A	92
First Revised Sheet No. 66B	93
First Revised Sheet No. 67	94
First Revised Sheet No. 67A	95
First Revised Sheet No. 67B	96
First Revised Sheet No. 68	97
First Revised Sheet No. 68A	98
First Revised Sheet No. 68B	99
First Revised Sheet No. 68C	100
First Revised Sheet No. 69	101
First Revised Sheet No. 69A	102
First Revised Sheet No. 69B	103

First Revised Sheet No. 70	104
First Revised Sheet No. 70A	105
First Revised Sheet No. 70B	106
First Revised Sheet No. 71	107
First Revised Sheet No. 71A	108
First Revised Sheet No. 71B	109
First Revised Sheet No. 72	110
First Revised Sheet No. 72A	111
First Revised Sheet No. 72B	112
First Revised Sheet No. 73	113
First Revised Sheet No. 73A	114
First Revised Sheet No. 73B	115
First Revised Sheet No. 74	116
First Revised Sheet No. 74A	117
First Revised Sheet No. 74B	118
First Revised Sheet No. 75	119
First Revised Sheet No. 75A	120
First Revised Sheet No. 75B	121
First Revised Sheet No. 76	122
First Revised Sheet No. 76A	123
First Revised Sheet No. 76B	124
First Revised Sheet No. 77	125
First Revised Sheet No. 77A	126
First Revised Sheet No. 77B	127
First Revised Sheet No. 78	128
First Revised Sheet No. 78A	129
First Revised Sheet No. 78B	130
First Revised Sheet No. 79	131
First Revised Sheet No. 79A	132
First Revised Sheet No. 79B	133
First Revised Sheet No. 80	134
First Revised Sheet No. 80A	135
First Revised Sheet No. 80B	136
First Revised Sheet No. 81	137
First Revised Sheet No. 81A	138
First Revised Sheet No. 81B	139
First Revised Sheet No. 82	140
First Revised Sheet No. 82A	141
First Revised Sheet No. 82B	142
First Revised Sheet No. 83	143
First Revised Sheet No. 83A	144
Second Revised Sheet No. 84	145
Second Revised Sheet No. 85	146
Second Revised Sheet No. 86	147
Fourth Revised Sheet No. 87	148
First Revised Sheet No. 88	149

Original Shoot No. 99A	150
Original Sheet No. 88A	150
Second Revised Sheet No. 89	151
First Revised Sheet No. 89A	152
Third Revised Sheet No. 90	153
Original Sheet No. 90A	154
Original Sheet No. 90B	155
First Revised Sheet No. 91	156
First Revised Sheet No. 92	157
First Revised Sheet No. 93	158
First Revised Sheet No. 94	159
First Revised Sheet No. 95	160
First Revised Sheet No. 96	161
First Revised Sheet No. 97	162
First Revised Sheet No. 98	163
First Revised Sheet No. 99	164
First Revised Sheet No. 100	165
First Revised Sheet No. 101	166
First Revised Sheet No. 102	167
First Revised Sheet No. 103	168
First Revised Sheet No. 104	169
First Revised Sheet No. 105	170
First Revised Sheet No. 106	171
First Revised Sheet No. 107	172
First Revised Sheet No. 108	172
First Revised Sheet No. 109	173
First Revised Sheet No. 110	175
First Revised Sheet No. 111	176
First Revised Sheet No. 112	177
Sheet Nos. 113 - 199	178
First Revised Sheet No. 200	179
Fifth Revised Sheet No. 201	180
Fourth Revised Sheet No. 201	181
First Revised Sheet No. 202	182
Fourth Revised Sheet No. 203	183
Fourth Revised Sheet No. 204	184
Second Revised Sheet No. 205	185
First Revised Sheet No. 206	186
Original Sheet No. 207	187
Sixth Revised Sheet No. 208	188
Fourth Revised Sheet No. 208	189
Sixth Revised Sheet No. 209	190
Fourth Revised Sheet No. 209	191
Sheet Nos. 210 - 220	192
Original Sheet No. 221	193
Sixth Revised Sheet No. 222	194
Fourth Revised Sheet No. 222	195
Today Novious Office 140. 222	100

First Revised Sheet No. 223	196
Fourth Revised Sheet No. 224	197
Fourth Revised Sheet No. 225	198
Second Revised Sheet No. 226	199
First Revised Sheet No. 227	200
First Revised Sheet No. 228	201
Original Sheet No. 229	202
Original Sheet No. 230	203
First Revised Sheet No. 231	204
Sixth Revised Sheet No. 232	205
Fourth Revised Sheet No. 232	206
Sixth Revised Sheet No. 233	207
Fourth Revised Sheet No. 233	208
Original Sheet No. 234	209
Sheet Nos. 235 - 240	210
Sixth Revised Sheet No. 241	211
Fourth Revised Sheet No. 241	212
First Revised Sheet No. 242	213
Second Revised Sheet No. 243	214
Fourth Revised Sheet No. 244	215
Fourth Revised Sheet No. 245	216
Second Revised Sheet No. 246	217
Original Sheet No. 247	218
First Revised Sheet No. 248	219
Sixth Revised Sheet No. 249	220
Fourth Revised Sheet No. 249	221
Fifth Revised Sheet No. 250	222
Fourth Revised Sheet No. 250	223
Sheet Nos. 251 - 260	224
First Revised Sheet No. 261	225
Fifth Revised Sheet No. 262	226
Fourth Revised Sheet No. 262	227
First Revised Sheet No. 263	228
First Revised Sheet No. 264	229
Fourth Revised Sheet No. 265	230
Fourth Revised Sheet No. 266	231
Second Revised Sheet No. 267	232
First Revised Sheet No. 268	233
Original Sheet No. 269	234
First Revised Sheet No. 270	235
First Revised Sheet No. 271	236
Sixth Revised Sheet No. 272	237
Fourth Revised Sheet No. 272	238
Sixth Revised Sheet No. 273	239
Fourth Revised Sheet No. 273	240
Original Sheet No. 274	241

Sheet Nos. 275 - 280	242
Fifth Revised Sheet No. 281	243
Fourth Revised Sheet No. 281	244
First Revised Sheet No. 282	245
Fifth Revised Sheet No. 283	246
Fourth Revised Sheet No. 284	247
First Revised Sheet No. 285	248
Fifth Revised Sheet No. 286	249
Fourth Revised Sheet No. 286	250
Fifth Revised Sheet No. 287	250 251
Fourth Revised Sheet No. 287	252
Sheet Nos. 288 - 300	252
Original Sheet No. 301	253 254
First Revised Sheet No. 302	25 4 255
Third Revised Sheet No. 302 Third Revised Sheet No. 303	256
Second Revised Sheet No. 304	257
Original Sheet No. 305	258
Original Sheet No. 306	259 260
Original Sheet No. 307 Sheet Nos. 308 - 320	
First Revised Sheet No. 321	261 262
First Revised Sheet No. 321	263
Second Revised Sheet No. 323	264
Second Revised Sheet No. 323 Second Revised Sheet No. 324	265
Original Sheet No. 325	266
Original Sheet No. 326	267
Original Sheet No. 327	268
Original Sheet No. 328	269
Sheet Nos. 329 - 330	270
Fifth Revised Sheet No. 331	271
Third Revised Sheet No. 332	272
First Revised Sheet No. 333	273
Fourth Revised Sheet No. 334	274
Second Revised Sheet No. 335	275
Fourth Revised Sheet No. 336	276
Original Sheet No. 337	277
Fifth Revised Sheet No. 338	278
First Revised Sheet No. 338A	279
First Revised Sheet No. 339	280
Fourth Revised Sheet No. 340	281
Sheet Nos. 341 - 350	282
First Revised Sheet No. 351	283
Second Revised Sheet No. 352	284
Original Sheet No. 353	285
First Revised Sheet No. 354	286
First Revised Sheet No. 355	287

First Davised Cheet No. 250	200
First Revised Sheet No. 356	288
Sheet Nos. 357 - 360	289
First Revised Sheet No. 361	290
Second Revised Sheet No. 362	291
First Revised Sheet No. 363	292
Original Sheet No. 364	293
Sheet Nos. 365 - 370	294
Sixth Revised Sheet No. 371	295
Fourth Revised Sheet No. 372	296
Sixth Revised Sheet No. 373	297
Fifth Revised Sheet No. 374	298
Sheet Nos. 375 - 499	299
Fourth Revised Sheet No. 500	300
Second Revised Sheet No. 501	301
Third Revised Sheet No. 502	302
Second Revised Sheet No. 503	303
Third Revised Sheet No. 504	304
Second Revised Sheet No. 505	305
First Revised Sheet No. 506	306
Sixth Revised Sheet No. 507	307
Second Revised Sheet No. 508	308
Third Revised Sheet No. 509	309
Third Revised Sheet No. 510	310
Second Revised Sheet No. 511	311
Third Revised Sheet No. 512	312
Fourth Revised Sheet No. 513	313
Original Sheet No. 513A	314
First Revised Sheet No. 513B	315
Third Revised Sheet No. 514	316
Third Revised Sheet No. 515	317
First Revised Sheet No. 516	318
Second Revised Sheet No. 517	319
Second Revised Sheet No. 517 Second Revised Sheet No. 518	320
Second Revised Sheet No. 518A	321
Third Revised Sheet No. 519	322
Third Revised Sheet No. 519 Third Revised Sheet No. 520	
	323
Second Revised Sheet No. 521	324
First Revised Sheet No. 522	325
First Revised Sheet No. 523	326
Original Sheet No. 524	327
Original Sheet No. 525	328
Original Sheet No. 526	329
Original Sheet No. 527	330
Second Revised Sheet No. 528	331
Third Revised Sheet No. 529	332
Third Revised Sheet No. 530	333

Original Chart No. 524	004
Original Sheet No. 531	334
Second Revised Sheet No. 532	335
Fifth Revised Sheet No. 533	336
Second Revised Sheet No. 534	337
Substitute Original Sheet No. 534A	338
Second Revised Sheet No. 535	339
First Revised Sheet No. 536	340
Fourth Revised Sheet No. 537	341
First Revised Sheet No. 538	342
Second Revised Sheet No. 539	343
First Revised Sheet No. 540	344
Second Revised Sheet No. 541	345
Second Revised Sheet No. 542	346
Second Revised Sheet No. 543	347
Fourth Revised Sheet No. 544	348
First Revised Sheet No. 545	349
Second Revised Sheet No. 546	350
Second Revised Sheet No. 547	351
Third Revised Sheet No. 548	352
Original Sheet No. 548A	353
Third Revised Sheet No. 549	354
Second Sub Original Sheet No. 549A	355
Fifth Revised Sheet No. 550	356
First Revised Sheet No. 550A	357
Third Revised Sheet No. 551	358
First Revised Sheet No. 552	359
Fourth Revised Sheet No. 553	360
First Revised Sheet No. 554	361
First Revised Sheet No. 555	362
First Revised Sheet No. 556	363
Original Sheet No. 557	364
Second Revised Sheet No. 558	365
First Revised Sheet No. 559	366
Second Revised Sheet No. 560	367
Second Revised Sheet No. 561	368
First Revised Sheet No. 562	369
Third Revised Sheet No. 563	370
First Revised Sheet No. 564	371
First Revised Sheet No. 565	372
	373
Original Sheet No. 566	
First Revised Sheet No. 567	374
Third Revised Sheet No. 568	375
Second Revised Sheet No. 569	376
Second Revised Sheet No. 570	377
Second Revised Sheet No. 571	378
Second Revised Sheet No. 572	379

0 15 101 111 570	000
Second Revised Sheet No. 573	380
Second Revised Sheet No. 574	381
Third Revised Sheet No. 575	382
Second Revised Sheet No. 576	383
Second Revised Sheet No. 577	384
Fourth Revised Sheet No. 578	385
Second Revised Sheet No. 579	386
Third Revised Sheet No. 580	387
Second Revised Sheet No. 581	388
First Revised Sheet No. 582	389
First Revised Sheet No. 583	390
Second Revised Sheet No. 584	391
First Revised Sheet No. 585	392
First Revised Sheet No. 586	393
First Revised Sheet No. 587	394
Second Revised Sheet No. 588	395
First Revised Sheet No. 589	396
First Revised Sheet No. 590	397
Second Revised Sheet No. 591	398
Second Revised Sheet No. 592	399
First Revised Sheet No. 593	400
First Revised Sheet No. 594	401
Second Revised Sheet No. 595	402
First Revised Sheet No. 596	403
First Revised Sheet No. 597	404
First Revised Sheet No. 598	405
First Revised Sheet No. 599	406
First Revised Sheet No. 600	407
First Revised Sheet No. 601	408
Second Revised Sheet No. 602	409
First Revised Sheet No. 603	410
First Revised Sheet No. 604	411
Fourth Revised Sheet No. 605	412
Third Revised Sheet No. 606	
	413
Sixth Revised Sheet No. 607	414
Third Revised Sheet No. 608	415
Fourth Revised Sheet No. 608A	416
Fifth Revised Sheet No. 608B	417
Second Revised Sheet No. 608C	418
First Revised Sheet No. 608D	419
Second Revised Sheet No. 608E	420
Original Sheet No. 608A.01	421
Ninth Revised Sheet No. 609	422
Second Revised Sheet No. 610	423
First Revised Sheet No. 611	424
First Revised Sheet No. 612	425

First Revised Sheet No. 613	426
Fourth Revised Sheet No. 614	427
Original Sheet No. 614A	428
Fourth Revised Sheet No. 615	429
Third Revised Sheet No. 615A	430
Fifth Revised Sheet No. 616	431
Original Sheet No. 616A	432
Third Revised Sheet No. 617	433
Second Revised Sheet No. 618	434
First Revised Sheet No. 619	435
Second Revised Sheet No. 620	436
First Revised Sheet No. 621	437
Second Revised Sheet No. 622	438
Third Revised Sheet No. 623	439
First Revised Sheet No. 624	440
Third Revised Sheet No. 625	441
Sheet Nos. 626 - 699	442
First Revised Sheet No. 700	443
Third Revised Sheet No. 701	444
Third Revised Sheet No. 702	445
Fifth Revised Sheet No. 703	446
First Revised Sheet No. 704	447
First Revised Sheet No. 705	448
First Revised Sheet No. 706	449
First Revised Sheet No. 707	450
First Revised Sheet No. 708	451
First Revised Sheet No. 709	452
First Revised Sheet No. 710	453
Sheet Nos. 711 - 720	454
Third Revised Sheet No. 721	455
Third Revised Sheet No. 722	456
Fifth Revised Sheet No. 723	457
First Revised Sheet No. 724	458
First Revised Sheet No. 725	459
First Revised Sheet No. 726	460
First Revised Sheet No. 727	461
First Revised Sheet No. 728	462
First Revised Sheet No. 729	463
First Revised Sheet No. 730	464
Sheet Nos. 731 - 740	465
Third Revised Sheet No. 741	466
Third Revised Sheet No. 742	467
Fifth Revised Sheet No. 743	468
First Revised Sheet No. 744	469
First Revised Sheet No. 745	470
First Revised Sheet No. 746	471

F: (B : 10) (M = 45	470
First Revised Sheet No. 747	472
First Revised Sheet No. 748	473
First Revised Sheet No. 749	474
First Revised Sheet No. 750	475
Sheet Nos. 751 - 760	476
Third Revised Sheet No. 761	477
Third Revised Sheet No. 762	478
Fifth Revised Sheet No. 763	479
First Revised Sheet No. 764	480
First Revised Sheet No. 765	481
First Revised Sheet No. 766	482
First Revised Sheet No. 767	483
First Revised Sheet No. 768	484
First Revised Sheet No. 769	485
First Revised Sheet No. 770	486
Sheet Nos. 771 - 780	487
Second Revised Sheet No. 781	488
Second Revised Sheet No. 782	489
Second Revised Sheet No. 783	490
First Revised Sheet No. 784	491
First Revised Sheet No. 785	492
First Revised Sheet No. 786	493
Sheet Nos. 787 - 790	494
First Revised Sheet No. 791	495
Original Sheet No. 792	496
Original Sheet No. 793	497
Original Sheet No. 794	498
•	499
Original Sheet No. 795	
Original Sheet No. 796	500
Original Sheet No. 797	501
Original Sheet No. 798	502
Sheet Nos. 799 - 810	503
First Revised Sheet No. 811	504
Fourth Revised Sheet No. 812	505
First Revised Sheet No. 813	506
First Revised Sheet No. 814	507
First Revised Sheet No. 815	508
First Revised Sheet No. 816	509
Original Sheet No. 817	510
Original Sheet No. 818	511
Original Sheet No. 819	512
Sheet No. 820	513
First Revised Sheet No. 821	514
First Revised Sheet No. 822	515
First Revised Sheet No. 823	516
First Revised Sheet No. 824	517

First Revised Sheet No. 825	518
First Revised Sheet No. 826	519
First Revised Sheet No. 827	520
Sheet Nos. 828 - 830	521
First Revised Sheet No. 831	522
First Revised Sheet No. 832	523
First Revised Sheet No. 833	524
Original Sheet No. 834	525
Original Sheet No. 835	526 526
Sheet Nos. 836 - 840	527
Fourth Revised Sheet No. 841	528
Third Revised Sheet No. 842	529
Fifth Revised Sheet No. 843	530
First Revised Sheet No. 844	531
First Revised Sheet No. 845	532
First Revised Sheet No. 846	533
Sheet Nos. 847 - 850	534
Third Revised Sheet No. 851	535
Seventh Revised Sheet No. 852	536
Fifth Revised Sheet No. 853	537
Third Revised Sheet No. 854	538
First Revised Sheet No. 855	539
First Revised Sheet No. 856	540
First Revised Sheet No. 857	541
First Revised Sheet No. 858	542
First Revised Sheet No. 859	543
Sheet No. 860	544
Third Revised Sheet No. 861	545
Second Revised Sheet No. 862	546
First Revised Sheet No. 863	547
Fourth Revised Sheet No. 864	548
Sheet Nos. 865 - 870	549
Second Revised Sheet No. 871	550
Second Revised Sheet No. 872	551
First Revised Sheet No. 873	552
First Revised Sheet No. 874	553
First Revised Sheet No. 875	554
First Revised Sheet No. 876	555
Sheet Nos. 877 - 890	556
Eighth Revised Sheet No. 891	557
Fourth Revised Sheet No. 892	558
Second Revised Sheet No. 893	559
Sheet Nos. 894 - 1000	560
	550

FERC Docket: RP04-361-000

Original Sheet No. 0 Algonquin Gas Transmission, LLC: Fifth Revised Volume No. 1

Original Sheet No. 0 : Effective

FERC GAS TARIFF

FIFTH REVISED VOLUME NO. 1

(Supersedes Algonquin Gas Transmission Company Fourth Revised Volume No. 1)

ALGONQUIN GAS TRANSMISSION, LLC

FILED WITH

FEDERAL ENERGY REGULATORY COMMISSION

Communications Concerning This Tariff Should Be Directed To:

Richard J. Kruse, Senior Vice President Industry Initiatives, Pricing and Regulatory Affairs Algonquin Gas Transmission, LLC 5400 Westheimer Court Houston, Texas 77056-5310 Telephone: (713) 627-5368 Facsimile: (713) 627-4027

FERC Docket: RP10-308-000

First Revised Sheet No. 1 First Revised Sheet No. 1 Superseding: Original Sheet No. 1

TABLE OF CONTENTS VOLUME NO. 1

			Sheet No.
Prelim	inary S	tatement	10
System	Map		15
Statem	ents of	Rates (Index)	20
Α.	AFT-1 AFT-1 AFT-E AFT-E AFT-1S AFT-ES AFT-ES AFT-ES AIT-1 T-1 AFT-4 AFT-CL PAL TTT	ortation Services Firm Transportation Service Capacity Release No-Notice Firm Transportation Service Capacity Release Small Customer Firm Transportation Service Capacity Release Small Customer No-Notice Firm Transportation Service Capacity Release Interruptible Transportation Service Transportation Service (New York and New Jersey) Firm Transportation Service Firm Transportation Service Capacity Release Park and Loan Service Title Transfer Tracking Service	22 25 27 28 29 25, 30 31 28, 33 34 35 36 38 39 41
В.		Interruptible Transportation Service	42 40
С.	Negotia	ated Rates	50
Rate S	chedules	s (Index)	200
	AFT-E AFT-1S AFT-ES	Firm Transportation Service No-Notice Firm Transportation Service Small Customer Firm Transportation Service Small Customer No-Notice Firm Transportation Service	201 221 241 261
	T-1 AFT-4 AFT-CL PAL	Firm Transportation Service Firm Transportation Service Park and Loan Service Title Transfer Tracking Service	281 301 321 331 351 361 371

FERC Docket: RP10-308-000

Original Sheet No. 1A Original Sheet No. 1A

TABLE OF CONTENTS VOLUME NO. 1 (continued)

Sheet No. 500 General Terms and Conditions (Index) Definitions 501 2. Request for Transportation Service 508 3. Credit Evaluation 517 4. Quality of Gas 519 5. 522 Measurement 6. Measuring Equipment 524 7. Receipt and Delivery Pressures 527 8. Term of Executed Service Agreements 9. Pregranted Abandonment and Right of First Refusal 10. Joint Obligations 531 11. Discretionary Waiver 531 12. Modification -531 13. Successors in Interest 531 Capacity Release 532 1.4. Termination for Default 15. 551 16. Force Majeure 552 17. 553 Governmental Regulation 18. Billing 553 19. Payment 554 20. Possession of Gas 557 21. Warranty of Title to Gas 557 22. Nominations 558 23. Scheduling of Service 562 566 24. Curtailment 25. Imbalance Resolution Procedures 572 26. 578 Actions Required To Protect System Operations 27. Determination of Receipts 580 Determination of Deliveries 28. 581 29. Operational Flow Orders 584 30. Operational Balancing Agreements 586 31. Unauthorized Contract Overrun Gas 32. Fuel Reimbursement Quantity 589 Gas Research Institute Charge Adjustment Provision 33. 593 34. Annual Charge Adjustment Clause Provision 595 35. [Reserved for Future Use] 596 36. Periodic Rate Adjustments 37. 602 [Reserved for Future Use] 38. Standards of Conduct Compliance Procedures 605 39. Complaint Procedures 606 40. Electronic Communication 607 41. Revenue Credits 609 42. Fees and Construction of New Facilities 612 43. Reports with Respect to Tariff 613 44. North American Energy Standards Board ("NAESB") 614 45. Discount Terms 46. Negotiated Rates 616 47. Off-system Pipeline Capacity 617 48. Segmentation, Secondary Point Rights and Related Scheduling 49. Materially Non-Conforming Service Agreements 625

FERC Docket: RP10-308-000

Original Sheet No. 1B Original Sheet No. 1B

TABLE OF CONTENTS VOLUME NO. 1 (continued)

Sheet No. 700 Forms of Service Agreements (Index) 702 AFT-1 Algonquin Firm Transportation Service AFT-E Algonquin No-Notice Firm Transportation Service 722 AFT-1S Small Customer Firm Transportation Service
AFT-ES Small Customer Firm No-Notice Transportation Service 742 762 AIT-1 Algonquin Interruptible Transportation Service 781 T-1 Transportation Service (New York and New Jersey) 791 AFT-4 Firm Transportation Service 811 821 Capacity Release Umbrella Agreement Operational Balancing Agreement 831 The LINK® System 841 AFT-CL Firm Transportation Service 852 PAL Park and Loan Service
TTT Title Transfer Tracking Service 861 871 AIT-2 Interruptible Transportation Service 891

Effective Date: 02/27/2007 Status: Effective FERC Docket: CP07- 23-001

First Revised Sheet No. 2 First Revised Sheet No. 2 : Effective Superseding: Original Sheet No. 2

TABLE OF CONTENTS VOLUME NO. 2

		Sheet No.
Rate Schedule X-1	Exchange of natural gas on emergency basis with Tennessee Gas Transmission Company	4
Rate Schedule X-2	Notice of Cancellation	8
Rate Schedule X-3	Notice of Cancellation	9
Rate Schedule X-4	Notice of Cancellation	10
Rate Schedule X-5	Notice of Cancellation	11
Rate Schedule X-6	Exchange Agreement with Tennessee Gas Transmission Company (Danbury)	15
Rate Schedule X-7	Notice of Cancellation	31
Rate Schedule X-8	Notice of Cancellation	32
Rate Schedule X-9	Exchange Agreement with Tennessee Gas Transmission Company (Thompsonville)	33
Rate Schedule X-10	Exchange Agreement with Texas Eastern Transmission Corporation (Bernards and Morris)	49
Rate Schedule X-11	Notice of Cancellation	53
Rate Schedule X-12	Exchange Agreement with Tennessee Gas Transmission Company (Milford)	54
Rate Schedule X-13	Notice of Cancellation	69
Rate Schedule X-14	Notice of Cancellation	70

Effective Date: 03/08/2007 Status: Effective FERC Docket: CP07- 21-001

First Revised Sheet No. 3 First Revised Sheet No. 3 : Effective Superseding: Original Sheet No. 3

TABLE OF CONTENTS VOLUME NO. 2 (continued)

		Sheet No.
Rate Schedule X-15	Notice of Cancellation	71
Rate Schedule X-16	Notice of Cancellation	72
Rate Schedule X-17	Notice of Cancellation	73
Rate Schedule X-18	Notice of Cancellation	74
Rate Schedule X-19	Notice of Cancellation	75
Rate Schedule X-20	Notice of Cancellation	75
Rate Schedule X-21	Notice of Cancellation	75
Rate Schedule X-22	Notice of Cancellation	75
Rate Schedule X-23	Notice of Cancellation	76
Rate Schedule X-24	Notice of Cancellation	77
Rate Schedule X-25	Notice of Cancellation	78
Rate Schedule X-26	Notice of Cancellation	79
Rate Schedule X-27	Notice of Cancellation	80

Effective Date: 03/08/2007 Status: Effective FERC Docket: CP07- 21-001

Second Revised Sheet No. 4 second Revised Sheet No. 4 : Effective Superseding: First Revised Sheet No. 4

TABLE OF CONTENTS VOLUME NO. 2 (continued)

				Sheet No.
Rate	Schedule	X-28	Notice of Cancellation	82
Rate	Schedule	X-29	Notice of Cancellation	95
Rate	Schedule	X-30	Notice of Cancellation	96
Rate	Schedule	X-31	Notice of Cancellation	97
Rate	Schedule	X-32	Notice of Cancellation	98
Rate	Schedule	X-33	Transportation Agreement between Algonquin Gas and Commonwealth Gas Company	99
Rate	Schedule	X-34	Notice of Cancellation	118
Rate	Schedule	X-35	Notice of Cancellation	119
Rate	Schedule	X-37	Notice of Cancellation	120
Rate	Schedule	X-38	Notice of Cancellation	121
Rate	Schedule	X-39	Notice of Cancellation	122

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000

Sheet Nos. 5 - 9 Sheet Nos. 5 - 9 : Effective

Sheet Nos. 5 - 9 are being reserved for future use

FERC Docket: RP10-77-000

First Revised Sheet No. 10 First Revised Sheet No. 10

Superseding: Original Sheet No. 10

PRELIMINARY STATEMENT

Algonquin Gas Transmission, LLC (referred to in this tariff as "Algonquin") is a corporation engaged in the business of transporting natural gas in interstate commerce under authorization granted by, and subject to the jurisdiction of, the Federal Energy Regulatory Commission, and its predecessor agency, the Federal Power Commission ("Commission" or "FERC"). Algonquin owns and operates a pipeline system in the states of New Jersey, New York, Connecticut, Rhode Island, and Massachusetts.

It is the policy of Algonquin to provide transportation service only under written contract acceptable to Algonquin after consideration of its commitments to others, delivery capacity and other factors deemed pertinent by Algonquin. If any such contract is to become operative only upon performance of certain precedent conditions, it is the policy of Algonquin to execute a separate written agreement covering the conditions to be satisfied before the contract for the service becomes operative.

This FERC Gas Tariff is filed in compliance with Part 154 of the Commission's Regulations under the Natural Gas Act. Nothing in this tariff is intended to inhibit the formation or use of market centers on Algonquin's system, or to inhibit the development of, or discriminate against the use of, Title Transfer Tracking or Imbalance Management Services provided by third parties or Algonquin's Customers. Any party interested in providing Title Transfer Tracking or Imbalance Management Services must coordinate with Algonquin.

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000

Sheet Nos. 11 - 14 Sheet Nos. 11 - 14 : Effective

Sheet Nos. 11 - 14 are being reserved for future use

FERC Docket: RP08-334-000

Second Revised Sheet No. 15 Second Revised Sheet No. 15: Pending Superseding: First Revised Sheet No. 15

MAPS

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000

Sheet Nos. 16 - 19 Sheet Nos. 16 - 19 : Effective

Sheet Nos. 16 - 19 are being reserved for future use

FERC Docket: RP10-308-000

Third Revised Sheet No. 20 Third Revised Sheet No. 20 Superseding: Second Revised Sheet No. 20 $\,$

STATEMENTS OF RATES SHOWING EFFECTIVE RATES INDEX

			Sheet No.
A.	Transport	ation Services	Sheet No.
	AFT-1	Firm Transportation Service	22
	AFT-1	Capacity Release	25
	AFT-E	No-Notice Firm Transportation Service	27
	AFT-E	Capacity Release	28
	AFT-1S	Small Customer Firm Transportation Service	29
	AFT-1S	Capacity Release	25, 30
	AFT-ES	Small Customer No-Notice Firm Transportation Service	31
	AFT-ES	Capacity Release	28, 32
	AIT-1	Interruptible Transportation Service	33
	T-1	Transportation Service (New York and New Jersey)	34
	AFT-4	Firm Transportation Service	35
	AFT-CL	Firm Transportation Service	36
	AFT-CL	Capacity Release	38
	PAL	Park and Loan Service	39
	TTT	Title Transfer Tracking Service	41
	AIT-2	Interruptible Transportation Service	42
В.	Fuel Reim	bursement Percentage	40
C.	Negotiate	d Rates	50

FERC Docket: RP05-208-000

First Revised Sheet No. 21 First Revised Sheet No. 21 : Effective Superseding: Original Sheet No. 21

Sheet No. 21 was previously issued, but is now reserved for future use.

FERC Docket: RP09-938-000

Tenth Revised Sheet No. 22 Tenth Revised Sheet No. 22 Superseding: Ninth Revised Sheet No. 22

Rate Schedule AFT-1 Firm Transportation Service

	Tim Transportation Betvice
Maximum Reservation Charge: Conversion From: (F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3) (FTP) (PSS-T) (AFT-2) (AFT-3) (AFT-5) (ITP) (X-35) (X-39)	Base\$/Dth
Minimum Reservation Charge:	\$0.0000 \$0.0000
Maximum Commodity Charge: (F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3) (FTP) (PSS-T) (AFT-2) (AFT-3) (AFT-5) (ITP) (X-35) (X-39) Minimum Commodity Charge: AFT-1 (F-1/WS-1) AFT-1 (F-2/F-3) AFT-1 (F-2/F-3) AFT-1 (STB/SS-3) AFT-1 (All other)	Base
Maximum Authorized Overrun Commodity Charge: (F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3) (FTP) (PSS-T) (AFT-2) (AFT-3) (AFT-5) (ITP) (X-35) (X-39) Minimum Authorized Overrun Commodity Charge: AFT-1 (F-1/WS-1) AFT-1 (F-2/F-3) AFT-1 (F-4) AFT-1 (STB/SS-3) AFT-1 (All other)	Base

Effective Date: 12/31/9999 Status: Accepted FERC Docket: RP09-864-000

Ninth Revised Sheet No. 22 Ninth Revised Sheet No. 22 Superseding: Eighth Revised Sheet No. 22

Rate Schedule AFT-1 Firm Transportation Service

	TITM TIAMOPOTEACTON DOIVICE
Maximum Reservation Charge: Conversion From: (F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3) (FTP) (PSS-T) (AFT-2) (AFT-3) (AFT-5) (ITP) (X-35) (X-39)	Base\$/Dth Tariff Total Rate 1/ Rate \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$6.5734 \$11.8368 \$11.8368 \$9.7854 \$9.7854 \$6.1138 \$6.1138 \$10.7554 \$10.7554 \$12.6265 \$12.6265 \$13.0110 \$13.0110 \$10.2027 \$10.2027 \$13.2089 \$13.2089
Minimum Reservation Charge:	\$0.0000 \$0.0000
Maximum Commodity Charge: (F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3) (FTP) (PSS-T) (AFT-2) (AFT-3) (AFT-5) (ITP) (X-35) (X-39) Minimum Commodity Charge: AFT-1 (F-1/WS-1) AFT-1 (F-2/F-3) AFT-1 (F-4) AFT-1 (STB/SS-3) AFT-1 (All other)	Base
Maximum Authorized Overrun Commodity Charge: (F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3) (FTP) (PSS-T) (AFT-2) (AFT-3) (AFT-5) (ITP) (X-35) (X-39) Minimum Authorized Overrun Commodity Charge: AFT-1 (F-1/WS-1) AFT-1 (F-2/F-3) AFT-1 (F-4) AFT-1 (STB/SS-3) AFT-1 (All other)	Base

FERC Docket: RP06-519-000

Second Revised Sheet No. 23 Second Revised Sheet No. 23 : Effective Superseding: First Revised Sheet No. 23

Rate Schedule AFT-1 Firm Transportation Service Continued

The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission. 1/

FERC Docket: RP09-938-000

Eighth Revised Sheet No. 24 Eighth Revised Sheet No. 24

Superseding: Seventh Revised Sheet No. 24

Rate Schedule AFT-1 Incremental Capacity Charges Firm Transportation Service

CHARGES \$/dth

APPLICABLE TO CUSTOMERS UTILIZING CAPACITY PURSUANT TO INCREMENTAL FACILITY EXPANSIONS AND IN ADDITION TO THE AFT-1 SYSTEM RATE: Maximum Minimum Docket No. CP99-46 (EMI-Tiverton Project): dth Customer 46,000 Calpine Energy Services LP 0.0000 RESERVATION CHARGE 1.6424 VOLUMETRIC RESERVATION CHARGE 0.0540 0.0000 2. Docket No. CP01-5 (HubLine Mainline): HUBLINE CHARGE APPLICABLE TO CUSTOMERS CONTRACTED FOR TRANSPORTATION PATH AND/OR UTILIZING PRIMARY RECEIPT AND/OR DELIVERY POINTS FROM BEVERLY TO WEYMOUTH, AND IN ADDITION TO THE AFT-1 SYSTEM RATE: Maximum Minimum RESERVATION CHARGE 1.8607 0.0000 VOLUMETRIC RESERVATION CHARGE 0.0612 0.0000 USAGE 2/ CHARGES \$/dth HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING SECONDARY RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING SECONDARY DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES: Maximum Minimum COMMODITY CHARGE 3/ 0.0612 0.0000 3. Docket No. CP06-76 (Ramapo Project): APPLICABLE TO CUSTOMERS CONTRACTED FOR TRANSPORTATION SERVICE CHARGES ON FACILITIES CONSTRUCTED UNDER THE RAMAPO PROJECT IN ADDITION \$/dth TO THE AFT-1 SYSTEM RATE: Maximum Minimum RESERVATION CHARGE 7.5608 0.0000 VOLUMETRIC RESERVATION CHARGE 0.2486 0.0000 AUTHORIZED OVERRUN CHARGE 0.2486 0.0000

- 1/ The Reservation Charge is the effective rate on file with the Commission excluding adjustments approved by the Commission.
- 2/ Excludes ACA of \$0.0019.
- 3/ HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

FERC Docket: RP09-938-000

Tenth Revised Sheet No. 25 Tenth Revised Sheet No. 25 Superseding: Ninth Revised Sheet No. 25

Rate Schedule AFT-1 Firm Transportation Service Capacity Release

Maximum	Base \$/Dth	Volumetri	Lc 2/
Reservation	Tariff	Base	Total
Charge:	Rate 1/	Rate	Rate
	(a)	(b)	(c) = (b)
(F-1/WS-1)	\$6.5734	\$0.2161	\$0.2161
(F-2/F-3)	\$6.5734	\$0.2161	\$0.2161
(F-4)	\$6.5734	\$0.2161	\$0.2161
(STB/SS-3)	\$6.5734	\$0.2161	\$0.2161
(FTP)	\$11.8368	\$0.3892	\$0.3892
(PSS-T)	\$9.7854	\$0.3217	\$0.3217
(AFT-2)	\$6.1138	\$0.2010	\$0.2010
(AFT-3)	\$10.7554	\$0.3536	\$0.3536
(AFT-5)	\$12.6265	\$0.4151	\$0.4151
(ITP)	\$13.0110	\$0.4278	\$0.4278
(X-35)	\$10.2027	\$0.3354	\$0.3354
(X-39)	\$13.2089	\$0.4343	\$0.4343
361			
Minimum	* 0.0000	\$0.0000	¢0.0000
Reservation Charge:	\$0.0000	\$0.0000	\$0.0000
	Rate Schedule AFT-1S Ca	pacity Release	
Maximum	Base \$/Dth	Volumetri	Lc 3/
Reservation	Tariff	Base	Total
Charge:	Rate 1/	Rate	Rate
-	(a)	(b)	(c) = (b)
(F-1/WS-1)	\$2.6294	\$0.0864	\$0.0864
(F-2/F-3)	\$2.6294	\$0.0864	\$0.0864
(F-4)	\$2.6294	\$0.0864	\$0.0864
(STB/SS-3)	\$2.6294	\$0.0864	\$0.0864
Art a toron			
Minimum	\$0.0000	\$0.0000	\$0.0000
Reservation Charge	20.0000	ŞU.UUUU	30.0000

Commodity Charge:	Tariff	ACA	Total
	Rate 1/	Adj.	Rate
(F-1/WS-1) Maximum $(F-2/F-3)$ Maximum $(F-4)$ Maximum $(STB/SS-3)$ Maximum	\$0.2273	\$0.0019	\$0.2292
	\$0.2273	\$0.0019	\$0.2292
	\$0.2273	\$0.0019	\$0.2292
	\$0.2273	\$0.0019	\$0.2292
(F-1/WS-1) Minimum	\$0.0112	\$0.0019	\$0.0131
(F-2/F-3) Minimum	\$0.0112	\$0.0019	\$0.0131
(F-4) Minimum	\$0.0112	\$0.0019	\$0.0131
(STB/SS-3) Minimum	\$0.0112	\$0.0019	\$0.0131

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Ninth Revised Sheet No. 25 Ninth Revised Sheet No. 25 Superseding: Eighth Revised Sheet No. 25 $\,$

Rate Schedule AFT-1 Firm Transportation Service Capacity Release

Maximum	Base \$/Dth	Volumetric 2/
Reservation	Tariff	Base Total
Charge:	Rate 1/	Rate Rate
-	(a)	(b) $(c) = (b)$
(F-1/WS-1)	\$6.5734	\$0.2161 \$0.2161
(F-2/F-3)	\$6.5734	\$0.2161 \$0.2161
(F-4)	\$6.5734	\$0.2161 \$0.2161
(STB/SS-3)	\$6.5734	\$0.2161 \$0.2161
(FTP)	\$11.8368	\$0.3892 \$0.3892
(PSS-T)	\$9.7854	\$0.3217 \$0.3217
(AFT-2)	\$6.1138	\$0.2010 \$0.2010
(AFT-3)	\$10.7554	\$0.3536 \$0.3536
(AFT-5)	\$12.6265	\$0.4151 \$0.4151
(ITP)	\$13.0110	\$0.4278 \$0.4278
(X-35)	\$10.2027	\$0.3354 \$0.3354
(X-39)	\$13.2089	\$0.4343 \$0.4343
261		
Minimum	*0.0000	\$0.0000
Reservation Charge:	\$0.0000	\$0.0000 \$0.0000
	Rate Schedule AFT-1S Ca	nacity Release
	nace concaute in 1 to ca	pacity Nerodoc
Maximum	Base \$/Dth	Volumetric 3/
Reservation	Tariff	Base Total
Charge:	Rate 1/	Rate Rate
	(a)	(b) $(c) = (b)$
(F-1/WS-1)	\$2.6294	\$0.0864 \$0.0864
(F-2/F-3)	\$2.6294	\$0.0864 \$0.0864
(F-4)	\$2.6294	\$0.0864 \$0.0864
(STB/SS-3)	\$2.6294	\$0.0864 \$0.0864

(F-4)	\$2.6294	\$0.0864	\$0.0864
(STB/SS-3)	\$2.6294	\$0.0864	\$0.0864
Minimum Reservation Charge	\$0.0000	\$0.0000	\$0.0000

Commodity Charge	:	Tar	e riff e 1/	\$/E ACA Adj.		Total Rate
(F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3)	Maximum Maximum	\$0 \$0	.2273	\$0.0017 \$0.0017 \$0.0017 \$0.0017	\$	0.2290 0.2290 0.2290 0.2290
(F-1/WS-1) (F-2/F-3) (F-4) (STB/SS-3)	Minimum Minimum	\$0 \$0	.0112	\$0.0017 \$0.0017 \$0.0017 \$0.0017	\$ \$	0.0129 0.0129 0.0129 0.0129

FERC Docket: RP06-519-000

Second Revised Sheet No. 26 Second Revised Sheet No. 26: Effective

Superseding: First Revised Sheet No. 26

Rate Schedule AFT-1
Firm Transportation Service
Capacity Release
Continued

- 1/ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.
- 2/ The volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a volumetric basis, and are exclusive of surcharges and commodity.
- 3/ Reservation charges and commodity charges applicable to capacity released under Rate Schedule AFT-1S and acquired under Rate Schedule AFT-1.

FERC Docket: RP09-938-000

Ninth Revised Sheet No. 27 Ninth Revised Sheet No. 27

Superseding: Eighth Revised Sheet No. 27

Rate Schedule AFT-E No-Notice Firm Transportation Service

	Base Tariff Rate 1/	\$/Dth ACA Adj.	Total Rate
Reservation Charge: Maximum	\$6.5734	-	\$6.5734
Minimum Commodity Charge:	\$0.0000	-	\$0.0000
Maximum Minimum	\$0.0112 \$0.0112		\$0.0131 \$0.0131
Authorized Overrun Commodity Charge Maximum Minimum	\$0.2273 \$0.0112		\$0.2292 \$0.0131
			, , , , , , ,

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING SECONDARY RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING SECONDARY DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

Commodity Charge: 2/

 Maximum
 \$0.0612
 \$0.0000
 \$0.0612

 Minimum
 \$0.0000
 \$0.0000
 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Eighth Revised Sheet No. 27 Eighth Revised Sheet No. 27

Superseding: Seventh Revised Sheet No. 27

Rate Schedule AFT-E No-Notice Firm Transportation Service

	Base	\$/Dth		
	Tariff	ACA	Total	
	Rate 1/	Adj.	Rate	
Reservation Charge:				
Maximum	\$6.5734	_	\$6.5734	
Minimum	\$0.0000	-	\$0.0000	
Commodity Charge:				
Maximum	\$0.0112	\$0.0017	\$0.0129	
Minimum	\$0.0112	\$0.0017	\$0.0129	
Authorized Overrun				
Commodity Charge				
Maximum	\$0.2273		\$0.2290	
Minimum	\$0.0112	\$0.0017	\$0.0129	

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING SECONDARY RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING SECONDARY DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

Commodity Charge: 2/

 Maximum
 \$0.0612
 \$0.0000
 \$0.0612

 Minimum
 \$0.0000
 \$0.0000
 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

FERC Docket: RP09-938-000

Ninth Revised Sheet No. 28 Ninth Revised Sheet No. 28

Superseding: Eighth Revised Sheet No. 28

Rate Schedule AFT-E No-Notice Firm Transportation Service Capacity Release

	Base	\$/Dth	Volumetric2/
	Tariff	Base	Total
	Rate 1/	Rate	Rate
	(a)	(b)	(c) = (b)
Reservation Charg	ge:		
Maximum	\$6.5734	\$0.2161	\$0.2161
Minimum	\$0.0000	\$0.0000	\$0.0000

Rate Schedule AFT-ES Capacity Release

	Base	\$/Dth	7	Volumetric	2/
	Tariff		Base		Total
	Rate 1/		Rate		Rate
	(a)		(b)		(c) = (b)
Reservation Charg	e:				
Maximum	\$2.6294		\$0.0864		\$0.0864
Minimum	\$0.0000		\$0.0000		\$0.0000
	Base	\$/Dth		3/	
	Tariff	ACA		Total	
	Rate	Adj.		Rate	
Commodity Charge:					
Maximum	\$0.2273	\$0.0019		\$0.2292	

- 1/ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.
- 2/ The volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a volumetric basis, and are exclusive of surcharges and commodity.
- 3/ Reservation charges and commodity charges applicable to capacity released under Rate Schedule AFT-1S and acquired under Rate Schedule AFT-1.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Eighth Revised Sheet No. 28 Eighth Revised Sheet No. 28

Superseding: Seventh Revised Sheet No. 28

Rate Schedule AFT-E No-Notice Firm Transportation Service Capacity Release

	Base	\$/Dth		Volumetric2/
	Tariff		Base	Total
	Rate 1/		Rate	Rate
	(a)		(b)	(c) = (b)
Reservation Charg	ge:			
Maximum	\$6.5734		\$0.2161	\$0.2161
Minimum	\$0.0000		\$0.0000	\$0.0000

Rate Schedule AFT-ES Capacity Release

	Base	\$/Dth	V	olumetric	2/
	Tariff		Base		Total
	Rate 1/		Rate		Rate
	(a)		(b)		(c) = (b)
Reservation Charg	e:				
Maximum	\$2.6294		\$0.0864		\$0.0864
Minimum	\$0.0000		\$0.0000		\$0.0000
	Base	\$/Dth		3/	
	Tariff	ACA		Total	
	Rate	Adj.		Rate	
Commodity Charge:					
Maximum	\$0.2273	\$0.0017		\$0.2290	
Minimum	\$0.0112	\$0.0017		\$0.0129	

- 1/ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.
- 2/ The volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a volumetric basis, and are exclusive of surcharges and commodity.
- 3/ Reservation charges and commodity charges applicable to capacity released under Rate Schedule AFT-1S and acquired under Rate Schedule AFT-1.

FERC Docket: RP09-938-000

Ninth Revised Sheet No. 29 Ninth Revised Sheet No. 29

Superseding: Eighth Revised Sheet No. 29

Rate Schedule AFT-1S Small Customer Firm Transportation Service

			Base	\$/Dth -	
			Tariff	ACA	Total
			Rate 1/	Adj.	Rate
Reserva	tion Charge:			_	
(F-1/WS-1)	Maximum	\$2.6294	_	\$2.6294
(F-2/F-3)	Maximum	\$2.6294	_	\$2.6294
(F-4)	Maximum	\$2.6294	_	\$2.6294
(STB/SS-3)	Maximum	\$2.6294	_	\$2.6294
(F-1/WS-1)	Minimum	\$0.0000	_	\$0.0000
(F-2/F-3)	Minimum	\$0.0000	_	\$0.0000
(F-4)	Minimum	\$0.0000	-	\$0.0000
(STB/SS-3)	Minimum	\$0.0000	-	\$0.0000
	ty Charge:				
	F-1/WS-1)	Maximum	\$0.2273	\$0.0019	\$0.2292
	F-2/F-3)	Maximum	\$0.2273	\$0.0019	\$0.2292
	F-4)	Maximum	\$0.2273	\$0.0019	\$0.2292
(STB/SS-3)	Maximum	\$0.2273	\$0.0019	\$0.2292
	_ a / a \		* 0 0110	** ***	** **
,	F-1/WS-1)	Minimum	\$0.0112	\$0.0019	\$0.0131
,	F-2/F-3)	Minimum	\$0.0112	\$0.0019	\$0.0131
,	F-4)	Minimum	\$0.0112	\$0.0019	\$0.0131
(STB/SS-3)	Minimum	\$0.0112	\$0.0019	\$0.0131
7					
	zed Overrun				
	ty Charge: F-1/WS-1)	Maximum	\$0.2273	\$0.0019	\$0.2292
,		Maximum	\$0.2273	\$0.0019	\$0.2292
,	. ,	Maximum	\$0.2273	\$0.0019	\$0.2292
	STB/SS-3)	Maximum	\$0.2273	\$0.0019	\$0.2292
(315/33-3)	Maximum	70.2273	70.0019	70.2292
(F-1/WS-1)	Minimum	\$0.0112	\$0.0019	\$0.0131
	F-2/F-3)	Minimum	\$0.0112	\$0.0019	\$0.0131
,	F-4)	Minimum	\$0.0112	\$0.0019	\$0.0131
	STB/SS-3)	Minimum	\$0.0112	\$0.0019	\$0.0131
'	,,		, , , , , , , , ,	,	+0.0101

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING SECONDARY RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING SECONDARY DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Eighth Revised Sheet No. 29 Eighth Revised Sheet No. 29

Superseding: Seventh Revised Sheet No. 29

Rate Schedule AFT-1S Small Customer Firm Transportation Service

			Base	\$/Dth ·	
			Tariff	ACA	Total
			Rate 1/	Adj.	Rate
Reserva	ation Charge:			3	
	(F-1/WS-1)	Maximum	\$2.6294	_	\$2.6294
	(F-2/F-3)	Maximum	\$2.6294	_	\$2.6294
	(F-4)	Maximum	\$2.6294	_	\$2.6294
	(STB/SS-3)	Maximum	\$2.6294	_	\$2.6294
	(F-1/WS-1)	Minimum	\$0.0000	_	\$0.0000
	(F-2/F-3)	Minimum	\$0.0000	_	\$0.0000
	(F-4)	Minimum	\$0.0000	-	\$0.0000
	(STB/SS-3)	Minimum	\$0.0000	-	\$0.0000
	ity Charge:				
	(F-1/WS-1)	Maximum	\$0.2273	\$0.0017	\$0.2290
	(F-2/F-3)	Maximum	\$0.2273	\$0.0017	\$0.2290
	(F-4)	Maximum	\$0.2273	\$0.0017	\$0.2290
	(STB/SS-3)	Maximum	\$0.2273	\$0.0017	\$0.2290
	·- • · • ·		* 0 0 1 1 0	** ***	** **
	(F-1/WS-1)	Minimum	\$0.0112	\$0.0017	\$0.0129
	(F-2/F-3)	Minimum	\$0.0112	\$0.0017	\$0.0129
	(F-4)	Minimum	\$0.0112	\$0.0017	\$0.0129
	(STB/SS-3)	Minimum	\$0.0112	\$0.0017	\$0.0129
7 10	a al Oa				
	ized Overrun ity Charge:				
	(F-1/WS-1)	Maximum	\$0.2273	\$0.0017	\$0.2290
	(F-2/F-3)	Maximum	\$0.2273	\$0.0017	\$0.2290
		Maximum	\$0.2273	\$0.0017	\$0.2290
	(STB/SS-3)	Maximum	\$0.2273	\$0.0017	\$0.2290
	(515/55 5)	Maximum	90.2275	QU.0017	70.2230
	(F-1/WS-1)	Minimum	\$0.0112	\$0.0017	\$0.0129
	(F-2/F-3)	Minimum	\$0.0112	\$0.0017	\$0.0129
	(F-4)	Minimum	\$0.0112	\$0.0017	\$0.0129
	(STB/SS-3)	Minimum	\$0.0112	\$0.0017	\$0.0129
	,,				,

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING SECONDARY RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING SECONDARY DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

Commodity Charge: 2/ Maximum \$0.0612 \$0.

Maximum \$0.0612 \$0.0000 \$0.0612 Minimum \$0.0000 \$0.0000 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

FERC Docket: RP06-519-000

First Revised Sheet No. 30 First Revised Sheet No. 30 : Effective Superseding: Original Sheet No. 30 $\,$

Rate Schedule AFT-1S Small Customer Firm Transportation Service Capacity Release

		Base\$/Dth	Volumetric2/	
		Tariff	Base	Total
		Rate 1/	Rate	Rate
Reservation	Charge:	(a)	(b)	(c) = (b)
(F-1/WS-1)	Maximum	\$2.6342	\$0.0866	\$0.0866
(F-2/F-3)	Maximum	\$2.6342	\$0.0866	\$0.0866
(F-4)	Maximum	\$2.6342	\$0.0866	\$0.0866
(STB/SS-3)	Maximum	\$2.6342	\$0.0866	\$0.0866
(F-1/WS-1)	Minimum	\$0.0000	\$0.0000	\$0.0000
(F-2/F-3)	Minimum	\$0.0000	\$0.0000	\$0.0000
(F-4)	Minimum	\$0.0000	\$0.0000	\$0.0000
(STB/SS-3)	Minimum	\$0.0000	\$0.0000	\$0.0000

- 1/ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.
- The Volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a volumetric basis, and are exclusive of surcharges and commodity.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Second Revised Sheet No. 30 Second Revised Sheet No. 30 Superseding: First Revised Sheet No. 30

Rate Schedule AFT-1S Small Customer Firm Transportation Service Capacity Release

		Base\$/Dth	Volumetric2/	
		Tariff	Base	Total
		Rate 1/	Rate	Rate
		(a)	(b)	(c) = (b)
Reservation (Charge:			
(F-1/WS-1)	Maximum	\$2.6294	\$0.0864	\$0.0864
(F-2/F-3)	Maximum	\$2.6294	\$0.0864	\$0.0864
(F-4)	Maximum	\$2.6294	\$0.0864	\$0.0864
(STB/SS-3)	Maximum	\$2.6294	\$0.0864	\$0.0864
(F-1/WS-1)	Minimum	\$0.0000	\$0.0000	\$0.0000
(F-2/F-3)	Minimum	\$0.0000	\$0.0000	\$0.0000
(F-4)	Minimum	\$0.0000	\$0.0000	\$0.0000
(STB/SS-3)	Minimum	\$0.0000	\$0.0000	\$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

The Volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a 2/ volumetric basis, and are exclusive of surcharges and commodity.

FERC Docket: RP09-938-000

Ninth Revised Sheet No. 31 Ninth Revised Sheet No. 31

Superseding: Eighth Revised Sheet No. 31

Rate Schedule AFT-ES
Small Customer
No-Notice Firm Transportation Service

	Base Tariff Rate 1/	\$/Dth ACA Adj.	Total Rate
Reservation Charge: Maximum Minimum	\$2.6294 \$0.0000	- -	\$2.6294 \$0.0000
Commodity Charge: Maximum Minimum	\$0.2273 \$0.0112		\$0.2292 \$0.0131
Authorized Overrun Commodity Charge: Maximum Minimum	\$0.2273 \$0.0112		\$0.2292 \$0.0131

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING SECONDARY RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING SECONDARY DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

Commodity Charge: 2/

Maximum \$0.0612 \$0.0000 \$0.0612 Minimum \$0.0000 \$0.0000 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Eighth Revised Sheet No. 31 Eighth Revised Sheet No. 31

Superseding: Seventh Revised Sheet No. 31

Rate Schedule AFT-ES
Small Customer
No-Notice Firm Transportation Service

	Base	\$/Dth	
	Tariff	ACA	Total
	Rate 1/	Adj.	Rate
Reservation Charge:			
Maximum	\$2.6294	-	\$2.6294
Minimum	\$0.0000	-	\$0.0000
Commodity Charge:			
Maximum	\$0.2273	\$0.0017	\$0.2290
Minimum	\$0.0112	\$0.0017	\$0.0129
Authorized Overrun			
Commodity Charge:			
Maximum	\$0.2273	\$0.0017	\$0.2290
Minimum	\$0.0112	\$0.0017	\$0.0129

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING SECONDARY RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING SECONDARY DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

Commodity Charge: 2/

Maximum \$0.0612 \$0.0000 \$0.0612 Minimum \$0.0000 \$0.0000 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

FERC Docket: RP06-519-000

First Revised Sheet No. 32 First Revised Sheet No. 32 : Effective Superseding: Original Sheet No. 32

Rate Schedule AFT-ES Small Customer No-Notice Firm Transportation Service Capacity Release

	Base\$/Dth Volumetric		
	Tariff	Base	Total
	Rate 1/	Rate	Rate
	(a)	(b)	(c) = (b)
Reservation Charge:			
Maximum	\$2.6342	\$0.0866	\$0.0866
Minimum	\$0.0000	\$0.0000	\$0.0000

- 1/ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.
- The Volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a volumetric basis, and are exclusive of surcharges and commodity.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Second Revised Sheet No. 32 Second Revised Sheet No. 32 Superseding: First Revised Sheet No. 32

Rate Schedule AFT-ES Small Customer No-Notice Firm Transportation Service Capacity Release

	Base	-\$/Dth Volumetric	2/
	Tariff	Base	Total
	Rate 1/	Rate	Rate
	(a)	(b)	(c) = (b)
Reservation Charge:			
Maximum	\$2.6294	\$0.0864	\$0.0864
Minimum	\$0.0000	\$0.0000	\$0.0000

- 1/ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.
- The Volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a volumetric basis, and are exclusive of surcharges and commodity.

FERC Docket: RP09-938-000

Ninth Revised Sheet No. 33 Ninth Revised Sheet No. 33

Superseding: Eighth Revised Sheet No. 33

Rate Schedule AIT-1 Interruptible Transportation Service

Campaditus Chausas	Base Tariff Rate 1/	\$/Dth ACA Adj.	Total Rate
Commodity Charge: Maximum	\$0.2421	\$0.0019	\$0.2440
Minimum	\$0.0076	\$0.0019	\$0.0095
Just have and Occasions			
Authorized Overrun Maximum	\$0.2421	\$0.0019	\$0.2440
Minimum	\$0.0076	\$0.0019	\$0.0095

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

Commodity Charge: 2/

 Maximum
 \$0.0612
 \$0.0000
 \$0.0000
 \$0.0612

 Minimum
 \$0.0000
 \$0.0000
 \$0.0000
 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Eighth Revised Sheet No. 33 Eighth Revised Sheet No. 33

Superseding: Seventh Revised Sheet No. 33

Rate Schedule AIT-1 Interruptible Transportation Service

	Zha uma	Base Tariff Rate 1/	\$/Dth ACA Adj.	Total Rate
Commodity (3	\$0.2421	\$0.0017	\$0.2438
Mini	num	\$0.0076	\$0.0017	\$0.0093
Authorized Maxim		\$0.2421	\$0.0017	\$0.2438
Mini	num	\$0.0076	\$0.0017	\$0.0093

HUBLINE SURCHARGE APPLICABLE TO ALL CUSTOMERS UTILIZING RECEIPT POINTS BETWEEN AND INCLUDING BEVERLY AND WEYMOUTH AND/OR UTILIZING DELIVERY POINTS BETWEEN BEVERLY AND WEYMOUTH, INCLUDING BEVERLY AND EXCLUDING WEYMOUTH, AND IN ADDITION TO OTHER APPLICABLE CHARGES:

Commodity Charge: 2/

 Maximum
 \$0.0612
 \$0.0000
 \$0.0000
 \$0.0612

 Minimum
 \$0.0000
 \$0.0000
 \$0.0000
 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} HubLine surcharges applicable to both the Commodity and Authorized Overrun Charges.

FERC Docket: RP09-938-000

Eighth Revised Sheet No. 34 Eighth Revised Sheet No. 34 Superseding: Seventh Revised Sheet No. 34

Rate Schedule T-1 Transportation Service (New York and New Jersey)

	Base Tariff Rate 1/	\$/Dth ACA Adj.	Total Rate
Reservation Charge	\$1.6480	-	\$1.6480
Commodity Charge	\$0.0039	\$0.0019	\$0.0058
Authorized Overrun Commodity Charge	\$0.0581	\$0.0019	\$0.0600

 $^{1/\,\,}$ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

FERC Docket: RP09-938-000

Eighth Revised Sheet No. 35 Eighth Revised Sheet No. 35 Superseding: Seventh Revised Sheet No. 35

Rate Schedule AFT-4 Firm Transportation Service

	Base Tariff Rate 1/	ACA Adj.	S/Dth Total Rate
Reservation Charge	\$3.5211	-	\$3.5211
Commodity Charge	\$0.0013	\$0.0019	\$0.0032
Authorized Overrun Commodity Charge	\$0.1171	\$0.0019	\$0.1190

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

FERC Docket: RP09-938-000

Seventh Revised Sheet No. 36 Seventh Revised Sheet No. 36 Superseding: Sixth Revised Sheet No. 36

Rate Schedule AFT-CL Firm Transportation Service

	Base	\$/Dth	
	Tariff	ACA	Total
	Rate 1/	Adj. 2/	Rate
CANAL LATERAL			
Reservation Charge:			
Maximum	\$2.0858		\$2.0858
Minimum Commodity Charge:	\$0.0000	-	\$0.0000
Maximum	\$0.0000	\$0.0019	\$0.0019
Minimum	\$0.0000		\$0.0019
Authorized Overrun			
Commodity Charge			
Maximum	\$0.0686		\$0.0705
Minimum	\$0.0000	\$0.0019	\$0.0019
MIDDLETOWN LATERAL			
Reservation Charge:			
Maximum	\$3.2764	_	\$3.2764
Minimum	\$0.0000	-	\$0.0000
Commodity Charge: Maximum	¢0 0000	\$0.0019	\$0.0019
Minimum	\$0.0000		\$0.0019
Authorized Overrun	40.0000	40.0013	40.0023
Commodity Charge			
Maximum		\$0.0019	\$0.1096
Minimum	\$0.0000	\$0.0019	\$0.0019
CLEARY LATERAL			
Reservation Charge:			
Maximum	\$1.4529		\$1.4529
Minimum	\$0.0000	-	\$0.0000
Commodity Charge: Maximum	¢0 0000	\$0.0019	\$0.0019
Minimum	\$0.0000		\$0.0019
Authorized Overrun	40.0000	+0.0013	40.0013
Commodity Charge			
Maximum	\$0.0478		\$0.0497
Minimum	\$0.0000	\$0.0019	\$0.0019
LAKE ROAD LATERAL			
Reservation Charge:			
Maximum	\$0.6476		\$0.6476
Minimum	\$0.0000	-	\$0.0000
Commodity Charge: Maximum	¢0 0000	\$0.0019	\$0.0019
Maximum Minimum	\$0.0000		\$0.0019
Authorized Overrun	+0.0000	+0.001J	40.00±3
Commodity Charge			
Maximum	\$0.0213		\$0.0232
Minimum	\$0.0000	\$0.0019	\$0.0019

The Base Tariff is the effective rate on file with the Commission excluding adjustments approved by the Commission.

The ACA Surcharge will only apply if the AFT-CL Customer has not paid an ACA Surcharge for the same gas volumes transported under another 2/ rate schedule.

FERC Docket: RP09-938-000

Seventh Revised Sheet No. 37 Seventh Revised Sheet No. 37

Superseding: Sixth Revised Sheet No. 37

Rate Schedule AFT-CL Firm Transportation Service

	Base\$/Dth			
	Tariff	ACA		
	Rate 1/	Adj. 2/	Adj. Rate	
BRAYTON POINT LATERAL				
Reservation Charge:	¢1 0700		¢1 0700	
Maximum Minimum	\$1.2700 \$0.0000		\$1.2700 \$0.0000	
Commodity Charge:	20.0000	_	20.0000	
Maximum	\$0.0000	\$0.0019	\$0.0019	
Minimum	\$0.0000	\$0.0019	\$0.0019	
Authorized Overrun				
Commodity Charge				
Maximum Minimum	\$0.0418		\$0.0437	
Minimum	\$0.0000	\$0.0019	\$0.0019	
BELLINGHAM LATERAL Reservation Charge:				
Maximum	\$0.9714	_	\$0.9714	
Minimum	\$0.0000		\$0.0000	
Commodity Charge:	40.0000		40.0000	
Maximum	\$0.0000	\$0.0019	\$0.0019	
Minimum	\$0.0000	\$0.0019	\$0.0019	
Authorized Overrun				
Commodity Charge Maximum	\$0.0319	\$0.0019	\$0.0338	
Minimum	\$0.0000		\$0.0019	
HHIMAN	40.0000	40.0013	40.0019	
PHELPS DODGE LATERAL				
Reservation Charge:				
Maximum	\$0.0000	_	\$0.0000	
Minimum	\$0.0000		\$0.0000	
Commodity Charge:				
Maximum		\$0.0019	\$0.0185	
Minimum	\$0.0000	\$0.0019	\$0.0019	
Authorized Overrun Commodity Charge				
Maximum	\$0.0166	\$0.0019	\$0.0185	
Minimum	\$0.0000		\$0.0019	
MANCHESTER STREET LATERAL				
Reservation Charge:				
Maximum	\$2.4500		\$2.4500	
Minimum	\$0.0000	-	\$0.0000	
Commodity Charge:	¢0 0000	¢0 0010	¢0 0010	
Maximum Minimum	\$0.0000 \$0.0000		\$0.0019 \$0.0019	
Minimum Authorized Overrun	70.0000	40.00TA	\$0.0019	
Commodity Charge				
Maximum	\$0.0805	\$0.0019	\$0.0824	
Minimum	\$0.0000	\$0.0019	\$0.0019	

^{1/} The Base Tariff is the effective rate on file with the Commission excluding adjustments approved by the Commission.

^{2/} The ACA Surcharge will only apply if the AFT-CL Customer has not paid an ACA Surcharge for the same gas volumes transported under another rate schedule.

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 37A Fifth Revised Sheet No. 37A Superseding: Fourth Revised Sheet No. 37A

Rate Schedule AFT-CL Firm Transportation Service

	Base	\$:/D+h	
	Tariff	ACA	GRI	Total
	Rate 1/	Adj.		Rate
CAPE COD LATERAL				
Reservation Charge:				
Maximum	\$9.0501	-	-	\$9.0501
Minimum	\$0.0000	-	-	\$0.0000
Commodity Charge:				
Maximum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Minimum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Authorized Overrun				
Commodity Charge Maximum	\$0.2975	\$0.0019	\$0.0000	\$0.2994
Minimum	\$0.0000	\$0.0019	\$0.0000	\$0.2994
PITITIUM	70.0000	Q0.0013	70.0000	70.0019
NORTHEAST GATEWAY LATERAL				
Reservation Charge:				
Maximum	\$4.3449	-	-	\$4.3449
Minimum	\$0.0000	-	-	\$0.0000
Commodity Charge:				
Maximum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Minimum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Authorized Overrun				
Commodity Charge Maximum	\$0.1428	\$0.0019	\$0.0000	\$0.1447
Minimum	\$0.1428	\$0.0019	\$0.0000	\$0.0019
FILITINGIII	40.0000	Q0.0019	40.0000	Q0.0019
J-2 FACILITY				
Reservation Charge:				
Maximum	\$4.9077	-	-	\$4.9077
Minimum	\$0.0000	-	-	\$0.0000
Commodity Charge:				
Maximum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Minimum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Authorized Overrun Commodity Charge				
Maximum	\$0.1613	\$0.0019	\$0.0000	\$0.1632
Minimum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
	,	, , , , , , ,	,	,
KLEEN ENERGY LATERAL				
Reservation Charge:				
Maximum	\$1.2247	-	-	\$1.2247
Minimum	\$0.0000	-	-	\$0.0000
Commodity Charge:	* 0 0000	** ***	** ***	** ***
Maximum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Minimum Authorized Overrun	\$0.0000	\$0.0019	\$0.0000	\$0.0019
Commodity Charge				
Maximum	\$0.0403	\$0.0019	\$0.0000	\$0.0422
Minimum	\$0.0000	\$0.0019	\$0.0000	\$0.0019
	, , , , , , , , ,	, , , , , , ,	,	, 0 . 0 0 ± 3

The Base Tariff is the effective rate on file with the Commission excluding adjustments approved by the Commission. 1/

The ACA Surcharge will only apply if the AFT-CL Customer has not paid an ACA Surcharge for the same gas volumes transported under another $\,$ rate schedule.

FERC Docket: CP05-383-001

Third Revised Sheet No. 38 Third Revised Sheet No. 38 : Effective Superseding: Second Revised Sheet No. 38

Rate Schedule AFT-CL Capacity Release

	Base Tariff Rate 1/ (a)	\$/DthVolum Base 2/ Rate (b)	metric Total Rate (c)=(b)
CANAL LATERAL Reservation Charge: Maximum Minimum	\$2.0858 \$0.0000	\$0.0686 \$0.0000	\$0.0686 \$0.0000
MIDDLETOWN LATERAL Reservation Charge: Maximum Minimum	\$3.2764 \$0.0000	\$0.1077 \$0.0000	\$0.1077 \$0.0000
CLEARY LATERAL Reservation Charge: Maximum Minimum	\$1.4529 \$0.0000	\$0.0478 \$0.0000	\$0.0478 \$0.0000
LAKE ROAD LATERAL Reservation Charge: Maximum Minimum	\$0.6476 \$0.0000	\$0.0213 \$0.0000	\$0.0213 \$0.0000
BRAYTON POINT LATERAL Reservation Charge: Maximum Minimum	\$1.2700 \$0.0000	\$0.0418 \$0.0000	\$0.0418 \$0.0000
BELLINGHAM LATERAL Reservation Charge: Maximum Minimum	\$0.9714 \$0.0000	\$0.0319 \$0.0000	\$0.0319 \$0.0000
PHELPS DODGE LATERAL Reservation Charge: Maximum Minimum	\$0.0000 \$0.0000	\$0.0000 \$0.0000	\$0.0000 \$0.0000
MANCHESTER STREET LATERAL Reservation Charge: Maximum Minimum	\$2.4500 \$0.0000	\$0.0805 \$0.0000	\$0.0805 \$0.0000
CAPE COD LATERAL Reservation Charge: Maximum Minimum	\$9.0501 \$0.0000	\$0.2975 \$0.0000	\$0.2975 \$0.0000
NORTHEAST GATEWAY LATERAL Reservation Charge: Maximum Minimum	\$4.3449 \$0.0000	\$0.1428 \$0.0000	\$0.1428 \$0.0000

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the

The volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides 2/ for bids on a volumetric basis, and are exclusive of surcharges and commodity.

FERC Docket: RP10-146-000

First Revised Sheet No. 38A First Revised Sheet No. 38A Superseding: Original Sheet No. 38A

Rate Schedule AFT-CL Capacity Release

	Base	\$/Dth		-Volumetric
	Tariff Rate 1/	2/	Base Rate	Total Rate
	(a)	_,	(b)	(c) = (b)
J-2 FACILITY Reservation Charge:				
Maximum Minimum	\$4.9077 \$0.0000		\$0.1613 \$0.0000	\$0.1613 \$0.0000
KLEEN ENERGY LATERAL Reservation Charge:				
Maximum Minimum	\$1.2247 \$0.0000		\$0.0403 \$0.0000	\$0.0403 \$0.0000

- The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the 1/
- The volumetric reservation charges are applicable to capacity releases where Releasing Customer's Notice provides for bids on a volumetric basis, and are exclusive of surcharges and commodity.

FERC Docket: RP09-938-000

Ninth Revised Sheet No. 39 Ninth Revised Sheet No. 39 Superseding: Eighth Revised Sheet No. 39

Rate Schedule PAL Park and Loan Service

		Base Tariff Rate 1/	\$/Dth ACA Adj.	Total Rate
	PAL Charge: Maximum	\$0.2421	\$0.0019	\$0.2440
I	Minimum	\$0.0000	\$0.0000	\$0.0000

The Base Tariff Rate is the effective rate on file with the Commission, excluding adjustments approved by the Commission. 1/

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Eighth Revised Sheet No. 39 Eighth Revised Sheet No. 39 Superseding: Seventh Revised Sheet No. 39

Rate Schedule PAL Park and Loan Service

	Base Tariff Rate 1/	\$/Dt ACA Adj.	th Total Rate
Daily PAL Charge: Maximum	\$0.2421	\$0.0017	\$0.2438
Minimum	\$0.0000	\$0.0000	\$0.0000

The Base Tariff Rate is the effective rate on file with the Commission, excluding adjustments approved by the Commission. 1/

FERC Docket: RP10-117-000

Fourth Revised Sheet No. 40 Fourth Revised Sheet No. 40 Superseding: Third Revised Sheet No. 40

Fuel Reimbursement Percentages

Period	Duration	FRP
SYSTEM SERVICES		
Winter	December 1 - March 31	1.35%
Spring, Summer And Fall	April 1 - November 30	0.84%
INCREMENTAL RAMAPO SERVICE		
Winter	December 1 - March 31	2.36%
Spring, Summer And Fall	April 1 - November 30	1.18%

Fuel Reimbursement Percentages (FRP) pursuant to Section 32 of the General Terms and Conditions of this FERC Gas Tariff.

FERC Docket: RP04-361-000

Rate Schedule TTT Title Transfer Tracking Service

		Base Tariff Rate 1/	\$/Transaction Adjustments	Total Rate
TTT Servio		\$5.3900	-	\$5.3900
Minimu	ım	\$0.0000	_	\$0.0000

 $^{1/\,}$ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

FERC Docket: RP09-938-000

Eleventh Revised Sheet No. 42 Eleventh Revised Sheet No. 42

Superseding: Tenth Revised Sheet No. 42

Rate Schedule AIT-2 Interruptible Transportation Service

	Base	\$/Dth -	
	Tariff Rate 1/	ACA Adj.2/	Total Rate
BRAYTON POINT LATERAL			
Commodity Charge Maximum Minimum Authorized Overrun	\$0.0418 \$0.0000	\$0.0019 \$0.0019	\$0.0437 \$0.0019
Commodity Charge Maximum Minimum		\$0.0019 \$0.0019	\$0.0437 \$0.0019
MANCHESTER STREET LATERAL Commodity Charge			
Maximum Minimum Authorized Overrun Commodity Charge		\$0.0019 \$0.0019	\$0.0824 \$0.0019
Maximum Minimum		\$0.0019 \$0.0019	\$0.0824 \$0.0019
CANAL LATERAL			
Commodity Charge Maximum Minimum Authorized Overrun		\$0.0019 \$0.0019	\$0.0705 \$0.0019
Commodity Charge Maximum Minimum	\$0.0686 \$0.0000	\$0.0019 \$0.0019	\$0.0705 \$0.0019
CAPE COD LATERAL			
Commodity Charge Maximum Minimum Authorized Overrun		\$0.0019 \$0.0019	\$0.2994 \$0.0019
Commodity Charge Maximum Minimum		\$0.0019 \$0.0019	\$0.2994 \$0.0019
NORTHEAST GATEWAY LATERAL Commodity Charge			
Maximum Minimum Authorized Overrun		\$0.0019 \$0.0019	\$0.1447 \$0.0019
Commodity Charge Maximum Minimum	\$0.1428 \$0.0000	\$0.0019 \$0.0019	\$0.1447 \$0.0019

^{1/} The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} The ACA Surcharge will only apply if the Customer has not paid an ACA Surcharge for the same gas volumes transported under another rate schedule.

FERC Docket: RP10-146-000

Second Revised Sheet No. 42A Second Revised Sheet No. 42A

Superseding: First Revised Sheet No. 42A

Rate Schedule AIT-2 Interruptible Transportation Service

		\$/Dth	
	Tariff Rate 1/		Total Rate
J-2 FACILITY Commodity Charge			
Maximum Minimum	\$0.1613 \$0.0000		\$0.1632 \$0.0019
Authorized Overrun Commodity Charge			
Maximum Minimum	\$0.1613 \$0.0000	\$0.0019 \$0.0019	\$0.1632 \$0.0019
KLEEN ENERGY LATERAL			
Commodity Charge Maximum Minimum	\$0.0403 \$0.0000		\$0.0422 \$0.0019
Authorized Overrun Commodity Charge			
Maximum Minimum	\$0.0403 \$0.0000		\$0.0422 \$0.0019

 $^{1/\,}$ $\,$ The Base Tariff is the effective rate on file with the Commission, excluding adjustments approved by the Commission.

^{2/} The ACA Surcharge will only apply if the Customer has not paid an ACA Surcharge for the same gas volumes transported under another rate schedule.

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000

Sheet Nos. 43 - 49 Sheet Nos. 43 - 49 : Effective

Sheet Nos. 43 - 49 are being reserved for future use

FERC Docket: RP09-823-000

First Revised Sheet No. 50 First Revised Sheet No. 50 Superseding: Original Sheet No. 50

Sheet No. 50 was previously issued, but is now reserved for future use.

FERC Docket: RP10-308-000

First Revised Sheet No. 51 First Revised Sheet No. 51

Superseding: Original Sheet No. 51

STATEMENT OF NEGOTIATED RATES 1/

 $\hbox{\tt Customer Name: Colonial Gas Company d/b/a Keyspan Energy Delivery New England}$

Term of Negotiated Rate: 5/

Rate Schedule: AFT-1
MDTQ: 25,000 dth/d

Reservation Rate: \$10.032

Usage Rate: 2/

Primary Receipt Point: 3/

Primary Delivery Point: M&R 18, Bourne, MA; M&R 51, Sagamore, MA 4/

The Recourse Rates applicable to this service are the maximum rates stated on Pipeline's currently effective Statement of Rates for Rate Schedule AFT-1. The Negotiated Rates are exclusive of fuel, shrinkage, lost and unaccounted for gas, which Customer shall be obligated to pay as effective from time to

- These negotiated rate transactions do not deviate in any material respect from the applicable form of service agreement set forth in Pipeline's FERC Gas Tariff. These negotiated rates are only applicable to transportation service up to Customer's MDTQ from the Primary Points of Receipt to the Primary Points of Delivery.
- 2/ The maximum applicable usage rate as effective from time to time based upon a straight fixed variable rate design.
- 3/ Customer may cause up to 25,000 Dth/day to be received on a primary basis at the interconnect of Algonquin's HubLine facilities and the facilities of the Maritimes and Northeast Pipeline, L.L.C. at Beverly, MA.
- 4/ Customer may cause up to 10,000 Dth/d to be delivered on a primary basis at M&R 18, Bourne, MA and up to 15,000 Dth/d on a primary basis at M&R 51, Sagamore, MA.
- The primary term commences on the later of October 1, 2003 or the HubLine Mainline service commencement date and extends until ten (10) years from the first day of the first full month following the HubLine Mainline service commencement date.

FERC Docket: RP10-308-000

First Revised Sheet No. 52 First Revised Sheet No. 52

Superseding: Original Sheet No. 52

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Fall River Gas Company d/b/a New England Gas Company - Fall

River

Term of Negotiated Rate: 5/

Rate Schedule: AFT-1
MDTQ: 5,000 dth/d

Reservation Rate: \$11.5583 dth/d

Usage Rate: 2/

Primary Receipt Point: 3/

Primary Delivery Point: M&R 00824, North Fall River, MA 4/

The Recourse Rates applicable to this service are the maximum rates stated on Pipeline's currently effective Statement of Rates for Rate Schedule AFT-1. The Negotiated Rates are exclusive of fuel, shrinkage, lost and unaccounted for gas, which Customer shall be obligated to pay as effective from time to time.

- These negotiated rate transactions do not deviate in any material respect from the applicable form of service agreement set forth in Pipeline's FERC Gas Tariff. These negotiated rates are only applicable to transportation service up to Customer's MDTQ from the Primary Points of Receipt to the Primary Points of Delivery.
- 2/ The maximum applicable usage rate as effective from time to time based upon a straight fixed variable rate design.
- 3/ Customer may cause up to 5,000 Dth/day to be received at the interconnect of Algonquin's HubLine facilities and the facilities of the Maritimes and Northeast Pipeline, L.L.C. at Beverly, MA.
- 4/ Customer may cause up to 5,000 Dth/day to be delivered on a primary basis at M&R 284.
- The primary term commences on the later of October 1, 2003 or the HubLine Mainline service commencement date and extends until the last day of the month in which the tenth (10th) anniversary of the HubLine Mainline commencement of service occurs.

FERC Docket: RP10-308-000

Second Revised Sheet No. 53 Second Revised Sheet No. 53

Superseding: First Revised Sheet No. 53

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: The Narragansett Electric Company d/b/a National Grid

Term of Negotiated Rate: 5/

Rate Schedule: AFT-1
MDTQ: 4,000 dth/d

Reservation Rate: \$11.5583 dth/d

Usage Rate: 2/

Primary Receipt Point: 3/

Primary Delivery Point: M&R 12, Warren, RI 4/

The Recourse Rates applicable to this service are the maximum rates stated on Pipeline's currently effective Statement of Rates for Rate Schedule AFT-1. The Negotiated Rates are exclusive of fuel, shrinkage, lost and unaccounted for gas, which Customer shall be obligated to pay as effective from time to time.

- These negotiated rate transactions do not deviate in any material respect from the applicable form of service agreement set forth in Pipeline's FERC Gas Tariff. These negotiated rates are only applicable to transportation service up to Customer's MDTQ from the Primary Points of Receipt to the Primary Points of Delivery.
- 2/ The maximum applicable usage rate as effective from time to time based upon a straight fixed variable rate design.
- 3/ Customer may cause up to 4,000 Dth/day to be received on a primary basis at the interconnect of Algonquin's HubLine facilities and the facilities of the Maritimes and Northeast Pipeline, L.L.C. at Beverly, MA.
- 4/ Customer may cause up to 4,000 Dth/day to be delivered on a primary basis at M&R 12.
- 5/ The primary term commences on the later of October 1, 2003 or the HubLine Mainline service commencement date and extends until the last day of the month in which the tenth (10th) anniversary of the HubLine Mainline commencement of service occurs.

FERC Docket: RP09-823-000

First Revised Sheet No. 54 First Revised Sheet No. 54 Superseding: Original Sheet No. 54

Sheet No. 54 was previously issued but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 55 First Revised Sheet No. 55 Superseding: Original Sheet No. 55

Sheet No. 55 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 56 First Revised Sheet No. 56 Superseding: Original Sheet No. 56

Sheet No. 56 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 57 First Revised Sheet No. 57 Superseding: Original Sheet No. 57

Sheet No. 57 was previously issued but is now reserved for future use.

FERC Docket: RP10-308-000

First Revised Sheet No. 58 First Revised Sheet No. 58

Superseding: Original Sheet No. 58

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Distrigas of Massachusetts LLC

Term of Negotiated Rate: 2/

Rate Schedule: AFT-1
MDTQ: 60,000 dth/d

Reservation Rate: \$6.5854 3/

Usage Rate: \$0.0133

Primary Receipt Point: 4/
Primary Delivery Point: 5/

The Recourse Rates applicable to this service are the maximum rates stated on Pipeline's currently effective Statement of Rates for Rate Schedule AFT-1..

- These negotiated rate transactions do not deviate in any material respect from the applicable form of service agreement set forth in Algonquin's FERC Gas Tariff. These negotiated rates are only applicable to transportation service up to Customer's MDTQ. In addition to the above rates, Customer shall pay all applicable surcharges and the applicable Fuel Retainage Percentage in effect from time to time and applicable to service under the Service Agreement.
- 2/ The primary term commences on October 1, 2004 and extends through October 31, 2014.
- In addition to the Reservation rate specified above, Customer shall pay a lump sum amount as set forth in Section 3.1 of the Service Agreement by the payment due date for the December invoice after service commences hereunder.
- 4/ Everett, MA (Meter # 00271).
- 5/ Waterbury (Meter # 00002); Stony Point, NY (Meter # 00040); Cromwell, CT (Meter # 00078); Bellingham (Meter # 00094); Mahwah (Meter # 00201); Mendon (Meter # 00205); Lambertville (Meter # 00210); Brookfield (Meter # 00251); ANP Bellingham (Meter # 00814)

FERC Docket: RP10-308-000

Second Revised Sheet No. 59 Second Revised Sheet No. 59

Superseding: First Revised Sheet No. 59

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: The Narragansett Electric Company d/b/a National Grid

Term of Negotiated Rate: 2/

Rate Schedule: AFT-1
MDTQ: 3,500 dth/d

Reservation Rate: \$6.992

Usage Rate: 3/

Primary Receipt Point: M&R 215, Beverly - Essex County, MA

Primary Delivery Point: M&R 59, Montville, CT

The Recourse Rates applicable to this service are the maximum rates stated on Algonquin's currently effective Statement of Rates for Rate Schedule AFT-1.

- These negotiated rate transactions do not deviate in any material respect from the applicable form of service agreement set forth in Algonquin's FERC Gas Tariff. This Negotiated Rate shall apply only to service under Contract No. 510209, up to Customer's specified MDTQ, and using the receipt and delivery points designated herein.
- 2/ The primary term commences on November 1, 2004 and extends until and including October 31, 2014.
- 3/ The maximum applicable commodity charge and the ACA, fuel and other surcharges applicable to Customer's service under Rate Schedule AFT-1.

FERC Docket: RP09-823-000

First Revised Sheet No. 60 First Revised Sheet No. 60 Superseding: Substitute Original Sheet No. 60

Sheet No. 60 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 60A First Revised Sheet No. 60A Superseding: Substitute Original Sheet No. 60A

Sheet No. 60A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 60B First Revised Sheet No. 60B Superseding: Substitute Original Sheet No. 60B

Sheet No. 60B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 61 First Revised Sheet No. 61 Superseding: Substitute Original Sheet No. 61

Sheet No. 61 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 61A First Revised Sheet No. 61A Superseding: Substitute Original Sheet No. 61A

Sheet No. 61A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 61B First Revised Sheet No. 61B Superseding: Substitute Original Sheet No. 61B

Sheet No. 61B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 62 First Revised Sheet No. 62 Superseding: Substitute Original Sheet No. 62

Sheet No. 62 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 62A First Revised Sheet No. 62A Superseding: Substitute Original Sheet No. 62A

Sheet No. 62A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 62B First Revised Sheet No. 62B Superseding: Substitute Original Sheet No. 62B

Sheet No. 62B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 63 First Revised Sheet No. 63 Superseding: Second Substitute Original Sheet No. 63

Sheet No. 63 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 63A First Revised Sheet No. 63A Superseding: Second Substitute Original Sheet No. 63A

Sheet No. 63A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 63B First Revised Sheet No. 63B Superseding: Second Revised Original Sheet No. 63B

Sheet No. 63B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 64 First Revised Sheet No. 64 Superseding: Original Sheet No. 64

Sheet No. 64 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 64A First Revised Sheet No. 64A Superseding: Second Substitute Original Sheet No. 64A

Sheet No. 64A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 64B First Revised Sheet No. 64B Superseding: Second Substitute Original Sheet No. 64B

Sheet No. 64B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 64C First Revised Sheet No. 64C Superseding: Original Sheet No. 64C

Sheet No. 64C was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 65 First Revised Sheet No. 65 Superseding: Substitute Original Sheet No. 65

Sheet No. 65 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 65A First Revised Sheet No. 65A Superseding: Substitute Original Sheet No. 65A

Sheet No. 65A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 65B First Revised Sheet No. 65B Superseding: Substitute Original Sheet No. 65B

Sheet No. 65B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 66 First Revised Sheet No. 66 Superseding: Original Sheet No. 66

Sheet No. 66 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 66A First Revised Sheet No. 66A Superseding: Substitute Original Sheet No. 66A

Sheet No. 66A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 66B First Revised Sheet No. 66B Superseding: Substitute Original Sheet No. 66B

Sheet No. 66B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 67 First Revised Sheet No. 67 Superseding: Substitute Original Sheet No. 67

Sheet No. 67 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 67A First Revised Sheet No. 67A Superseding: Substitute Original Sheet No. 67A

Sheet No. 67A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 67B First Revised Sheet No. 67B Superseding: Substitute Original Sheet No. 67B

Sheet No. 67B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 68 First Revised Sheet No. 68 Superseding: Original Sheet No. 68

Sheet No. 68 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 68A First Revised Sheet No. 68A Superseding: Substitute Original Sheet No. 68A

Sheet No. 68A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 68B First Revised Sheet No. 68B Superseding: Substitute Original Sheet No. 68B

Sheet No. 68B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 68C First Revised Sheet No. 68C Superseding: Substitute Original Sheet No. 68C

Sheet No. 68C was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 69 First Revised Sheet No. 69 Superseding: Substitute Original Sheet No. 69

Sheet No. 69 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 69A First Revised Sheet No. 69A Superseding: Substitute Original Sheet No. 69A

Sheet No. 69A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 69B First Revised Sheet No. 69B Superseding: Original Sheet No. 69B

Sheet No. 69B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 70 First Revised Sheet No. 70 Superseding: Substitute Original Sheet No. 70

Sheet No. 70 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 70A First Revised Sheet No. 70A Superseding: Substitute Original Sheet No. 70A

Sheet No. 70A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 70B First Revised Sheet No. 70B Superseding: Substitute Original Sheet No. 70B

Sheet No. 70B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 71 First Revised Sheet No. 71 Superseding: Substitute Original Sheet No. 71

Sheet No. 71 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 71A First Revised Sheet No. 71A Superseding: Substitute Original Sheet No. 71A

Sheet No. 71A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 71B First Revised Sheet No. 71B Superseding: Substitute Original Sheet No. 71B

Sheet No. 71B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 72 First Revised Sheet No. 72 Superseding: Substitute Original Sheet No. 72

Sheet No. 72 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 72A First Revised Sheet No. 72A Superseding: Substitute Original Sheet No. 72A

Sheet No. 72A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 72B First Revised Sheet No. 72B Superseding: Substitute Original Sheet No. 72B

Sheet No. 72B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 73 First Revised Sheet No. 73 Superseding: Substitute Original Sheet No. 73

Sheet No. 73 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 73A First Revised Sheet No. 73A Superseding: Substitute Original Sheet No. 73A

Sheet No. 73A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 73B First Revised Sheet No. 73B Superseding: Substitute Original Sheet No. 73B

Sheet No. 73B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 74 First Revised Sheet No. 74 Superseding: Substitute Original Sheet No. 74

Sheet No. 74 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 74A First Revised Sheet No. 74A Superseding: Substitute Original Sheet No. 74A

Sheet No. 74A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 74B First Revised Sheet No. 74B Superseding: Original Sheet No. 74B

Sheet No. 74B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 75 First Revised Sheet No. 75 Superseding: Substitute Original Sheet No. 75

Sheet No. 75 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 75A First Revised Sheet No. 75A Superseding: Substitute Original Sheet No. 75A

Sheet No. 75A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 75B First Revised Sheet No. 75B Superseding: Original Sheet No. 75B

Sheet No. 75B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 76 First Revised Sheet No. 76 Superseding: Original Sheet No. 76

Sheet No. 76 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 76A First Revised Sheet No. 76A Superseding: Substitute Original Sheet No. 76A

Sheet No. 76A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 76B First Revised Sheet No. 76B Superseding: Substitute Original Sheet No. 76B

Sheet No. 76B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 77 First Revised Sheet No. 77 Superseding: Substitute Original Sheet No. 77

Sheet No. 77 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 77A First Revised Sheet No. 77A Superseding: Substitute Original Sheet No. 77A

Sheet No. 77A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 77B First Revised Sheet No. 77B Superseding: Substitute Original Sheet No. 77B

Sheet No. 77B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 78 First Revised Sheet No. 78 Superseding: Substitute Original Sheet No. 78

Sheet No. 78 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 78 \mbox{A} First Revised Sheet No. 78 \mbox{A} Superseding: Substitute Original Sheet No. 78 \mbox{A}

Sheet No. 78A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 78B First Revised Sheet No. 78B Superseding: Original Sheet No. 78B

Sheet No. 78B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 79 First Revised Sheet No. 79 Superseding: Substitute Original Sheet No. 79

Sheet No. 79 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 79A First Revised Sheet No. 79A Superseding: Substitute Original Sheet No. 79A

Sheet No. 79A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 79B First Revised Sheet No. 79B Superseding: Original Sheet No. 79B

Sheet No. 79B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 80 First Revised Sheet No. 80 Superseding: Substitute Original Sheet No. 80

Sheet No. 80 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 80A First Revised Sheet No. 80A Superseding: Substitute Original Sheet No. 80A

Sheet No. 80A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 80B First Revised Sheet No. 80B Superseding: Original Sheet No. 80B

Sheet No. 80B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 81 First Revised Sheet No. 81 Superseding: Substitute Original Sheet No. 81

Sheet No. 81 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 81A First Revised Sheet No. 81A Superseding: Substitute Original Sheet No. 81A

Sheet No. 81A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 81B First Revised Sheet No. 81B Superseding: Substitute Original Sheet No. 81B

Sheet No. 81B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 82 First Revised Sheet No. 82 Superseding: Substitute Original Sheet No. 82

Sheet No. 82 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 82A First Revised Sheet No. 82A Superseding: Substitute Original Sheet No. 82A

Sheet No. 82A was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 82B First Revised Sheet No. 82B Superseding: Substitute Original Sheet No. 82B

Sheet No. 82B was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 83 First Revised Sheet No. 83 Superseding: Substitute Original Sheet No. 83

Sheet No. 83 was previously issued, but is now reserved for future use.

FERC Docket: RP09-823-000

First Revised Sheet No. 83A First Revised Sheet No. 83A Superseding: Substitute Original Sheet No. 83A

Sheet No. 83A was previously issued, but is now reserved for future use.

FERC Docket: RP10-308-000

Second Revised Sheet No. 84 Second Revised Sheet No. 84

Superseding: First Revised Sheet No. 84

STATEMENT OF NEGOTIATED RATES 1/4/

Customer Name: Mirant Energy Trading LLC

Service Agreement: 510336

Rate Schedule: AFT-CL

MDTQ: 75,000 dth

Reservation Rate: 3/

Primary Receipt Point: Meter No. 8004
Primary Delivery Point: Meter No. 00806

The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for firm transportation on the Canal Lateral under Rate Schedule AFT-CL.

- This negotiated rate transaction does not deviate in any material respect from the applicable form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designated herein. Algonquin and Customer, by agreeing to this negotiated rate, acknowledge that the otherwise generally applicable maximum Recourse Rate(s) shall not apply or be available to Customer for service under the Service Agreement during the Negotiated Rate Term, notwithstanding any adjustment to such generally applicable maximum Recourse Rate(s) which may become effective during the Negotiated Rate Term. Customer shall remain eligible for revenue credits determined pursuant to Section 41.1 of the General Terms and Conditions of Algonquin's FERC Gas Tariff.
- The term of this negotiated rate ("Negotiated Rate Term") commences on the date service commences under the Service Agreement specified above and continues through the end of the Initial Primary Term of such Service Agreement. In the event Customer elects to extend the Primary Term under the Service Agreement for a period of five (5) years pursuant to Section 8.3 of the General Terms and Conditions of Algonquin's FERC Gas Tariff, then Algonquin and Customer agree that the Negotiated Rate Term shall also be extended for such five (5) year period.
- 3/ Customer shall pay a negotiated reservation rate of \$0.056 per Dth, per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term.
- The usage rate to be paid by Customer under the Service Agreement specified above during the Negotiated Rate Term shall be zero. Customer shall also pay the applicable Annual Charge Adjustment ("ACA"), other applicable surcharges and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above.

FERC Docket: RP10-308-000

Second Revised Sheet No. 85 Second Revised Sheet No. 85

Superseding: First Revised Sheet No. 85

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Colonial Gas Company D/B/A National Grid

Service Agreement: 510365R1

Term of Negotiated Rate: The term of this negotiated rate ("Negotiated Rate Term") commences on date service commences under the Service Agreement specified above and continues through the end of the Primary Term of such Service Agreement.

Rate Schedule: AFT-CL (Canal Lateral)

MDTQ: 38,000 dth/d

Reservation Rate: Customer shall pay a negotiated reservation rate of \$1.0646 per Dth, per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term.

Usage Rate: The usage rate to be paid by customer under the Service Agreement specified above during the Negotiated Rate Term shall be zero. Customer shall also pay the applicable Annual Charge Adjustment ("ACA"), other applicable surcharges, and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above.

Primary Receipt Point: M&R 80004, Interconnect AGT G SYS & Canal Lateral in Bourne, MA

Primary Delivery Point: M&R 80806, G-24 Mainline Head of Lateral, Sandwich, MA

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for Rate Schedule AFT-CL for the Algonquin Canal Lateral.

FOOTNOTES:

1/This negotiated rate transaction does not deviate in any material respect from the form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designate herein.

FERC Docket: RP10-308-000

Second Revised Sheet No. 86 Second Revised Sheet No. 86

Superseding: First Revised Sheet No. 86

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Colonial Gas Company D/B/A National Grid

Service Agreement: 510366R1

Term of Negotiated Rate: The term of this negotiated rate ("Negotiated Rate Term") commences on date service commences under the Service Agreement specified above and continues through the end of the Primary Term of such Service Agreement.

Rate Schedule: AFT-CL (Cape Cod Lateral)

MDTQ: 38,000 dth/d

Reservation Rate: Customer shall pay a negotiated reservation rate of \$8.3654 per Dth, per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term.

Usage Rate: The usage rate to be paid by customer under the Service Agreement specified above during the Negotiated Rate Term shall be zero. Customer shall also pay the applicable Annual Charge Adjustment ("ACA"), other applicable surcharges, and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above.

Primary Receipt Point: M&R 80060, G-24 Lateral, Head of Lateral, Sandwich, MA

Primary Delivery Point: M&R 00829, Cape Cod Expansion - Sandwich MS, Sandwich, MA

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for Rate Schedule AFT-CL for the Algonquin Cape Cod Project.

FOOTNOTES:

1/This negotiated rate transaction does not deviate in any material respect from the form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designate herein.

FERC Docket: RP10-582-000

Fourth Revised Sheet No. 87 Fourth Revised Sheet No. 87

Superseding: Third Revised Sheet No. 87

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Excelerate Gas Marketing, Limited Partnership

Service Agreement: 510414

Term of Negotiated Rate: The 25-year term of this negotiated rate ("Negotiated Rate Term") commences on the date service commences under the Service Agreement specified above and continues through the end of the 25-year Primary Term of such Service Agreement.

Rate Schedule: AFT-CL (Northeast Gateway Lateral)

MDTQ: 800,000 dth/d

Reservation Rate: Customer shall pay a negotiated reservation rate per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term as follows:

Year 1 (Dec 2007 - June 2008)	\$3.49/dth
Year 1 (July 2008 - Nov 2008)	\$3.79/dth
Years 2 - 5	\$3.79/dth
Years 6 - 10	\$3.59/dth
Years 11 - 15	\$3.39/dth
Years 16 - 20	\$2.99/dth
Years 21 - 25	\$2.40/dth

Usage Rate: The usage rate to be paid by Customer under the Service Agreement specified above during the Negotiated Rate Term shall be zero. Customer shall also pay the applicable Annual Charge Adjustment ("ACA"), other applicable surcharges, and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above.

Primary Receipt Point: M&R 89847, NORTHEAST GATEWAY DEEPWATER PORT (DWP1), MA

Primary Delivery Point: M&R 88203, INTERCONNECTION BETWEEN NORTHEAST GATEWAY AND HUBLINE, MA

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for Rate Schedule AFT-CL for the Algonquin Northeast Gateway Lateral.

FOOTNOTES:

1/ This negotiated rate transaction does not deviate in any material respect from the form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designated herein.

FERC Docket: RP10-308-000

First Revised Sheet No. 88 First Revised Sheet No. 88

Superseding: Original Sheet No. 88

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Fore River Development, LLC Service Agreement: Contract No. 510008 2/

Term of Negotiated Rate: The term of this negotiated rate ("Negotiated Rate Term") commences on February 1, 2008 or such later date as approved by the Commission, and continues through the Primary Term of the above-designated Service Agreement, which date is November 30, 2023.

Rate Schedule: AFT-1(F-1/WS-1), Hubline Project

MDTQ: 140,000 dth/d

Primary Receipt Points: (1) Interconnection between the facilities of Algonquin and Maritimes & Northeast Pipeline, L.L.C., MDRO = 70,000 dth/d; (2) Interconnection between Algonquin's Q System and I-3 System, MDRO = 70,000 dth/d.

Primary Delivery Point: Algonquin Line I-9 tie-in with the Fore River Plant in Weymouth, MA.

Applicable Rates: 3/4/5/

- (a) For firm transportation from Primary Receipt Point No. 1 and secondary points of receipt, Customer shall pay the following negotiated rates: (i) a reservation charge rate of (1) \$4.101 per dekatherm per month, for the first 3 months of the Negotiated Rate Term, and (2) \$7.000 per dekatherm per month, for the remainder of the Negotiated Rate Term, for 70,000 dekatherms of Customer's MDTQ as specified in the above-designated Service Agreement; (ii) a usage charge rate equal to the then currently effective AFT-1(F-1/WS-1) usage rate per dekatherm of gas delivered for Customer's account under the above-designated Service Agreement; and (iii) other currently effective surcharges and fuel.
- (b) For firm transportation from Primary Receipt Point No. 2 to the Primary Point of Delivery, Customer shall pay the following negotiated rates: (i) a reservation charge rate of (1) \$0.371 per dekatherm per month, for the first 3 months of the Negotiated Rate Term, and (2) \$3.270 per dekatherm per month, for the remainder of the Negotiated Rate Term, for 70,000 dekatherms of Customer's MDTQ as specified in the above-designated Service Agreement, and (ii) other currently effective surcharges.
- (c) The negotiated rates set forth in (a) and (b) above apply only to transportation under the above-designated Service Agreement from the point(s) of receipt to the point(s) of delivery as expressly set forth in (a) and (b) above respectively. In the event that Customer utilizes points of receipt or delivery other than those specified in (a) and (b) above, then the maximum recourse reservation and commodity rates in effect for Hubline Service shall apply to Customer's entire MDTQ under the above-designated Service Agreement for the month in which such other points are utilized, except that, for the first 3 months of the Negotiated Rate Term, a reservation rate of \$5.5471 per dekatherm per month shall apply in lieu of the maximum recourse reservation rate.

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for Rate Schedule AFT-1(F-1/WS-1) plus the incremental reservation surcharge for the Hubline Project.

FOOTNOTES:

- 1/ This negotiated rate transaction does not deviate in any material respect from the form of service agreement set forth in Algonquin's FERC Gas Tariff.
- 2/ This negotiated rate agreement replaces in its entirety the negotiated rate letter agreement dated August 2, 2000, and associated with Contract No. 99025 (which has been replaced and superseded by Contract No. 510008), which was accepted by the Commission on November 7, 2003.

FERC Docket: RP00- 70-018

Original Sheet No. 88A Original Sheet No. 88A: Effective

STATEMENT OF NEGOTIATED RATES (Continued)

- 3/ The negotiated rates shall not apply to any other service or to capacity released pursuant to the capacity release provisions of Section 14 of Algonquin's FERC Gas Tariff.
- 4/ The intent of Algonquin and Customer is that Algonquin will charge the rates described above under its Commission-approved negotiated rate authority. In the event that Algonquin and Customer are unable to fulfill the intent of this agreement pursuant to the terms outlined above, or Algonquin and Customer are unable to fulfill any other term of this agreement or any term is deemed illegal or unenforceable, Algonquin and Customer shall negotiate an alternative mechanism that will produce the same economic result for both Algonquin and Customer.
- 5/ If, during the Negotiated Rate Term, Algonquin's effective maximum rates are being collected by Algonquin subject to refund, Algonquin shall have no refund obligation to Customer even if the final maximum rates are reduced to a level below the negotiated rates provided herein. In the event Customer releases its firm transportation rights under the above-designated Service Agreement Customer shall continue to be obligated to pay Algonquin for the difference, if any, by which the negotiated rate herein exceeds the release rate. In addition, in the event that Customer releases its firm transportation rights under the above-designated Service Agreement at a recourse rate in excess of the negotiated rate agreed to herein, Algonquin shall retain any and all credits for the difference between the release rate and the negotiated rate agreed to herein.

FERC Docket: RP10-609-000

Second Revised Sheet No. 89 Second Revised Sheet No. 89

Superseding: First Revised Sheet No. 89

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: KeySpan Gas East Corporation D/B/A/ National Grid

Service Agreement: Contract No. 510369

Negotiated Rate Term: The term of this negotiated rate ("Negotiated Rate Term") commences on the date service commences under the Service Agreement specified above and continues through the end of the 15 year Primary Term of such Service Agreement. Algonquin and Customer agree that such Service Agreement will be a ROFR Agreement pursuant to the terms of Algonquin's FERC Gas Tariff and will remain a ROFR Agreement following the end of the Negotiated Rate Term.

Rate Schedule: AFT-1 (Docket No. CP06-76, Ramapo Project)

MDTO:

For the Period: Maximum Daily Transportation Quantity

Nov 1, 2008 - Oct 31, 2009 171,500 dth/d Nov 1, 2009 - end of Primary Term 196,000 dth/d

Reservation Rate: Customer shall pay a rate of \$10.71 per Dth, per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term.

Usage Rate: The usage rate to be paid by Customer under the Service Agreement specified above during the Negotiated Rate Term shall be zero. Customer shall also pay the applicable Annual Charge Adjustment ("ACA") and other FERC-prescribed surcharges of general applicability, and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above. The usage rate set forth herein shall apply to any replacement customer pursuant to a capacity release in accordance with Section 14 of the General Terms and Conditions of Algonquin's FERC Gas Tariff.

Primary Receipt Point: M&R 00202, Ramapo, NY

Primary Delivery Point(s):

For the Period: Delivery Point Maximum Daily
Delivery Obligation

Nov 1, 2008 - Oct 31, 2009 Lateral C System Mainline, CT 23,000 dth/d M&R 00851, Brookfield Delivery, CT 148,500 dth/d

Nov 1, 2009 - end of Primary Term M&R 00851, Brookfield Delivery, CT 196,000 dth/d

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for Rate Schedule AFT-1 for the Ramapo Project.

FOOTNOTES:

1/ This negotiated rate transaction does not deviate in any material respect from the applicable form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designated herein, and any secondary receipt and delivery points available under Rate Schedule AFT-1.

FERC Docket: RP10-308-000

First Revised Sheet No. 89A First Revised Sheet No. 89A

Superseding: Original Sheet No. 89A

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Consolidated Edison Company of New York, Inc.

Service Agreement: Contract No. 510371

Negotiated Rate Term: The term of this negotiated rate ("Negotiated Rate Term") commences on the date service commences under the Service Agreement specified above and continues through the end of the Primary Term of such Service Agreement, which Primary Term will extend for a period of either 10 years or 15 years, based upon Customer's election pursuant to its one-time option to extend the Primary Term by five years. Customer's election to extend the Primary Term to 15 years is exercisable no later than five years prior to the end of the initial 10 year term. In the event Customer exercises its one-time option to extend the Primary Term of the Service Agreement specified above for a period of five years, then Algonquin and Customer agree that such Service Agreement will be a ROFR Agreement pursuant to the terms of Algonquin's FERC Gas Tariff and will remain a ROFR Agreement following the end of the Negotiated Rate Term.

Rate Schedule: AFT-1 (Docket No. CP06-76, Ramapo Project)

MDTQ: 125,000 dth/d

Reservation Rate: Customer shall pay a rate of \$11.38 per Dth, per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term. In the event Customer exercises its one-time option to extend the Primary Term of such Service Agreement for a period of five years, then Customer shall pay a negotiated reservation rate of \$10.37 per Dth, per month of Customer's MDTQ under such Service Agreement for the extended five year period of the Primary Term.

Usage Rate: The usage rate to be paid by Customer under the Service Agreement specified above during the Negotiated Rate Term and/or any extended term, shall be zero. Customer shall also pay the applicable Annual Charge Adjustment ("ACA") and other FERC-prescribed surcharges of general applicability, and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above.

Primary Receipt Point: M&R 00202, Ramapo, NY

Primary Delivery Point: M&R 00851, Brookfield Delivery, CT

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for Rate Schedule AFT-1 for the Ramapo Project.

FOOTNOTES:

1/ Except for Customer's one-time option to extend the term described above, this negotiated rate transaction does not deviate in any material respect from the applicable form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designated herein, and any secondary receipt and delivery points available under Rate Schedule AFT-1.

FERC Docket: RP10-308-000

Third Revised Sheet No. 90 Third Revised Sheet No. 90

Superseding: Second Revised Sheet No. 90

STATEMENT OF NEGOTIATED RATES 1/ 2/

Customer Name: NSTAR Gas Company

Service Agreement: 510482

Term of Negotiated Rate: The term of this negotiated rate ("Negotiated Rate Term") commences on the date service commences under the Service Agreement specified above and continues through the end of the twenty-year term of such Service Agreement.

Rate Schedule: AFT-CL (J-2 Facility)

MDTQ: 140,000 dth/d

Reservation Rate: Customer shall pay a negotiated reservation rate of \$3.7544 per Dth, per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term. To the extent necessary to reflect final, actual costs to construct the J-2 Loop, Algonquin will file a one-time adjustment to this negotiated rate, with the updated negotiated rate to become effective as of the commencement of the Negotiated Rate Term. Following the issuance of a Federal Energy Regulatory Commission ("Commission") order accepting the revised negotiated rate agreement, Algonquin will adjust its invoice to NSTAR, with the appropriate offsets or surcharges as necessary and with interest as applicable calculated at the FERC rate for refunds, to reflect this effective date for the updated negotiated rate.

Usage Rate: The usage rate to be paid by Customer under the Service Agreement specified above during the Negotiated Rate Term shall be zero. Customer shall also pay the applicable Annual Charge Adjustment ("ACA"), lost and unaccounted for gas, and other applicable surcharges for service under the Service Agreement specified above.

Primary Receipt Point: M&R 80094, J-2 Facility

Primary Delivery Point: M&R 00826, Mystic Street Station, Middlesex County, MA

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Statement of Rates for firm transportation service on the J-2 Facility under Rate Schedule AFT-CL.

FOOTNOTES:

- 1/ This negotiated rate transaction does not deviate in any material respect from the applicable form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designated herein. Algonquin and Customer, by agreeing to this negotiated rate, acknowledge that the otherwise generally applicable maximum Recourse Rate(s) shall not apply or be available to Customer for service under the Service Agreement during the Negotiated Rate Term, notwithstanding any adjustment to such generally applicable maximum Recourse Rate(s) which may become effective during the Negotiated Rate Term. Customer shall receive revenue credits in the amount of fifty percent of revenues actually received by Algonquin under Rate Schedule AIT-2 (J-2) that are attributable to commodity charges and authorized overrun charges but not including imbalance resolution charges, scheduling penalties, unauthorized contract overrun penalties, ACA, revenues attributable to Section 8 of Rate Schedule PAL, or any other such charges or surcharges.
- 2/ This negotiated rate transaction is being implemented pursuant to the Stipulation and Agreement ("Settlement") between Algonquin and Customer, dated October 16, 2007, and the order of the Commission on the Settlement, dated February 14, 2008, 122 FERC ¶ 61,131 (2008). This negotiated rate tariff sheet reflects the terms of the Settlement effective as of the commencement of the Negotiated Rate Term.

FERC Docket: RP10-228-000

Original Sheet No. 90A Original Sheet No. 90A

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Kleen Energy Systems, LLC

Service Agreement: 510505

Term of Negotiated Rate: The fifteen (15) year term of this negotiated rate ("Negotiated Rate Term") commences on the date service commences under the Service Agreement specified above and continues through the end of the 15-year Primary Term of such Service Agreement.

Rate Schedule: AFT-CL (Middletown Lateral)

MDTQ: 101,000 dth/d

Reservation Rate: Customer shall pay a negotiated reservation rate per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term as follows:

Years 1 - 15 \$1.8020/dth

Usage Rate: The usage rate to be paid by Customer under the Service Agreement specified above during the Negotiated Rate Term shall be \$0.00. Customer shall also pay the applicable Annual Charge Adjustment ("ACA"), other applicable surcharges, and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above.

Primary Receipt Point: M&R 80001 Primary Delivery Point: M&R 80100

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Rate Tariff Sheet for Rate Schedule AFT-CL for the Algonquin Middletown Lateral.

FOOTNOTES:

1/This negotiated rate transaction does not deviate in any material respect from the form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designated herein.

FERC Docket: RP10-228-000

Original Sheet No. 90B Original Sheet No. 90B

STATEMENT OF NEGOTIATED RATES 1/

Customer Name: Kleen Energy Systems, LLC

Service Agreement: 510506

Term of Negotiated Rate: The fifteen (15) year term of this negotiated rate ("Negotiated Rate Term") commences on the date service commences under the Service Agreement specified above and continues through the end of the 15-year Primary Term of such Service Agreement.

Rate Schedule: AFT-CL (Kleen Lateral)

MDTQ: 131,000 dth/d

Reservation Rate: Customer shall pay a negotiated reservation rate per month of Customer's MDTQ under the Service Agreement specified above during the Negotiated Rate Term as follows:

Years 1 - 15 \$1.2176/dth

Usage Rate: The usage rate to be paid by Customer under the Service Agreement specified above during the Negotiated Rate Term shall be \$0.00. Customer shall also pay the applicable Annual Charge Adjustment ("ACA"), other applicable surcharges, and Fuel Reimbursement Quantity ("FRQ") charges for service under the Service Agreement specified above.

Primary Receipt Point: M&R 80101 Primary Delivery Point: M&R 00833

Recourse Rate(s): The Recourse Rate(s) are the maximum rate(s) stated on Algonquin's currently effective Rate Tariff Sheet for Rate Schedule AFT-CL for the Algonquin Kleen Lateral.

FOOTNOTES:

1/This negotiated rate transaction does not deviate in any material respect from the form of service agreement set forth in Algonquin's FERC Gas Tariff. This negotiated rate shall apply only to service under the Service Agreement specified above, up to Customer's specified MDTQ, using the receipt and delivery points designated herein.

FERC Docket: RP05-208-000

First Revised Sheet No. 91 First Revised Sheet No. 91 : Effective Superseding: Original Sheet No. 91

Sheet No. 91 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 92 First Revised Sheet No. 92 : Effective Superseding: Original Sheet No. 92

Sheet No. 92 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 93 First Revised Sheet No. 93 : Effective Superseding: Original Sheet No. 93

Sheet No. 93 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 94 First Revised Sheet No. 94 : Effective Superseding: Original Sheet No. 94

Sheet No. 94 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 95 First Revised Sheet No. 95 : Effective Superseding: Original Sheet No. 95

Sheet No. 95 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 96 First Revised Sheet No. 96 : Effective Superseding: Original Sheet No. 96

Sheet No. 96 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 97 First Revised Sheet No. 97 : Effective Superseding: Original Sheet No. 97

Sheet No. 97 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 98 First Revised Sheet No. 98 : Effective Superseding: Original Sheet No. 98

Sheet No. 98 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 99 First Revised Sheet No. 99 : Effective Superseding: Original Sheet No. 99

Sheet No. 99 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 100 First Revised Sheet No. 100 : Effective Superseding: Original Sheet No. 100 $\,$

Sheet No. 100 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 101 First Revised Sheet No. 101 : Effective Superseding: Original Sheet No. 101

Sheet No. 101 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 102 First Revised Sheet No. 102 : Effective Superseding: Original Sheet No. 102

Sheet No. 102 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 103 First Revised Sheet No. 103 : Effective Superseding: Original Sheet No. 103

Sheet No. 103 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 104 First Revised Sheet No. 104 : Effective Superseding: Original Sheet No. 104

Sheet No. 104 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 105 First Revised Sheet No. 105 : Effective Superseding: Original Sheet No. 105

Sheet No. 105 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 106 First Revised Sheet No. 106 : Effective Superseding: Original Sheet No. 106

Sheet No. 106 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 107 First Revised Sheet No. 107 : Effective Superseding: Original Sheet No. 107

Sheet No. 107 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 108 First Revised Sheet No. 108 : Effective Superseding: Original Sheet No. 108

Sheet No. 108 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 109 First Revised Sheet No. 109 : Effective Superseding: Original Sheet No. 109

Sheet No. 109 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 110 First Revised Sheet No. 110 : Effective Superseding: Original Sheet No. 110

Sheet No. 110 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 111 First Revised Sheet No. 111 : Effective Superseding: Original Sheet No. 111

Sheet No. 111 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 112 First Revised Sheet No. 112 : Effective Superseding: Original Sheet No. 112

Sheet No. 112 was previously issued, but is now reserved for future use.

FERC Docket: RP04-361-000

Sheet Nos. 113 - 199 Sheet Nos. 113 - 199 : Effective

Sheet Nos. 113 - 199 are being reserved for future use

FERC Docket: RP06- 75-000

First Revised Sheet No. 200 First Revised Sheet No. 200: Effective Superseding: Original Sheet No. 200

RATE SCHEDULES

INDEX

Rate Schedule	Designation and Title	Sheet No.	Rate Sheet No.
AFT-1	Firm Transportation Service	201	22, 24
AFT-1	Capacity Release		25
AFT-E	No-Notice Firm Transportation Service	221	27
AFT-E	Capacity Release		28
AFT-1S	Small Customer Firm Transportation Service	241	29
AFT-1S	Capacity Release		30
AFT-ES	Small Customer No-Notice Firm Transportation Service	261	31
AFT-ES	Capacity Release		32
AIT-1	Interruptible Transportation Service	281	33
T-1	Transportation Service (New York and New Jersey)	301	34
AFT-4	Firm Transportation Service	321	35
AFT-CL	Firm Transportation Service	331	36
AFT-CL	Capacity Release		38
PAL	Park and Loan Service	351	39
TTT	Title Transfer Tracking Service	361	41
AIT-2	Interruptible Transportation Service	371	42

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 201 Fifth Revised Sheet No. 201

Superseding: Fourth Revised Sheet No. 201

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

1. AVATLABILITY

This rate schedule is available for firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer has made a valid request for firm transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part;
- Sufficient firm capacity is available to effectuate such transportation without any construction of facilities or other investment by Algonquin, or Algonquin has waived this requirement in writing;
- c. The Primary Point(s) of Receipt and Primary Point(s) of Delivery requested by Customer are acceptable to Algonquin from the viewpoint of adequacy of Algonquin's existing facilities to receive and transport Customer's gas with Algonquin's existing firm service taking precedence; and
- d. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- e. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline (M&R 80095), and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available under this Rate Schedule AFT-1.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 201 Fourth Revised Sheet No. 201

Superseding: Third Revised Sheet No. 201

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

AVATTABITATY

This rate schedule is available for firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer has made a valid request for firm transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part;
- Sufficient firm capacity is available to effectuate such transportation without any construction of facilities or other investment by Algonquin, or Algonquin has waived this requirement in writing;
- The Primary Point(s) of Receipt and Primary Point(s) of Delivery requested by Customer are acceptable to Algonquin from the viewpoint of adequacy of Algonquin's existing facilities to receive and transport Customer's gas with Algonquin's existing firm service taking precedence; and
- d. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- e. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline (M&R 80095) are available under this Rate Schedule AFT-1.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined

FERC Docket: RP10-77-000

First Revised Sheet No. 202 First Revised Sheet No. 202

Superseding: Original Sheet No. 202

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

APPLICABILITY AND CHARACTER OF SERVICE (continued)

in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement, except as provided in Section 2.6 herein below, plus amounts reflecting the Fuel Reimbursement Quantity; provided, however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any Day a quantity of gas in excess of the applicable Maximum Daily Receipt Obligation ("MDRO") plus any applicable Fuel Reimbursement Quantity, provided that, if more than one Customer requests receipts in excess of its MDRO at a Point of Receipt, and the sum of all such requests exceeds the available capacity at such Point of Receipt, Algonquin shall apportion such receipts in excess of MDRO among such Customers pro rata according to the Customers' firm MDROs at the relevant Point of Receipt.

2.2 Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on Algonquin's system as are specified in an executed AFT-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"); provided, however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ, and on any Day a quantity of gas in excess of the applicable Maximum Daily Delivery Obligation ("MDDO"), provided that, if more than one Customer requests deliveries in excess of its MDDO at a Point of Delivery, and the sum of all such requests exceeds the available capacity at such Point of Delivery, Algonquin shall apportion such deliveries in excess of MDDO among such Customers pro rata according to the Customers' firm MDDOs at the relevant Point of Delivery.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 203 Fourth Revised Sheet No. 203

Superseding: Third Revised Sheet No. 203

RATE SCHEDULE AFT-1
FIRM TRANSPORTATION SERVICE

2. APPLICABILITY AND CHARACTER OF SERVICE (continued)

- Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.
- 2.4 Algonquin shall not be obligated to add any facilities or expand the capacity of Algonquin's pipeline system in any manner in order to provide transportation service to Customer pursuant to this rate schedule; provided, however, Algonquin may, at its option, and with Customer's consent, add facilities or expand capacity to provide such transportation service, subject to Section 42.
- 2.5 Unless otherwise specified in the applicable service agreement, services hereunder shall be available on any day of the year, subject to Customer's MATQ and MDTQ limitations.
- With respect to existing service agreements resulting from the conversion pursuant to 2.6 Order No. 636 of sales and storage service under former Rate Schedules F-1, WS-1, STB and SS-3, subject to Algonquin's firm service obligations from primary points of receipt to primary points of delivery Algonquin shall provide service above Customer's MATQ up to but not in excess of the highest MDTQ specified in the executed service agreement multiplied by 365 (366 for a leap year) and, on any Day, Algonquin shall provide service above Customer's MDTQ up to but not in excess of the highest MDTQ specified in the executed service agreement; provided, however, that such additional quantities shall be provided with the same priority for purposes of Scheduling and Curtailment accorded to service from Secondary Points of Receipt and Secondary Points of Delivery, and provided further, that for purposes of Curtailment Customer's maximum daily entitlement shall be the highest MDTQ specified in Customer's executed service agreement during the period of Curtailment. Such additional quantities shall be subject to fuel, the applicable commodity rate and any applicable commodity surcharges.

3. RATE

- 3.1 Unit Rates. The applicable maximum and minimum unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AFT-1 of this tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 33 and Section 34 of the General Terms and Conditions of this tariff. The applicable unit rates to be charged on any Day by Algonquin for gas delivered to Customer shall not be in excess of the maximum unit rate nor less than the minimum unit rate.
- 3.2 Monthly Bill for Forwardhaul Rates. For Customers executing an AFT-1 Service Agreement for transportation which is not solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-1 Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable service agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 204 Fourth Revised Sheet No. 204

Superseding: Third Revised Sheet No. 204

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

- 3. RATE (continued)
 - (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in Customer's executed AFT-1 Service Agreement; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions: less
 - (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 3.3 Monthly Bill for Backhaul Rates. For Customers executing an AFT-1 Service Agreement for transportation which is solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-1 Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable Service Agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:

FERC Docket: RP10-77-000

Second Revised Sheet No. 205 Second Revised Sheet No. 205

Superseding: First Revised Sheet No. 205

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

3. RATE (continued)

- (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in Customer's executed AFT-1 Service Agreement; plus
- (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
- (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
- (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
- (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
- (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less
- (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.

FERC Docket: RP10-77-000

First Revised Sheet No. 206 First Revised Sheet No. 206

Superseding: Original Sheet No. 206

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES

- 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.
- 4.2 Delivery of Gas. Based upon the daily quantity scheduled, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions of this tariff. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.
- 4.3 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.

FERC Docket: RP04-361-000

Original Sheet No. 207 Original Sheet No. 207: Effective

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

5. OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

- 5.1 Customer shall make all necessary arrangements with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 Algonquin shall schedule receipts at a Secondary Point of Receipt or deliveries at a Secondary Point of Delivery pursuant to the provisions of Sections 48.2 and 48.3 of the General Terms and Conditions.
- 5.3 To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. Prior to any reduction or interruption in service due to the failure of the upstream entity to deliver gas on behalf of Customer, Algonquin shall provide notice in a time and manner that is reasonable under then existing conditions. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
- 5.4 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt at an hourly rate of 1/24th of the scheduled daily quantity. The daily quantities of natural gas transported shall be accepted at the Point(s) of Delivery at a substantially constant hourly rate or, in the case of service agreements resulting from the conversion pursuant to Order No. 636 of sales entitlements under former Rate Schedules F-1 and WS-1, at a rate no greater than .06 multiplied by the scheduled daily quantity.

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 208 Sixth Revised Sheet No. 208

Superseding: Fifth Revised Sheet No. 208

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

- POINT(S) OF RECEIPT AND DELIVERY
 - (A) Primary Points of Receipt: The Primary Point(s) of Receipt at which Algonquin shall receive gas for transportation under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Receipt the MDRO and receipt pressure obligations. Such exhibit by mutual written agreement may be superseded by a new exhibit which may add or delete specific points or make other changes thereto that the parties deem appropriate. Algonquin shall not accept any proposed Primary Point(s) of Receipt, or quantity at any Primary Point(s) of Receipt, or change in quantities among Primary Point(s) of Receipt if (a) the resulting aggregate MDROs at all of Customer's Primary Point(s) of Receipt would exceed Customer's MDTQ, or (b) in doing so, in Algonquin's reasonable judgment, Algonquin would impair its ability to satisfy its existing firm obligations to receive gas pursuant to other firm service agreements under which such Point(s) of Receipt are Primary Points of Receipt and to purchase and receive its Company Use Gas at maximum deliverability levels, as such Company Use Gas arrangements exist under agreements effective at the date of Customer's request or reasonably expected by Algonquin to be effective within six months of the request.
 - A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Receipt in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Receipt to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Receipt that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Receipt, shall revert to the Releasing Customer, and any Primary Points of Receipt granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections 6.2 between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Secondary Points of Receipt under this Rate Schedule AFT-1. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 208 Fourth Revised Sheet No. 208

Superseding: Third Revised Sheet No. 208

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

- POINT(S) OF RECEIPT AND DELIVERY
 - (A) Primary Points of Receipt: The Primary Point(s) of Receipt at which Algonquin shall receive gas for transportation under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Receipt the MDRO and receipt pressure obligations. Such exhibit by mutual written agreement may be superseded by a new exhibit which may add or delete specific points or make other changes thereto that the parties deem appropriate. Algonquin shall not accept any proposed Primary Point(s) of Receipt, or quantity at any Primary Point(s) of Receipt, or change in quantities among Primary Point(s) of Receipt if (a) the resulting aggregate MDROs at all of Customer's Primary Point(s) of Receipt would exceed Customer's MDTQ, or (b) in doing so, in Algonquin's reasonable judgment, Algonquin would impair its ability to satisfy its existing firm obligations to receive gas pursuant to other firm service agreements under which such Point(s) of Receipt are Primary Points of Receipt and to purchase and receive its Company Use Gas at maximum deliverability levels, as such Company Use Gas arrangements exist under agreements effective at the date of Customer's request or reasonably expected by Algonquin to be effective within six months of the request.
 - A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Receipt in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Receipt to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Receipt that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Receipt, shall revert to the Releasing Customer, and any Primary Points of Receipt granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections 6.2 between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline, are available as Secondary Points of Receipt under this Rate Schedule AFT-1. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 209 Sixth Revised Sheet No. 209

Superseding: Fifth Revised Sheet No. 209

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

POINT(S) OF RECEIPT AND DELIVERY (continued)

- 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Primary Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
- Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections 6.4 between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the ${\ J-2\ }$ Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Secondary Points of Delivery under this Rate Schedule AFT-1. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 209 Fourth Revised Sheet No. 209

Superseding: Third Revised Sheet No. 209

RATE SCHEDULE AFT-1 FIRM TRANSPORTATION SERVICE

6. POINT(S) OF RECEIPT AND DELIVERY (continued)

- (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Primary Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - (B) A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
- Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Secondary Points of Delivery under this Rate Schedule AFT-1. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 210 - 220 Sheet Nos. 210 - 220 : Effective

Sheet Nos. 210 - 220 are being reserved for future use

FERC Docket: RP04-361-000

Original Sheet No. 221 Original Sheet No. 221: Effective

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

AVATTABITATY

This rate schedule is available for firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer has made a valid request for firm transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part;
- Sufficient firm capacity is available to effectuate such transportation without any construction of facilities or other investment by Algonquin, or Algonquin has waived this requirement in writing;
- The Primary Point(s) of Receipt and Primary Point(s) of Delivery requested by Customer are acceptable to Algonquin from the viewpoint of adequacy of Algonquin's existing facilities to receive and transport Customer's gas with Algonquin's existing firm service taking precedence;
- d. Customer has contracted with one or more operators of upstream facilities that interconnect with the Algonquin system (the "Upstream Entity(ies)"), for a sufficient quantity of no-notice transportation services that encompass the right on the part of Customer to increase or decrease its receipts from the Upstream Entity(ies) into designated Points of Receipt on Algonquin's system with no advance notice, up to Customer's MDTQ under this Rate Schedule AFT-E plus fuel as provided for in Section 32 of the General Terms and Conditions of this tariff in the case of increases in receipts, or down to zero in the case of decreases in receipts, without regard to any quantities that Customer previously scheduled on the Upstream Entities, and with no requirement that Customer adhere to a constant rate of flow on the Upstream Entities (the "Upstream Arrangements").

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 222 Sixth Revised Sheet No. 222

Superseding: Fifth Revised Sheet No. 222

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

1. AVAILABILITY (continued)

- e. Customer has named a contact party (the "Notice Agent") for the receipt of orders issued in accordance with Section 5 ("Section 5 Orders") of this rate schedule;
- f. Customer has authorized Algonquin to exercise Customer's nomination and scheduling rights under the Upstream Arrangements, if necessary by executing agency instruments satisfactory to the Upstream Entity, to the extent required to permit service under this rate schedule; and
- g. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- h. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095), and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available under this Rate Schedule AFT-E.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-E Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement, except as provided in Section 2.7 herein below, plus amounts reflecting the Fuel Reimbursement Quantity; provided however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any Day a quantity of gas in excess of the applicable Maximum

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 222 Fourth Revised Sheet No. 222

Superseding: Third Revised Sheet No. 222

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

1. AVAILABILITY (continued)

- e. Customer has named a contact party (the "Notice Agent") for the receipt of orders issued in accordance with Section 5 ("Section 5 Orders") of this rate schedule;
- f. Customer has authorized Algonquin to exercise Customer's nomination and scheduling rights under the Upstream Arrangements, if necessary by executing agency instruments satisfactory to the Upstream Entity, to the extent required to permit service under this rate schedule; and
- g. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- h. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095) are available under this Rate Schedule AFT-E.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-E Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement, except as provided in Section 2.7 herein below, plus amounts reflecting the Fuel Reimbursement Quantity; provided however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any day a quantity of gas in excess of the applicable Maximum

FERC Docket: RP10-77-000

First Revised Sheet No. 223 First Revised Sheet No. 223

Superseding: Original Sheet No. 223

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

2. APPLICABILITY AND CHARACTER OF SERVICE (continued)

Daily Receipt Obligation ("MDRO") plus any applicable Fuel Reimbursement Quantity, provided that, if more than one Customer requests receipts in excess of its MDRO at a Point of Receipt, and the sum of all such requests exceeds the available capacity at such Point of Receipt, Algonquin shall apportion such receipts in excess of MDRO among such Customers pro rata according to the Customers' firm MDROs at the relevant Point of Receipt.

- 2.2 Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on Algonquin's system as are specified in an executed AFT-E Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"); provided however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ, and on any Day a quantity of gas in excess of the applicable Maximum Daily Delivery Obligation ("MDDO"), provided that, if more than one Customer requests deliveries in excess of its MDDO at a Point of Delivery, and the sum of all such requests exceeds the available capacity at such Point of Delivery, Algonquin shall apportion such deliveries in excess of MDDO among such Customers pro rata according to the Customers' firm MDDOs at the relevant Point of Delivery.
- 2.3 Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 224 Fourth Revised Sheet No. 224

Superseding: Third Revised Sheet No. 224

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

- APPLICABILITY AND CHARACTER OF SERVICE (continued)
 - 2.4 Algonquin shall not be obligated to add any facilities or expand the capacity of Algonquin's pipeline system in any manner in order to provide transportation service to Customer pursuant to this rate schedule; provided, however, Algonquin may, at its option, and with Customer's consent, add facilities or expand capacity to provide such transportation service, subject to Section 42.
 - 2.5 Subject to the conditions specified in Section 5 of this Rate Schedule, service hereunder shall encompass the right on the part of Customer to increase deliveries at the Point(s) of Delivery up to the MDTQ and to decrease deliveries at the Point(s) of Delivery without advance notice to Algonquin and without Customer's previously having provided for a concurrent increase or decrease in receipts at the Point(s) of Receipt.
 - 2.6 Unless otherwise specified in the applicable service agreement, service hereunder shall be available on any day of the year, subject to Customer's MATQ and MDTQ limitations.
 - 2.7 With respect to existing service agreements resulting from the conversion pursuant to Order No. 636 of sales and storage service under former Rate Schedules F-1, WS-1, STB and SS-3, subject to Algonquin's firm service obligations from primary points of receipt to primary points of delivery Algonquin shall provide service above Customer's MATQ up to but not in excess of the highest MDTQ specified in the executed service agreement multiplied by 365 (366 for a leap year) and, on any Day, Algonquin shall provide service above Customer's MDTQ up to but not in excess of the highest MDTQ specified in the executed service agreement; provided, however, that such additional quantities shall be provided with the same priority for purposes of Scheduling and Curtailment accorded to service from Secondary Points of Receipt and Secondary Points of Delivery, and provided further, that for purposes of Curtailment Customer's maximum daily entitlement shall be the highest MDTQ specified in Customer's executed service agreement during the period of Curtailment. Such additional quantities shall be subject to fuel, the applicable commodity rate and any applicable commodity surcharges.

RATE

- 3.1 Unit Rates. The applicable maximum and minimum unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AFT-E of this tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 33 and Section 34 of the General Terms and Conditions of this tariff. The applicable unit rates to be charged on any Day by Algonquin for gas delivered to Customer shall not be in excess of the maximum unit rate nor less than the minimum unit rate.
- 3.2 Monthly Bill for Forwardhaul Rates. For Customers executing an AFT-E Service Agreement for transportation which is not solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-E Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable service agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 225 Fourth Revised Sheet No. 225

Superseding: Third Revised Sheet No. 225

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

- 3. RATE (continued)
 - (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in Customer's executed AFT-E Service Agreement; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions: less
 - (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 3.3 Monthly Bill for Backhaul Rates. For Customers executing an AFT-E Service Agreement for transportation which is solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-E Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable Service Agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:

FERC Docket: RP10-77-000

Second Revised Sheet No. 226 Second Revised Sheet No. 226

Superseding: First Revised Sheet No. 226

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

3. RATE (continued)

- (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in Customer's executed AFT-E Service Agreement; plus
- (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
- (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
- (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
- (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
- (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less
- (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.

FERC Docket: RP10-77-000

First Revised Sheet No. 227 First Revised Sheet No. 227

Superseding: Original Sheet No. 227

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES

- 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.
- 4.2 Delivery of Gas. Based upon the daily quantity scheduled and in accordance with Section 5 herein, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions of this tariff. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.

FERC Docket: RP10-77-000

First Revised Sheet No. 228 First Revised Sheet No. 228

Superseding: Original Sheet No. 228

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES (continued)
 - 4.3 No-Notice Service. Notwithstanding the quantities nominated by Customer and scheduled by Algonquin hereunder, Customer shall be entitled to increase its deliveries up to the MDDO at any point, up to the MHTQ during any Hour, and up to the MDTQ, or to decrease its deliveries. Provided that all of the operational conditions specified in Section 5 of this rate schedule (the "Section 5 Conditions") are met, Algonquin shall consent to such increase or decrease in deliveries, thereby nullifying any daily scheduling or hourly scheduling penalty that would otherwise be applicable pursuant to Section 23 of the General Terms and Conditions. Furthermore, if the Section 5 Conditions are met, Algonquin will forbear from taking action pursuant to Section 26 of the General Terms and Conditions to reduce deliveries to Customer. If any of the Section 5 Conditions are not met, Algonquin shall not be required to receive or deliver gas in amounts other than the transportation quantities scheduled by Customer, and variations between scheduled quantities and actual deliveries to Customer's meter shall be subject to the assessment of penalties pursuant to Section 23 of the General Terms and Conditions.
 - 4.4 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.
- 5. OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

FERC Docket: RP04-361-000

Original Sheet No. 229 Original Sheet No. 229: Effective

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

- OTHER OPERATING CONDITIONS (continued)
 - 5.1 Customer shall make all necessary arrangements, including the Upstream Arrangements, with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
 - 5.2 Algonquin shall schedule receipts at a Secondary Point of Receipt or deliveries at a Secondary Point of Delivery pursuant to the provisions of Sections 48.2 and 48.3 of the General Terms and Conditions.
 - 5.3 To the extent that any Upstream Entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. Prior to any reduction or interruption in service due to the failure of the Upstream Entity to deliver gas on behalf of Customer, Algonquin shall provide notice in a time and manner that is reasonable under then existing conditions. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
 - 5.4 The Upstream Arrangements must be in full force and effect; Customer shall not be in default of the Upstream Arrangements; and Customer's rights under the Upstream Arrangements shall be enforceable by Customer or its agents without impairment due to curtailment, force majeure or other reasons.

FERC Docket: RP04-361-000

Original Sheet No. 230 original Sheet No. 230 : Effective

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

5. OTHER OPERATING CONDITIONS (continued)

- 5.5 The Upstream Entity(ies) shall recognize Algonquin's rights as Customer's agent to issue orders ("Section 5 Orders") on Customer's behalf nominating or directing for Customer's account no-notice service as encompassed within the Upstream Arrangements at such levels and at such times as Algonquin determines in its reasonable discretion is required by the Algonquin system to meet the needs of Customer under this rate schedule.
- 5.6 All Upstream Entities involved in all Upstream Arrangements with Customer served under this rate schedule shall deliver to the various points of interconnection between Algonquin's facilities and the facilities of the various Upstream Entities such quantities of gas at such times as Algonquin determines in its reasonable discretion are required by the Algonquin system to meet the needs of Customer under this rate schedule.
- 5.7 Customer's Notice Agent must have complete authority to call upon supplies for Customer's account at such times, in such quantities, and at such locations as Algonquin deems necessary, in its reasonable discretion, to enable Algonquin to provide service as contemplated for Customer under this rate schedule.
- 5.8 All Notice Agents named by Customer under this rate schedule shall execute Section 5 Orders at such locations and at such times as Algonquin determines in its reasonable discretion are necessary to enable Algonquin to provide service as contemplated under Rate Schedules AFT-E and AFT-ES.
- 5.9 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt at an hourly rate of 1/24th of the scheduled daily quantity. The daily quantities of natural gas transported shall be accepted at the Point(s) of Delivery at a substantially constant hourly rate or, in the case of service agreements resulting from the conversion pursuant to Order No. 636 of sales entitlements under former Rate Schedules F-1 and WS-1, at an hourly rate no greater than .06 multiplied by the scheduled daily quantity.

FERC Docket: RP10-77-000

First Revised Sheet No. 231 First Revised Sheet No. 231

Superseding: Original Sheet No. 231

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

5. OTHER OPERATING CONDITIONS (continued)

- 5.10 In the event Customer has multiple upstream no-notice services pursuant to more than one contract, from more than one pipeline, or under more than one rate schedule, Customer shall provide Algonquin a predetermined order of preference for these upstream services. Algonquin shall follow this order of preference in scheduling Customer's upstream no-notice service. Customer may, upon twenty-four hours' written notice to Algonquin, modify this specification of scheduling preferences.
- 5.11 In the event Algonquin issues a Section 5 Order on Customer's behalf, Algonquin shall provide notice as soon as possible to Customer and Customer's Notice Agent by telephone or facsimile and by posting on the Internet Web Site that the Section 5 Order has been issued.
- 6. POINT(S) OF RECEIPT AND DELIVERY
 - Primary Points of Receipt: The Primary Point(s) of Receipt at which 6.1 (A) Algonquin shall receive gas for transportation under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Receipt the MDRO and receipt pressure obligations. Such exhibit by mutual written agreement may be superseded by a new exhibit which may add or delete specific points or make other changes thereto that the parties deem appropriate. Algonquin shall not accept any proposed Primary Point(s) of Receipt, or quantity at any Primary Point(s) of Receipt, or change in quantities among Primary Point(s) of Receipt if (a) the resulting aggregate MDROs at all of Customer's Primary Point(s) of Receipt would exceed Customer's MDTQ, or (b) in doing so, in Algonquin's reasonable judgment, Algonquin would impair its ability to satisfy its existing firm obligations to receive gas pursuant to other firm service agreements under which such Point(s) of Receipt are Primary Points of Receipt and to purchase and receive its Company Use Gas at maximum deliverability levels, as such Company Use Gas arrangements exist under agreements effective at the date of Customer's request or reasonably expected by Algonquin to be effective within six months of the request.
 - A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Receipt in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Receipt to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Receipt that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Receipt, shall revert to the Releasing Customer, and any Primary Points of Receipt granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 232 Sixth Revised Sheet No. 232

Superseding: Fifth Revised Sheet No. 232

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral, are available as Secondary Points of Receipt under this Rate Schedule AFT-E. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.
 - 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Primary Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - (B) A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - 6.4 Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral,

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 232 Fourth Revised Sheet No. 232

Superseding: Third Revised Sheet No. 232

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Secondary Points of Receipt under this Rate Schedule AFT-E. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.
 - 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Primary Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - A Replacement Customer that acquired capacity pursuant to the capacity (B) release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 233 Sixth Revised Sheet No. 233

Superseding: Fifth Revised Sheet No. 233

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Secondary Points of Delivery under this Rate Schedule AFT-E. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. INDEMNIFICATION

- 7.1 Customer acknowledges that as a result of Section 5 Orders issued to Customer's Upstream Entity or Notice Agent hereunder, Customer may incur imbalances between receipts and deliveries that will be cashed out at Month's end pursuant to Section 25 of the General Terms and Conditions of this tariff. Algonquin shall provide Customer with notice of an imbalance in a timely manner. Customer shall hold Algonquin harmless and indemnify Algonquin against any claim by Customer or otherwise arising out of the incurrence of imbalances as a result of Algonquin's issuance of Section 5 Orders.
- 7.2 Customer acknowledges that the exercise by Algonquin of its agency authority with respect to the Upstream Arrangements may cause the Upstream Entity to assess imbalance cash-out charges and transportation usage, storage injection, storage withdrawal or other charges against Customer. Customer shall hold Algonquin harmless and indemnify Algonquin against any liability whatsoever from any party whatsoever as a result of Algonquin's exercise of its agency authority with respect to the Upstream Arrangements hereunder.

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 233 Fourth Revised Sheet No. 233

Superseding: Third Revised Sheet No. 233

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Secondary Points of Delivery under this Rate Schedule AFT-E. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. INDEMNIFICATION

- 7.1 Customer acknowledges that as a result of Section 5 Orders issued to Customer's Upstream Entity or Notice Agent hereunder, Customer may incur imbalances between receipts and deliveries that will be cashed out at month's end pursuant to Section 25 of the General Terms and Conditions of this tariff. Algonquin shall provide Customer with notice of an imbalance in a timely manner. Customer shall hold Algonquin harmless and indemnify Algonquin against any claim by Customer or otherwise arising out of the incurrence of imbalances as a result of Algonquin's issuance of Section 5 Orders.
- 7.2 Customer acknowledges that the exercise by Algonquin of its agency authority with respect to the Upstream Arrangements may cause the Upstream Entity to assess imbalance cash-out charges and transportation usage, storage injection, storage withdrawal or other charges against Customer. Customer shall hold Algonquin harmless and indemnify Algonquin against any liability whatsoever from any party whatsoever as a result of Algonquin's exercise of its agency authority with respect to the Upstream Arrangements hereunder.

FERC Docket: RP04-361-000

Original Sheet No. 234 Original Sheet No. 234: Effective

RATE SCHEDULE AFT-E FIRM TRANSPORTATION SERVICE

7. INDEMNIFICATION (continued)

- 7.3 Customer acknowledges that Notice Agent's execution of Section 5 Orders may cause upstream transporters or suppliers to assess charges against Customer for gas costs, transportation services, storage services, imbalance cash out or otherwise. Customer shall hold Algonquin harmless and indemnify Algonquin against any liability whatsoever from any party whatsoever as a result of Algonquin's issuance of Section 5 Orders to the Notice Agent, and the Notice Agent's performance or non-performance of its duties.
- 7.4 Notwithstanding the other provisions of this Section 7, Algonquin is not absolved from liability arising as a result of negligence on the part of Algonquin.

8. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 235 - 240 Sheet Nos. 235 - 240 : Effective

Sheet Nos. 235 - 240 are being reserved for future use

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 241 Sixth Revised Sheet No. 241

Superseding: Fifth Revised Sheet No. 241

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

1. AVATTABITATY

This rate schedule is available for firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer's total MDTQ under all Algonquin firm rate schedules is 10,000 Dth or less per Day;
- b. Customer's sole source of transportation service deliveries is Algonquin;
- c. Customer has made a valid request for firm transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part;
- d. Sufficient firm capacity is available to effectuate such transportation without any construction of facilities or other investment by Algonquin, or Algonquin has waived this requirement in writing;
- e. The Primary Point(s) of Receipt and Primary Point(s) of Delivery requested by Customer are acceptable to Algonquin from the viewpoint of adequacy of Algonquin's existing facilities to receive and transport Customer's gas with Algonquin's existing firm service taking precedence; and
- f. Customer has executed a Service Agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- g. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095), and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available under this Rate Schedule AFT-1S.
- 2. APPLICABILITY AND CHARACTER OF SERVICE
- 2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-1S Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily

Effective Date: 12/31/9999 Status: Accepted

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 241 Fourth Revised Sheet No. 241

Superseding: Third Revised Sheet No. 241

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

AVATTABITATY

This rate schedule is available for firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer's total MDTQ under all Algonquin firm rate schedules is 10,000 Dth or less per day;
- b. Customer's sole source of transportation service deliveries is Algonquin;
- c. Customer has made a valid request for firm transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part;
- d. Sufficient firm capacity is available to effectuate such transportation without any construction of facilities or other investment by Algonquin, or Algonquin has waived this requirement in writing;
- e. The Primary Point(s) of Receipt and Primary Point(s) of Delivery requested by Customer are acceptable to Algonquin from the viewpoint of adequacy of Algonquin's existing facilities to receive and transport Customer's gas with Algonquin's existing firm service taking precedence; and
- f. Customer has executed a Service Agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- g. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095) are available under this Rate Schedule AFT-1S.
- 2. APPLICABILITY AND CHARACTER OF SERVICE
- 2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-1S Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily

FERC Docket: RP10-77-000

First Revised Sheet No. 242 First Revised Sheet No. 242

Superseding: Original Sheet No. 242

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

2. APPLICABILITY AND CHARACTER OF SERVICE (continued)

quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement, except as provided in Section 2.7 herein below, plus amounts reflecting the Fuel Reimbursement Quantity; provided however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any Day a quantity of gas in excess of the applicable Maximum Daily Receipt Obligation ("MDRO") plus any applicable Fuel Reimbursement Quantity, provided that, if more than one Customer requests receipts in excess of its MDRO at a Point of Receipt, and the sum of all such requests exceeds the available capacity at such Point of Receipt, Algonquin shall apportion such receipts in excess of MDRO among such Customers pro rata according to the Customers' firm MDROs at the relevant Point of Receipt.

Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on Algonquin's system as are specified in an executed AFT-1S Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"); provided however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ, and on any Day a quantity of gas in excess of the applicable Maximum Daily Delivery Obligation ("MDDO"), provided that, if more than one Customer requests deliveries in excess of its MDDO at a Point of Delivery, and the sum of all such requests exceeds the available capacity at such Point of Delivery, Algonquin shall apportion such deliveries in excess of MDDO among such Customers pro rata according to the Customers' firm MDDOs at the relevant Point of Delivery.

FERC Docket: RP10-308-000

Second Revised Sheet No. 243 Second Revised Sheet No. 243

Superseding: First Revised Sheet No. 243

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

APPLICABILITY AND CHARACTER OF SERVICE (continued)

- 2.3 Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.
- 2.4 A Customer executing a service agreement under this Rate Schedule AFT-1S has the right to make an election to convert its service in its entirety from firm transportation service under this rate schedule to firm transportation service under Rate Schedule AFT-1 by providing written notice of such election on or before June 1 of any year. As part of such conversion, such converting Customer has the right to reduce its MDTQ to be applicable under Rate Schedule AFT-1 upon such written notice of its election to convert. Such conversion and reduction shall be effective as of November 1 of that year.
- 2.5 Algonquin shall not be obligated to add any facilities or expand the capacity of Algonquin's pipeline system in any manner in order to provide transportation service to Customer pursuant to this rate schedule; provided, however, Algonquin may, at its option, and with Customer's consent, add facilities or expand capacity to provide such transportation service, subject to Section 42.
- 2.6 Unless otherwise specified in the applicable service agreement, service hereunder shall be available on any day of the year, subject to Customer's MATQ and MDTQ limitations.
- 2.7 With respect to existing service agreements resulting from the conversion pursuant to Order No. 636 of sales and storage service under former Rate Schedules F-1, WS-1, STB and SS-3, subject to Algonquin's firm service obligations from primary points of receipt to primary points of delivery Algonquin shall provide service above Customer's MATQ up to but not in excess of the highest MDTQ specified in the executed service agreement multiplied by 365 (366 for a leap year) and, on any Day, Algonquin shall provide service above Customer's MDTQ up to but not in excess of the highest MDTQ specified in the executed service agreement; provided, however, that such additional quantities shall be provided with the same priority for purposes of Scheduling and Curtailment accorded to service from Secondary Points of Receipt and Secondary Points of Delivery, and provided further, that for purposes of Curtailment Customer's maximum daily entitlement shall be the highest MDTQ specified in Customer's executed service agreement during the period of Curtailment. Such additional quantities shall be subject to fuel, the applicable commodity rate and any applicable commodity surcharges.

3. RATE

3.1 Unit Rates. The applicable maximum and minimum unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AFT-1S of this tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 33 and Section 34 of the General Terms and Conditions of this tariff.

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 244 Fourth Revised Sheet No. 244

Superseding: Third Revised Sheet No. 244

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

3. RATE (continued)

The applicable unit rates to be charged on any Day by Algonquin for gas delivered to Customer shall not be in excess of the maximum unit rate nor less than the minimum unit rate

- 3.2 Monthly Bill for Forwardhaul Rates. For Customers executing an AFT-1S Service Agreement for transportation which is not solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-1S Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable Service Agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in Customer's executed AFT-1S Service Agreement; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 245 Fourth Revised Sheet No. 245

Superseding: Third Revised Sheet No. 245

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

- 3. RATE (continued)
 - (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 3.3 Monthly Bill for Backhaul Rates. For Customers executing an AFT-1S Service Agreement for transportation which is solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-1S Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable Service Agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in Customer's executed AFT-1S Service Agreement; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less

FERC Docket: RP10-77-000

Second Revised Sheet No. 246 Second Revised Sheet No. 246

Superseding: First Revised Sheet No. 246

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

3. RATE (continued)

- (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES
 - 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.
 - 4.2 Delivery of Gas. Based upon the daily quantity scheduled, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions of this tariff. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.

FERC Docket: RP04-361-000

Original Sheet No. 247 Original Sheet No. 247: Effective

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES (continued)
 - 4.3 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.
- 5. OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

- 5.1 Customer shall make all necessary arrangements with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 Algonquin shall schedule receipts at a Secondary Point of Receipt or deliveries at a Secondary Point of Delivery pursuant to the provisions of Sections 48.2 and 48.3 of the General Terms and Conditions.
- To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. Prior to any reduction or interruption in service due to the failure of the upstream entity to deliver gas on behalf of Customer, Algonquin shall provide notice in a time and manner that is reasonable under then existing conditions. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.

FERC Docket: RP10-77-000

First Revised Sheet No. 248 First Revised Sheet No. 248

Superseding: Original Sheet No. 248

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

- 5. OTHER OPERATING CONDITIONS (continued)
 - 5.4 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt at an hourly rate of 1/24th of the scheduled daily quantity. The daily quantities of natural gas transported shall be accepted at the Point(s) of Delivery at a substantially constant hourly rate or, in the case of service agreements resulting from the conversion pursuant to Order No. 636 of sales entitlements under former Rate Schedules F-1 and WS-1, at a rate no greater than .06 multiplied by the scheduled daily quantity.
- POINT(S) OF RECEIPT AND DELIVERY
 - Primary Points of Receipt: The Primary Point(s) of Receipt at which Algonquin shall receive gas for transportation under this rate schedule shall 6.1 (A) be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Receipt the MDRO and receipt pressure obligations. Such exhibit by mutual written agreement may be superseded by a new exhibit which may add or delete specific points or make other changes thereto that the parties deem appropriate. Algonquin shall not accept any proposed Primary Point(s) of Receipt, or quantity at any Primary Point(s) of Receipt, or change in quantities among Primary Point(s) of Receipt if (a) the resulting aggregate MDROs at all of Customer's Primary Point(s) of Receipt would exceed Customer's MDTQ, or (b) in doing so, in Algonquin's reasonable judgment, Algonquin would impair its ability to satisfy its existing firm obligations to receive gas pursuant to other firm service agreements under which such Point(s) of Receipt are Primary Points of Receipt and to purchase and receive its Company Use Gas at maximum deliverability levels, as such Company Use Gas arrangements exist under agreements effective at the date of Customer's request or reasonably expected by Algonquin to be effective within six months of the request.
 - A Replacement Customer that acquired capacity pursuant to the capacity (B) release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Receipt in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Receipt to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Receipt that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Receipt, shall revert to the Releasing Customer, and any Primary Points of Receipt granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 249 Sixth Revised Sheet No. 249

Superseding: Fifth Revised Sheet No. 249

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Secondary Points of Receipt under this Rate Schedule AFT-1S. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.
 - 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Primary Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - (B) A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - 6.4 Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral,

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 249 Fourth Revised Sheet No. 249

Superseding: Third Revised Sheet No. 249

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Secondary Points of Receipt under this Rate Schedule AFT-1S. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.
 - 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Primary Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - A Replacement Customer that acquired capacity pursuant to the capacity (B) release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 250 Fifth Revised Sheet No. 250

Superseding: Fourth Revised Sheet No. 250

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Secondary Points of Delivery under this Rate Schedule AFT-1S. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 250 Fourth Revised Sheet No. 250

Superseding: Third Revised Sheet No. 250

RATE SCHEDULE AFT-1S FIRM TRANSPORTATION SERVICE

Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Secondary Points of Delivery under this Rate Schedule AFT-1S. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 251 - 260 Sheet Nos. 251 - 260 : Effective

Sheet Nos. 251 - 260 are being reserved for future use

FERC Docket: RP10-77-000

First Revised Sheet No. 261 First Revised Sheet No. 261

Superseding: Original Sheet No. 261

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

AVATTABITATY

This rate schedule is available for firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer's total MDTQ under all Algonquin firm rate schedules is 10,000 Dth or less per Day;
- b. Customer's sole source of transportation service deliveries is Algonquin;
- c. Customer has made a valid request for firm transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part;
- d. Sufficient firm capacity is available to effectuate such transportation without any construction of facilities or other investment by Algonquin, or Algonquin has waived this requirement in writing;
- e. The Primary Point(s) of Receipt and Primary Point(s) of Delivery requested by Customer are acceptable to Algonquin from the viewpoint of adequacy of Algonquin's existing facilities to receive and transport Customer's gas with Algonquin's existing firm service taking precedence;
- f. Customer has contracted with one or more operators of upstream facilities that interconnect with the Algonquin system (the "Upstream Entity(ies)"), for a sufficient quantity of no-notice transportation services that encompass the right on the part of Customer to increase or decrease its receipts from the Upstream Entity(ies) into designated Points of Receipt on Algonquin's system with no advance notice, up to Customer's MDTQ under this Rate Schedule AFT-ES plus fuel as provided for in Section 32 of the General Terms and Conditions of this tariff in the case of

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 262 Fifth Revised Sheet No. 262

Superseding: Fourth Revised Sheet No. 262

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

1. AVAILABILITY (continued)

increases in receipts, or down to zero in the case of decreases in receipts, without regard to any quantities that Customer previously scheduled on the Upstream Entity(ies), and with no requirement that Customer adhere to a constant rate of flow on the Upstream Entity(ies) (the "Upstream Arrangements");

- g. Customer has named a contact party (the "Notice Agent") for the receipt of orders issued in accordance with Section 5 ("Section 5 Orders") of this rate schedule;
- h. Customer has authorized Algonquin to exercise Customer's nomination and scheduling rights under the Upstream Arrangements, if necessary by executing agency instruments satisfactory to the Upstream Entity, to the extent required to permit service under this rate schedule; and
- i. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- j. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095), and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available under this Rate Schedule AFT-ES.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-ES Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 262 Fourth Revised Sheet No. 262

Superseding: Third Revised Sheet No. 262

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

1. AVAILABILITY (continued)

increases in receipts, or down to zero in the case of decreases in receipts, without regard to any quantities that Customer previously scheduled on the Upstream Entity(ies), and with no requirement that Customer adhere to a constant rate of flow on the Upstream Entity(ies) (the "Upstream Arrangements");

- g. Customer has named a contact party (the "Notice Agent") for the receipt of orders issued in accordance with Section 5 ("Section 5 Orders") of this rate schedule;
- h. Customer has authorized Algonquin to exercise Customer's nomination and scheduling rights under the Upstream Arrangements, if necessary by executing agency instruments satisfactory to the Upstream Entity, to the extent required to permit service under this rate schedule; and
- i. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- j. Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095) are available under this Rate Schedule AFT-ES.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-ES Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the

FERC Docket: RP10-77-000

First Revised Sheet No. 263 First Revised Sheet No. 263

Superseding: Original Sheet No. 263

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

APPLICABILITY AND CHARACTER OF SERVICE (continued)

service agreement, except as provided in Section 2.8 herein below, plus amounts reflecting the Fuel Reimbursement Quantity; provided however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any Day a quantity of gas in excess of the applicable Maximum Daily Receipt Obligation ("MDRO") plus any applicable Fuel Reimbursement Quantity, provided that, if more than one Customer requests receipts in excess of its MDRO at a Point of Receipt, and the sum of all such requests exceeds the available capacity at such Point of Receipt, Algonquin shall apportion such receipts in excess of MDRO among such Customers pro rata according to the Customers' firm MDROs at the relevant Point of Receipt.

- 2.2 Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on Algonquin's system as are specified in an executed AFT-ES Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"); provided however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ, and on any Day a quantity of gas in excess of the applicable Maximum Daily Delivery Obligation ("MDDO"), provided that, if more than one Customer requests deliveries in excess of its MDDO at a Point of Delivery, and the sum of all such requests exceeds the available capacity at such Point of Delivery, Algonquin shall apportion such deliveries in excess of MDDO among such Customers pro rata according to the Customers' firm MDDOs at the relevant Point of Delivery.
- 2.3 Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas

FERC Docket: RP10-77-000

First Revised Sheet No. 264 First Revised Sheet No. 264

Superseding: Original Sheet No. 264

RATE SCHEDULE AFT-ES
FIRM TRANSPORTATION SERVICE

2. APPLICABILITY AND CHARACTER OF SERVICE (continued)

can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.

- 2.4 A Customer executing a service agreement under this Rate Schedule AFT-ES has the right to make an election to convert its service in its entirety from firm transportation service under this rate schedule to firm transportation service under Rate Schedule AFT-E or Rate Schedule AFT-1 by providing written notice of such election on or before June 1 of any year. As part of such conversion, such converting Customer has the right to reduce its MDTQ to be applicable under Rate Schedule AFT-E or Rate Schedule AFT-1 upon such written notice of its election to convert. Such conversion and reduction shall be effective as of November 1 of that year.
- 2.5 Algonquin shall not be obligated to add any facilities or expand the capacity of Algonquin's pipeline system in any manner in order to provide transportation service to Customer pursuant to this rate schedule; provided, however, Algonquin may, at its option, and with Customer's consent, add facilities or expand capacity to provide such transportation service, subject to Section 42.
- 2.6 Subject to the conditions specified in Section 5 of this Rate Schedule, service hereunder shall encompass the right on the part of Customer to increase deliveries at the Point(s) of Delivery up to the MDTQ and to decrease deliveries at the Point(s) of Delivery without advance notice to Algonquin and without Customer's previously having provided for a concurrent increase or decrease in receipts at the Point(s) of Receipt.
- 2.7 Unless otherwise specified in the applicable service agreement, service hereunder shall be available on any day of the year, subject to Customer's MATQ and MDTQ limitations.
- 2.8 With respect to existing service agreements resulting from the conversion pursuant to Order No. 636 of sales and storage service under former Rate Schedules F-1, WS-1, STB and SS-3, subject to Algonquin's firm service obligations from primary points of receipt to primary points of delivery Algonquin shall provide service above Customer's MATQ up to but not in excess of the highest MDTQ specified in the executed service agreement multiplied by 365 (366 for a leap year) and, on any Day, Algonquin shall provide service above Customer's MDTQ up to but not in excess of the highest MDTQ specified in the executed service agreement; provided, however, that such additional quantities shall be provided with the same priority for purposes of Scheduling and Curtailment accorded to service from Secondary Points of Receipt and Secondary Points of Delivery, and provided further, that for purposes of Curtailment Customer's maximum daily entitlement shall be the highest MDTQ specified in Customer's executed service agreement during the period of Curtailment. Such additional quantities shall be subject to fuel, the applicable commodity rate and any applicable commodity surcharges.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 265 Fourth Revised Sheet No. 265

Superseding: Third Revised Sheet No. 265

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

- 3. RATE
- 3.1 Unit Rates. The applicable maximum and minimum unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AFT-ES of this tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 33 and Section 34 of the General Terms and Conditions of this tariff. The applicable unit rates to be charged on any Day by Algonquin for gas delivered to Customer shall not be in excess of the maximum unit rate nor less than the minimum unit rate.
- 3.2 Monthly Bill for Forwardhaul Rates. For Customers executing an AFT-ES Service Agreement for transportation which is not solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-ES Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable Service Agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in Customer's executed AFT-ES Service Agreement; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 266 Fourth Revised Sheet No. 266

Superseding: Third Revised Sheet No. 266

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

- 3. RATE (continued)
 - (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less
 - (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 3.3 Monthly Bill for Backhaul Rates. For Customers executing an AFT-ES Service Agreement for transportation which is solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AFT-ES Service Agreement is effective and for each Month thereafter unless otherwise specified in the applicable Service Agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Reservation Charge: The charge per Month per Dth of Customer's highest MDTQ during the Contract Year, as specified in the applicable Service Agreement; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus

FERC Docket: RP10-77-000

Second Revised Sheet No. 267 Second Revised Sheet No. 267

Superseding: First Revised Sheet No. 267

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

- 3. RATE (continued)
 - (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less
 - (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES
 - 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.

FERC Docket: RP10-77-000

First Revised Sheet No. 268 First Revised Sheet No. 268

Superseding: Original Sheet No. 268

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES (continued)
 - 4.2 Delivery of Gas. Based upon the daily quantity scheduled and in accordance with Section 5 herein, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions of this tariff. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.
 - 4.3 No-Notice Service. Notwithstanding the quantities nominated by Customer and scheduled by Algonquin hereunder, Customer shall be entitled to increase its deliveries up to the MDDO at any point, up to the MHTQ during any Hour, and up to the MDTQ, or to decrease its deliveries. Provided that all of the operational conditions specified in Section 5 of this rate schedule (the "Section 5 Conditions") are met, Algonquin shall consent to such increase or decrease in deliveries, thereby nullifying any daily scheduling or hourly scheduling penalty that would otherwise be applicable pursuant to Section 23 of the General Terms and Conditions. Furthermore, if the Section 5 Conditions are met, Algonquin will forbear from taking action pursuant to Section 26 of the General Terms and Conditions to reduce deliveries to Customer. If any of the Section 5 Conditions are not met, Algonquin shall not be required to receive or deliver gas in amounts other than the transportation quantities scheduled by Customer, and variations between scheduled quantities and actual deliveries to Customer's meter shall be subject to the assessment of penalties pursuant to Section 23 of the General Terms and Conditions.
 - 4.4 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.

FERC Docket: RP04-361-000

Original Sheet No. 269 Original Sheet No. 269: Effective

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

- 5.1 Customer shall make all necessary arrangements, including the Upstream Arrangements, with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 Algonquin shall schedule receipts at a Secondary Point of Receipt or deliveries at a Secondary Point of Delivery pursuant to the provisions of Sections 48.2 and 48.3 of the General Terms and Conditions.
- 5.3 To the extent that any Upstream Entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. Prior to any reduction or interruption in service due to the failure of the Upstream Entity to deliver gas on behalf of Customer, Algonquin shall provide notice in a time and manner that is reasonable under then existing conditions. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
- 5.4 The Upstream Arrangements must be in full force and effect; Customer shall not be in default of the Upstream Arrangements; and Customer's rights under the Upstream Arrangements shall be enforceable by Customer or its agents without impairment due to curtailment, force majeure or other reasons.
- 5.5 The Upstream Entity(ies) shall recognize Algonquin's rights as Customer's agent to issue orders ("Section 5 Orders") on Customer's behalf nominating or directing for Customer's account no-notice service as encompassed within the Upstream Arrangements at such levels and at such times as Algonquin

FERC Docket: RP10-77-000

First Revised Sheet No. 270 First Revised Sheet No. 270

Superseding: Original Sheet No. 270

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

5. OTHER OPERATING CONDITIONS (continued)

determines in its reasonable discretion is required by the Algonquin system to meet the needs of Customer under this rate schedule.

- 5.6 All Upstream Entities involved in all Upstream Arrangements with Customer served under this rate schedule shall deliver to the various points of interconnection between Algonquin's facilities and the facilities of the various Upstream Entities such quantities of gas at such times as Algonquin determines in its reasonable discretion are required by the Algonquin system to meet the needs of Customer under this rate schedule.
- 5.7 Customer's Notice Agent must have complete authority to call upon supplies for Customer's account at such times, in such quantities, and at such locations as Algonquin deems necessary, in its reasonable discretion, to enable Algonquin to provide service as contemplated for Customer under this rate schedule.
- 5.8 All Notice Agents named by Customer under this rate schedule shall execute Section 5 Orders at such locations and at such times as Algonquin determines in its reasonable discretion are necessary to enable Algonquin to provide service as contemplated under Rate Schedules AFT-ES and AFT-E.
- 5.9 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt at an hourly rate of 1/24th of the scheduled daily quantity. The daily quantities of natural gas transported shall be accepted at the Point(s) of Delivery at a substantially constant hourly rate or, in the case of service agreements resulting from the conversion pursuant to Order No. 636 of sales entitlements under former Rate Schedules F-1 and WS-1, at an hourly rate no greater than .06 multiplied by the scheduled daily quantity.
- 5.10 In the event Customer has multiple upstream no-notice services pursuant to more than one contract, from more than one pipeline, or under more than one rate schedule, Customer shall provide Algonquin a predetermined order of preference for these upstream services. Algonquin shall follow this order of preference in scheduling Customer's upstream no-notice service. Customer may,

FERC Docket: RP10-77-000

First Revised Sheet No. 271 First Revised Sheet No. 271

Superseding: Original Sheet No. 271

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

5. OTHER OPERATING CONDITIONS (continued)

upon twenty-four hours' written notice to Algonquin, modify this specification of scheduling preferences.

- 5.11 In the event Algonquin issues a Section 5 Order on Customer's behalf, Algonquin shall provide notice as soon as possible to Customer and Customer's Notice Agent by telephone or facsimile and by posting on its Internet Web Site that the Section 5 Order has been issued.
- 6. POINT(S) OF RECEIPT AND DELIVERY
 - Primary Points of Receipt: The Primary Point(s) of Receipt at which Algonquin shall receive gas for transportation under this rate schedule shall 6.1 (A) be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Receipt the MDRO and receipt pressure obligations. Such exhibit by mutual written agreement may be superseded by a new exhibit which may add or delete specific points or make other changes thereto that the parties deem appropriate. Algonquin shall not accept any proposed Primary Point(s) of Receipt, or quantity at any Primary Point(s) of Receipt, or change in quantities among Primary Point(s) of Receipt if (a) the resulting aggregate MDROs at all of Customer's Primary Point(s) of Receipt would exceed Customer's MDTQ, or (b) in doing so, in Algonquin's reasonable judgment, Algonquin would impair its ability to satisfy its existing firm obligations to receive gas pursuant to other firm service agreements under which such Point(s) of Receipt are Primary Points of Receipt and to purchase and receive its Company Use Gas at maximum deliverability levels, as such Company Use Gas arrangements exist under agreements effective at the date of Customer's request or reasonably expected by Algonquin to be effective within six months of the request.
 - A Replacement Customer that acquired capacity pursuant to the capacity (B) release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Receipt in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Receipt to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Receipt that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Receipt, shall revert to the Releasing Customer, and any Primary Points of Receipt granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 272 Sixth Revised Sheet No. 272

Superseding: Fifth Revised Sheet No. 272

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Secondary Points of Receipt under this Rate Schedule AFT-ES. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.
 - 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - (B) A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - 6.4 Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 272 Fourth Revised Sheet No. 272

Superseding: Third Revised Sheet No. 272

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Secondary Points of Receipt under this Rate Schedule AFT-ES. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.
 - 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate MDDOs at all of Customer's Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - (B) A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 273 Sixth Revised Sheet No. 273

Superseding: Fifth Revised Sheet No. 273

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Secondary Points of Delivery under this Rate Schedule AFT-ES. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. INDEMNIFICATION

- 7.1 Customer acknowledges that as a result of Section 5 Orders issued to Customer's Upstream Entity or Notice Agent hereunder, Customer may incur imbalances between receipts and deliveries that will be cashed out at Month's end pursuant to Section 25 of the General Terms and Conditions of this tariff. Algonquin shall provide Customer with notice of an imbalance in a timely manner. Customer shall hold Algonquin harmless and indemnify Algonquin against any claim by Customer or otherwise arising out of the incurrence of imbalances as a result of Algonquin's issuance of Section 5 Orders.
- 7.2 Customer acknowledges that the exercise by Algonquin of its agency authority with respect to the Upstream Arrangements may cause the Upstream Entity to assess imbalance cash-out charges and transportation usage, storage injection, storage withdrawal or other charges against Customer. Customer shall hold Algonquin harmless and indemnify Algonquin against any liability whatsoever from any party whatsoever as a result of Algonquin's exercise of its agency authority with respect to the Upstream Arrangements hereunder.
- 7.3 Customer acknowledges that Notice Agent's execution of Section 5 Orders may cause upstream transporters or suppliers to assess charges against Customer for gas costs, transportation services, storage services, imbalance cash out or otherwise. Customer shall hold Algonquin harmless and indemnify Algonquin against any liability whatsoever from any party whatsoever as a result of Algonquin's issuance of Section 5 Orders to the Notice Agent, and the Notice Agent's performance or non-performance of its duties.

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 273 Fourth Revised Sheet No. 273

Superseding: Third Revised Sheet No. 273

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Secondary Points of Delivery under this Rate Schedule AFT-ES. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. INDEMNIFICATION

- 7.1 Customer acknowledges that as a result of Section 5 Orders issued to Customer's Upstream Entity or Notice Agent hereunder, Customer may incur imbalances between receipts and deliveries that will be cashed out at month's end pursuant to Section 25 of the General Terms and Conditions of this tariff. Algonquin shall provide Customer with notice of an imbalance in a timely manner. Customer shall hold Algonquin harmless and indemnify Algonquin against any claim by Customer or otherwise arising out of the incurrence of imbalances as a result of Algonquin's issuance of Section 5 Orders.
- 7.2 Customer acknowledges that the exercise by Algonquin of its agency authority with respect to the Upstream Arrangements may cause the Upstream Entity to assess imbalance cash-out charges and transportation usage, storage injection, storage withdrawal or other charges against Customer. Customer shall hold Algonquin harmless and indemnify Algonquin against any liability whatsoever from any party whatsoever as a result of Algonquin's exercise of its agency authority with respect to the Upstream Arrangements hereunder.
- 7.3 Customer acknowledges that Notice Agent's execution of Section 5 Orders may cause upstream transporters or suppliers to assess charges against Customer for gas costs, transportation services, storage services, imbalance cash out or otherwise. Customer shall hold Algonquin harmless and indemnify Algonquin against any liability whatsoever from any party whatsoever as a result of Algonquin's issuance of Section 5 Orders to the Notice Agent, and the Notice Agent's performance or non-performance of its duties.

FERC Docket: RP04-361-000

Original Sheet No. 274 Original Sheet No. 274: Effective

RATE SCHEDULE AFT-ES FIRM TRANSPORTATION SERVICE

- 7. INDEMNIFICATION (continued)
 - 7.4 Notwithstanding the other provisions of this Section 7, Algonquin is not absolved from liability arising as a result of negligence on the part of Algonquin.
- 8. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 275 - 280 Sheet Nos. 275 - 280 : Effective

Sheet Nos. 275 - 280 are being reserved for future use

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 281 Fifth Revised Sheet No. 281

Superseding: Fourth Revised Sheet No. 281

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

AVATLABILITY

This rate schedule is available for interruptible transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer has made a valid request for interruptible transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part; and
- b. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095), and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available under this Rate Schedule AIT-1.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on an interruptible basis. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AIT-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement plus amounts reflecting the Fuel Reimbursement Quantity.

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 281 Fourth Revised Sheet No. 281

Superseding: Third Revised Sheet No. 281

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

AVATLABILITY

This rate schedule is available for interruptible transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer has made a valid request for interruptible transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part; and
- b. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.
- Transportation service effectuated through capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL is not available under this rate schedule; provided, however that the interconnection between the Brayton Point Lateral and Algonquin's mainline (M&R No. 80035), the interconnection between the Manchester Street Lateral and Algonquin's mainline (M&R No. 80071), the interconnection between the Canal Lateral and Algonquin's mainline (M&R No. 8004), the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline (M&R No. 80095) are available under this Rate Schedule AIT-1.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on an interruptible basis. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AIT-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement plus amounts reflecting the Fuel Reimbursement Quantity.

FERC Docket: RP10-77-000

First Revised Sheet No. 282 First Revised Sheet No. 282

Superseding: Original Sheet No. 282

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

- 2. APPLICABILITY AND CHARACTER OF SERVICE (continued)
 - Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on Algonquin's system as are specified in an executed AIT-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"), provided, however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ.
 - 2.3 Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.
 - 2.4 Algonquin shall not be obligated to add any facilities or expand the capacity of Algonquin's pipeline system in any manner in order to provide transportation service to Customer pursuant to this rate schedule; provided, however, Algonquin may, at its option, and with Customer's consent, add facilities or expand capacity to provide such transportation service, subject to Section 42.

FERC Docket: RP10-308-000

Fifth Revised Sheet No. 283 Fifth Revised Sheet No. 283

Superseding: Fourth Revised Sheet No. 283

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

3. RATE

- 3.1 Unit Rates. The applicable maximum and minimum unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AIT-1 of this FERC Gas Tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 33 and Section 34 of the General Terms and Conditions of this tariff. The applicable unit rates to be charged on any Day by Algonquin for gas delivered to Customer shall not be in excess of the maximum unit rate nor less than the minimum unit rate.
- 3.2 Monthly Bill for Forwardhaul Rates. For Customers executing an AIT-1 Service Agreement for transportation which is not solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AIT-I Service Agreement is effective and for each Month thereafter, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (b) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (c) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (d) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (e) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions.

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 284 Fourth Revised Sheet No. 284

Superseding: Third Revised Sheet No. 284

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

- 3. RATE (continued)
- 3.3 Monthly Bill for Backhaul Rates. For Customers executing an AIT-1 Service Agreement for transportation which is solely a Backhaul transportation arrangement as defined in Section 1 of the General Terms and Conditions, and commencing for the Month in which the AIT-1 Service Agreement is effective and for each Month thereafter, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (b) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (c) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (d) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (e) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions.

FERC Docket: RP10-77-000

First Revised Sheet No. 285 First Revised Sheet No. 285

Superseding: Original Sheet No. 285

RATE SCHEDULE AIT-1
INTERRUPTIBLE TRANSPORTATION SERVICE

4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES

- 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.
- 4.2 Delivery of Gas. Based upon the daily quantity scheduled, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.
- 4.3 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 286 Fifth Revised Sheet No. 286

Superseding: Fourth Revised Sheet No. 286

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

- 5.1 Customer shall make all necessary arrangements with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
- 5.3 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt and accepted at the Point(s) of Delivery at a substantially constant hourly rate, or such other hourly rate as may be acceptable to Algonquin.

6. POINT(S) OF RECEIPT AND DELIVERY

All interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Point(s) of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Points of Receipt under this Rate Schedule AIT-1.

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 286 Fourth Revised Sheet No. 286

Superseding: Third Revised Sheet No. 286

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

- 5.1 Customer shall make all necessary arrangements with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
- 5.3 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt and accepted at the Point(s) of Delivery at a substantially constant hourly rate, or such other hourly rate as may be acceptable to Algonquin.

6. POINT(S) OF RECEIPT AND DELIVERY

6.1 All interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Point(s) of Receipt, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Points of Receipt under this Rate Schedule AIT-1.

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 287 Fifth Revised Sheet No. 287

Superseding: Fourth Revised Sheet No. 287

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - All interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Point(s) of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and HubLine offshore system in Massachusetts Bay, Massachusetts, the interconnection between the J-2 Facility and Algonquin's mainline, and the interconnection between the Middletown Lateral and the Kleen Energy Lateral are available as Points of Delivery under this Rate Schedule AIT-1.
- 7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP09-864-000

Fourth Revised Sheet No. 287 Fourth Revised Sheet No. 287

Superseding: Third Revised Sheet No. 287

RATE SCHEDULE AIT-1 INTERRUPTIBLE TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - All interconnections between the facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Point(s) of Delivery, with the exception of interconnections with the facilities of other operators accessible only through the utilization of capacity on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, or the J-2 Facility, as such lateral facilities are defined in Rate Schedule AFT-CL; provided, however, that the interconnection between the Brayton Point Lateral and Algonquin's mainline, the interconnection between the Manchester Street Lateral and Algonquin's mainline, the interconnection between the Canal Lateral and Algonquin's mainline, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts, and the interconnection between the J-2 Facility and Algonquin's mainline are available as Points of Delivery under this Rate Schedule AIT-1.
- 7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 288 - 300 Sheet Nos. 288 - 300 : Effective

Sheet Nos. 288 - 300 are being reserved for future use

AVAILABILITY

FERC Docket: RP04-361-000

1.

Original Sheet No. 301 Original Sheet No. 301: Effective

RATE SCHEDULE T-1 Transportation Service - New Jersey and New York

Transportation borvios non corbe, and m

This rate schedule is available to any gas distribution system serving gas consumers in New Jersey or New York (hereinafter called "Customer") for the transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin"):

- (a) When such natural gas is transported through the portion of Algonquin's pipeline located in New Jersey and New York;
- (b) When such natural gas is tendered for transportation by Customer at a point or points on the main pipeline system of Algonquin in New Jersey or New York;
- (c) When, after transportation by Algonquin, Customer takes delivery of natural gas at a point or points on Algonquin's main pipeline system in New Jersey or New York;
- (d) When Customer has an effective service agreement for such transportation service between specified points of receipt and delivery up to a specific Maximum Daily Transportation Quantity of natural gas which shall be stated in said service agreement.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed T-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a

FERC Docket: RP10-77-000

First Revised Sheet No. 302 First Revised Sheet No. 302

Superseding: Original Sheet No. 302

RATE SCHEDULE T-1

Transportation Service - New Jersey and New York

2. APPLICABILITY AND CHARACTER OF SERVICE (continued)

basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement plus amounts reflecting the Fuel Reimbursement Quantity; provided, however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any Day a quantity of gas in excess of the applicable Maximum Daily Receipt Obligation ("MDRO") plus any applicable Fuel Reimbursement Quantity, provided that, if more than one Customer requests receipts in excess of its MDRO at a Point of Receipt, and the sum of all such requests exceeds the available capacity at such Point of Receipt, Algonquin shall apportion such receipts in excess of MDRO among such Customers pro rata according to the Customers' firm MDROs at the relevant Point of Receipt.

2.2 Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on Algonquin's system as are specified in an executed T-1 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"); provided, however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ, and on any Day a quantity of gas in excess of the applicable Maximum Daily Delivery Obligation ("MDDO"), provided that, if more than one Customer requests deliveries in excess of its MDDO at a Point of Delivery, and the sum of all such requests exceeds the available capacity at such Point of Delivery, Algonquin shall apportion such deliveries in excess of MDDO among such Customers pro rata according to the Customers' firm MDDOs at the relevant Point of Delivery.

FERC Docket: RP10-308-000

Third Revised Sheet No. 303 Third Revised Sheet No. 303

Superseding: Second Revised Sheet No. 303

RATE SCHEDULE T-1
Transportation Service - New Jersey and New York

2. APPLICABILITY AND CHARACTER OF SERVICE (continued)

2.3 Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.

3. RATE

- 3.1 Unit Rates. The applicable unit rates are set forth in the currently effective Statement of Rates for Rate Schedule T-1 of this tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 34 of the General Terms and Conditions of this tariff.
- 3.2 Monthly Bill. Commencing for the Month in which the T-1 Service Agreement is effective and for each Month thereafter, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Reservation Charge: The charge per Month per Dth of Customer's effective MDTQ under this rate schedule; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus

FERC Docket: RP10-77-000

Second Revised Sheet No. 304 Second Revised Sheet No. 304

Superseding: First Revised Sheet No. 304

RATE SCHEDULE T-1

Transportation Service - New Jersey and New York

- 3. RATE (continued)
 - (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less
 - (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES
 - 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.

FERC Docket: RP04-361-000

Original Sheet No. 305 Original Sheet No. 305: Effective

4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES (continued)

- 4.2 Delivery of Gas. Based upon the daily quantity scheduled, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions of this tariff. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.
- 4.3 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.

5. OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

5.1 Customer shall make all necessary arrangements with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.

FERC Docket: RP04-361-000

Original Sheet No. 306 Original Sheet No. 306: Effective

RATE SCHEDULE T-1 Transportation Service - New Jersey and New York

5. OTHER OPERATING CONDITIONS (continued)

- 5.2 To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
- 5.3 The daily quantities of natural gas transported shall be delivered at the Points of Receipt and accepted at the Points of Delivery at a substantially constant hourly rate, or such other hourly rate as may be acceptable to Algonquin.

6. POINT(S) OF RECEIPT AND DELIVERY

- 6.1 The Point(s) of Receipt at which Algonquin shall receive gas for transportation under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Receipt the MDRO and receipt pressure obligations.
- 6.2 The Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Delivery the MDDO and delivery pressure obligations.

FERC Docket: RP04-361-000

 $\textbf{Original Sheet No. 307} \; \texttt{Original Sheet No. 307} \; : \; \; \texttt{Effective}$

7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff, with the exception of Section 14 (Capacity Release), are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 308 - 320 Sheet Nos. 308 - 320 : Effective

Sheet Nos. 308 - 320 are being reserved for future use

FERC Docket: RP10-77-000

First Revised Sheet No. 321 First Revised Sheet No. 321

Superseding: Original Sheet No. 321

RATE SCHEDULE AFT-4
Firm Transportation Service

AVATTABITATY

This rate schedule is available to Columbia Gas Transmission Corporation (hereinafter called "Customer") for the firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") when Customer has entered into a service agreement with Algonquin providing for the receipt, transportation, and delivery of gas on a firm basis pursuant to this rate schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-4 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement plus amounts reflecting the Fuel Reimbursement Quantity; provided, however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any Day a quantity of gas in excess of the applicable Maximum Daily Receipt Obligation ("MDRO") plus any applicable Fuel Reimbursement Quantity, provided that, if more than one Customer requests receipts in excess of its MDRO at a Point of Receipt, and the sum of all such requests exceeds the available capacity at such Point of Receipt, Algonquin shall apportion such receipts in excess of MDRO among such Customers pro rata according to the Customers' firm MDROs at the relevant Point of Receipt.

FERC Docket: RP10-77-000

First Revised Sheet No. 322 First Revised Sheet No. 322

Superseding: Original Sheet No. 322

RATE SCHEDULE AFT-4
Firm Transportation Service

- 2. APPLICABILITY AND CHARACTER OF SERVICE (continued)
 - 2.2 Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on Algonquin's system as are specified in an executed AFT-4 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"); provided, however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ, and on any Day a quantity of gas in excess of the applicable Maximum Daily Delivery Obligation ("MDDO"), provided that, if more than one Customer requests deliveries in excess of its MDDO at a Point of Delivery, and the sum of all such requests exceeds the available capacity at such Point of Delivery, Algonquin shall apportion such deliveries in excess of MDDO among such Customers pro rata according to the Customers' firm MDDOs at the relevant Point of Delivery.
 - 2.3 Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Ouantities.

FERC Docket: RP10-308-000

Second Revised Sheet No. 323 Second Revised Sheet No. 323

Superseding: First Revised Sheet No. 323

RATE SCHEDULE AFT-4
Firm Transportation Service

- 3. RATE
- 3.1 Unit Rates. The applicable unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AFT-4 of this tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 34 of the General Terms and Conditions of this tariff.
- 3.2 Monthly Bill. Commencing for the Month in which the AFT-4 Service Agreement is effective and for each Month thereafter, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Reservation Charge: The charge per Month per Dth of Customer's effective MDTQ under this rate schedule; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus

FERC Docket: RP10-77-000

Second Revised Sheet No. 324 Second Revised Sheet No. 324

Superseding: First Revised Sheet No. 324

RATE SCHEDULE AFT-4
Firm Transportation Service

3. RATE (continued)

- (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
- (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
- (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less
- (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES
 - 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.

FERC Docket: RP04-361-000

Original Sheet No. 325 Original Sheet No. 325 : Effective

RATE SCHEDULE AFT-4 Firm Transportation Service

- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES (continued)
 - Delivery of Gas. Based upon the daily quantity scheduled, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions of this tariff. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.
 - 4.3 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.

FERC Docket: RP04-361-000

Original Sheet No. 326 original Sheet No. 326: Effective

RATE SCHEDULE AFT-4
Firm Transportation Service

5. OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

- 5.1 Customer shall make all necessary arrangements with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.

FERC Docket: RP04-361-000

Original Sheet No. 327 Original Sheet No. 327: Effective

RATE SCHEDULE AFT-4
Firm Transportation Service

- 5. OTHER OPERATING CONDITIONS (continued)
 - 5.3 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt and accepted at the Point(s) of Delivery at a substantially constant hourly rate, or such other hourly rate as may be acceptable to Algonquin.
- 6. POINT(S) OF RECEIPT AND DELIVERY
 - 6.1 The Point(s) of Receipt at which Algonquin shall receive gas for transportation under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Receipt the MDRO and receipt pressure obligations.

FERC Docket: RP04-361-000

Original Sheet No. 328 Original Sheet No. 328 : Effective

RATE SCHEDULE AFT-4
Firm Transportation Service

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - 6.2 The Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Delivery the MDDO and delivery pressure obligations.

7. WARRANTY OF TITLE

Customer warrants that it, its designee ("Designee"), or any assignee under Section 284.242 of the Commission's regulations will at the time of delivery of the gas to Algonquin have good and merchantable title to all of the gas free and clear of all liens, encumbrances and claims whatsoever. Title to the gas received by Algonquin for Customer's, Designee's or assignee's account hereunder shall remain with Customer, Designee or assignee during transportation of the gas by Algonquin. Customer, its designee ("Designee"), or any assignee under Section 284.242 of the Commission's regulations agrees to indemnify and save Algonquin harmless from all suits, actions, debts, accounts, damages, costs, losses and expenses arising out of adverse claims of any and all persons to the gas other than claims arising solely through acts of Algonquin.

8. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff, with the exception of Section 14 (Capacity Release), are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 329 - 330 Sheet Nos. 329 - 330 : Effective

Sheet Nos. 329 - 330 are being reserved for future use

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 331 Fifth Revised Sheet No. 331

Superseding: Fourth Revised Sheet No. 331

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

AVATLABILITY

This rate schedule is available for firm transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer") on that portion of Algonquin's pipeline system known as (1) the Canal Lateral, which shall mean pipeline facilities including pipeline lateral facilities, meter station, and appurtenant facilities which extend from a point on Algonquin's existing mainline interstate natural gas pipeline system in Bourne, Massachusetts, along the north side of the Cape Cod Canal, under the Cape Cod Canal at or near the Bourne/Sandwich town line, and terminating at a point in the town of Sandwich in Barnstable County, Massachusetts, (2) the Middletown Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station, and appurtenant facilities which will extend from a point on Algonquin's existing interstate natural gas pipeline system in Glastonbury, Connecticut to a point of interconnection, at the outlet side of the Middletown Meter Station, with facilities constructed by Connecticut Light and Power Company, (3) the Cleary Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station, and appurtenant facilities which will extend from a point on Algonquin's existing interstate natural gas pipeline system in Berkley, Massachusetts, under the Taunton River to a point of interconnection, at the outlet side of the Cleary Meter Station, with facilities constructed by Taunton Municipal Lighting Plant, (4) the Lake Road Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station, and appurtenant facilities which will extend from a point on Algonquin's existing interstate natural gas pipeline system in Windham County, Connecticut to a point of interconnection, at the outlet side of the Lake Road Meter Station, with facilities constructed by Lake Road Generating Co. LP, (5) the Brayton Point Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station, and appurtenant facilities which will extend from a point on Algonquin's existing interstate natural gas pipeline system in Dighton, MA (M&R No. 80034) to a point of interconnection, at the outlet side of the Brayton Point Meter Station, with facilities owned by US Gen New England, Inc. (M&R No. 00090), (6) the Bellingham Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station and appurtenant facilities which will extend from a point on Algonquin's existing interstate natural gas pipeline system in Norfolk County, Massachusetts to a point of interconnection, at the outlet side of the Bellingham Meter Station, with facilities constructed by ANP Bellingham Energy Company, (7) the Phelps Dodge Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station and appurtenant facilities which will extend from a point on Algonquin's existing interstate natural gas pipeline system in New London County, Connecticut to a point of interconnection at the outlet side of the Phelps Dodge Meter Station, with facilities owned by Phelps Dodge Copper Products Company, (8) the Manchester Street Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station and appurtenant facilities that extend from a point on Algonquin's existing interstate natural gas pipeline system at the head of the G-12 Lateral (M&R No. 80070) to a point of interconnection with facilities owned by USGen New England, Inc. at the outlet side of the Manchester Street power plant (M&R No. 00087), (9) the Cape Cod Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station and appurtenant facilities that extend from the terminus of Algonquin's Canal Lateral in the town of Sandwich in Barnstable County, Massachusetts, to a point of interconnection with facilities owned by Colonial Gas Company d/b/a KeySpan Energy Delivery New England in the town of Sandwich in Barnstable County, Massachusetts, (10) the Northeast Gateway Lateral, which shall mean pipeline facilities including a pipeline lateral and appurtenant facilities that extend from a point on Algonquin's existing HubLine offshore system in Massachusetts Bay, Massachusetts, to a point of interconnection with the offshore deepwater port facilities owned by Northeast Gateway Energy Bridge, L.L.C., (11) the J-2 Facility, which shall mean pipeline facilities including two parallel pipeline laterals, meter stations and appurtenant facilities that extend from a point on Algonquin's existing interstate natural gas pipeline system at the head of the J-2 Facility (M&R No. 80094) to a point of interconnection with facilities owned by NSTAR Gas Company downstream of the Mystic Street Station (M&R No. 00826), or (12) the Kleen Energy Lateral, which shall mean pipeline facilities including a pipeline lateral, meter station and appurtenant facilities that extend from a point on Algonquin's existing Middletown Lateral in the City of Middletown, Connecticut, to a point of interconnection, at the outlet side of the Kleen Energy Meter Station (M&R No. 00833), with the Kleen Energy Power Plant facilities when:

FERC Docket: RP10-146-000

Third Revised Sheet No. 332 Third Revised Sheet No. 332

Superseding: Second Revised Sheet No. 332

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

- a. Algonquin has placed the Canal Lateral, Middletown Lateral, Cleary Lateral, Lake Road Lateral, Brayton Point Lateral, Bellingham Lateral, Phelps Dodge Lateral, Manchester Street Lateral, Cape Cod Lateral, Northeast Gateway Lateral, J-2 Facility, or Kleen Energy Lateral, as applicable, in service;
- b. Customer has made a valid request for firm transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part;
- c. Sufficient firm capacity is available to effectuate such transportation without any construction of facilities or other investment by Algonquin, or Algonquin has waived this requirement in writing;
- d. The Primary Point(s) of Receipt and Primary Point(s) of Delivery requested by Customer are acceptable to Algonquin from the viewpoint of adequacy of Algonquin's existing facilities to receive and transport Customer's gas with Algonquin's existing firm service taking precedence; and
- e. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Transportation service hereunder will be on a firm basis except as provided herein and in Sections 16 and 24 of the General Terms and Conditions of this tariff. Algonquin shall receive from Customer, or for the account of Customer, at those points on Algonquin's system as specified in an executed AFT-CL Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas

FERC Docket: RP10-77-000

First Revised Sheet No. 333 First Revised Sheet No. 333

Superseding: Original Sheet No. 333

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

APPLICABILITY AND CHARACTER OF SERVICE (continued)

tendered for account of Customer up to Customer's Maximum Daily Transportation Quantity the ("MDTQ") and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement; provided, however, Algonquin shall not be obligated to, but may at its option, receive at any Point(s) of Receipt on any Day a quantity of gas in excess of the applicable Maximum Daily Receipt Obligation ("MDRO"), provided that, if more than one Customer requests receipts in excess of its MDRO at a Point of Receipt, and the sum of all such requests exceeds the available capacity at such Point of Receipt, Algonquin shall apportion such receipts in excess of MDRO among such Customers pro rata according to the Customers' firm MDROs at the relevant Point of Receipt.

- 2.2 Upon receipt of such natural gas for Customer's account, Algonquin shall transport and deliver hourly quantities of gas required by Customer up to Customer's MDTQ and, on a cumulative basis in any year, up to Customer's MATQ at those points on Algonquin's system as are specified in an executed AFT-CL Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"); provided, however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery a daily quantity exceeding the applicable Maximum Daily Delivery Obligation ("MDDO"), provided that, if more than one Customer requests deliveries in excess of its MDDO at a Point of Delivery, and the sum of all such requests exceeds the available capacity at such Point of Delivery, Algonquin shall apportion such deliveries in excess of MDDO among such Customers pro rata according to the Customers' firm MDDOs at the relevant Point of Delivery.
- 2.3 Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 334 Fourth Revised Sheet No. 334

Superseding: Third Revised Sheet No. 334

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

- APPLICABILITY AND CHARACTER OF SERVICE (continued)
 - 2.4 Algonquin shall not be obligated to add any facilities or expand the capacity of Algonquin's pipeline system in any manner in order to provide transportation service to Customer pursuant to this rate schedule; provided, however, Algonquin may, at its option, and with Customer's consent, add facilities or expand capacity to provide such transportation service, subject to Section 3 below.
 - 2.5 Unless otherwise specified in the applicable Service Agreement, services hereunder shall be available on any day of the year, subject to Customer's MATQ and MDTQ limitations.

RATE

- 3.1 Unit Rates. The applicable maximum and minimum unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AFT-CL of this tariff and are hereby incorporated herein. Such rates are subject to change under Sections 33 and 34 of the General Terms & Conditions as well as subject to the provisions of Section 4.3 of this rate schedule. The applicable unit rates to be charged on any Day by Algonquin for gas delivered to Customer shall not be in excess of the maximum unit rate nor less than the minimum unit rate.
- 3.2 Monthly Bill. Commencing for the Month in which the AFT-CL Service Agreement is effective, and for each Month thereafter unless otherwise specified in the applicable Service Agreement, Algonquin shall charge and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Reservation Charge: The charge per Month per Dth of Customer's MDTQ as specified in Customer's executed AFT-CL Service Agreement; plus
 - (b) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month under this rate schedule (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (c) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus

FERC Docket: RP10-77-000

Second Revised Sheet No. 335 Second Revised Sheet No. 335

Superseding: First Revised Sheet No. 335

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

3. RATE (continued)

- (d) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
- (e) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
- (f) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions; less
- (g) Revenue Credit: The revenue credit provided for in Section 41 of the General Terms and Conditions.
- 3.3 Customer Reimbursement. Customer shall, in addition to the charges referenced above, reimburse Algonquin for the following:
 - (a) The costs of any facilities installed by Algonquin with Customer's consent to receive, measure, transport or deliver natural gas for the account of Customer; and
 - (b) Any and all filing and approval fees required in connection with Customer's service agreement that Algonquin is obligated to pay to the FERC or any other governmental authority having jurisdiction.

Any reimbursement due Algonquin by Customer pursuant to this Section 3.3 shall be due and payable to Algonquin within ten days of the date of Algonquin's invoice(s) for same.

- 4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES
 - 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of

FERC Docket: RP10-146-000

Fourth Revised Sheet No. 336 Fourth Revised Sheet No. 336

Superseding: Third Revised Sheet No. 336

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES (continued)

the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.

- 4.2 Delivery of Gas. Based upon the daily quantity scheduled, Algonquin shall make daily delivery of Customer's scheduled quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder. Any excess or deficiency in such receipts and deliveries shall be resolved in accordance with the General Terms and Conditions of this tariff. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.
- 4.3 Responsibility for Imbalances. Any imbalance resulting from transportation pursuant to this rate schedule on the Canal Lateral, Middletown Lateral, Cleary Lateral, Lake Road Lateral, Brayton Point Lateral, Bellingham Lateral, Phelps Dodge Lateral, Manchester Street Lateral, Cape Cod Lateral, Northeast Gateway Lateral, J-2 Facility, or Kleen Energy Lateral, as applicable, shall be accounted for under the upstream transportation agreement(s) pursuant to which such gas was delivered to the inlet of the Canal Lateral, Middletown Lateral, Cleary Lateral, Lake Road Lateral, Brayton Point Lateral, Bellingham Lateral, Phelps Dodge Lateral, Manchester Street Lateral, Cape Cod Lateral, Northeast Gateway Lateral, J-2 Facility, or Kleen Energy Lateral, as applicable. No imbalance resolution charges, unauthorized overrun penalties, or scheduling penalties shall be assessed under this rate schedule to the extent that Algonquin has assessed any such charges or penalties under another agreement with respect to the same gas or transaction.
- 4.4 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.
- 5. OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

FERC Docket: RP04-361-000

Original Sheet No. 337 Original Sheet No. 337: Effective

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

- 5. OTHER OPERATING CONDITIONS (continued)
 - 5.1 Customer shall make all necessary arrangements with other parties at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
 - 5.2 Algonquin shall schedule receipts at a Secondary Point of Receipt or deliveries at a Secondary Point of Delivery pursuant to the provisions of Sections 48.2 and 48.3 of the General Terms and Conditions.
 - 5.3 To the extent that any upstream entity involved in handling Customer's gas (other than Algonquin) refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. Prior to any reduction or interruption in service due to the failure of the upstream entity to deliver gas on behalf of Customer, Algonquin shall provide notice in a time and manner that is reasonable under then existing conditions. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
- 6. POINT(S) OF RECEIPT AND DELIVERY
 - 6.1 (A) Primary Points of Receipt: The Primary Point(s) of Receipt at which Algonquin shall receive gas for transportation under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Primary Point of Receipt the MDRO and receipt pressure obligations. Such exhibit by mutual written agreement may be superseded by a new exhibit which may add or delete specific points or make other changes thereto that the parties deem appropriate. Algonquin shall not accept any proposed Primary Point(s) of Receipt, or quantity at any Primary Point(s) of Receipt, or change in quantities among Primary Point(s) of Receipt if (a) the resulting aggregate MDROs at all of Customer's Primary Point(s) of Receipt would exceed Customer's MDTQ, or (b) in doing so, in Algonquin's reasonable judgment, Algonquin would impair its ability to satisfy its existing firm obligations to receive gas pursuant to other firm service agreements under which such Point(s) of Receipt are Primary Points of Receipt and to purchase and receive its Company Use Gas at maximum deliverability levels, as such Company Use Gas arrangements exist under agreements effective at the date of Customer's request or reasonably expected by Algonquin to be effective within six months of the request.

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338

Superseding: Fourth Revised Sheet No. 338

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators on the designated AFT-CL lateral for use as a Primary Point of Receipt in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Receipt to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment to the Primary Points of Receipt under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In addition, a Replacement Customer may choose only those primary points along the lateral segment on which it contracts for transportation service under the replacement contract, as those lateral segments are set forth in the description of Secondary Points in Section 6.2 below. In the event that Replacement Customer selects a new Primary Point of Receipt that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Receipt, shall revert to the Releasing Customer, and any Primary Points of Receipt granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - Secondary Points of Receipt: Notwithstanding the foregoing, all interconnections between the Canal Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Canal Lateral, all interconnections between the Middletown Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Middletown Lateral, all interconnections between the Cleary Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Cleary Lateral, all interconnections between the Lake Road Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Lake Road Lateral, all interconnections between the Brayton Point Lateral facilities of Algonquin and the facilities of other operators including, but not limited to, the tap on the Algonquin G-1 System in Dighton, MA (M&R No. 80034) shall be available for use by Customer as Secondary Points of Receipt for service on the Brayton Point Lateral, all interconnections between the Bellingham Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Bellingham Lateral, all interconnections between the Phelps Dodge Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Phelps Dodge Lateral, all interconnections between the Manchester Street Lateral facilities of Algonquin and the facilities of other operators, including, but not limited to, the head of the G-12 Lateral (M&R No. 80070), shall be available for use by Customer as Secondary Points of Receipt for service on the Manchester Street Lateral, all interconnections between the Cape Cod Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Cape Cod Lateral, all interconnections between the Northeast Gateway Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Northeast Gateway Lateral,

FERC Docket: RP10-146-000

First Revised Sheet No. 338A First Revised Sheet No. 338A

Superseding: Original Sheet No. 338A

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

6. POINT(S) OF RECEIPT AND DELIVERY (continued)

all interconnections between the J-2 Facility of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the J-2 Facility, and all interconnections between the Kleen Energy Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the Kleen Energy Lateral. Such use of Secondary Points of Receipt is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

FERC Docket: RP10-77-000

First Revised Sheet No. 339 First Revised Sheet No. 339

Superseding: Original Sheet No. 339

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

- 6. POINT(S) OF RECEIPT AND DELIVERY (continued)
 - 6.3 (A) Primary Points of Delivery: The Primary Point(s) of Delivery at which Algonquin shall deliver gas for Customer's account under this rate schedule shall be specified in an exhibit to the service agreement executed by Algonquin and Customer. Such exhibit shall specify for each Point of Delivery the MDDO and delivery pressure obligations. Algonquin shall not accept any proposed Primary Point(s) of Delivery, or quantity at any Primary Point(s) of Delivery, or change in quantities among Primary Point(s) of Delivery if the resulting aggregate Primary Point(s) of Delivery if the resulting aggregate Primary Point(s) of Delivery would exceed Customer's MDTQ except to the extent Customer's MDDOs reflect the conversion of pre-existing entitlements under firm sales agreements.
 - A Replacement Customer that acquired capacity pursuant to the capacity release mechanism set forth in Section 14 of the General Terms and Conditions (B) of this tariff or a releasing Customer that has released capacity pursuant to the capacity release mechanism set forth in Section 14 of the GT&C of this tariff may request, subject to the availability of point and path capacity, any interconnection between the facilities of Algonquin and the facilities of other operators on the designated AFT-CL lateral for use as a Primary Point of Delivery in a segmented transaction, provided, however, that Algonquin shall not accept any proposed Primary Point of Delivery to the extent that (a) the resulting aggregate contractual entitlements under the related releasing and replacement contracts along any segment would exceed the MDTQ of the original contract, or (b) the quantities transported along any segment to the Primary Points of Delivery under the resulting aggregate related releasing and replacement contracts would exceed the MDTQ of the original contract. In addition, a Replacement Customer may choose only those primary points along the lateral segment on which it contracts for transportation service under the replacement contract, as those lateral segments are set forth in the description of Secondary Points in Section 6.4 below. In the event that Replacement Customer selects a new Primary Point of Delivery that is located within the acquired contract path, the portion of the path no longer covered by that contract is deemed to be unsubscribed capacity that may be sold by Algonquin for the term of the capacity release agreement. Upon termination of the capacity release agreement, all capacity covered by the original release, including the original Primary Points of Delivery, shall revert to the Releasing Customer, and any Primary Points of Delivery granted during the term of the capacity release agreement shall revert to Algonquin as unsubscribed capacity.
 - 6.4 Secondary Points of Delivery: Notwithstanding the foregoing, all interconnections between the Canal Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Canal Lateral, all interconnections between the Middletown facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Middletown Lateral, all interconnections between the Cleary Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Cleary Lateral, all interconnections between the Lake Road Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Lake Road Lateral, all interconnections between the Brayton Point Lateral facilities of Algonquin and the facilities of other operators including, but not limited to, the tap on the Algonquin G-1 System in Dighton, MA (M&R 80034) shall be available for use by Customer as Secondary Points of Delivery for service on the Brayton Point Lateral, all interconnections between the Bellingham Lateral facilities of Algonquin and facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Bellingham Lateral,

FERC Docket: RP10-146-000

Fourth Revised Sheet No. 340 Fourth Revised Sheet No. 340

Superseding: Third Revised Sheet No. 340

RATE SCHEDULE AFT-CL FIRM TRANSPORTATION SERVICE

all interconnections between the Phelps Dodge Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Phelps Dodge Lateral, all interconnections between the Manchester Street Lateral facilities of Algonquin and the facilities of other operators, including, but not limited to, the head of the G-12 Lateral (M&R No. 80070), shall be available for use by Customer as Secondary Points of Delivery for service on the Manchester Street Lateral, all interconnections between the Cape Cod Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Cape Cod Lateral, all interconnections between the Northeast Gateway Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Northeast Gateway Lateral, all interconnections between the J-2 Facility of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Receipt for service on the J-2 Facility, and all interconnections between the Kleen Energy Lateral facilities of Algonquin and the facilities of other operators shall be available for use by Customer as Secondary Points of Delivery for service on the Kleen Energy Lateral. Such use of Secondary Points of Delivery is subject to and pursuant to Section 48.2 of the General Terms and Conditions of this FERC Gas Tariff.

7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 341 - 350 Sheet Nos. 341 - 350 : Effective

Sheet Nos. 341 - 350 are being reserved for future use

FERC Docket: RP08-268-000

First Revised Sheet No. 351 First Revised Sheet No. 351 : Effective

Superseding: Original Sheet No. 351

RATE SCHEDULE PAL

PARK AND LOAN SERVICE

1. AVAILABILITY

This rate schedule is available for parking and loaning of natural gas on an interruptible basis by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer"), when:

- a. Customer has made a valid request for Park and Loan (PAL) Service pursuant to Section 2 of the General Terms and Conditions of the FERC Gas Tariff of which this rate schedule is a part; and
- b. After review and acceptance of such request by Algonquin, Customer and Algonquin have executed a PAL Service Agreement, in the form contained in the FERC Gas Tariff of which this rate schedule is a part, with Algonquin for service under this Rate Schedule; and
- c. Customer and Algonquin have executed a service agreement under any of Algonquin's open-access transportation Rate Schedules which provide for the transportation of gas quantities to or from the PAL Point(s) of Transaction, provided however, that no transportation service agreement is required if gas to be parked and/or loaned pursuant to this Rate Schedule is not transported by Algonquin. Such service agreements shall be in the form contained in the FERC Gas Tariff of which this rate schedule is a part.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Park and Loan Service shall be interruptible, as provided herein, in the effective PAL Service Agreement and in Algonquin's General Terms and Conditions. Individual Park and/or Loan Services provided under this rate schedule shall be provided for a minimum of one (1) Day or such longer period as agreed to by Customer and Algonquin, as specified in Exhibit(s) A to the Customer's PAL Service Agreement. Services shall be properly nominated pursuant to Section 22 of the General Terms and Conditions and such nominations shall include, without limitation, the term of the transaction, the quantity to be parked or loaned, as applicable, the PAL Point(s) of Transaction, and all other information required for a nomination under the General Terms and Conditions. Service under this Rate Schedule does not constitute "no-notice" service as that term is used in Order No. 636.
- 2.2 Park Service is an interruptible service pursuant to which Algonquin will: (a) receive quantities of gas for a Customer's account, at the nominated PAL Point(s) of Transaction, (b) hold such parked quantities, and (c) return such quantities of gas to the Customer at the PAL Point(s) of Transaction where Algonquin received the gas. The maximum quantity of gas in Customer's account as Park Service on any Day shall not exceed the Maximum Park Quantity ("MPQ") shown in Exhibit(s) A to the Customer's PAL Service Agreement.
- 2.3 Loan Service is an interruptible service pursuant to which Algonquin will: (a) deliver quantities of gas for a Customer's account, from the nominated PAL Point(s) of Transaction, and (b) subsequently receive such quantities of gas at the PAL Point(s) of Transaction where Algonquin delivered the gas. The maximum quantity of gas in Customer's account as Loan Service on any Day shall not exceed the Maximum Loan Quantity ("MLQ") shown in Exhibit(s) A to the Customer's PAL Service Agreement.
- 2.4 Service under this Rate Schedule shall be made available on a first come, first served basis, to any Customer willing to pay maximum rates or such other rate mutually agreed upon by Customer and Algonquin, subject to all applicable provisions of the General Terms and Conditions of the FERC Gas Tariff of which this rate schedule is a part.

FERC Docket: RP10-308-000

Second Revised Sheet No. 352 Second Revised Sheet No. 352

Superseding: First Revised Sheet No. 352

RATE SCHEDULE PAL

PARK AND LOAN SERVICE

- 2.5 Interruptible parking and loaning of natural gas under this Rate Schedule shall be provided when and to the extent that Algonquin determines, using its reasonable discretion, that capacity is available on its existing facilities and that it has the operational flexibility to provide the service without detriment or disadvantage to Algonquin's firm obligations or other interruptible services during the transaction period as specified in Exhibit(s) A to the Customer's PAL Service Agreement. Existing quantities of gas parked or loaned cannot be bumped by new requests for park and/or loan service.
- 2.6 Algonquin shall not be obligated to add any facilities or expand the capacity of its pipeline system in any manner in order to provide Park Service and/or Loan Service to Customer pursuant to this Rate Schedule. Algonquin is free to contract at any time with other parties for new storage or transportation services (whether firm or interruptible) and/or new service under this Rate Schedule without liability to Customer for any resulting interruption or reduction of service hereunder.

RATE

- 3.1 The applicable rates for service hereunder are set forth in the currently effective Statement of Rates for Rate Schedule PAL of this FERC Gas Tariff and hereby incorporated herein. The rates in this Rate Schedule are subject to adjustment pursuant to Section 36 of Algonquin's General Terms and Conditions. Unless Algonquin and Customer agree in writing upon a rate for service provided hereunder, the rate applicable to a Customer for service hereunder shall be the applicable maximum rate(s) as set forth on the effective Sheet No. 39. In the event a rate less than the applicable maximum rate(s) and not less than the applicable minimum rate(s) is agreed upon, such rate shall be applicable for the period agreed upon by Customer and Algonquin.
- 3.2 Algonquin shall submit and Customer shall pay a monthly invoice reflecting the sum of all Daily Charges for each Day during the applicable Month that PAL service is provided to the Customer by Algonquin. The Daily Charges shall be the product of the quantities of gas in Customer's PAL account for each separate Park and Loan transaction and the maximum or mutually agreed upon rate.

4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES

- 4.1 If Customer desires parking and loaning service of natural gas on any Day under this Rate Schedule, Customer shall submit a nomination to Algonquin in accordance with Section 22 of Algonquin's General Terms and Conditions. Customer shall nominate the agreed upon park or loan quantities and term at the PAL Point(s) of Transaction specified in Exhibit(s) A to the Customer's PAL Service Agreement. Such nominated quantities and term shall be subject to confirmation by Algonquin. Algonquin shall schedule the quantities nominated hereunder in accordance with Section 23 of Algonquin's General Terms and Conditions.
- 4.2 The park or loan quantity of gas in Customer's account on any Day shall not exceed the MPQ and/or the MLQ set forth in Exhibit(s) A to the Customer's PAL Service Agreement, as applicable. Algorquin shall reject any nomination if Customer nominates quantities when the balance(s) in Customer's account exceeds the specified MPQ and/or MLQ or when giving effect to the nomination would result in a balance(s) in Customer's account in excess of the specified MPQ and/or MLQ.

FERC Docket: RP04-361-000

Original Sheet No. 353 Original Sheet No. 353: Effective

RATE SCHEDULE PAL

PARK AND LOAN SERVICE

- 4.3 PAL services shall be provided on an interruptible basis. Interruption of PAL services may include decreasing, temporarily suspending, or discontinuing the receipt or delivery of gas if Algonquin in its reasonable discretion determines that such decrease, suspension or discontinuance is necessary to prevent the impairment of reliable service or when a higher priority service so requires. Nothing in this rate schedule shall limit Algonquin's right to take action pursuant to Section 26 of the General Terms and Conditions of this tariff.
- 4.4 Commingling of Gas. From the time the natural gas is received by Algonquin at the PAL Point(s) of Transaction, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.

5. OTHER OPERATING CONDITIONS

- 5.1 Customer shall make all necessary arrangements with other parties at or upstream of the PAL Point(s) of Transaction where Customer delivers gas to Algonquin for Park Service or where Customer delivers to decrease the balance in its account for Loan Service, and at or downstream of the PAL Point(s) of Transaction where Algonquin delivers gas to decrease the balance in Customer's account for Park Service or where Algonquin delivers gas to Customer for Loan Service, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue to receive gas on behalf of Customer, as contemplated in Section 5.1 of this Rate Schedule. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce an equal quantity of the deliveries of gas on behalf of Customer, as contemplated in Section 5.1 of this Rate Schedule.

6. PARK AND LOAN (PAL) POINT(S) OF TRANSACTION

Any physical or accounting point that exists from time to time on Algonquin's system shall be available for use by PAL Customers as Point(s) of Transaction. Algonquin shall permit a PAL Customer to park at the Point(s) of Receipt specified in Exhibit(s) A of Customer's PAL Service Agreement, and/or Algonquin shall loan to Customer at the Point(s) of Delivery specified in Exhibit(s) A of Customer's PAL Service Agreement, the quantity specified in Exhibit(s) A of Customer's PAL Service Agreement.

Customer shall provide a nomination, in accordance with Section 22 of the General Terms and Conditions, for each transaction identified in Exhibit(s) A to the Customer's PAL Service Agreement.

FERC Docket: RP10-77-000

First Revised Sheet No. 354 First Revised Sheet No. 354

Superseding: Original Sheet No. 354

RATE SCHEDULE PAL

PARK AND LOAN SERVICE

7. IMPAIRMENT OF RECEIPTS AND DELIVERIES

- 7.1 If Algonquin, on any Day is unable to receive and/or deliver the total requests of all Customers on Algonquin's system, Algonquin shall limit receipts and/or deliveries of gas hereunder in accordance with Sections 16 and 24 of Algonquin's General Terms and Conditions.
- 7.2 Customer may be required, upon notification from Algonquin, to suspend or reduce receipts or deliveries for the agreed upon Park Service and/or the agreed upon Loan Service. Further, Customer may be required, upon notification from Algonquin, to remove quantities of gas previously provided to Algonquin under the Park service, or return quantities of gas previously loaned to Customer under the Loan service. Notwithstanding anything to the contrary in Section 40 of the General Terms and Conditions, such notification shall be provided to affected parties through the affected party's choice of Electronic Notice Delivery mechanism(s).
- 7.3 Should Algonquin notify Customer to remove or return quantities of gas pursuant to Section 7.2, Algonquin's notification shall specify the time frame, which shall be no less than twenty-four (24) hours, within which Park service quantities shall be removed, or Loan service quantities shall be returned. Such notifications shall be consistent with Algonquin's operating conditions, unless Algonquin and Customer mutually agree to a different specific time frame. The obligation of Customer to comply with the issued notification shall continue until such time as Algonquin is able to recommence the PAL services. In the event that Customer does not comply with Algonquin's notice, the parked balance or the loaned balance will be resolved in accordance with Sections 7.5 and 7.6, as applicable, of this Rate Schedule PAL.
 - To the extent that Customer is unable to comply with Algonquin's notice to remove parked quantities or return loaned quantities either because Algonquin is not able to schedule Customer's nomination for such transaction's), the PAL penalties set forth in Sections 7.5 and 7.6, as applicable, shall not be assessed for those quantities that Algonquin is unable to schedule during the period that Algonquin is unable to schedule the Customer's nominations required by this Section 7.3.
- 7.4 If Customer requests delivery of parked quantities and Algonquin is unable to return the gas to Customer, then Customer has the option to: (i) suspend the Daily Charge and extend the term of the transaction until Algonquin is able to return the gas to Customer; or (ii) trade the balance with another Customer that has an offsetting loan balance at the same PAL Point of Transaction.
 - If Customer requests to return loaned quantities to Algonquin and Algonquin is unable to accept such receipts, then Customer has the option to: (i) suspend the Daily Charge and extend the term of the transaction until Algonquin is able to accept such receipts from Customer; or (ii) trade the balance with another Customer that has an offsetting park balance at the same PAL Point of Transaction.

FERC Docket: RP10-77-000

First Revised Sheet No. 355 First Revised Sheet No. 355

Superseding: Original Sheet No. 355

RATE SCHEDULE PAL

PARK AND LOAN SERVICE

- 7.5 Parked quantities shall become the property of Algonquin at no cost to Algonquin, free and clear of any adverse claims, in the event that any of the following situations occur:
 - (a) Algonquin's operating conditions require Algonquin to notify Customer that receipts of parked quantities must be suspended or reduced, and Customer fails to comply with such notification within the specified time frame; and/or
 - (b) Algonquin's operating conditions require Algonquin to notify Customer that all or a part of Customer's parked quantities must be removed, and Customer fails to comply with such notification within the specified time frame; and/or
 - (c) The PAL account reflects a balance at the termination date of the associated Exhibit A to the Customer's PAL Service Agreement or at the end of any mutually agreed upon extension of the termination date, and Algonquin's operating conditions require the removal of the parked quantities.

If Algonquin notifies Customer, pursuant to Sections 7.2 and 7.5(a) herein, that receipts of parked quantities must be suspended or reduced, only those quantities parked by the Customer in violation of the notification shall become the property of Algonquin at no cost to Algonquin, free and clear of any adverse claims.

Algonquin shall credit the Fuel Reimbursement Quantity Deferred Account as provided in Section 32 of the General Terms and Conditions of this FERC Gas Tariff with the value of the gas retained pursuant to this Section 7.5. The value of the gas retained by Algonquin shall be based on the average of the Gas Daily postings for the high "Common" price for the "AGT Citygate" posting for the seven Day period beginning three Days before the Day on which the park occurred.

- 7.6 Title to loaned quantities shall be transferred to Customer and Customer shall be obligated to pay for each Dekatherm of such gas at a price equal to 150% of the average of the Gas Daily postings for the "Midpoint" price for the "AGT Citygate" posting for the seven Day period beginning three Days before the Day on which the loan occurred, in the event that any of the following situations occur:
 - (a) Algonquin's operating conditions require Algonquin to notify Customer that deliveries of Customer's loaned quantities must be suspended or reduced, and Customer fails to comply with such notification; and/or
 - (b) Algonquin's operating conditions require Algonquin to notify Customer that all or a part of Customer's loaned quantities must be returned to Algonquin, and Customer fails to comply within the specified time frame; and/or
 - (c) The PAL account reflects a balance at the termination date of the associated Exhibit A to the Customer's PAL Service Agreement or at the end of any mutually agreed upon extension of the termination date.

If Algonquin notifies Customer, pursuant to Sections 7.2 and 7.6(a) herein, that deliveries of Customer's loaned quantities must be suspended or reduced, only those quantities taken by the Customer in violation of the notification shall be transferred to Customer and Customer shall be obligated to pay for each Dekatherm of such gas at a price calculated as described in this Section 7.6. Algonquin shall credit the Fuel Reimbursement Quantity Deferred Account as provided in Section 32 of the General Terms and Conditions, with the net proceeds from the operation of this Section 7.6.

FERC Docket: RP10-308-000

First Revised Sheet No. 356 First Revised Sheet No. 356

Superseding: Original Sheet No. 356

RATE SCHEDULE PAL

PARK AND LOAN SERVICE

8. BALANCES

8.1 All PAL balance quantities reside at the PAL Point of Transaction at which the original park or loan quantity was nominated, and must be resolved at that PAL Point of Transaction by taking delivery of the Park Balance or returning the Loan Balance to Algonquin, as applicable. Balances incurred under this Rate Schedule PAL shall be resolved via in-kind balancing during the remaining term of the transaction, which term is specified in Exhibit(s) A to the Customer's PAL Service Agreement. Customer may resolve park and/or loan balances by transporting such balances between PAL Points of Transaction. The transportation shall be accomplished by Customer's submission of a nomination under Customer's executed transportation service agreement and the subsequent scheduling and confirmation processes pursuant to Sections 22 and 23 of the General Terms and Conditions.

Customer shall be required to reduce to zero each park and/or loan quantity of gas at each PAL Point of Transaction no later than the termination date of the Exhibit(s) A to the Customer's PAL Service Agreement, or prior to any mutually agreed upon extension of that term, applicable to that quantity. Any balances remaining upon the termination date of the Exhibit(s) A to the Customer's PAL Service Agreement, or any mutually agreed upon extension of that term, shall be resolved in accordance with Sections 7.5 and 7.6, as applicable, of this rate Schedule PAL.

Algonquin shall debit or credit, as appropriate, the Fuel Reimbursement Quantity Deferred Account as provided in Section 32 of the General Terms and Conditions, with the net proceeds, from the operation of this Section 8.

8.2 If Customer's PAL account reflects a park balance at the termination date of the associated Exhibit A to Customer's PAL Service Agreement or at the end of any mutually agreed upon extension of the termination date, and the balance has not been resolved pursuant to Section 7.5(c) of this Rate Schedule PAL, the term of the transaction will be extended until the earlier of (i) such time as the balance is reduced to zero or (ii) operating conditions require the removal of the balance pursuant to Section 7.5(c) of this Rate Schedule PAL. For any balance remaining beyond such agreed upon period set forth for a PAL transaction in Exhibit A of Customer's PAL Service Agreement, the applicable maximum rate(s) as set forth on the effective Statement of Rates for Rate Schedule PAL shall be applied on a daily basis until the balance is reduced to zero.

9. GOVERNMENTAL AUTHORIZATIONS

Park and Loan Services under executed PAL Service Agreements shall be implemented pursuant to any applicable self-implementing authorization or program of the FERC for which Algonquin has filed or in which Algonquin has agreed to participate.

10. GENERAL TERMS AND CONDITIONS

All of the General Terms and Conditions of Algonquin's FERC Gas Tariff of which this Rate Schedule is a part are applicable to this Rate Schedule and are made a part hereof to the extent that such terms and conditions are not contradicted by any provision herein.

FERC Docket: RP04-361-000

Sheet Nos. 357 - 360 Sheet Nos. 357 - 360 : Effective

Sheet Nos. 357 - 360 are being reserved for future use

FERC Docket: RP10-77-000

First Revised Sheet No. 361 First Revised Sheet No. 361

Superseding: Original Sheet No. 361

RATE SCHEDULE TTT
TITLE TRANSFER TRACKING SERVICE

1. AVAILABILITY

- 1.1 This Open-access Rate Schedule is available to any party (hereinafter called "TTT Party") which has requested Title Transfer Tracking ("TTT") service pursuant to Section 2 of Algonquin Gas Transmission, LLC's (hereinafter called "Algonquin") General Terms and Conditions and, after review and acceptance of such request by Algonquin, has entered into a service agreement with Algonquin for service under Rate Schedule TTT. Such service agreement shall be in the form contained in Algonquin's FERC Gas Tariff of which this Rate Schedule TTT is a part. In addition, this Rate Schedule is available to any Third Party Account Administrator ("3PAD") who has either (i) been designated as an agent for a TTT Party, or (ii) executed a service agreement with Algonquin for service under this Rate Schedule TTT. For the purposes of this rate schedule, the term "TTT Party" shall also refer to a 3PAD.
- 1.2 Title Transfer Tracking Service will be made available pursuant to the scheduling provisions of Sections 22, 23 and 24 of the General Terms and Conditions of this FERC Gas Tariff, to any TTT Party that is willing and able to pay either the maximum rates hereunder or another rate to which Algonquin and TTT Party mutually agree, in accordance with the General Terms and Conditions. Algonquin is not required to provide any requested TTT service under this Rate Schedule for which Algonquin determines that administrative capability is not available, or that would interfere with Algonquin's performance of its firm service obligations.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 This Rate Schedule shall apply to quantities of natural gas for which Algonquin provides TTT service on behalf of TTT Party, subject to an executed TTT Agreement that conforms to the form of Agreement contained in this Tariff. TTT service is an administrative service, under which Algonquin provides TTT Party with the ability to submit nominations for Title Transfer transactions at any receipt location on Algonquin's system, provided that the point is covered by an active Operational Balancing Agreement. These receipt locations shall be referred to in this Rate Schedule TTT as "TTT Points". Upon receipt of a nomination from the TTT Party, Algonquin will process the nomination at the TTT Point as directed by the TTT Party.
- 2.2 TTT Party or its customers must arrange for transportation service in accordance with Algonquin's FERC Gas Tariff under separate transportation service agreements (the "Subject Service Agreements"), for physical delivery of natural gas quantities subsequent to Title Transfers that are tracked pursuant to nominations under this Rate Schedule, or for additional TTT service under separate TTT Service Agreements. This Rate Schedule does not entitle TTT Party to transportation service under Algonquin's FERC Gas Tariff separate from the entitlements under the Subject Service Agreements, nor does it entitle TTT Party to maintain imbalances on Algonquin's system.

FERC Docket: RP10-308-000

Second Revised Sheet No. 362 Second Revised Sheet No. 362

Superseding: First Revised Sheet No. 362

RATE SCHEDULE TTT TITLE TRANSFER TRACKING SERVICE

2. APPLICABILITY AND CHARACTER OF SERVICE (continued)

- 2.3 Physical deliveries of the quantities of gas for which the title was tracked pursuant to this Rate Schedule shall be performed in accordance with provisions of the Subject Service Agreements and the terms of this FERC Gas Tariff. Quantities nominated under the TTT Service Agreement shall have the same scheduling priority as the Subject Service Agreement(s) used to physically transport the gas away from the TTT Point.
- 2.4 Nothing herein shall limit Algonquin's right to take action as may be required to adjust receipts of gas in order to alleviate conditions which threaten the integrity of its system, or the ability of Algonquin to transport firm quantities scheduled by any Customer, including actions taken pursuant to Sections 23 and 24 of the General Terms and Conditions.

RATES

The applicable rates and charges for service provided to TTT Party under this Rate Schedule shall include those uniform rates set forth in the currently effective Statement of Rates for Rate Schedule TTT of this FERC Gas Tariff. These rates and charges are incorporated herein by reference. Unless Algonquin and TTT Party agree in writing upon a rate for service provided hereunder, the rate applicable to a TTT Party for service hereunder shall be the applicable maximum rate(s). In the event a rate less than the applicable maximum rate(s) and not less than the applicable minimum rate(s) is agreed upon, such rate shall be applicable for the period agreed upon by Algonquin and TTT Party.

4. MONTHLY BILL

- 4.1 For TTT service provided under this Rate Schedule, TTT Party shall pay Algonquin each month the sum of the following:
 - A. TTT Service Charge. A charge for transaction nominated and scheduled for TTT service;
 - B. Any other applicable rates, charges and penalties as set forth in the General Terms and Conditions of this FERC Gas Tariff.
- 4.2 Notwithstanding the general provisions of Sections 3 and 4.1, above, if Algonquin and TTT Party mutually agree to Negotiated Rates for service hereunder, then such Negotiated Rates shall apply in lieu of the otherwise applicable charges identified in Section 4.1.A of this Rate Schedule.

FERC Docket: RP10-77-000

First Revised Sheet No. 363 First Revised Sheet No. 363

Superseding: Original Sheet No. 363

RATE SCHEDULE TTT
TITLE TRANSFER TRACKING SERVICE

5. TTT POINTS

Any physical receipt point located on Algonquin's system ("TTT Point"), provided that the point is covered by an active Operational Balancing Agreement, can be utilized for the nomination of TTT service. Algonquin will permit TTT Party to nominate Title Transfers of quantities of natural gas at any of the TTT Points, subject to the existence of adequate corresponding transportation rights under Subject Service Agreement(s) to transport the quantities away from the TTT Point.

Algonquin shall provide one meter identification number to correspond to each location at which Title Transfer Tracking services for that location can be effected. This meter identification will be used exclusively for nominations of transactions related to Title Transfer Tracking service on Algonquin's system. Algonquin shall post the meter identification numbers on its Internet Web Site.

6. TITLE TRANSFER TRACKING PROCEDURES

6.1 General Procedure.

- A. For any Day on which a TTT Party desires to utilize the TTT service to track a Title Transfer for TTT Party's account under this Rate Schedule, such TTT Party shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions, specifying:
 - (1) the quantity of gas for which title is to be transferred,
 - (2) the upstream party participating in the TTT transaction, and the upstream party's associated upstream contract identifier,
 - (3) the downstream party participating in the TTT transaction, and the downstream party's associated downstream contract identifier,
 - (4) the meter identification number that corresponds to the physical receipt location at which the TTT service is being performed, and
 - (5) such other information as may be required by Algonquin to effect the TTT service.
- B. When TTT Party's nomination is confirmed as required by the General Terms and Conditions and subject to the limitations set forth in this Rate Schedule, such quantity shall constitute a TTT Scheduled Quantity at that TTT Point.
- 6.2 TTT Party is required to ensure that quantities transferred to the transferee(s) of TTT Party and scheduled under a Subject Service Agreement or another TTT Service Agreement equal the quantities received from TTT Party's transferor(s) at each TTT Point on each Day.

FERC Docket: RP04-361-000

Original Sheet No. 364 original Sheet No. 364: Effective

RATE SCHEDULE TTT TITLE TRANSFER TRACKING SERVICE

6. TITLE TRANSFER TRACKING PROCEDURES (continued)

- 6.3 On or before the nomination deadlines applicable to Algonquin's service under the Subject Service Agreement(s), TTT Party shall provide nomination data to Algonquin consistent with the requirements of Section 22 of the General Terms and Conditions, for each of TTT Party's TTT Points. Algonquin shall confirm that the nomination data received from TTT Party is equivalent to valid nominations under another TTT Service Agreement(s) or under Subject Service Agreements that will physically transport the quantities away from the TTT Point(s). The quantities determined through this process shall be referred to as the "TTT Scheduled Quantities."
- 6.4 After confirmation and scheduling, any changes to TTT Scheduled Quantities shall be effective only if a corresponding nomination change has been made under a Subject Service Agreement and the change has been confirmed by Algonquin, the affected point operator(s) and any other affected TTT Party.
- 6.5 If any nominated service under a Subject Service Agreement fails to be scheduled for any reason, then Algonquin shall notify TTT Party, and TTT Party shall revise the nomination data on a prospective basis to reflect the revised TTT Quantities.
- 6.6 TTT Party shall make personnel available to Algonquin at all times, for purposes of the nomination and confirmations to be performed hereunder.
- 6.7 With respect to the TTT service, Algonquin shall be entitled to rely upon the information set forth in the nomination of a TTT Party made in accordance with Section 22 of the General Terms and Conditions, and Algonquin shall have no obligation to verify the accuracy of such information.

7. GENERAL TERMS AND CONDITIONS

All of the General Terms and Conditions of Algonquin's FERC Gas Tariff of which this Rate Schedule is a part are applicable to this Rate Schedule and are made a part hereof to the extent that such terms and conditions are not contradicted by any provision herein. For purposes of the applicable General Terms and Conditions of Algonquin's FERC Gas Tariff, references to Customer shall refer to the TTT Party.

FERC Docket: RP04-361-000

Sheet Nos. 365 - 370 Sheet Nos. 365 - 370 : Effective

Sheet Nos. 365 - 370 are being reserved for future use

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 371 Sixth Revised Sheet No. 371

Superseding: Fifth Revised Sheet No. 371

RATE SCHEDULE AIT-2 INTERRUPTIBLE TRANSPORTATION SERVICE

AVATTABITATY

This rate schedule is available for interruptible transportation of natural gas by Algonquin Gas Transmission, LLC (hereinafter called "Algonquin") for any party (hereinafter called "Customer") on the Brayton Point Lateral, on the Manchester Street Lateral, on the Canal Lateral, on the Cape Cod Lateral, on the Northeast Gateway Lateral, on the J-2 Facility, or on the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL, when:

- a. Customer has made a valid request for interruptible transportation pursuant to Section 2 of the General Terms and Conditions of this FERC Gas Tariff of which this rate schedule is a part; and
- b. Customer has executed a service agreement in the form contained in the FERC Gas Tariff of which this rate schedule is a part; and
- C. Customer's service agreement specifies that either the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as such lateral facilities are defined in Rate Schedule AFT-CL, shall be utilized to effectuate service hereunder.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Transportation service hereunder will be on an interruptible basis. Algonquin shall receive from Customer, or for the account of Customer, at those points on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as specified in Customer's executed AIT-2 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Receipt") for transportation hereunder daily quantities of gas tendered for the account of Customer up to Customer's Maximum Daily Transportation Quantity ("MDTQ") plus an amount reflecting the Fuel Reimbursement Quantity, if applicable, as defined in Section 32 of the General Terms and Conditions of this tariff and, on a cumulative basis in any year, up to Customer's Maximum Annual Transportation Quantity ("MATQ") as specified in the service agreement plus amounts reflecting the Fuel Reimbursement Quantity, if applicable.
- 2.2 Upon receipt of such natural gas for Customer's account, Algonquin shall, after making allowance for the Fuel Reimbursement Quantity, if applicable, transport and deliver hourly quantities of gas required by Customer up to Customer's Maximum Hourly Transportation Quantity ("MHTQ") at those points on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as specified in Customer's executed AIT-2 Service Agreement between Customer and Algonquin (hereinafter referred to as "Point(s) of Delivery"), provided, however, Algonquin shall not be obligated to, but may at its option, deliver at any Point(s) of Delivery an hourly quantity exceeding the MHTQ.
- Provided such quantities have been scheduled in accordance with Section 23 of the General Terms and Conditions, Customer may tender quantities of gas in excess of the MDTQ plus any applicable Fuel Reimbursement Quantity on any Day if in Algonquin's reasonable judgment transportation of such gas can be accomplished by Algonquin without detriment to any other Customer under any of Algonquin's rate schedules. Such excess quantities shall be deemed to be Authorized Overrun Quantities.
- 2.4 Algonquin shall not be obligated to add any facilities or expand the capacity of Algonquin's pipeline system in any manner in order to provide transportation service to Customer pursuant to this rate schedule; provided, however, Algonquin may, at its option, and with Customer's consent, add facilities or expand capacity to provide such transportation service, subject to Section 42 of the General Terms and Conditions of this tariff.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 372 Fourth Revised Sheet No. 372

Superseding: Third Revised Sheet No. 372

RATE SCHEDULE AIT-2 INTERRUPTIBLE TRANSPORTATION SERVICE

3. RATE

- 3.1 Unit Rates. The applicable maximum and minimum unit rates are set forth in the currently effective Statement of Rates for Rate Schedule AIT-2 of this FERC Gas Tariff and are hereby incorporated herein. Such rates are subject to adjustment pursuant to Section 33 and Section 34 of the General Terms and Conditions of this tariff. The applicable unit rates to be charged on any Day by Algonquin for gas delivered to Customer shall not be in excess of the maximum unit rate nor less than the minimum unit rate, except as provided in Section 46 of the General Terms and Conditions of Algonquin's FERC Gas Tariff.
- 3.2 Monthly Bill. Commencing for the Month in which the AIT-2 Service Agreement is effective and for each Month thereafter, Algonquin shall charge Customer and Customer shall pay Algonquin the sum of the following amounts:
 - (a) Commodity Charge: The applicable commodity rate multiplied by the quantity of gas delivered in the Month (excluding Authorized Overrun Quantities) at the Point(s) of Delivery; plus
 - (b) Authorized Overrun Charge: The applicable authorized overrun charge per Dth of Authorized Overrun Quantity delivered to Customer for the Month under this rate schedule; plus
 - (c) Imbalance Resolution Charges: The applicable imbalance resolution charges assessed pursuant to Section 25 of the General Terms and Conditions; plus
 - (d) Scheduling Penalties: The applicable scheduling penalties assessed pursuant to Section 23 of the General Terms and Conditions; plus
 - (e) Unauthorized Contract Overrun Penalties: The applicable unauthorized contract overrun penalties assessed pursuant to Section 31 of the General Terms and Conditions.

FERC Docket: RP10-146-000

Sixth Revised Sheet No. 373 Sixth Revised Sheet No. 373

Superseding: Fifth Revised Sheet No. 373

RATE SCHEDULE AIT-2 INTERRUPTIBLE TRANSPORTATION SERVICE

4. NOMINATIONS AND SCHEDULING OF RECEIPTS AND DELIVERIES

- 4.1 Nominations and Scheduling. If Customer desires transportation of natural gas on any Day under this rate schedule, Customer shall provide a nomination to Algonquin in accordance with Section 22 of the General Terms and Conditions of this tariff. Based upon the nomination of Customer, Algonquin shall schedule receipts and deliveries of gas in accordance with the General Terms and Conditions. It is the responsibility of Customer to adjust its deliveries and receipts to conform to the scheduled quantities.
- 4.2 Delivery of Gas. Based upon the daily quantity scheduled, Algonquin shall make daily delivery of Customer's scheduled quantity taking into account the Fuel Reimbursement Quantity. It is the intention of Algonquin that daily deliveries of gas at the Point(s) of Delivery by Algonquin hereunder shall be as nearly equal as possible to daily receipts of gas at the Point(s) of Receipt by Algonquin for transportation hereunder, less the applicable Fuel Reimbursement Quantity. Any excess or deficiency in such receipts, less the applicable Fuel Reimbursement Quantity, and deliveries shall be resolved in accordance with the General Terms and Conditions. Nothing in this rate schedule shall limit Algonquin's right to take actions pursuant to Section 26 of the General Terms and Conditions of this tariff.
- 4.3 Responsibility for Imbalances. Any imbalance resulting from transportation transactions pursuant to this rate schedule on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2 Facility, or the Kleen Energy Lateral, as specified on Customer's AIT-2 Service Agreement for which natural gas was received under an upstream transportation agreement(s) on Algonquin's system shall be accounted for under the upstream transportation agreement(s). No imbalance resolution charges, unauthorized overrun penalties, or scheduling penalties shall be assessed under this rate schedule for such transactions to the extent that Algonquin has assessed any such charges or penalties under the upstream transportation agreement(s) with respect to the same gas or transactions.
- 4.4 Commingling of Gas. From the time the natural gas is received by Algonquin at the Point(s) of Receipt, Algonquin shall have the unqualified right to commingle such natural gas with other gas in Algonquin's system.

FERC Docket: RP10-146-000

Fifth Revised Sheet No. 374 Fifth Revised Sheet No. 374

Superseding: Fourth Revised Sheet No. 374

RATE SCHEDULE AIT-2 INTERRUPTIBLE TRANSPORTATION SERVICE

5. OTHER OPERATING CONDITIONS

Algonquin's obligation to provide service under this rate schedule is subject to the following conditions being satisfied:

- 5.1 Customer shall make all necessary arrangements with Algonquin or other parties, as applicable, at or upstream of the Point(s) of Receipt where Customer tenders gas to Algonquin for transportation, and at or downstream of the Point(s) of Delivery where Algonquin delivers gas for Customer's account, and such arrangements must be compatible with Algonquin's system operations.
- 5.2 To the extent that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer. To the extent that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer.
- 5.3 The daily quantities of natural gas transported shall be delivered at the Point(s) of Receipt and accepted at the Point(s) of Delivery at a substantially constant hourly rate, or such other hourly rate as may be acceptable to Algonquin.

POINT(S) OF RECEIPT AND DELIVERY

- All interconnections between the Brayton Point Lateral facilities, including, but not limited to, the tap on Algonquin's G-1 System in Dighton, MA (M&R No. 80034); the Manchester Street Lateral facilities, including, but not limited to the head of the G-12 Lateral (M&R No. 80070); the Canal Lateral facilities, including, but not limited to, the interconnection between the Canal Lateral and Algonquin's mainline; the Cape Cod Lateral facilities, including, but not limited to, the interconnection between the Cape Cod Lateral and the Canal Lateral; the Northeast Gateway Lateral facilities, including, but not limited to, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts; the J-2 Facility, including, but not limited to the head of the J-2 Facility (M&R No. 80094); or the Kleen Energy Lateral facilities, including, but not limited to, the interconnection between the Kleen Energy Lateral and the Middletown Lateral, as specified in Customer's executed AIT-2 Service Agreement, and the facilities of other operators shall be available for use by Customer as Point(s) of Receipt.
- All interconnections between the Brayton Point Lateral facilities, including, but not limited to, the tap on Algonquin's G-1 System in Dighton, MA (M&R No. 80034); the Manchester Street Lateral facilities, including, but not limited to the head of the G-12 Lateral (M&R No. 80070); the Canal Lateral facilities, including, but not limited to, the interconnection between the Canal Lateral and Algonquin's mainline; the Cape Cod Lateral facilities, including, but not limited to, the interconnection between the Cape Cod Lateral and the Canal Lateral; the Northeast Gateway Lateral facilities, including, but not limited to, the interconnection between the Northeast Gateway Lateral and the HubLine offshore system in Massachusetts Bay, Massachusetts; the J-2 Facility, including, but not limited to the head of the J-2 Facility (M&R No. 80094); or the Kleen Energy Lateral facilities, including, but not limited to, the interconnection between the Kleen Energy Lateral and the Middletown Lateral, as specified in Customer's executed AIT-2 Service Agreement, and the facilities of other operators shall be available for use by Customer as Point(s) of Delivery.

7. GENERAL TERMS AND CONDITIONS

The applicable General Terms and Conditions of this FERC Gas Tariff are hereby made a part of this rate schedule.

FERC Docket: RP04-361-000

Sheet Nos. 375 - 499 Sheet Nos. 375 - 499: Effective

Sheet Nos. 375 - 499 are being reserved for future use

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 500 Fourth Revised Sheet No. 500 Superseding: Third Revised Sheet No. 500

GENERAL TERMS AND CONDITIONS

INDEX

0	Danadakina Hasilan	Thest No
Section 1.	Descriptive Heading Definitions	Sheet No. 501
2.	Request for Transportation Service	508
3.	Credit Evaluation	517
4.	Quality of Gas	519
5.	Measurement	522
6.	Measuring Equipment	524
7.	Receipt and Delivery Pressures	527
8.	Term of Executed Service Agreements	528
9.	Pregranted Abandonment and Right of First Refusal	528
10.	Joint Obligations	531
11.	Discretionary Waiver	531
12.	Modification	531
13.	Successors in Interest	531
14.	Capacity Release	532
15.	Termination for Default	551
16.	Force Majeure	552
17.	Governmental Regulation	553
18.	Billing	553
19.	Payment	554
20.	Possession of Gas	557
21.	Warranty of Title to Gas	557
22.	Nominations	558
23.	Scheduling of Service	562
24.	Curtailment	566
25.	Imbalance Resolution Procedures	572
26.	Actions Required To Protect System Operations	578
27.	Determination of Receipts	580
28. 29.	Determination of Deliveries	581
∠9. 30.	Operational Flow Orders	584 586
31.	Operational Balancing Agreements Unauthorized Contract Overrun Gas	588
32.	Fuel Reimbursement Quantity	589
33.	Gas Research Institute Charge Adjustment Provision	
34.	Annual Charge Adjustment Clause Provision	595
35.	[Reserved for Future Use]	596
36.	Periodic Rate Adjustments	602
37.	[Reserved for Future Use]	602
38.	Standards of Conduct Compliance Procedures	605
39.	Complaint Procedures	606
40.	Electronic Communication	607
41.	Revenue Credits	609
42.	Fees and Construction of New Facilities	612
43.	Reports with Respect to Tariff	613
44.	North American Energy Standards Board ("NAESB")	614
45.	Discount Terms	615
46.	Negotiated Rates	616
47.	Off-system Pipeline Capacity	617
48.	Segmentation, Secondary Point Rights and Related Scheduling	618
49.	Materially Non-Conforming Service Agreements	625

FERC Docket: RP10-77-000

Second Revised Sheet No. 501 Second Revised Sheet No. 501

Superseding: First Revised Sheet No. 501

GENERAL TERMS AND CONDITIONS

1. DEFINITIONS

The following terms and abbreviations, when used herein or in any agreement incorporating these General Terms and Conditions, are intended and used and shall be construed to have meanings as follows:

Algonquin. The term "Algonquin" refers to Algonquin Gas Transmission, LLC in its role as a provider of transportation services.

Backhaul. The term "Backhaul" shall mean movement of gas from a Point of Receipt to a Point of Delivery such that the contractual direction of movement on the mainline is at all times and at all points along the path in a direction opposite to the actual flow of gas in the pipeline.

Base Flow Path. The term "Base Flow Path" shall mean the path determined by the assumed gas flow from the Primary Point of Receipt to the Primary Point of Delivery as such points are specified in the service agreement under a firm transportation rate schedule executed by Algonquin and Customer.

Btu. The term "Btu" or "British thermal unit" shall mean the amount of heat required to raise the temperature of one avoirdupois pound of pure water from 58.5° Fahrenheit to 59.5° Fahrenheit at a constant pressure of 14.73 dry psia.

Business Day. The term "Business Day" shall mean Monday through Friday excluding Federal Banking Holidays for transactions in the U.S. and similar holidays for transactions occurring in Canada and Mexico.

Cash-out Party. The term "Cash-out Party" shall mean any Customer or other contractually liable entity who has an imbalance under any agreement with Algonquin, which imbalance will be resolved in accordance with Section 25 of the General Terms and Conditions.

Commission or FERC. The term "Commission" or "FERC" shall mean the Federal Energy Regulatory Commission or its predecessor agency, the Federal Power Commission, or any successor agency.

Company Use Gas. The term "Company Use Gas" shall mean the amount of gas used for fuel, including compressor and heater fuel; gas used for maintenance; line pack, gas lost as a result of force majeure events, the ownership of which cannot be reasonably identified; and unaccounted for gas.

Contract Year. The term "Contract Year" shall mean a period of one year commencing each November $1. \,$

CT. The term "CT" or "CCT" shall mean Central Clock Time, which includes the recognition of Daylight Savings Time, whereas "CST" shall mean Central Standard Time.

FERC Docket: RP10-77-000

Third Revised Sheet No. 502 Third Revised Sheet No. 502

Superseding: Second Revised Sheet No. 502

GENERAL TERMS AND CONDITIONS

DEFINITIONS (continued)

Cubic Foot. The term "Cubic Foot" shall mean the volume of gas which occupies 1 Cubic Foot of space, measured according to Boyle's and Charles's Law for the measurement of gas under varying pressures with deviation therefrom as provided in Section 5 of these General Terms and Conditions and on the measurement basis likewise specified in such

Customer. The term "Customer" shall mean the entity that has executed a Service Agreement with Algonquin under one or more of Algonquin's rate schedules.

Day. The term "Day" or "Gas Day" shall mean a period of 24 consecutive Hours, beginning at 9:00 a.m. Central Clock Time.

Dekatherm. The term "Dth" or "Dekatherm" shall mean the quantity of heat energy which is equivalent to 1,000,000 British Thermal Units. One "Dekatherm" of gas shall mean the quantity of gas which contains one Dekatherm of heat energy.

Discount Confirmation. The term "Discount Confirmation" shall mean an electronic mail (e-mail) message sent by Algonquin to Customer to confirm the terms of the discount granted pursuant to Section 45 of the General Terms and Conditions of Algonquin's FERC Gas Tariff.

Elapsed Prorata Capacity. The term "Elapsed Prorata Capacity" shall mean that portion of the capacity that would have theoretically been available for use prior to the effective time of the intraday recall based upon a cumulative uniform hourly use of the capacity.

Hour. The term "Hour" shall mean a period of 60 consecutive minutes.

Imbalance Management Services. The term "Imbalance Management Services" shall mean the options available to Customers for resolution of imbalances prior to the application of the cash-out mechanism set forth in Section 25 of the General Terms and Conditions. These options include, but are not limited to, OBA (where applicable), Park and Loan Service, and Imbalance Netting and Trading.

FERC Docket: RP10-77-000

Second Revised Sheet No. 503 Second Revised Sheet No. 503

Superseding: First Revised Sheet No. 503

GENERAL TERMS AND CONDITIONS (continued)

1. DEFINITIONS (continued)

Maximum Annual Transportation Quantity. The term "Maximum Annual Transportation Quantity" or "MATQ" shall mean the maximum quantity of natural gas in Dth which Algonquin is willing to receive for transportation for the account of Customer during any Contract Year at all Point(s) of Receipt specified in an executed Service Agreement, exclusive of any applicable Fuel Reimbursement Quantity. The MATQ under any Service Agreement that provides for an MATQ that is equal to 365 times the MDTQ shall be increased by 1/365 during any leap year. For all other agreements, the MATQ shall be increased during any leap year by an amount equal to the MATQ multiplied by the quotient derived by dividing the MDTQ by the MATQ.

Maximum Daily Delivery Obligation. The term "Maximum Daily Delivery Obligation" or "MDDO" shall mean the maximum quantity of natural gas which Algonquin is willing to deliver on any Day at any indicated Point(s) of Delivery as specified in an executed Service Agreement for the applicable period.

Maximum Daily Receipt Obligation. The "Maximum Daily Receipt Obligation" or "MDRO" shall mean the maximum quantity of natural gas which Algonquin is willing to receive on any Day at any indicated Point(s) of Receipt as specified in an executed Service Agreement for the applicable period, exclusive of any applicable Fuel Reimbursement Quantity.

Maximum Daily Transportation Quantity. The term "Maximum Daily Transportation Quantity" or "MDTQ" shall mean the maximum quantity of natural gas in Dth which Algonquin is willing to receive for transportation for the account of Customer on any Day at all Point(s) of Receipt specified in an executed Service Agreement for the applicable period, exclusive of any applicable Fuel Reimbursement Quantity.

Maximum Hourly Transportation Quantity. The term "Maximum Hourly Transportation Quantity" or "MHTQ" shall mean the maximum quantity of gas in Dth which Algonquin is willing to deliver to Customer during any Hour at all Points of Delivery specified in an executed Service Agreement. Such quantity shall equal 1/24th of the MDTQ for all Service Agreements except Service Agreements resulting from the conversion pursuant to Order No. 636 of sales entitlements under former Rate Schedules F-1 and WS-1, under which the MHTQ shall be equal to the MDTQ for the applicable period multiplied by .06.

FERC Docket: RP10-77-000

Third Revised Sheet No. 504 Third Revised Sheet No. 504

Superseding: Second Revised Sheet No. 504

GENERAL TERMS AND CONDITIONS (continued)

1. DEFINITIONS (continued)

Maximum Loan Quantity. The term "Maximum Loan Quantity" or "MLQ" shall mean the maximum quantity of gas permitted to be in Customer's account on any Day as Loan Service as specified in Exhibit(s) A to the Customer's PAL Service Agreement.

Maximum Park Quantity. The term "Maximum Park Quantity" or "MPQ" shall mean the maximum quantity of gas permitted to be in Customer's account on any Day as Park Service as specified in Exhibit(s) A to the Customer's PAL Service Agreement.

Mcf. The term "Mcf" shall mean 1,000 Cubic Feet of gas.

 ${\tt MMBtu.}$ The term "MMBtu" shall mean one million Btu. One MMBtu equals one Dekatherm.

Month. The term "Month" shall mean the period beginning at 9:00 a.m. Central Clock Time on the first "Day" of the calendar month and ending at the same hour on the first "Day" of the next succeeding calendar month.

Negotiated Rate. The term "Negotiated Rate" shall mean a rate or rate formula mutually agreed upon by Algonquin and a Customer pursuant to Section 46 of the General Terms and Conditions with respect to rates, rate components, charges, or credits differing from the otherwise applicable Recourse Rate. The Negotiated Rate may be less than, equal to, or greater than the maximum Recourse Rate; shall not be less than the minimum tariff rate; may be a rate design other than the straight fixed-variable; and may include a minimum quantity.

Netting. The term "Netting" is used to describe the process of resolving imbalances for a Service Requester within an Operational Impact Area. There are two types of Netting:

- a. Summing is the accumulation of all imbalances above any applicable tolerances for a Service Requester or agent.
- b. Offsetting is the combination of positive and negative imbalances above any applicable tolerances for a Service Requester or agent. For Algonquin, the definition in (a) above is applicable to the Imbalance Netting provisions set forth in Section 25.9 of the General Terms and Conditions.

New Interruptible Agreement. The term "New Interruptible Agreement" shall mean all agreements for interruptible transportation service other than those specified defined as an Old Interruptible Agreement.

Nomination Period. The term "Nomination Period" shall mean a period of time Customer includes in a nomination for gas service.

FERC Docket: RP10-77-000

Second Revised Sheet No. 505 Second Revised Sheet No. 505

Superseding: First Revised Sheet No. 505

GENERAL TERMS AND CONDITIONS (continued)

1. DEFINITIONS (continued)

Old Interruptible Agreement. The term "Old Interruptible Agreement" shall mean an agreement for interruptible transportation service that resulted from the conversion of entitlements under Algonquin's former Rate Schedules T-LG, STB and SS-III.

Operational Impact Area. The term "Operational Impact Area" shall describe a Transportation Service Provider's designation of the largest possible area(s) on its system in which imbalances have a similar operational effect. For Algonquin, the entire pipeline system shall comprise a single Operational Impact Area.

Posted Point of Restriction. The term "Posted Point of Restriction" shall mean any point or segment on Algonquin's pipeline system for which Algonquin has posted on its Web Site a notice requiring a reduction of scheduled capacity notice, a notice that the point or segment is scheduled at its capacity for the Day, or a notice of expected restrictions due to weather, operating conditions or maintenance.

Primary Point of Delivery. The term "Primary Point of Delivery" shall mean a Point of Delivery specified in a Service Agreement under a firm transportation rate schedule, at which Algonquin has a firm MDDO.

Primary Point of Receipt. The term "Primary Point of Receipt" shall mean a Point of Receipt specified in a Service Agreement under a firm transportation service rate schedule, at which Algonquin has a firm MDRO.

FERC Docket: RP07- 66-000

First Revised Sheet No. 506 First Revised Sheet No. 506 : Effective Superseding: Original Sheet No. 506

GENERAL TERMS AND CONDITIONS (continued)

This sheet was previously issued, but is now reserved for future use.

FERC Docket: RP10-77-000

Sixth Revised Sheet No. 507 Sixth Revised Sheet No. 507

Superseding: Fifth Revised Sheet No. 507

GENERAL TERMS AND CONDITIONS (continued)

DEFINITIONS (continued)

Recourse Rate. The term "Recourse Rate" shall mean the rates (other than a Negotiated Rate as defined in this Section 1) applicable to Part 284 services as effective from time to time.

ROFR Agreement. The term "ROFR Agreement" shall mean one of the following:

- (1) a firm service agreement under one of Algonquin's Open-Access Rate Schedules contracted at the applicable maximum rate for that service either (a) for service for twelve (12) consecutive months or more or (b) for more than one (1) year for a service which is not available for twelve (12) consecutive months;
- (2) a firm service agreement which would otherwise qualify as a ROFR Agreement but for the fact that Customer has been charged and paid a negotiated or discounted rate during the term of such firm service agreement if (a) Customer was charged and has paid a rate no less than the effective maximum Recourse Rate during the twelve (12) months immediately prior to the latest applicable date on which Customer or Algonquin is required to give notice of termination (notice deadline) of the firm service agreement, and (b) Customer pays a rate no less than the effective maximum Recourse Rate from such notice deadline until the termination date of the service agreement; provided, however, in no event shall the aggregate period of time contemplated by 2(a) and 2(b) extend more than twenty-four months prior to the termination date of the service agreement; or (3) a firm service agreement under one of Algonquin's Open-Access Rate Schedules which firm service agreement is subject to a negotiated or discounted rate that Algonquin agrees, on a not unduly discriminatory basis, shall qualify as a ROFR Agreement, in which case such negotiated or discounted rate agreement shall expressly provide that the firm service agreement qualifies as a ROFR Agreement.

Third Party Account Administrator. The term "Third Party Account Administrator" shall mean a Title Transfer Tracking Service Provider other than Algonquin.

Title Transfer. The term "Title Transfer" shall mean the change of title to gas between parties at a location.

Title Transfer Tracking. The term "Title Transfer Tracking" shall mean the process of accounting for the progression of title changes from party to party that does not effect a physical transfer of the gas.

Title Transfer Tracking Service Provider. The term "Title Transfer Tracking Service Provider" shall mean a party conducting the Title Transfer Tracking activity.

Total Heating Value. The term "Total Heating Value" shall mean the number of Btu evolved by the complete combustion with air, at constant pressure, of one anhydrous (dry) Cubic Foot of gas under a pressure of 14.73 psia and at a temperature of 60° Fahrenheit and when the products of combustion are cooled to the initial temperature of the gas and air and the water formed by combustion is condensed to the liquid state.

Web Site. The term "Web Site" shall mean Algonquin's HTML site accessible via the Internet's World Wide Web located at http://www.link.spectraenergy.com.

FERC Docket: RP10-77-000

Second Revised Sheet No. 508 Second Revised Sheet No. 508

Superseding: First Revised Sheet No. 508

GENERAL TERMS AND CONDITIONS (continued)

2. REQUEST FOR TRANSPORTATION SERVICE

2.1 General. Specific requests for information concerning service(s) should be directed to:

Algonquin Gas Transmission, LLC Attention: Marketing Department P.O. Box 1642 Houston, Texas 77251-1642

Telephone: 1-800-827-LINK, or in Houston (713) 989-LINK

Algonquin shall provide the requested information orally, or in writing, as appropriate.

- (a) Persons desiring a new service or an amendment to existing service under one of Algonquin's Rate Schedules set forth in Volume No. 1 of Algonquin's FERC Gas Tariff must be a LINK® System User pursuant to Section 40 of these General Terms and Conditions and must submit a request for service electronically via the LINK® System. Commencing January 1, 2006, persons submitting a bid for firm service under one of Algonquin's Open Access Rate Schedules pursuant to Section 9 of the General Terms and Conditions must submit the bid electronically via the LINK® System; until that date, such bids must be submitted as described in the applicable posting.
- (b) A request for a new service or an amendment to an existing service shall contain the information identified on the Request for Service Information List posted on Algonquin's public Web Site, as such list may be amended from time to time. Requests to amend existing service that will affect a Customer's financial obligations to Algonquin, without regard to the impact of any applicable discount or Negotiated Rates, are referred to as Billing Amendments. Requests to amend existing service that will not affect a Customer's financial obligations to Algonquin, without regard to the impact of any applicable discount or Negotiated Rates, are referred to as Non-Billing Amendments. A Customer requesting a new service or an amendment to existing service shall also provide the following to Algonquin:
 - (1) Either at the time of the request for new service or an amendment to existing service is submitted to Algonquin or at the time of execution of the service agreement, such other information (if any), in writing, as may be required to comply with regulatory reporting or filing requirements; and
 - (2) Within ten (10) Business Days of the submittal of the request for new service or a request for a Billing Amendment, any credit information required to be provided pursuant to Section 3 of the General Terms and Conditions.
- (c) Neither a request for new service nor a request that would result in a Billing Amendment shall be deemed to have been received by Algonquin until Customer has submitted such request online via the LINK® System and Algonquin has received the information required or requested pursuant to this Section 2.1 and Section 3 of the General Terms and Conditions. A request that would result in a Non-Billing Amendment shall be deemed to have been received on the date such request is submitted in the LINK® System. If Algonquin requests additional information or assurance in accordance with this Section 2.1, and such additional information or assurance is received within ten (10) Business Days of Algonquin's request, Customer's request for service shall be deemed to have been received on the date on which Customer's additional financial information is received by Algonquin; otherwise, Customer's request for service shall be deemed to be null and void.
- (d) If Customer does not submit the information required in Section 2.1(b) above within the required timeframes, the request for service shall be considered to be null and void. In addition, Algonquin shall reject any request for service created in the LINK® System by Customer, but not submitted to Algonquin within ninety (90) days of Customer's creation of such request.

FERC Docket: RP09-97-000

Third Revised Sheet No. 509 Third Revised Sheet No. 509

Superseding: Second Revised Sheet No. 509

GENERAL TERMS AND CONDITIONS (continued)

- 2. REOUEST FOR TRANSPORTATION SERVICE (continued)
- 2.2 Conditions of Firm Transportation. All firm transportation requests shall be subject to the following conditions:
 - (a) No request for transportation from a Point of Receipt or to a Point of Delivery shall be considered valid or be granted if to do so would impair Algonquin's ability to render existing services pursuant to Algonquin's firm service rate schedules.
 - (b) Subject to the provisions of (a) above, any amendment to any firm service agreement or exhibit to add or change Point(s) of Receipt or Point(s) of Delivery that requires additional capacity pursuant to an applicable firm rate schedule will be addressed in accordance with Sections 2.4 and 2.5.
 - (c) In certain situations, Algonquin may use an accounting meter number to represent a physical location on its pipeline system. A Point of Receipt and/or a Point of Delivery identified on Customer's executed service agreement(s) may be designated in the LINK® System by means of an accounting meter number and description that differs from the physical meter number and description specified on the service agreement. The same rights and obligations exist for both Algonquin and Customer regardless of whether a location is identified in Customer's executed service agreement by means of a physical meter number or an accounting meter number.
- 2.3 Execution of Service Agreement and Amendments. A service agreement and/or an amendment to an existing service agreement shall be executed, as specified in this Section 2.3, by Customer and Algonquin following the completion of the approval process.
 - (a) All interruptible service agreements, all interruptible service agreement amendments, firm service agreements with a term of one (1) year or less and all amendments for firm service agreements with a term of one (1) year or less shall be executed electronically via the LINK® System by Customer and Algonquin; any agreement that is executed in full utilizing electronic transmission through the LINK® System is a valid and enforceable contract that is binding on all parties. All firm service agreements with a term of more than one (1) year and all amendments to firm service agreements with a term of more than one (1) year shall be executed in writing. A Service Agreement shall be executed and, if executed in writing returned to Algonquin, within fifteen (15) days of the tender of a service agreement by Algonquin. In the event Customer fails to submit a valid nomination for transportation pursuant to an interruptible service agreement within ninety (90) days after the later of (i) the date service is to commence, (ii) the date the service agreement is fully executed by Customer and Algonquin, or (iii) the date that the facilities, if any, to be constructed are ready for service, the service agreement and the corresponding transportation request for service shall be considered null and void.

FERC Docket: RP10-77-000

Third Revised Sheet No. 510 Third Revised Sheet No. 510

Superseding: Second Revised Sheet No. 510

GENERAL TERMS AND CONDITIONS (continued)

- 2. REQUEST FOR TRANSPORTATION SERVICE (continued)
 - (b) For each of Algonquin's firm Rate Schedules, the Service Agreement executed in writing or electronically via the LINK® System, as applicable, by Customer and Algonquin, the Exhibit(s) executed by Customer and Algonquin, the applicable rate schedule, the General Terms and Conditions of this FERC Gas Tariff, and any applicable Negotiated Rate or discount agreement will comprise the entire agreement between Customer and Algonquin.
 - (c) For each of Algonquin's interruptible Rate Schedules, the Service Agreement executed electronically via the LINK® System by Customer and Algonquin, the Exhibit(s) executed by Customer and Algonquin, the applicable rate schedule, the General Terms and Conditions of this FERC Gas Tariff, and any applicable Negotiated Rate or discount agreement will comprise the entire agreement between Customer and Algonquin.

Effective Date: 07/19/2007 Status: Effective

FERC Docket: RP07-139-000

Second Revised Sheet No. 511 second Revised Sheet No. 511 : Suspended

Superseding: First Revised Sheet No. 511

GENERAL TERMS AND CONDITIONS (continued)

- 2. REQUEST FOR TRANSPORTATION SERVICE (continued)
- 2.4 Prospective Sale of Available Capacity and Minimum Terms of Any Awards.
 - (a) Unless otherwise agreed to by Algonquin, Customer may request available capacity for service to commence at a future date only within the following timeframes:
 - (1) For service with a primary contract term of less than ninety (90) days, the requested service shall commence no later than five (5) days from the date the capacity is awarded;
 - (2) For service with a primary contract term equal to or longer than ninety (90) days but less than one (1) year, the requested service shall commence no later than thirty (30) days from the date the capacity is awarded;
 - (3) For service with a primary contract term of one (1) year or longer, the requested service shall commence no later than one hundred and eighty (180) days from the date the capacity is awarded.
 - (b) Algonquin may consider, on a not unduly discriminatory basis, a request for firm service outside the time periods specified in Section 2.4(a), if the request involves any of the following circumstances:
 - (1) The request is for capacity pursuant to an open season initiated by Algonquin;
 - (2) The request involves capacity that is available due to the termination of an existing contract or the reduction of contracted volume under an existing contract;
 - (3) The request involves the modification or construction of facilities or the issuance of any necessary certificate authorization.

In the event that Algonquin allows a variation from the time periods specified in Section $2.4\,(a)$, in accordance with the circumstances described in Section $2.4\,(b)\,(1)-(3)$ or otherwise, the details of the variation will be included in the notice of the open season. In addition, unless otherwise agreed to by Algonquin, all awards of capacity must be for continuous service at a constant MDTQ and at maximum rates for the entire term of the service. Any deviations from the time periods specified in Section $2.4\,(a)$ shall only be done in a not unduly discriminatory manner consistent with Commission regulations.

FERC Docket: RP10-77-000

Third Revised Sheet No. 512 Third Revised Sheet No. 512 Superseding: Substitute Second Revised Sheet No. 512

GENERAL TERMS AND CONDITIONS (continued)

- 2. REQUEST FOR TRANSPORTATION SERVICE (continued)
- 2.5 Open Season for Available Capacity.

Upon receipt of a valid request for firm service pursuant to Section 2.4 with a term greater than thirty (30) days for which Algonquin determines capacity is available to satisfy any such request, Algonquin shall post notices for solicitation of bids in an open season for service to start immediately or in the future. For valid requests for firm service pursuant to Section 2.4 with a term of thirty (30) days or less for which Algonquin determines capacity is available to satisfy any such request, Algonquin shall have the right, but shall not be obligated, to post notices for solicitation of bids in an open season on such capacity. In addition, Algonquin shall post notices for solicitation of bids in an open season for significant mainline or lateral capacity expansion projects on its system; provided, however, that an open season shall not be required for minor or auxiliary facilities, additions of receipt and/or delivery facilities or facility modifications which do not significantly alter the operational characteristics of the pipeline system.

For any open season conducted pursuant to this Section 2.5 for service with a term of less than ninety (90) days, the open season will be conducted for a minimum of one (1) Business Day. For an open season under this Section 2.5 for service with a term equal to or greater than ninety (90) days that is not otherwise associated with an expansion project, the open season will be conducted for a minimum of (i) five (5) Business Days from the posting of the notice of request for service for the capacity or (ii) fifteen (15) Business Days from the date the capacity in question was first posted as being available for contracting, whichever is the later calendar date. In no event shall the open season be for a period greater than one (1) calendar month. All open seasons shall end at 2:00 p.m. CT not less than one (1) Business Day prior to the date service would be available. The open season may include generally available unsubscribed capacity; capacity under expiring or terminating service agreements, to the extent such agreements do not have a right of first refusal or the subject shipper does not exercise the right of first refusal; and/or capacity resulting from one or more expansion projects. Algonquin will post available capacity before it provides such information to any potential customer.

- (a) The open season notice will include the following:
 - (1) the location of the capacity or proposed expansion;
 - (2) the total quantity, unless such quantity is not known or knowable;
 - (3) the date capacity is available or proposed to be available; and
 - (4) the dates on which the open season begins and ends.

Algonquin will attempt to structure any such open season posting pursuant to this Section 2.5 so as not to identify specifically the Customer or potential customer submitting the request and/or the specific location of the market(s) to be served.

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 513 Fourth Revised Sheet No. 513

Superseding: Third Revised Sheet No. 513

GENERAL TERMS AND CONDITIONS (continued)

- 2. REQUEST FOR TRANSPORTATION SERVICE (continued)
 - (b) Bidders who desire service to be provided in whole or in part by the capacity posted pursuant to this Section 2.5 must submit bids for such capacity online via the LINK® System. To be a valid bid, a bid must comply with the bid requirements set forth in Section 2.5(c).
 - (c) To be a valid bid, the bidder must provide all information and data required by Sections 2.1 and 3 of these General Terms and Conditions. Algonquin reserves the right to reject any bid (i) at less than maximum rate, (ii) that may adversely affect the operational integrity of Algonquin's system, (iii) that seeks to change a Point or Points of Delivery specified in an executed Service Agreement at which the MHTQ equals the applicable MDTQ multiplied by 0.6, (iv) that does not satisfy all the terms of the applicable posting and/ or (v) that contains terms and conditions other than those provided for in Algonquin's FERC Gas Tariff. If Algonquin rejects any request for available capacity posted pursuant to this Section 2.5, Algonquin will notify the bidder via e-mail of its reason(s) for such rejection.
 - (d) At the close of the bidding period for any open season held pursuant to this Section 2.5, Algonquin shall select from among all valid bids the "best bid," as determined pursuant to this Section 2.5(d) and, if applicable, Section 2.5(e). Algonquin shall review all bids received from bidders that have not been rejected by Algonquin pursuant to Section 2.5(c) above, to determine which bid is the "best bid." For purposes of this Section 2.5, the "best bid" shall be the bid which yields to Algonquin the highest net present value. Net present value shall be calculated on the basis of the present value of the Reservation Charge per unit to Algonquin except that under a Negotiated Rate agreement with a minimum quantity, the net present value evaluation shall also include the fixed cost component of the usage revenue at the minimum quantity. In making the determination of net present value Algonquin shall apply the rate, as of the date of the review, stated in accordance with Section 154.501(d) of the Commission's Regulations, to all bids. In determining the highest net present value in connection with a Customer paying a Negotiated Rate higher than the maximum Recourse Rate, such Negotiated Rate Customer paying a rate higher than the maximum Recourse Rate will be deemed to be paying a rate equal to the maximum Recourse Rate.

In determining the "best bid," any request to add or change a Point of Receipt and/or Point of Delivery will be considered to have a net present value of zero (0) when comparing such requests to other requests for service and awarding capacity unless the Customer has agreed in conjunction with its request to (i) increase its MDTQ, (ii) increase the rate Customer is currently paying to Algonquin, if such rate is less than the applicable maximum rate, and/or (iii) extend the term of its firm service agreement in which case Algonquin shall consider the terms of such MDTQ increase, rate increase and/or contract extension when calculating the net present value. In the event that a request to add or change a Point of Receipt and/or a Point of Delivery results in a bid(s) that yields a net present value that is greater than zero ("Positive NPV Bid"), Algonquin shall award the capacity to the bid which yields to Algonquin the highest net present value. In the event that no Positive NPV Bid(s) is received by Algonquin or point capacity remains after Algonquin has awarded capacity to or among the Positive NPV Bid(s), Algonquin shall award the remaining point capacity to the firm Customer whose request to change its primary point(s) was received first-in-time by Algonquin.

(e) The Risk of Default Factor shall be one (1) minus the differential between (i) the bidder's probability of default which is calculated by extrapolating to the maximum bid term to be used for bid evaluation purposes, using Standard & Poor's ("S&P") most recent "Cumulative Average Default Rates by Rating Modifier" table ("S&P Table"), and (ii) the indicated probability of default for a bidder with a credit rating at or above the credit ratings listed in Section 2.5(e)(1) for a like term. In addition, the Risk of Default Factor shall only be applied to

FERC Docket: RP07-139-006

Original Sheet No. 513A Original Sheet No. 513A

GENERAL TERMS AND CONDITIONS (continued)

2. REQUEST FOR TRANSPORTATION SERVICE (continued)

existing unsubscribed capacity and shall not be applied to any bidders that have a credit rating at or above those credit ratings listed in Section $2.5\,(e)\,(1)$ below, or whose parent company or guarantor has such a credit rating, provided that such parent company or guarantor has provided a guarantee for all of the obligations under the specific service agreement at issue pursuant to Section $2.5\,(e)\,(3)\,(a)$. The application of the Risk of Default Factor is governed by the following:

- A bidder with a credit rating below (a) an S&P rating of BBB-, (b) a Moody's rating of Baa3, (c) a Fitch rating of BBB-, or (d) a Dominion Bond Rating Service rating of BBB (Low), and who does not have a parent company or quarantor with a credit rating at or above those credit ratings just listed, provided that such parent company or guarantor provides a guarantee for all of the obligations under the specific service agreement at issue pursuant to Section 2.5(e)(3)(a), shall have a probability of default of zero for bidding purposes if such bidder indicates on its bid form that it will post sufficient collateral to equal the difference between (i) the net present value of the revenue in the bid, adjusted by the bidder's Risk Default Factor (assuming that it would be applied), and (ii) the net present value of a bid from a bidder with a minimum rating(s) equal to the above rating(s) herein with the same term, and such bidder does subsequently post such collateral as specified in Section 2.5(e)(7). For these bid evaluation purposes, collateral is defined as a letter of credit from a financial institution with at least an A- rating, or a cash deposit. Collateral will be capped, for purposes of the bid evaluation, at 50% of the revenue for the term included in the bidder's bid.
- (2) For a bidder described in Section 2.5(e)(1) that does not indicate on its bid form that it will post such collateral, Algonquin will adjust the bid by applying the Risk of Default Factor derived from the S&P Table, as set forth in Section 2.5(e). Although a bidder under this Section 2.5(e)(2) is not posting sufficient collateral to satisfy the requirement set forth in Section 2.5(e)(1) above, such a bidder nevertheless may increase the net present value of its bid by posting some collateral but less than the amount described above in Section 2.5(e)(1) which must be in a form allowed under Section 2.5(e)(1) contemporaneously with submitting its bid in the open season, provided that such bidder cannot increase the net present value of the bid to greater than the net present value of the same bid if the bidder had posted the collateral as described in Section 2.5(e)(1).
- (3) The following rules will apply to ratings for purposes of evaluating multiple bids as part of the NPV bid evaluation process:
 - (a) Substitute Ratings. Algonquin will substitute the credit rating of a bidder's guarantor, or parent company, whichever is higher, provided that such guarantor or parent company has provided a guarantee for all of the obligations under the specific service agreement at issue and satisfies the requirements of Section 3.2(d).
 - (b) Equivalent Ratings. For any bidder that does not have, or whose parent company or guarantor does not have, a credit rating from one or more of the credit rating agencies noted above, Algonquin will assign an equivalent rating using a credit scoring methodology, applied on a nondiscriminatory basis, provided such parent company or guarantor has provided a guarantee for all of the obligations under the specific service agreement at issue.
- (4) Releasing shippers who propose to release capacity for a term of more than one year will have the option of specifying, pursuant to the capacity release procedures in Section 14.4, that the same NPV and Risk of Default processes as set forth herein will be used by Algonquin when evaluating bids by replacement customers for such released capacity. If a releasing shipper makes such an election, Algonquin will follow the time line for non-standard capacity release bid evaluation.

FERC Docket: RP10-77-000

First Revised Sheet No. 513B First Revised Sheet No. 513B

Superseding: Original Sheet No. 513B

GENERAL TERMS AND CONDITIONS (continued)

2. REQUEST FOR TRANSPORTATION SERVICE (continued)

- (5) All credit ratings shall be determined as of the last Day of the Open Season for the pipeline capacity at issue, and in the event a bidder is rated by two or more rating agencies and there is a split rating between rating agencies, the lowest rating applies.
- (6) For any bid submitted with a term of fifteen (15) years or longer, Algonquin will apply the 15-year Risk of Default factor from the S&P Table.
- (7) It is a condition of any capacity award that the winning bidder post any collateral indicated on its bid form within five (5) business days of the notice of award, and Algonquin shall have the right to reject and/or terminate any award for failure of a bidder to do so. If a bidder fails to post such collateral, Algonquin shall have the right to require that such bidder post collateral contemporaneously with any future bid(s) or provide other credit assurance of a type reasonably acceptable to Algonquin given the financial obligations associated with the particular award.
- (8) Algonquin will provide interest on the collateral in accordance with Section $\frac{3}{2}$

FERC Docket: RP10-77-000

Third Revised Sheet No. 514 Third Revised Sheet No. 514 Superseding: Substitute Second Revised Sheet No. 514

GENERAL TERMS AND CONDITIONS (continued)

- 2. REQUEST FOR TRANSPORTATION SERVICE (continued)
 - (f) In the event Algonquin receives two (2) or more bids that qualify as the "best bid" (i.e., two (2) or more bids that produce an equivalent net present value), capacity subject to the open season will be allocated between or among such requests pro rata based on the MDTQ requested; provided, however, if a bidder offered capacity on a pro rata basis pursuant to this Section 2.5(f) fails to execute a service agreement within (i) five (5) Business Days for service with a contract term of less than one (1) year or (ii) fifteen (15) days of the tender of a service agreement by Algonquin for service with a contract term of one (1) year or longer, Algonquin shall deem such bidder's request to be null and void and shall reallocate the available capacity pro rata based on the MDTQ requested to such remaining best bids in accordance with this Section 2.5(f).
 - (g) Within twenty-four (24) hours after an award of capacity under this Section 2.5, Algonquin shall post on the LINK® System the best bid(s), showing the net present value for such best bid(s). In the event the best bid is a discounted rate that is less than the applicable maximum tariff rate and Algonquin accepts a bid at less than the maximum rate as the winning bid, then the bidder must submit a discount request form online via the LINK® System and Algonquin must approve such discount request pursuant to the provisions of Section 45 of these General Terms and Conditions in order for such rate to become effective.
 - (h) To the extent Algonquin has (i) available capacity or (ii) capacity under expiring or terminating service agreements where such capacity is not subject to a right of first refusal or shipper does not exercise its right of first refusal, Algonquin reserves the right, but shall not be obligated, to reserve such capacity for a future expansion project. Algonquin may reserve capacity for a future expansion project for which an open season has been held or will be held within twelve (12) months of the date that Algonquin posts such capacity as being reserved. Any capacity reserved pursuant to this Section 2.5(h) must first be posted as available capacity on Algonquin's Web Site for at least five (5) Business Days. Any minimum terms and conditions imposed in an open season for capacity to be reserved must not materially differ so as to be more restrictive than the terms and conditions imposed in the expansion project open season. In the event that a subsequent expansion project open season imposes minimum terms and conditions that are materially different from the minimum terms and conditions imposed for the reserved capacity open season, Algonquin shall hold another open season for the reserved capacity that uses the same minimum terms and conditions as were imposed for the expansion project open season. Any capacity reserved under this Section 2.5(h) may be reserved for up to twelve (12) months prior to the time Algonquin files for certificate approval for proposed construction of a related project and thereafter until all expansion facilities are placed into service. Any capacity reserved under this Section 2.5(h) shall be made available, pursuant to the provisions of Section 2.6 of these General Terms and Conditions, for transportation service on an interim basis up to, but not including, the inservice date of the related expansion project(s). For such interim service, Algonquin reserves the right to limit any Customer's renewal rights that might otherwise apply to such service, including Customer's right of first refusal, if applicable. Any capacity reserved for a future expansion project that does not go forward for any reason shall be reposted as generally available within thirty (30) days of the date the capacity becomes available.

FERC Docket: RP10-77-000

Third Revised Sheet No. 515 Third Revised Sheet No. 515

Superseding: Second Revised Sheet No. 515

GENERAL TERMS AND CONDITIONS (continued)

- 2. REQUEST FOR TRANSPORTATION SERVICE (continued)
- 2.6 Interim Service. Capacity that is under contract for a future period pursuant to Section 2.5(h) or this Section 2.6 may be made available on an interim basis up to the service commencement date of such contract for a future period ("Interim Capacity"). The availability of Interim Capacity, including any limitations on the renewal rights for such capacity, will be posted on Algonquin's LINK® System in accordance with Section 2.5(a). Any party desiring to submit a bid for such Interim Capacity must submit its bid online via the LINK® System in accordance with the provisions of Sections 2.5(b) and 2.5(c). Such Interim Capacity shall be available for bidding for at least five (5) Business Days. Algonquin shall award the Interim Capacity and post a notice of the winning bid(s) on the LINK® System, in accordance with Sections 2.5(d) through 2.5(g). The right of first refusal will not be applicable to any service agreement entered into pursuant to this Section 2.6.

Effective Date: 09/12/2005 Status: Effective

FERC Docket: RP05-559-000

First Revised Sheet No. 516 First Revised Sheet No. 516 : Effective Superseding: Original Sheet No. 516

Sheet No. 516 was previously issued, but is now reserved for future use.

FERC Docket: RP09-600-000

Second Revised Sheet No. 517 Second Revised Sheet No. 517

Superseding: First Revised Sheet No. 517

GENERAL TERMS AND CONDITIONS (continued)

3. CREDIT EVALUATION

- 3.1 Standards Used in Appraising Credit. Algonquin's acceptance of a request for new service or a request resulting in a Billing Amendment under any Rate Schedule set forth in Algonquin's FERC Gas Tariff is contingent upon Customer satisfying a credit appraisal by Algonquin. Algonquin shall apply, on a non-discriminatory basis, consistent financial evaluation standards to determine the acceptability of Customer's overall financial condition. Algonquin shall not use any criteria, formula, ranking system or other methodology that would give any preference or advantage to an affiliate of Algonquin. Such credit appraisal and any further or ongoing credit appraisal as may be necessary shall be based upon the following information and criteria:
 - (a) Customer shall provide current financial statements, annual reports, 10-K reports or other filings with regulatory agencies which discuss Customer's financial status; a list of all corporate affiliates, parent companies and subsidiaries; and any reports from credit reporting and bond rating agencies which are available. Algonquin shall determine the acceptability of the Customer's overall financial condition.
 - (b) Customer shall provide a bank reference and at least two trade references. The results of reference checks and any credit reports submitted in accordance with Section 3.1(a) above must show that Customer's obligations are being paid on a reasonably prompt basis.
 - (c) Customer shall confirm in writing that Customer is not operating under any chapter of the bankruptcy laws and is not subject to liquidation or debt reduction procedures under state laws, such as an assignment for the benefit of creditors, or any informal creditors' committee agreement. An exception can be made for a Customer who is a debtor in possession operating under Chapter 11 of the Federal Bankruptcy Act but only with adequate assurance that the service billing will be paid promptly as a cost of administration under the federal court's jurisdiction.
 - (d) Customer shall confirm in writing that Customer is not aware of any change in business conditions which would cause a substantial deterioration in its financial condition, a condition of insolvency, or the inability to exist as an ongoing business entity.

FERC Docket: RP10-77-000

Second Revised Sheet No. 518 Second Revised Sheet No. 518

Superseding: Second Sub First Revised Sheet No. 518

GENERAL TERMS AND CONDITIONS (continued)

3. CREDIT EVALUATION (continued)

- (e) Customers with an ongoing business relationship with Algonquin shall have no delinquent balances outstanding for services provided previously by Algonquin, and Customer must have paid its account according to the established terms and not made deductions or withheld payment for claims not authorized by contract.
- (f) Customer shall confirm in writing that no significant collection lawsuits or judgments are outstanding which would materially affect Customer's ability to remain solvent as a business entity.
- 3.2 Alternatives to Satisfaction of Credit Criteria. Upon notification by Algonquin that Customer has failed to satisfy the credit criteria or subsequently during the term of the service agreement no longer satisfies the credit criteria, such Customer may still obtain credit approval by Algonquin if it pays any outstanding balances due Algonquin for service rendered or has complied with Section 19.3 of the General Terms and Conditions with regard to such balances and elects to provide one of the following forms of security:
 - (a) an advance deposit;
 - (b) a standby irrevocable letter of credit, acceptable to Algonquin, issued by a financial institution that satisfies Algonquin's credit appraisal;
 - (c) security interest in collateral found to be satisfactory to Algonquin; or
 - (d) a guarantee, acceptable to Algonquin, by another person or entity that satisfies Algonquin's credit appraisal.

Algonquin shall provide such Customer with a written statement supporting Algonquin's request for the security amount requested at the time such security is requested. If pipeline rejects the security provided by Customer in accordance with Section $3.2\,(b)-(d)$ above, Algonquin shall re-issue its request for the security and include a written explanation for the rejection of the security previously provided by Customer. For a new Customer, or a Customer requesting a Billing Amendment, such security shall be calculated as follows:

- (i) For firm service, an amount equal to the three (3) highest Months' worth of reservation charges at the applicable maximum Recourse Rate during a Contract Year;
- (ii) For interruptible service, an amount equal to fifteen (15) Days of usage per Month for three (3) Months multiplied by the arithmetic average of the applicable commodity rate(s), multiplied by Customer's Maximum Daily Quantity;
- (iii) For Park and Loan Service, the applicable maximum Park and Loan Daily Charge multiplied by Customer's Maximum Park Quantity or Maximum Loan Quantity, as applicable, plus the value of any quantity to be loaned to Customer;
- (iv) For Capacity Release Umbrella agreements, Algonquin will accept any collateral amount submitted by Customer in relation to the request for the capacity release umbrella agreement; however, Customer will be required to comply with all of Algonquin's credit requirements at such time as Customer submits a bid to acquire released capacity pursuant to Section 14 of the General Terms and Conditions of this FERC Gas Tariff;
- (v) For Operational Balancing Agreements, an estimated imbalance quantity of 5,000 Dth per Month for three (3) Months multiplied by the average of Algonquin's cashout prices for the latest three (3) Months.

For an existing Customer, such security shall be equal to the highest three (3) Months of activity (based on usage of in-kind and loan agreements and the billed amounts, including cashout amounts, for all other agreements) for all of Customer's active service agreements during the previous twelve (12) Months. For the purposes of this section, the term "in-kind agreement" does not include fuel reimbursement.

FERC Docket: RP10-77-000

Second Revised Sheet No. 518A Second Revised Sheet No. 518A

Superseding: First Revised Sheet No. 518A

GENERAL TERMS AND CONDITIONS (continued)

3. CREDIT EVALUATION (continued)

With respect to a request for new service or a request resulting in a Billing Amendment, the security required by Algonquin pursuant to this Section 3.2 must be received by Algonquin within ten (10) Business Days of Algonquin's written notification to Customer, otherwise, such request will be deemed to be null and void. With respect to an existing service agreement, Customer must tender (i) an advance payment equal to the highest Month of activity (based on usage of in-kind and loan agreements and the billed amounts, including cashout amounts, for all other agreements) for all of Customer's active service agreements during the previous twelve (12) months within five (5) Business Days of Algonquin's notification, and (ii) the security required by this Section 3.2 within thirty (30) days of Algonquin's notification. If such advance payment or security is not received within the specified time period(s), then Algonquin may immediately suspend service, and may terminate Customer's service agreement(s) upon thirty (30) days' notice to Customer and the Commission.

Security in the form of an advance deposit shall accrue interest to the benefit of Customer from the date Algonquin receives such deposit. Interest on such advance deposits shall be calculated monthly at the most recently established 91-day Treasury Bill auction rate, as published in The Wall Street Journal. Alternatively, a Customer providing security in the form of an advance deposit may deposit such security into an interest-bearing escrow account, established by Customer at Customer's expense, to which account Algonquin shall have unrestricted access in the amount of an invoice upon presentment of an invoice for the payment of services provided to Customer.

- 3.3 Once every twelve (12) months, Customer shall provide Algonquin with updated financial information in the form required in Section 3.1 above. In addition, if Algonquin becomes aware that there may be a potentially material change in the financial condition of an existing Customer, Algonquin may require Customer to promptly resubmit all of the financial information required in Section 3.1 above. Algonquin shall re-evaluate Customer's creditworthiness based upon the updated information provided by Customer pursuant to Section 3.1 above. If Customer's credit standing ceases to meet Algonquin's credit requirements at any time during the period of service, then Algonquin has the right to require security as specified in Section 3.2 above. If the credit standing of any entity issuing a letter of credit and/or guaranty in support of Customer's obligations ceases to meet Algonquin's credit appraisal at any time during the period of service, then Algonquin has the right to require Customer to provide replacement security that satisfies the requirements of Section 3.2 above.
- 3.4 In the event any information provided by Customer pursuant to Sections 2.1 and 3.1 with respect to a request for new service or a request that would result in a Billing Amendment materially changes prior to execution of the new service agreement or Billing Amendment, as applicable, or Customer learns that such information may materially change prior to execution of the service agreement or Billing Amendment, as applicable, Customer shall be obligated to provide written notice of such material changes to Algonquin. Algonquin shall require Customer to resubmit all of the financial information required in this Section 3 within ten (10) Business Days of the identification of the material change. If such updated financial information is not received within ten (10) Business Days, the request for service will be deemed null and void. Algonquin shall re-evaluate Customer's creditworthiness based upon the updated information provided by Customer pursuant to this Section 3.

FERC Docket: RP10-77-000

Third Revised Sheet No. 519 Third Revised Sheet No. 519

Superseding: Second Revised Sheet No. 519

GENERAL TERMS AND CONDITIONS (continued)

4. OUALITY OF GAS

- 4.1 Composition of Gas. The gas received by Algonquin for Customer's account, or delivered by Algonquin to Customer, shall be a combustible gas consisting wholly of, or a mixture of:
 - (a) Natural gas of the quality and composition produced in its natural state except that Algonquin may extract or permit the extraction of any of the constituents thereof except methane.
 - (b) Gas generated by vaporization of liquefied natural gas (LNG).
 - (c) Manufactured, reformed, synthesized, or mixed gas consisting essentially of hydrocarbons of the quality and character produced by nature in the petroleum, oil, and gas fields with physical properties such that when such gas is commingled with natural gas, the two become indistinguishable.
- 4.2 Heating Value and Wobbe Number. Gas received by Algonquin for Customer's account at each Point of Receipt, or delivered by Algonquin to Customer at the Point(s) of Delivery shall have a Total Heating Value of at least 967 Btu and not greater than 1,110 Btu per Cubic Foot of gas volume. The gas shall have a Wobbe Number of not less than 1314 nor greater than 1400 (calculated using Total Heating Value (THV)), dry, under standard conditions at 14.73 psia at 60 degrees Fahrenheit based on the following mathematical definition and in accordance with Section 5 of these GT&C:

THV / Sqrt SGgas

Where:

= Total Heating Value (Btu/standard Cubic Foot)

SGgas = Specific Gravity
Sgrt = Square Root of

- 4.3 Objectionable Properties. The gas received and delivered hereunder:
 - (a) Shall be commercially free from dust, gum, gum-forming constituents, and free liquids under continuous gas flow conditions at the pressure and temperature conditions in Algonquin's pipeline at a point approximately fifty feet in advance of the interconnection into Algonquin's system that is serving as the Point of Receipt and the meter inlet header located at or near any Point of Delivery;

FERC Docket: RP10-77-000

Third Revised Sheet No. 520 Third Revised Sheet No. 520

Superseding: Second Revised Sheet No. 520

GENERAL TERMS AND CONDITIONS (continued)

4. QUALITY OF GAS (continued)

- (b) Shall not contain an amount of water vapor exceeding seven pounds per 1,000,000 Cubic Feet of gas volume as measured by methods in accordance with accepted industry practice, or by other methods mutually agreed upon by Customer and Algonquin;
- (c) Shall contain less than one-half (0.5) grain of hydrogen sulphide per 100 Cubic Feet of gas volume as measured by methods in accordance with accepted industry practice, such as, but not limited to, lead acetate testing, analysis by titrator, analysis by chromatograph, or by other methods mutually agreed upon by Customer and Algonquin;
- (d) Shall not contain more than ten (10) grains of total sulphur, excluding any mercaptan sulphur, per 100 Cubic Feet of gas volume as measured by methods in accordance with accepted industry practice, such as, but not limited to, analysis by titrator, analysis by chromatograph, or by other methods mutually agreed upon by Customer and Algonquin;
- (e) Shall be of a flowing temperature which is adequate to prevent interference with the proper operation of lines, regulators, meters and other equipment of Algonquin. Algonquin may impose restrictions on the temperature of the flowing gas that it receives when, in Algonquin's reasonable judgment, these restrictions are necessary to insure the proper operation of Algonquin's facilities;
- (f) Shall not contain more than four percent (4.0%) by volume of a combined total of any non-hydrocarbon gas including, without limitation carbon dioxide, nitrogen, krypton, helium, argon, xenon, and neon. Within this volume of non-hydrocarbons, the total carbon dioxide content shall not exceed two percent (2.0%) by volume, and the total combined nitrogen and oxygen content shall not exceed two and three quarters percent (2.75%) by volume;
- (g) Shall not have uncombined oxygen content in excess of two-tenths of one percent (0.2%) by volume; and
- (h) Shall not contain, either in the gas or in any liquids with the gas, any microbiological organism, active bacteria or bacterial agent capable of contributing to or causing corrosion and/or operational and/or other problems. Microbiological organisms, bacteria or bacterial agents include, but are not limited to, sulfate reducing bacteria (SRB) and acid producing bacteria (APB). Tests for bacteria or bacterial agents shall be conducted on samples taken from the meter run or the appurtenant piping using American Petroleum Institute (API) test method API-RP38 or any other test method acceptable to Algonquin and Customer which is currently available or may become available at any time.
- (i) The non-methane hydrocarbon content shall contain no more than twelve percent (12.0%) of ethanes and heavier hydrocarbons (C2+), of which the content of butanes and heavier hydrocarbons (C4+) shall not exceed one and one half percent (1.5%).

FERC Docket: RP07-504-002

Second Revised Sheet No. 521 Second Revised Sheet No. 521

Superseding: First Revised Sheet No. 521

GENERAL TERMS AND CONDITIONS (continued)

4. OUALITY OF GAS (continued)

- 4.4 Verification of Gas Quality. At Algonquin's request, Customer shall use all reasonable efforts to obtain and provide to Algonquin all records regarding gas quality kept by upstream pipelines transporting the gas received by Algonquin for Customer's account. Customer shall use all reasonable efforts to ensure and verify for Algonquin that such upstream pipelines are using appropriate equipment to monitor compliance with the gas quality specifications applicable on Algonquin's system as stated in this Section 4.
- 4.5 Failure To Conform to Specifications. If the gas tendered by Customer for receipt by Algonquin, or offered for delivery by Algonquin to Customer, shall fail at any time to conform to any of the specifications set forth in this Section 4, then the party asserting such deficiency shall notify the other of such deficiency and may, at its option, refuse to accept nonconforming gas pending correction by Algonquin or Customer as appropriate. Algonquin may refuse to accept gas or may impose additional gas quality specifications and restrictions if Algonquin, in its reasonable judgment, determines that harm to Algonquin's facilities or operations could reasonably be expected to occur if it receives gas that fails to meet such additional specifications and restrictions.
- 4.6 Odorization. Algonquin shall have no obligation to odorize the gas tendered by Customer other than to conform to the regulations of appropriate governmental authorities having jurisdiction. However, if Algonquin odorizes the gas, such odorization shall be by use of a malodorant agent of such character as to indicate by a distinctive odor the presence of gas. Whenever odorized gas is delivered, the quality and specifications, as set forth in this Section 4, of such gas shall be determined prior to the addition of malodorant or with proper allowance for changes or additions due to such malodorant. Such odorization of the gas by Algonquin shall be for the purpose of detection of the gas only during the time when the gas is in the possession of Algonquin, prior to delivery to the Customer.
- 4.7 Waiver of Requirements. Algonquin may waive the requirements set forth in this Section 4 in order to allow Customer to tender or cause to be tendered gas which does not, when injected into Algonquin's pipeline, meet the quality specifications set forth in Section 4; provided that acceptance of such gas shall not adversely affect Algonquin's system facilities or operations, and further provided that once such gas has been blended, to the extent blending occurs, the commingled gas stream at any delivery point on Algonquin's system shall be compliant with the quality specifications set forth in Section 4. Algonquin shall post on LINK® any waiver of Algonquin's gas quality requirements. Algonquin shall implement this Section 4.7 on a non-discriminatory basis and may cancel any waiver at any time if necessary to assure that the commingled gas stream is compliant with the quality specifications set forth in Section 4 at any delivery point on Algonquin's system.

FERC Docket: RP10-77-000

First Revised Sheet No. 522 First Revised Sheet No. 522

Superseding: Original Sheet No. 522

GENERAL TERMS AND CONDITIONS (continued)

5. MEASUREMENT

- 5.1 Measurement Unit. The unit of measurement of gas delivered by Algonquin to Customer shall be the MMBtu.
- 5.2 Determination of MMBtu Delivered. The number of MMBtu delivered at a Point of Delivery during any Day shall be determined as the product of (a) and (b) below, divided by 1,000:
 - (a) The volume of gas delivered during such Day at such point in Mcf, determined in accordance with the procedures set forth in Section 5.3 below.
 - (b) The Total Heating Value of the gas delivered during such Day at such Point of Delivery.

The Total Heating Value shall be measured, unless otherwise agreed by the parties concerned, by methods in accordance with accepted industry practice, such as, but not limited to, recording calorimeter(s) or gas chromatograph(s) located at appropriate points. The arithmetic average of the Total Heating Value recorded each 24-hour day or other periodic measurements within a shorter time frame as Algonquin shall select shall be used for computations.

- 5.3 Determination of Volume Delivered. The volume of gas delivered at any point shall be measured according to Boyle's and Charles's Law for the measurement of gas under varying pressures with deviations therefrom as provided in paragraph (g) below, on the measurement basis hereinafter specified, and shall be determined as follows:
 - (a) The unit of volume for the purpose of measurement shall be one Cubic Foot of gas at a temperature of 60° Fahrenheit and an absolute pressure of 14.73 pounds per square inch.
 - (b) The average absolute atmospheric pressure shall be assumed to be 14.4 pounds per square inch.

FERC Docket: RP10-77-000

First Revised Sheet No. 523 First Revised Sheet No. 523

Superseding: Original Sheet No. 523

GENERAL TERMS AND CONDITIONS (continued)

5. MEASUREMENT (continued)

- (c) The static pressure of the gas passing through the meters shall be determined by the use of electronic measurement equipment or by the use of another pressure recording device. The instantaneous static pressure measurements from the electronic measurement equipment or the arithmetic average of the temperature recorded each Day shall be used in computing gas volumes.
- (d) The temperature of the gas passing through the meters shall be determined by the use of electronic measurement equipment or other temperature recording device. The instantaneous temperature measurements from the electronic measurement equipment or the arithmetic average of the temperature recorded each Day shall be used in computing gas volumes.
- (e) The specific gravity of the gas flowing through the meters, when necessary for computing gas volumes, shall be measured by methods Algonquin deems to be in accordance with accepted industry practice, such as, but not limited to, recording gravitometer(s) or gas chromatograph(s) located at appropriate point(s). The arithmetic average of the specific gravity recorded each Day or other periodic measurements within a shorter time frame as Algonquin shall select may be applied to metering instruments to provide the volume computation.
- (f) The volume of gas shall be determined in accordance with, but not limited to, the published recommendations of the American Gas Association and the American National Standards Institute as Algonquin deems to be in accordance with accepted industry practice.
- (g) The deviation of the gas from Boyle's and Charles's Law shall be determined in accordance with, but not limited to, the published recommendations of the American Gas Association and the American National Standards Institute as Algonquin deems to be in

FERC Docket: RP04-361-000

Original Sheet No. 524 Original Sheet No. 524: Effective

GENERAL TERMS AND CONDITIONS (continued)

5. MEASUREMENT (continued)

accordance with accepted industry practice. The amount of nitrogen and carbon dioxide contained in the gas shall be determined at a time interval not exceeding three months for the purpose of determining the deviation factors.

6. MEASURING EQUIPMENT

6.1 Measuring Stations.

- (a) At each Point of Receipt, Customer shall cause the operator of the connecting facilities to install, maintain, and operate, at that party's expense, a measuring station properly equipped with measuring equipment which shall consist of the necessary metering devices, designed and installed in accordance with, but not limited to, the published recommendations of the American Gas Association and the American National Standards Institute as Algonquin deems to be in accordance with accepted industry practice.
- (b) At each Point of Delivery, Algonquin will install, maintain, and operate a measuring station properly equipped with measuring equipment which shall consist of the necessary metering devices, designed and installed in accordance with, but not limited to, the published recommendations of the American Gas Association and the American National Standards Institute as Algonquin deems to be in accordance with accepted industry practice. Customer shall reimburse Algonquin for all expenses incurred by Algonquin in connection with the design and construction of such measuring station, together with all associated piping and equipment.

FERC Docket: RP04-361-000

Original Sheet No. 525 Original Sheet No. 525: Effective

GENERAL TERMS AND CONDITIONS (continued)

6. MEASURING EQUIPMENT (continued)

- 6.2 Check Measuring Equipment. Customer may install, operate and maintain, at its own expense, such check measuring equipment as it shall desire, provided that such equipment shall be so installed, maintained and operated as not to interfere with the operation of Algonquin's measuring equipment at or near the Point of Delivery or other facilities or operations. Algonquin shall have access to such check measuring equipment at all reasonable hours, but the operation and maintenance thereof shall be done only by Customer.
- 6.3 Access to Equipment and Records. Both Customer and Algonquin shall have the right to be present at the time of any installing, reading, cleaning, changing, repairing, inspecting, testing, calibrating, or adjusting done in connection with the other's measuring equipment used in measuring deliveries hereunder. The records from such measuring equipment shall remain the property of their owner, but upon request each will submit to the other its records and charts, together with calculations therefrom, for inspection and verification, subject to return within ten days after receipt thereof.
- 6.4 Test of Meters. The accuracy of all measuring equipment shall be verified by the operator at reasonable intervals, and, if requested, in the presence of representatives of either Customer or Algonquin, as applicable, but neither Customer nor Algonquin shall be required to verify the accuracy of such equipment more frequently than once in any thirty-day period. If either Customer or Algonquin at any time observes an error in any such measuring equipment, it will promptly notify the other party, and the parties shall then cooperate to secure a prompt verification of the accuracy of such equipment.

FERC Docket: RP04-361-000

Original Sheet No. 526 Original Sheet No. 526: Effective

GENERAL TERMS AND CONDITIONS (continued)

6. MEASURING EQUIPMENT (continued)

- 6.5 Measuring Equipment Inaccuracy and Failure. The quantity of gas delivered hereunder during periods when the measuring equipment is out of service or registering inaccurately shall be estimated as follows:
 - (a) If, upon test, any measuring equipment, the readings of which are used in the registration, integration, or computation of quantities of gas hereunder, is found to be in error to the extent that it introduces not more than 2% measurement error in the individual measuring equipment affected, previous records of such equipment shall be considered accurate.
 - (b) If, upon test, any such above measuring equipment is found to be inaccurate to the extent that it causes the registered or computed quantities of the instrument(s) so affected to be in error by an amount exceeding 2% at a recording corresponding to the average hourly rate of flow through the instrument(s) in the period since the last preceding test, then any registrations, integration, or computed quantities of such instruments affected shall be corrected to zero error for any part of the period since the last test in which such error is known to have existed or which may be agreed upon by representatives of the parties. In case the period of such error is not known definitely or agreed upon, such correction shall be for a period of one-half of the elapsed time since the date of the last test, but not exceeding a correction period of sixteen days.

FERC Docket: RP04-361-000

Original Sheet No. 527 Original Sheet No. 527: Effective

GENERAL TERMS AND CONDITIONS (continued)

6. MEASURING EQUIPMENT (continued)

- (c) If no tests have been performed to determine the degree of inaccuracy, or if the measuring equipment is out of service, the quantity of gas shall be estimated:
 - (1) By using the registration of any existing and agreed upon check equipment considered by parties concerned to be registering accurately, or, in the absence of such facilities,
 - (2) By correcting the error if the quantity or percentage of error is ascertainable by calibration, test, or mathematical calculation; or if neither of the two foregoing procedures is applicable,
 - (3) By relating the quantity of deliveries to deliveries during periods under similar conditions when the measuring equipment was deemed to have been registering accurately.
- 6.6 Preservation of Records. Each party shall preserve for a period of at least two years all test data, charts and other similar records.

7. RECEIPT AND DELIVERY PRESSURES

Gas shall be received by Algonquin for the account of Customer at each Point of Receipt at a pressure sufficient to enter Algonquin's system at the required rate of flow without requiring a reduction in the pressure on Algonquin's system unless Algonquin, in its reasonable discretion, expressly agrees to do so in the effective service agreement between Algonquin and Customer. In no event shall gas be received at a pressure in excess of Algonquin's Maximum Allowable Operating Pressure.

Algonquin shall deliver gas to Customer at each Point of Delivery at a pressure equal to or greater than the pressure set forth in the effective service agreement between Algonquin and Customer, unless otherwise agreed.

FERC Docket: RP10-77-000

Second Revised Sheet No. 528 Second Revised Sheet No. 528

Superseding: First Revised Sheet No. 528

GENERAL TERMS AND CONDITIONS (continued)

8. TERM OF EXECUTED SERVICE AGREEMENTS

- 8.1 Firm Service Agreements. The term to be covered by an executed firm service agreement shall be as mutually agreed at the time of execution.
- 8.2 Interruptible Service Agreements. The primary term to be covered by an executed interruptible service agreement shall be as mutually agreed at the time of execution.
- 8.3 Extension of Service Agreements. Prior to the expiration of the term of a Part 284 service agreement and prior to Algonquin's posting the availability of capacity under Algonquin's Right of First Refusal provisions, if applicable, Algonquin and Customer may mutually agree to an extension of the term of the service agreement (the exact length of which is to be negotiated on a case-by-case basis, in a not unduly discriminatory manner).
- 9. PREGRANTED ABANDONMENT AND RIGHT OF FIRST REFUSAL
 - 9.1 Except as provided in Section 9.2, Algonquin shall have all necessary abandonment authorizations under the Natural Gas Act upon the (i) expiration of the contractual term, or (ii) the partial reduction of service levels pursuant to either any contractual right for such reduction or a right of first refusal under Section 9.2, or (iii) upon termination of the service agreement under an Open-access Rate Schedule.
 - 9.2 (a) If Customer satisfies the bid matching requirements of this Section 9.2, such Customer may retain its capacity with respect to any capacity under a Service Agreement which qualifies as a ROFR Agreement and which is to be unsubscribed as a result of (i) the expiration of the contractual term, (ii) the partial reduction of service levels, either pursuant to any contractual right for such reduction or pursuant to this Section 9, or (iii) the termination by either Customer or Algonquin of the service agreement, and continue to receive firm service under a new service agreement.
 - (b) Eleven (11) months prior to the effective date of the partial reduction pursuant to a ROFR Agreement or the termination of a ROFR Agreement, Algonquin shall post on the LINK® System and the Web Site the following information:
 - the specific quantity available under the service agreement to be terminated,
 - (2) Point(s) of Receipt and Point(s) of Delivery,
 - (3) the date of expiration,
 - (4) the current maximum rate applicable to the service agreement to be terminated.

FERC Docket: RP10-77-000

Third Revised Sheet No. 529 Third Revised Sheet No. 529

Superseding: Second Revised Sheet No. 529

GENERAL TERMS AND CONDITIONS (continued)

- 9. PREGRANTED ABANDONMENT AND RIGHT OF FIRST REFUSAL (continued)
 - (c) Replacement Customers who desire service to be provided in whole or in part by the capacity posted pursuant to Section 9.2(b) must submit bids for such capacity as described in the applicable posting no later than the last Day of the fourth Month following the Month in which Algonquin posts an applicable notice pursuant to Section 9.2(b); commencing January 1, 2006, such bids must be submitted online via the LINK® System. To be a valid bid, a bid must comply with the bid requirements set forth in Section 9.2(d). At the close of such bidding period, Algonquin shall select among the valid bids the "best bid(s)", as determined pursuant to Section 9.2(e), and shall relay the relevant terms of such "best bid(s)" to the Customer whose capacity has been posted pursuant to Section 9.2(b) within 3 Business Days of the close of such bidding period. If Customer elects to match, as determined by Section 9.2(f), the "best bid(s)", the Customer shall be entitled to retain its capacity and continue to receive firm service under a new service agreement which reflects the matching of the relevant contractual provisions of the "best bid(s)". If Customer does not match the "best bid(s)", then Customer's right of first refusal for the capacity shall terminate and Algonquin shall have all necessary abandonment authorization under the Natural Gas Act. In the event there is no "best bid(s)" for Customer to match, Section 9.2(g) shall determine whether Customer shall be entitled to retain its capacity and continue to receive service.
 - (d) To be a valid bid, the Replacement Customer must provide all information and data required by Sections 2.1 and 3 of the General Terms and Conditions. Algonquin may reject all bids which would require Algonquin to discount below a rate and for discount period agreeable to Algonquin. In the event Algonquin agrees to accept such a discounted rate that is less than the applicable maximum tariff rate, Customer must submit a discount request online via the LINK® System and Algonquin must approve such request pursuant to the provisions of Section 45 of these General Terms and Conditions in order for such rate to become effective.
 - (e) Algonquin shall review all bids from Replacement Customers received pursuant to Section 9.2(c), which have not been rejected by Algonquin, to determine which bid is the "best bid(s)". For purposes of this Section 9, the "best bid(s)" shall be the bids which yields to Algonquin the highest net present value. Net present value shall be calculated on the basis of the present value of the Reservation Charge per unit to Algonquin except that under a Negotiated Rate agreement with a minimum quantity, the net present value evaluation shall also include the fixed cost component of the usage revenue at the minimum quantity. In making the determination of net present value Algonquin shall apply the rate, as of the date of the review, stated in accordance with Section 154.501(d) of the Commission's regulations, to all bids.

FERC Docket: RP10-77-000

Third Revised Sheet No. 530 Third Revised Sheet No. 530

Superseding: Second Revised Sheet No. 530

GENERAL TERMS AND CONDITIONS (continued)

- 9. PREGRANTED ABANDONMENT AND RIGHT OF FIRST REFUSAL (continued)
 - (f) Upon receipt from Algonquin of the "best bid(s)", Customer shall have the right for a thirty (30) day period in which to notify Algonquin in writing whether the Customer is willing to match the "best bid(s)" for the capacity in whole or in part, made available by (i) the termination of such service agreement, (ii) expiration of the contract term by its own terms, or (iii) partial reduction pursuant to contractual right or the reduction of service level pursuant to a right of first refusal. Commencing January 1, 2006, such notification must be submitted online via the LINK® System. Failure to notify Algonquin within said thirty (30) day period constitutes a non-revocable waiver of Customer's right to match the "best bid(s)" and termination of the right of first refusal for the capacity. In order to match the "best bid(s)", Customer must agree to a rate up to the maximum rate and contract term that provide Algonquin with at least the same net present value, for an equivalent amount of capacity, as the valid "best bid(s)" submitted by the Replacement Customer(s); provided, however, the maximum rate a Customer must match is the maximum rate Algonquin can charge for delivery to the Customer's Point(s) of Delivery under the Agreement which is subject to the Customer's right to match the "best bid".
 - (g) In the event Algonquin does not receive any bids pursuant to Section 9.2(c) or Algonquin rejects all bids received due to the fact that such bids were premised on rate discount levels or rate discount periods unacceptable to Algonquin, Algonquin and Customer may mutually agree upon the terms and conditions under which Customer shall be entitled to retain its capacity and continue to receive service. In no event shall Algonquin and Customer agree upon terms which yield to Algonquin a net present value less than any bid received pursuant to Section 9.2(c) and rejected by Algonquin and in no event shall Algonquin be obligated to sell capacity at less than the maximum lawful price for such capacity. In the event Algonquin and Customer have not reached agreement on the terms and conditions under which service will be extended, at Customer's election, such election to be exercised prior to the date which is five months before the termination date, Algonquin shall tender and Customer shall execute within 20 days of receipt, a new service agreement reflecting service for all or part of the contractual quantity, so long as such Customer agrees to pay the maximum rate; provided, however, if Algonquin and Customer mutually agree to a lower rate which yields to Algonquin a net present value equal to or greater than any bid received pursuant to Section 9.3 and rejected by Algonquin, the Customer does not have to pay the maximum rate to retain the capacity.
 - (h) In the event Algonquin and Customer do not execute a new service agreement pursuant to the provisions of this Section 9.2, Algonquin shall have all necessary pregranted abandonment authorization as to any part of the contractual quantity which is not covered by a new service agreement. The new service agreement must meet all the requirements of the definition of ROFR Agreement for Customer to continue to have the right of first refusal pursuant to Section 9.2.
 - 9.3 Capacity which is not subject to the right of first refusal claim pursuant to this Section 9 shall be posted as available capacity on Algonquin's Web Site and shall be marketed in accordance with Section 2 of Algonquin's General Terms and Conditions.

FERC Docket: RP04-361-000

Original Sheet No. 531 Original Sheet No. 531: Effective

GENERAL TERMS AND CONDITIONS (continued)

10. JOINT OBLIGATIONS

When Customer under a service agreement consists of two or more persons, the obligations of such persons under such agreement shall be joint and several and, except as otherwise provided, any action provided to be taken by Customer shall be taken by such persons jointly.

11. DISCRETIONARY WAIVER

Algonquin may waive any of its rights hereunder or any obligations of Customer on a basis that is not unduly discriminatory; provided that no waiver by either Customer or Algonquin of any one or more defaults by the other in the performance of any provision of the service agreement between Customer and Algonquin shall operate or be construed as a waiver of any future default or defaults, whether of a like or of a different character.

12. MODIFICATION

No modification to the terms and provisions of any service agreement or exhibit thereto shall be or become effective except by the execution of a superseding service agreement or exhibit thereto.

13. SUCCESSORS IN INTEREST

Any company which shall succeed by purchase, merger, consolidation or otherwise to the properties substantially as an entirety, of Algonquin or of Customer, used or intended to be used for rendering gas service authorized by the Commission, shall be entitled to the rights and shall be subject to the obligations of its predecessors in title under a service agreement. In accordance with the capacity release procedures set forth in Section 14 below, Customer may, without relieving itself of its obligations under such agreement, assign any of its rights and obligations thereunder to another shipper, but otherwise no assignment of such agreement, or of any of the rights or obligations thereunder shall be made unless there first shall have been obtained the consent thereto of Algonquin, in the event of any assignment by Customer, or the consent thereto of Customer, in the event of an assignment by Algonquin. These restrictions on assignment shall not in any way prevent any party from pledging or mortgaging its rights under a service agreement as security for its indebtedness.

FERC Docket: RP10-77-000

Second Revised Sheet No. 532 Second Revised Sheet No. 532

Superseding: First Revised Sheet No. 532

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE

- 14.1 General. This Section 14 sets forth the sole means by which a Customer under any firm, open-access Part 284 Service Agreement ("Releasing Customer") may release its capacity rights under a Service Agreement ("Existing Service Agreement"). Subject to the following provisions, firm capacity rights shall be released to Algonquin for resale by it on a firm basis. This Section 14 is not applicable to any Customer whose Service Agreement has been executed pursuant to an individually certificated service.
- 14.2 Capacity Eligible for Release. A Customer under any firm rate schedule under Part 284 may release all or a part of its capacity under an Existing Service Agreement, on a full Day or a partial Day basis, on a permanent or temporary basis, subject to recall if so specified. A Releasing Customer may propose a Customer ("Prearranged Customer") to obtain the Releasing Customer's released capacity, provided the Prearranged Customer is an Approved Bidder, as set forth in Section 14.3 below.

In addition to releasing a portion of all capacity rights under a firm service agreement under Rate Schedules AFT-1, AFT-E, AFT-1S, AFT-ES and AFT-CL, capacity rights under these rate schedules may be released in discrete packages representing all or a fraction of the firm (within Base Flow Path) capacity rights in any segment of Algonquin's system, subject to the provisions regarding overlapping segments set forth in Section 48.1(B) of the General Terms and Conditions. Releasing Customers and Replacement Customers shall not have rights through this Section 14.2 to use overlapping segments where, in the overlapped portion, the total capacity released and scheduled exceeds the MDTQ for the original Service Agreement.

14.3 Approved Bidders. To qualify as an Approved Bidder, a party must have been accepted by Algonquin as satisfying the credit standards of Section 3 of these General Terms and Conditions and must continue to satisfy the credit standards of Section 3 when its bid is made and accepted. Algonquin shall process applications for credit approval with diligence. Names of Approved Bidders shall be placed on a list that is posted on the LINK® System. Any previously listed Approved Bidder that fails to continue to satisfy the standards of Section 3 shall be deleted from the list of Approved Bidders. Algonquin will waive the creditworthiness requirement on a non-discriminatory basis for Replacement and Prearranged Customers and permit them to submit bids, if the Releasing Customer provides Algonquin with a guarantee or other form of credit assurance in form and substance satisfactory to Algonquin of the financial obligations of the Replacement or Prearranged Customer with respect to the capacity being released by Releasing Customer for the specific transaction guaranteed by the Releasing Customer prior to the commencement of service to the Replacement or Prearranged Customer.

14.4 Release Procedures.

(a) Notice by Releasing Customer. A Releasing Customer wanting to release capacity shall notify Algonquin via the LINK® System or, where applicable, by electronic data interchange (hereinafter called "Releasing Customer's Notice"). Releasing Customer's Notice may include reasonable terms and conditions that are objectively stated, not contrary to any applicable provision of these General Terms and Conditions, applicable to all potential bidders, and non-discriminatory, and shall include the following:

FERC Docket: RP10-77-000

Fifth Revised Sheet No. 533 Fifth Revised Sheet No. 533

Superseding: Fourth Revised Sheet No. 533

GENERAL TERMS AND CONDITIONS (continued)

- 14. CAPACITY RELEASE (continued)
 - (1) Releasing Customer's Service Agreement number;
 - whether the release is recallable or non-recallable; and if recallable,

 (i) whether the recall rights are on a full Day or a partial Day basis,

 (ii) the specifics of any recall right, (iii) whether the Releasing

 Customer's recall notification must be provided exclusively on a Business

 Day, and (iv) any reput methods and rights associated with returning the

 previously recalled capacity to the Replacement Customer. These rights

 and methods may be either: (i) reput must be accepted by the original

 Replacement Customer for the original terms of the release, or (ii)

 reput may be accepted at the option of the original Replacement Customer

 for the original terms of the release.
 - (3) the MDTQ, expressed as a numeric quantity, to be released, subject to the calculation described in Section 14.4(q) below;
 - the primary Point(s) of Receipt and the MDRO associated with each such point, and/or the primary Point(s) of Delivery and the MDDO associated with such point, subject to the calculation described in Section 14.4(g) below, if the Releasing Customer is proposing to release any of its primary points and the associated point quantities;
 - (5) the period of the release;
 - (6) any other conditions of the release;
 - (7) identity of any Prearranged Customer;
 - (8) any minimum reservation charge requirements or, if Releasing Customer elects to entertain bids on a volumetric rate basis, the minimum reservation charge stated on a volumetric basis. For releases that become effective on or after July 30, 2008, the Releasing Customer may specify a minimum Reservation Charge requirement which is in excess of the maximum tariff rate for the applicable service if (i) the term of the proposed release is one (1) year or less, and (ii) the effective date of the proposed release is on or before one (1) year from the date on which Algonquin is notified of the release.
 - (9) whether bids should be submitted in dollars and cents or percentages of maximum tariff rate;
 - (10) for biddable releases, the bid evaluation method which shall be, at the Releasing Customer's option, one of the following three standard evaluation methods: highest rate, net revenue or present value; or an alternative Releasing Customer defined bid evaluation method, including, but not limited to, the application of the Risk of Default Factor as set forth in Section 2.5(e); provided, however, Releasing Customer's specified alternative bid evaluation method must be set forth with sufficient specificity that Algonquin's evaluation of the bids to determine the "best bid" is a purely ministerial matter that does not require any discretionary exercise of judgment by Algonquin; in addition, Releasing Customer must specify the Tie Break Method that Algonquin will apply to award capacity among multiple bids that yield the same value;

FERC Docket: RP09-72-000

Second Revised Sheet No. 534 Second Revised Sheet No. 534

Superseding: First Revised Sheet No. 534

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

- (11) whether contingent bids may be submitted and if so, the date by which each contingent bidder will be required to satisfy or eliminate the contingency if Releasing Customer elects to allow the bidder additional time beyond the time period specified in Section 14.4(b) to satisfy or eliminate the contingency;
- (12) the Business Day on which the Bid Period will expire, if the Releasing Customer desires to establish a Bid Period that is longer than that determined in Section 14.4(b);
- (13) the expiration of the period within which the Prearranged Customer may match the best bid, if the Releasing Customer desires to establish a match period that is longer than that determined in Section 14.4(b). Releasing Customer shall not be allowed to specify an extension of the original Bid Period or the Prearranged deal Matching Period without posting a new release;
- (14) e-mail address for the Releasing Customer contact person. It is the Releasing Customer's responsibility to update e-mail address information provided to Algonquin as necessary;
- (15) the recall notification period(s), as identified in Section 14.4(h) below, that will be available for use by the parties;
- (16) the priorities that Algonquin is authorized to utilize in the event that overlapping nominations submitted by the Releasing Customer and any Replacement Customer are in excess of the Releasing Customer's original MDTQ;
- (17) whether the Releasing Customer will allow the Replacement or Prearranged Customer to amend the Primary Point(s) of Receipt and/or Primary Point(s) of Delivery after the capacity is awarded;
- (18) whether the Prearranged Customer is affiliated with the Releasing Customer;
- (19) whether the Releasing Customer will allow the Replacement or Prearranged Customer to re-release the capacity acquired from the Releasing Customer;

FERC Docket: RP09-72-001

Substitute Original Sheet No. 534A Substitute Original Sheet No. 534A

Superseding: Original Sheet No. 534A

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

- (20) whether the proposed release is to an asset manager as part of an asset management arrangement as defined in Section 284.8(h)(3) of the Commission's regulations or to a marketer participating in a state-mandated retail access program as defined in Section 284.8(h)(4) of the Commission's regulations, and, if the proposed release is part of an asset management arrangement, the volumetric level of the asset manager's delivery or purchase obligation and the time period during which that obligation is in effect;
- (21) for purposes of bidding and awarding, any maximum and/or minimum rates specified by Releasing Customer shall include the tariff reservation rate and all demand surcharges, specified as a total number or as stated separately. For releases that become effective on or after July 30, 2008, the maximum and/or minimum rates specified by Releasing Customer may exceed the maximum tariff rate for the applicable service if (i) the term of the proposed release is one (1) year or less, and (ii) the effective date of the proposed release is on or before one (1) year from the date on which Algonquin is notified of the release.

Releasing Customer may elect not to reveal minimum conditions to anyone other than Algonquin.

(b) Posting Requirements and Bid Periods for Standard Releases Subject to Bidding. The Capacity Release timeline is applicable to all parties involved in the Capacity Release process; however, it is only applicable if 1) all information provided by the parties to the transaction is valid and the Replacement Customer has been determined to be creditworthy before the capacity release bid is tendered and 2) there are no special terms or conditions of the release. Algonquin shall post offers and bids, including prearranged deals, upon receipt; provided, however, offers and bids must be complete before posting. Only posted offers and bids will be available electronically. If a Releasing Customer requests a later posting time, Algonquin shall support such request insofar as it comports with the standard timeline set forth in this Section 14.4(b). The term "Bid Period" shall mean the period during which bids may be submitted, commencing upon the posting of Releasing Customer's Notice and ending at the time specified below.

FERC Docket: RP10-77-000

Second Revised Sheet No. 535 Second Revised Sheet No. 535

Superseding: First Revised Sheet No. 535

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

- For short term releases (one year or less),
 offers shall be tendered by 12:00 p.m. CT on a Business Day.
- The Bid Period shall end at $1:00 \ \mathrm{p.m.}$ CT on the same Business Day or at 1:00 p.m. CT on any subsequent Business Day as specified in Releasing Customer's Notice, whichever is elected by Releasing Customer.
- The evaluation period begins at 1:00 p.m. CT on the Day the Bid Period ends, during which contingency is eliminated, the determination of the best bid is made, and ties are broken.
- The evaluation period ends and award is posted if no match is required at 2:00 p.m. CT the Day the bid is evaluated.
- If the Prearranged Customer's bid is not the "best bid," Algonquin shall allow such Prearranged Customer a Matching Period of thirty (30) minutes from 2:00 p.m. CT to 2:30 p.m. CT on the Day the Bid Period ends to determine whether to match the "best bid" and to notify Algonquin of its decision.
- If a match is required, Algonquin will post the award by 3:00 p.m. CT.
- Algonquin will issue a contract within one Hour of the award posting (with a new contract number, when applicable).

Nominations for gas flow are possible beginning at the next available nomination cycle for the effective date of the contract. Such nominations will be processed in accordance with the nomination and scheduling requirements of Sections 22 and 23 of the General Terms and Conditions; however, in no circumstances will gas flow prior to the effective date of the release as specified in Releasing Customer's Notice.

- For long term releases (more than one year),
 - offers shall be tendered by 12:00 p.m. CT four (4) Business Days before the award.
 - The Bid Period shall end at 1:00 p.m. CT on the Business Day before timely nominations are due (open season is three Business Days) or at 1:00 p.m. CT on any subsequent Business Day as specified in Releasing Customer's Notice, whichever is elected by Releasing Customer.
 - The evaluation period begins at 1:00 p.m. CT on the Day the Bid Period ends, during which contingency is eliminated, the determination of the best bid is made, and ties are broken.
 - The evaluation period ends and award is posted if no match is required at 2:00 p.m. CT the same Day.
 - If the Prearranged Customer's bid is not the "best bid," Algonquin shall allow such Prearranged Customer a Matching Period of thirty (30) minutes from 2:00 p.m. CT to 2:30 p.m. CT on the Day the Bid Period ends to determine whether to match the "best bid" and to notify Algonquin of its decision.
 - If a match is required, Algonquin will post the award by 3:00 p.m. CT.
 - Algonquin will issue a contract within one Hour of the award posting (with a new contract number, when applicable).

Nominations for gas flow are possible beginning at the next available nomination cycle for the effective date of the contract. Such nominations will be processed in accordance with the nomination and scheduling requirements of Sections 22 and 23 of the General Terms and Conditions; however, in no circumstances will gas flow prior to the effective date of the release as specified in Releasing Customer's Notice.

FERC Docket: RP10-77-000

First Revised Sheet No. 536 First Revised Sheet No. 536

Superseding: Original Sheet No. 536

GENERAL TERMS AND CONDITIONS (continued)

- 14. CAPACITY RELEASE (continued)
 - (c) Timeline for Non-Standard Releases

If the Releasing Customer specifies a bid evaluation methodology other than highest rate, net revenue or present value, or any other special terms or conditions, the above timeline shall apply; provided, however, one additional Business Day will be added to the evaluation period. Such extended evaluation period shall cause gas flow to be at least one Day later than gas could flow under the timeline set forth in Section $14.4\,(b)\,(1)$ or Section $14.4\,(b)\,(2)$.

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 537 Fourth Revised Sheet No. 537

Superseding: Substitute Third Revised Sheet No. 537

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

- (d) Notice by Algonquin. Releasing Customer's Notice, including the identity of the Releasing Customer shall be posted on the LINK® System and by electronic data interchange until the expiration of the applicable Bid Period; however, Pipeline shall not post any minimum conditions that Releasing Customer has elected not to disclose. At the Releasing Customer's request Algonquin may, but shall not be obligated to take other action to market any capacity which the Releasing Customer desires to release, and the Releasing Customer may search for any party to take its capacity.
- Bids. Only Approved Bidders that have executed a Capacity Release Umbrella Agreement, the form of which is contained in this tariff, may bid on capacity which the Releasing Customer proposes to release. Bids shall be made on the LINK® System or, where applicable, through electronic data interchange during the Bid Period on an open bidding basis, provided that the bidders' names shall not be posted except for the winning bidder. The Approved Bidder's bid submitted via the LINK® System or electronic data interchange shall include an e-mail address for at least one contact person, which will not be posted. It is the Approved Bidder's responsibility to update e-mail address information provided to Algonquin as necessary. Bids must meet the minimum terms of Releasing Customer's Notice in order to be considered a valid bid. Invalid bids shall be rejected. For releases that become effective on or after July 30, 2008, an Approved Bidder may submit a bid in excess of the maximum tariff rate for the applicable service if (i) the term of the proposed release is one (1) year or less, and (ii) the effective date of the proposed release is on or before one (1) year from the date on which Algonquin is notified of the release. Algonquin shall select the bid to be accepted in accordance with Section 14.5(c). If permitted by Releasing Customer's Notice, and subject to the provisions thereof, the bidder may submit a contingent bid by including the details of the contingency in its bid; provided, however, that such contingency must not be contrary to any applicable provision of these General Terms and Conditions.
- (f) If there are awards of capacity for which there have been multiple bids meeting minimum conditions, the best bid shall be awarded first, until all offered capacity is awarded.
- (g) Partial Day Release Quantity
 - (1) The daily contractual entitlement that can be released by a Releasing Customer for a partial Day release is limited to the lesser of:
 - (i) the quantity contained in the Releasing Customer's Notice submitted by the Releasing Customer; or
 - (ii) a quantity equal to the difference between the MDTQ for the contract to be released by the Releasing Customer and the quantity scheduled for that period of the Day prior to the effective time of the release of the capacity, based upon the elapsed-prorated-scheduled quantity.

This allocated daily contractual entitlement shall be used for purposes of nominations, billing, and if applicable, for overrun calculations.

(2) If on the Day of a partial Day release the Releasing Customer's existing scheduled quantity exceeds the MDTQ remaining on the original contract after the award of the partial Day release, then the Releasing Customer must reduce its nominated quantity to a quantity that is equal to or less than the MDTQ remaining on the original contract.

FERC Docket: RP10-77-000

First Revised Sheet No. 538 First Revised Sheet No. 538

Superseding: Original Sheet No. 538

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

(h) Recall Provisions

Releasing Customer's rights to recall capacity on a full Day or partial Day basis shall be stated clearly in Releasing Customer's Notice. The purchase of gas by a Releasing Customer from a Replacement Customer at the Releasing Customer's city gate point(s) shall not be deemed to be the exercise of a recall by the Releasing Customer.

The Releasing Customer shall provide capacity recall notification to Algonquin via the LINK® System. The recall notification shall specify the recall notification period for the specified effective Gas Day, as well as any other information needed to uniquely identify the capacity being recalled.

Algonquin shall support the following recall notification periods for all released capacity subject to recall rights:

(i) Timely Recall Notification:

- (a) A Releasing Customer recalling capacity should provide notice of such recall to Algonquin and the first Replacement Customer no later than 8:00 A.M. CCT on the Day that Timely Nominations are due;
- (b) Algonquin shall provide notification of such recall to all affected Replacement Customers no later than 9:00 A.M. CCT on the Day that Timely Nominations are due;
- (ii) Early Evening Recall Notification:
 - (a) A Releasing Customer recalling capacity should provide notice of such recall to Algonquin and the first Replacement Customer no later than 3:00 P.M. CCT on the Day that Evening Nominations are due;
 - (b) Algonquin shall provide notification of such recall to all affected Replacement Customers no later than 4:00 P.M. CCT on the Day that Evening Nominations are due;
- (iii) Evening Recall Notification:
 - (a) A Releasing Customer recalling capacity should provide notice of such recall to Algonquin and the first Replacement Customer no later than 5:00 P.M. CCT on the Day that Evening Nominations are due;
 - (b) Algonquin shall provide notification of such recall to all affected Replacement Customers no later than 6:00 P.M. CCT on the Day that Evening Nominations are due;
- (iv) Intraday 1 Recall Notification:
 - (a) A Releasing Customer recalling capacity should provide notice of such recall to Algonquin and the first Replacement Customer no later than 7:00 A.M. CCT on the Day that Intraday 1 Nominations are due;
 - (b) Algonquin shall provide notification of such recall to all affected Replacement Customers no later than 8:00 A.M. CCT on the Day that Intraday 1 Nominations are due; and
- (v) Intraday 2 Recall Notification:
 - (a) A Releasing Customer recalling capacity should provide notice of such recall to Algonquin and the first Replacement Customer no later than 2:30 P.M. CCT on the Day that Intraday 2 Nominations are due;
 - (b) Algonquin shall provide notification of such recall to all affected Replacement Customers no later than 3:30 P.M. CCT on the Day that Intraday 2 Nominations are due.

FERC Docket: RP10-77-000

Second Revised Sheet No. 539 Second Revised Sheet No. 539

Superseding: First Revised Sheet No. 539

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

For recall notification provided to Algonquin prior to the recall notification deadline specified above and received between 7:00 A.M. CCT and 5:00 P.M. CCT, Algonquin shall provide notification to all affected Replacement Customers no later than one Hour after receipt of such recall notification. For recall notification provided to Algonquin after 5:00 P.M. CCT and prior to 7:00 A.M. CCT, Algonquin shall provide notification to all affected Replacement Customers no later than 8:00 A.M. CCT after receipt of such recall notification.

Algonquin's notices of recalled capacity to all affected Replacement Customers shall be provided via the LINK® System, along with written notice via e-mail communication to those Replacement Customer contact person(s) identified in the Replacement Customer's bid submitted pursuant to Section 14.4(e) of these General Terms and Conditions. Such notices shall contain the information required to uniquely identify the capacity being recalled, and shall indicate whether penalties will apply for the Gas Day for which quantities are reduced due to a capacity recall. Upon receipt of notification of the recall from Algonquin, each affected Replacement Customer shall revise its nominations within the applicable nomination cycle in order to implement the recall. Each affected Replacement Customer will be solely responsible for adjusting its supply and transportation arrangements, which may be necessary as a result of such recall. Replacement Customers involved in rerelease transactions may receive notice slightly after the first Replacement Customer receives notice. The recalling Releasing Customer may nominate the recalled capacity consistent with the applicable nomination cycle, as defined in Section 22.5 of these General Terms and Conditions.

For recall notifications provided to Algonquin during the Timely, Early Evening or Evening recall notification period, the recall shall be effective as of the beginning of the specified effective Gas Day. For recall notifications provided during the Intraday 1 or Intraday 2 recall notification period, the recall shall be effective at 5:00 p.m. CCT or 9:00 p.m. CCT, respectively, on the specified effective Gas Day.

FERC Docket: RP10-77-000

First Revised Sheet No. 540 First Revised Sheet No. 540

Superseding: Original Sheet No. 540

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

(i) Partial Day Recall Quantity

The daily contractual entitlement that can be recalled by a Releasing Customer for a partial Day recall is a quantity equal to the lesser of:

- (1) The quantity specified in the Releasing Customer's notice to recall capacity; or
- (2) The difference between the quantity released by the Releasing Customer and the Elapsed Prorata Capacity.

In the recall notification provided to Algonquin by the Releasing Customer, the quantity to be recalled shall be expressed in terms of the adjusted total released capacity entitlements based upon the Elapsed Prorata Capacity. In the event of an intra-day capacity recall, Algonquin shall determine the allocation of capacity between the Releasing Customer and the Replacement Customer(s) based upon the Elapsed Prorata Capacity.

The amount of capacity allocated to the Replacement Customer(s) shall equal the original released quantity less the recalled capacity. This allocated daily contractual quantity shall be used for purposes of nominations, billing, and if applicable, for overrun calculations. As a result of the allocation of capacity described in this section, Algonquin shall not be obligated to deliver a combined quantity to the Releasing Customer and the Replacement Customer(s) that is in excess of the total daily contract quantity of the release.

(j) Reput Provisions

Algonquin shall support the function of reputting by the Releasing Customer. The Releasing Customer may reput previously recalled capacity to the Replacement Customer pursuant to the reput rights and methods identified in the Releasing Customer's Notice to release capacity, as required by Section 14.4(a)(2) above. When capacity is recalled, such capacity may not be reput for the same Gas Day. The deadline for the Releasing Customer to notify Algonquin of a reput of capacity is 8:00 A.M. CCT to allow the Replacement Customer to submit timely nominations for gas to flow on the next Gas Day.

FERC Docket: RP09-72-000

Second Revised Sheet No. 541 Second Revised Sheet No. 541

Superseding: First Revised Sheet No. 541

GENERAL TERMS AND CONDITIONS (continued)

- 14. CAPACITY RELEASE (continued)
- 14.5 Effect of Notices and Bids.
 - (a) The posting of Releasing Customer's Notice shall constitute an invitation to entertain bids on the Releasing Customer's firm transportation contract rights and obligations pursuant to its terms and the terms herein, but shall not constitute an offer by Algonquin. A bid shall constitute an offer by the bidding party to accept from Algonquin such rights and obligations, and shall remain open until Algonquin accepts the bid or a competing bid, unless retracted by the bidder prior to the close of the Bid Period by any means that meets the requirements of the tariff and the Releasing Customer's terms and conditions. Bids will be legally binding on the Replacement Customer or Prearranged Customer until written or electronic notice of withdrawal is received by Algonquin; provided, however, bids cannot be withdrawn after the bid period ends. Bids may be revised during the Bid Period, and shall be deemed received as of the time of the last revision; provided, however, that no bidder shall maintain more than one bid for the same capacity. No bid may be resubmitted or revised in order to submit a lower bid.
 - (b) Releasing Customer's Notice shall constitute authorization for Algonquin to accept bids by posting notice of such acceptance on the LINK® System, and where applicable, by electronic data interchange. Releasing Customer may withdraw Releasing Customer's Notice at any time prior to the close of the Bid Period associated with Releasing Customer's Notice, where unanticipated circumstances justify the withdrawal and no bids meeting the minimum conditions have been made. The Releasing Customer's notice will be legally binding on the Releasing Customer until the written or electronic notice of withdrawal is received by Algonquin. The reason for withdrawal must be submitted to Algonquin for posting.
 - (c) Algonquin shall evaluate bids in accordance with the bid evaluation method identified, pursuant to Section 14.4(a)(10) of these General Terms and Conditions, in the Releasing Customer's Notice.
 - (1) If there is only one valid bid, Algonquin shall award the capacity to the bidder that submitted such bid, subject to any Prearranged Customer's exercise of its matching rights.
 - (2) If there is more than one valid bid, then, subject to any Prearranged Customer's exercise of its matching rights, Algonquin shall award the capacity to the bidder whose bid yields the highest value based on the bid evaluation method specified in the Releasing Customer's Notice, taking into account the price, volume and term of the bid, as applicable. If more than one bid yields the same value under this subsection (2), Algonquin shall award the capacity according to the method specified in the Releasing Customer's Notice ("Tie Break Method") as a means for awarding the released capacity among multiple equal bids.

FERC Docket: RP09-72-000

Second Revised Sheet No. 542 Second Revised Sheet No. 542

Superseding: First Revised Sheet No. 542

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

- (3) For purposes of determining the value of a bid, Algonquin shall use only the reservation charge (including reservation charges stated on a volumetric basis). If the bid evaluation method specified by the Releasing Customer is present value, Algonquin shall use a discount rate of ten (10) percent.
- (4) In the event of a tie, a bid with no contingency shall be accepted over a bid with a contingency even if the bid with no contingency was received later in time.
- (5) If the winning bid is a contingent bid, the bidder that submitted such contingent bid will be required to satisfy or eliminate any contingency in accordance with the capacity release timeline set forth in Section 14.4(b) or in the Releasing Customer's Notice, as applicable, and shall confirm to Algonquin via e-mail to link-help@spectraenergy.com that the contingency has been satisfied or eliminated. In the event that such bidder fails to satisfy or eliminate its contingency pursuant to this subsection (5), the capacity will be awarded to the next highest bidder(s) as determined pursuant to this Section 14.5(c).
- (d) In the event that the Prearranged Customer desires to exercise its right to match the "best bid," the Prearranged Customer must notify Algonquin via the LINK® System.
- (e) Algonquin shall not award capacity release offers to the Replacement Customer until and unless the Replacement Customer meets Algonquin's creditworthiness requirements applicable to all services it receives from Algonquin, including the service represented by the capacity release.
- (f) Algonquin shall post on the LINK® System and by electronic data interchange the winning bid and the identity of the winning bidder immediately upon accepting the winning bid.
- 14.6 Effect of Acceptance of Bid. Algonquin's acceptance of a bid shall, subject to the other provisions hereof, result in and constitute a binding agreement (a) between Algonquin and the Releasing Customer pursuant to which the Releasing Customer releases the rights described in the accepted bid without being released from any of its obligations under the Existing Service Agreement (the "Release") and (b) between Algonquin and the bidder whose bid was accepted ("Successful Bidder") pursuant to which the Successful Bidder acquires from Algonquin all of the Releasing Customer's released rights under its firm Service Agreement with Algonquin that are described in the accepted bid and the Successful Bidder agrees to perform timely and irrevocably all obligations under the terms of the Service Agreement applicable to the released capacity, in the form of an addendum to the Capacity Release Umbrella Agreement incorporating the specific terms of the accepted bid ("New

FERC Docket: RP09-72-000

Second Revised Sheet No. 543 Second Revised Sheet No. 543

Superseding: First Revised Sheet No. 543

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

Service Agreement"). Algonquin shall tender electronically to the Successful Bidder a New Service Agreement in the form of an Addendum to the Capacity Release Umbrella Agreement incorporating the terms of the accepted bid. A party that acquires released capacity from Algonquin under this Section 14 shall be a Customer for all purposes under this tariff, including capacity release under this Section 14, subject to the re-release rights specified by Releasing Customer pursuant to Section 14.2(a)(19) above. Algonquin will notify the Releasing Customer with recall rights of the name of new Replacement Customers who subsequently obtain all or a portion of such capacity after the Addendum to the Capacity Release Umbrella Agreement is tendered. Nothing in this Section 14 shall operate as a waiver of any other requirements of this tariff applicable to Customers, including requirements regarding Customer eligibility and scheduling of service. Upon acceptance of its bid by Algonquin a Replacement Customer may schedule service and gas may flow upon the effective date of the release, subject to compliance with the other provisions of this tariff.

- 14.7 Obligations of Releasing Customer and Other Matters. Notwithstanding anything to the contrary herein, in any Releasing Customer's Notice, in any bid or accepted bid, in any Release, in any New Service Agreement or otherwise:
 - (a) The Release or the New Service Agreement relating thereto shall not constitute a release of any of the obligations of the Releasing Customer under the Existing Service Agreement and the Releasing Customer shall remain fully liable for all of those obligations, including, without limitation, all reservation charges, reservation surcharges, and other fixed costs owing under that Service Agreement in respect of capacity rights released pursuant to the Release, as if the Release had never been entered into; provided, however, Algonquin shall credit the Releasing Customer with all reservation charges, reservation surcharges, and other fixed costs ("Fixed Charges"), but not charges related to usage, actually

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 544 Fourth Revised Sheet No. 544

Superseding: Substitute Third Revised Sheet No. 544

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

paid by a Replacement Customer and received by Algonquin which are owed in respect of the capacity rights released by that Releasing Customer and acquired by that Replacement Customer from Algonquin under a New Service Agreement pursuant to Releasing Customer's release (but not pursuant to subsequent releases of the same capacity), even if greater than the Fixed Charges obligation of the Releasing Customer, up to the maximum rate for the released capacity. Under Negotiated Rate agreements, the Releasing Customer is obligated to pay Algonquin the difference by which the Negotiated Rate exceeds the rate paid by the Replacement Customer. Under Negotiated Rate agreements, Algonquin and the Releasing Customer may agree upon payment obligations and crediting mechanisms which may vary from or are different from those set forth in Algonquin's capacity release provisions. Replacement Customer under a New Service Agreement shall be liable to Algonquin for all obligations incurred thereunder, including Fixed Charges and charges related to usage. Any Replacement Customer shall be entitled to receive refunds to the extent the rate paid by it is subsequently determined to be in excess of a just and reasonable rate; provided, however, that if the Releasing Customer is subject to a discounted rate and received credits attributable to Fixed Charges that Algonquin is required to refund, Algonquin shall be entitled to reverse such credits and shall be required to refund such amounts to the Replacement Customer only to the extent actually collected back from the Releasing Customer. In addition, for releases that become effective on or after July 30, 2008, any rate paid by a Replacement Customer in any capacity release transaction with a term of one (1) year or less which is not subject to the maximum rate cap is deemed to be a final rate and is not subject to refund if the effective date of the release is on or before one (1) year from the date on which Algonquin was notified of the release.

- (b) The service agreement between Algonquin and each Releasing Customer shall remain in full force and effect and unmodified, as if the Release from that Releasing Customer and related New Service Agreement had never been entered into, except (i) the Releasing Customer shall not be entitled to any of the rights under the service agreement which are released by it for the period set forth in the Release and (ii) the Releasing Customer shall be entitled to a credit as provided in this subsection. Notwithstanding the above, in the event there is a permanent release of Releasing Customer's Capacity pursuant to Section 14.14 of these General Terms and Conditions, Algonquin will agree to discharge Releasing Customer of liability, on a prospective basis, in whole or in part.
- (c) Algonquin shall bill the Releasing Customer the amount it is obligated to pay for Fixed Charges and shall simultaneously credit such invoice with the Fixed Charges that Algonquin is entitled to recover under all New Service Agreements which are owed in respect of the capacity rights released by the Releasing Customer and acquired pursuant to such release (but not pursuant to subsequent releases of the same capacity); provided, however, that such credit shall be reversed, and

FERC Docket: RP07- 13-000

First Revised Sheet No. 545 First Revised Sheet No. 545 : Effective

Superseding: Original Sheet No. 545

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

Algonquin shall be entitled to interest from Releasing Customer in accordance with Section 19 of these General Terms and Conditions, to the extent Algonquin does not recover such Fixed Charges from the Replacement Customer on the due date. To the extent Releasing Customer makes payment in satisfaction of any such obligations it shall be subrogated to Algonquin's rights to collect such amounts from the defaulting Customer under the New Service Agreement.

- (d) Algonquin may rely upon and shall be protected in acting or refraining from acting upon any resolution, certificate, statement, instrument, notice, or other document believed by it to be genuine and to have been signed or presented by the proper party or parties.
- (e) Algonquin shall not be bound to make any investigation into the facts or matters stated in any resolution, certificate, statement, instrument, notice, or other document, but Algonquin, in its reasonable discretion, may make such further inquiry or investigation into such facts or matters as it may see fit.
- (f) Except as expressly set forth in Section 14.7(a), each Releasing Customer agrees to protect and indemnify Algonquin against, and to release and hold Algonquin harmless against, any loss, liability or expense (including, without limitation, court costs and attorneys' fees) incurred or suffered by Algonquin or such Releasing Customer arising out of or in connection with the provisions of this Section 14 (including, without limitation, Algonquin's performance or failure of performance of any of its obligations under this Section 14).
- (g) Without in any way limiting the other provisions of this Section 14, each Approved Bidder, by delivery of a bid to Algonquin, shall be deemed to have agreed:
 - (i) to the provisions of (d) and (e) above;
 - (ii) to protect and indemnify Algonquin against, and to release and hold Algonquin harmless against, any loss,

FERC Docket: RP09-72-000

Second Revised Sheet No. 546 Second Revised Sheet No. 546

Superseding: First Revised Sheet No. 546

GENERAL TERMS AND CONDITIONSPRIVATE (continued)

14. CAPACITY RELEASE (continued)

liability or expense (including, without limitation, court costs and attorneys' fees) incurred or suffered by Algonquin or such Approved Bidder arising out of or in connection with the provisions of this Section 14 (including, without limitation, Algonquin's performance or failure of performance of any of its obligations under this Section 14); and

- (iii) that Algonquin shall have disclaimed all express and implied representations and warrants concerning the released rights acquired by the bidder, the New Service Agreement and the related Release.
- (h) [Reserved for Future Use]
- (i) Notwithstanding the other provisions of this Section 14.7, Algonquin is not absolved from liability arising as a result of negligence on the part of Algonquin.

14.8 Terms of Release.

(a) Capacity released and available for resale by Algonquin under this Section 14 shall consist of the MDTQ and MATQ associated therewith and, if the Releasing Customer is proposing to release any of its primary points, the MDRO specified in Releasing Customer's Notice for the Point(s) of Receipt specified therein and/or the MDDO specified in such notice for the Point(s) of Delivery specified therein. A Customer's release of a portion of the MDTQ under any service agreement shall operate as a release of a proportionate amount of the MATQ thereunder. The Customer acquiring capacity released under this Section 14 may

FERC Docket: RP10-308-000

Second Revised Sheet No. 547 Second Revised Sheet No. 547

Superseding: First Revised Sheet No. 547

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

use other Point (s) of Receipt and Point(s) of Delivery only as allowed under the applicable rate schedule and Service Agreement. Unless the Release is permanent and with no rights of recall, the New Service Agreement may not be amended to change Point(s) of Receipt or Point(s) of Delivery without the written consent of Releasing Customer.

- (b) If capacity is released during any Contract Year, the Customer acquiring such capacity shall have its MATQ reduced for such Contract Year by the quantity received during such Contract Year prior to the release and associated with the MDTQ so acquired. If a capacity release terminates during any Contract Year, the Releasing Customer shall have its MATQ reduced for such Contract Year by the quantity received during such Contract Year during the period of the release under the acquiring Customer's agreement.
- (c) Any Releasing Customer retaining any rights to recall capacity during the term of the New Service Agreement may exercise such right by providing notification to Algonquin in accordance with the notice requirements set forth in Section 14.4(h) herein. Algonquin may rely on such notification by the Releasing Customer, and Releasing Customer shall indemnify Algonquin and hold it harmless for any liability or costs Algonquin incurs as a result of such reliance, and the Replacement Customers affected by such recall shall release Algonquin and hold it harmless for such reliance.
- (d) Capacity resold under this Section 14, including capacity resold under the bidding exception of Section 14.10(a), shall be subject to the maximum rates, including any applicable surcharges, set forth in Algonquin's Statement of Rates for the

FERC Docket: RP10-308-000

Third Revised Sheet No. 548 Third Revised Sheet No. 548 Superseding: Substitute Second Revised Sheet No. 548

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

applicable rate schedule; provided, however, for releases that become effective on or after July 30, 2008, any proposed capacity release transaction with a duration greater than thirty-one (31) days and less than or equal to one (1) year, unless such proposed release is to an asset manager or a marketer participating in a state-mandated retail access program, must be posted for bidding regardless of the proposed rate; and provided further that, for releases that become effective on or after July 30, 2008, the rate for any release transaction with a term of one (1) year or less may exceed the maximum tariff rate for the applicable rate schedule, if the effective date of the release is on or before one (1) year from the date on which Algonquin is notified of the release. Capacity released under Rate Schedule AFT-1S or AFT-ES and acquired under Rate Schedule AFT-1 or AFT-E for a term of more than one (1) year shall be subject to the maximum and minimum reservation and usage charges set forth in Algonquin's Statement of Rates applicable to such Rate Schedule AFT-1 or AFT-E with respect to capacity released under Rate Schedule AFT-1S or AFT-ES. If Releasing Customer specifies that the capacity may be released on a volumetric-rate basis, the volumetric rate for the release shall apply only to the reservation portion of the rate and shall be subject to the maximum volumetric rates set forth in Algonquin's Statement of Rates for the applicable rate schedule. Capacity released subject to recall is firm service subject to the same maximum rates as capacity released not subject to recall.

For releases that become effective on or after July 30, 2008, the rate for any capacity release transaction with a term of one (1) year or less may exceed the maximum tariff rate for the applicable rate schedule if the effective date of the release is on or before one (1) year from the date on which Algonquin is notified of the release.

- (e) Capacity released hereunder shall be acquired by the Replacement Customer under the same rate schedule as that applying to the Existing Service Agreement, with the following exceptions:
 - (1) To the extent that capacity released under Rate Schedules AFT-E and AFT-ES has delivery points that are different from the Points of Delivery specified in the Releasing Customer's Service Agreement under such rate schedules as of the time of conversion to no-notice service pursuant to Order No. 636, the capacity shall be acquired under Rate Schedules AFT-1 and AFT-1S, respectively (subject to subsection (2) of this subsection (e)); and
 - (2) Capacity released under Rate Schedules AFT-1S and AFT-ES shall be acquired under Rate Schedules AFT-1 and AFT-E (subject to subsection (1) of this subsection (e)), respectively, if the acquiring Customer does not meet the requirements of Sections 1.a and 1.b of Rate Schedules AFT-1S or AFT-ES, respectively.

FERC Docket: RP09-72-000

Original Sheet No. 548A Original Sheet No. 548A

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

- 14.9 Capacity Request Notice. Any party desiring to acquire firm capacity under this Section 14 must submit the following information to Algonquin via e-mail to link-help@spectraenergy.com.
 - (a) the Replacement Customer's name and contact information;
 - (b) the Maximum Daily Transportation Quantity desired;
 - (c) the desired commencement date and term of the service;
 - (d) the desired primary receipt and delivery points and the associated Maximum Daily Receipt Obligation and Maximum Daily Delivery Obligation for the service;
 - (e) the maximum rate(s) that Replacement Customer will pay for the service;
 - (f) whether Replacement Customer will accept a release with recall rights, and if so, what recall rights would be acceptable; and
 - (g) whether Replacement Customer's request is contingent, and if so, the basis for the contingency.

Algonquin shall post such information on the LINK® System for a period of one month or until a transaction is effected, whichever is shorter.

FERC Docket: RP10-77-000

Third Revised Sheet No. 549 Third Revised Sheet No. 549 Superseding: Substitute Second Revised Sheet No. 549

> GENERAL TERMS AND CONDITIONS (continued)

- 14. CAPACITY RELEASE (continued)
- 14.10 Exceptions to Bidding Requirement for Pre-arranged Releases.
 - For releases that become effective on or after July 30, 2008, the following need not comply with the bidding requirements of this Section 14, but shall be subject to all other provisions of this Section 14:

 - (1) a release for any period of thirty-one (31) days or less,(2) a release for more than one (1) year at the maximum tariff rate,
 - (3) a release of capacity to an asset manager (as defined in Section 284.8(h)(3) of the Commission's regulations), and
 - (4) a release of capacity to a marketer participating in a state-regulated retail access program (as defined in Section 284.8(h)(4) of the Commission's regulations).

For releases that become effective on or after July 30, 2008, any release, with the exception of releases to an asset manager or to a marketer participating in a state-regulated retail access program, with a term that is greater than thirty-one (31) days and less than or equal to one (1) year must be posted for bidding pursuant to Section 14.4(b) above, regardless of the proposed rate. A release under this Section 14.10 may be effected by the Releasing Customer's satisfying the notice requirements of Section 14.4 and the selected Approved Bidder's submitting a bid satisfying the requirements of Section 14.4. Such release shall become effective immediately upon Algonquin's receipt of both such notice and such bid, but no sooner than the date specified in Releasing Customer's Notice, subject to compliance with the other provisions of this tariff. The information provided by the Releasing Customer pursuant to Section 14.4(a) will be posted on the LINK® System and by electronic data interchange as required by Section 14.4(d).

The standard timelines applicable to pre-arranged capacity releases that are not subject to the competitive bidding requirements are as follows:

Timely Cycle:

Posting of prearranged deals not subject to bid are due by 10:30 a.m. CCT.

Evening Cycle:

Posting of prearranged deals not subject to bid are due by 5:00 p.m. CCT.

Intraday 1 Cycle:

Posting of prearranged deals not subject to bid are due by 9:00 a.m. CCT.

Intraday 2 Cycle:

Posting of prearranged deals not subject to bid are due by 4:00 p.m. CCT.

Prior to the nomination deadline for the chosen cycle for the begin date specified in the Releasing Customer's Notice, the Prearranged Customer must initiate confirmation of prearranged deals electronically. Algonquin will issue a contract within one Hour of the award posting (with a new contract number, when applicable).

Nominations for gas flow are possible beginning at the next available nomination cycle for the effective date of the contract. Such nominations will be processed in accordance with the nomination and scheduling requirements of Sections 22 and 23 of the General Terms and Conditions; however, in no circumstances will gas flow prior to the effective date of the release as specified in Releasing Customer's Notice.

FERC Docket: RP09-72-003

Second Sub Original Sheet No. 549A Second Sub Original Sheet No. 549A

Superseding: Substitute Original Sheet No. 549A

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

- (c) When a release of capacity for a period of thirty-one (31) days or less is not subject to the bidding requirements under this Section 14.10, a Releasing Customer may not rollover, extend, or in any way continue the capacity release to the same Replacement Customer which utilizes the same capacity or overlaps such capacity using the thirty-one (31) days or less bidding exemption described in Section 14.10(a)(1) above until twenty-eight (28) days after the first release period has ended. The twenty-eight (28) day hiatus does not apply to any re-release to the same Replacement Customer that is posted for bidding or that qualifies for any of the other exemptions from bidding described in Section 14.10(a) above.
- (d) Notwithstanding the standard timelines specified in Section 14.10(b) above, Algonquin shall support a process to allow the Releasing Customer and the Pre-arranged Customer to create and finalize pre-arranged non-biddable capacity release transactions to be effective for a given Gas Day at any time prior to 7:00 a.m. CCT on the calendar day on which that Gas Day ends.

FERC Docket: RP10-77-000

Fifth Revised Sheet No. 550 Fifth Revised Sheet No. 550

Superseding: Fourth Revised Sheet No. 550

GENERAL TERMS AND CONDITIONS (continued)

- 14. CAPACITY RELEASE (continued)
 - (e) Algonquin shall post notice of transactions under Section 14.10 on the LINK® System and by electronic data interchange in accordance with the timelines specified in Section 14.10(a) for notice purposes only, and such posting shall trigger no right in any other party to bid.
- 14.11 Compensation. If Algonquin and a Releasing Customer so agree, Algonquin may receive a mutually agreeable fee for taking action to market released capacity other than merely posting the notice of availability. Such fee may be deducted from the credit due Releasing Customer under Section 14.7.
- 14.12 Algonquin's Rights to Terminate Temporary Capacity Releases. In the event of a temporary release for which (1) Algonquin has given notice of termination of the Releasing Customer's contract because the Releasing Customer no longer satisfies Algonquin's credit requirements as outlined in Sections 3.1 and 3.2 of Algonquin's General Terms and Conditions and (2) the reservation charge specified in the effective Addendum to the Replacement or Prearranged Customer's Capacity Release Umbrella Agreement is less than the level of the reservation charge which the Releasing Customer was obligated to pay Algonquin (or, if Releasing Customer is paying a Negotiated Rate, the sum of all reservation-type and commodity-type charges), then Algonquin shall be entitled to terminate the Addendum, upon 30 Days' written notice to the Replacement or Prearranged Customer, unless the Replacement or Prearranged Customer agrees prior to the end of said 30-Day notice period to pay for the remainder of the term of the Addendum one of the following: (i) the reservation and commodity charges at levels which the Releasing Customer was obligated to pay Algonquin, (ii) the applicable maximum tariff rate, or (iii) such rate as mutually agreed to by Algonquin and Replacement or Prearranged Customer. The Replacement or Prearranged Customer may elect to pay the lesser of the foregoing three options. the subject release was a segmented release, Algonquin shall not be required to permit Replacement or Prearranged Customer to retain its geographic segment of capacity, and may require Replacement or Prearranged Customer to pay for the full capacity path of the defaulting Releasing Customer at the lower of the rate the defaulting Releasing Customer paid or the applicable maximum tariff rate. Algonquin's right to terminate the Addendum is subject to Algonquin providing written notice of termination to the Replacement or Prearranged Customer within 60 Days of the determination by Algonquin that the Releasing Customer no longer satisfies Algonquin's credit requirements. Termination of the Addendum shall not occur prior to termination of the Releasing Customer's contract.

FERC Docket: RP07- 13-000

First Revised Sheet No. 550A First Revised Sheet No. 550A: Effective

Superseding: Original Sheet No. 550A

GENERAL TERMS AND CONDITIONS (continued)

14. CAPACITY RELEASE (continued)

14.13 Notices to Releasing Customers.

Algonquin shall provide the original Releasing Customer with Internet E-mail notification reasonably proximate in time with any of the following formal notices given by Algonquin to the Releasing Customer's Replacement Customer(s), of the following:

- (1) Notice to the Replacement Customer regarding the Replacement Customer's past due, deficiency, or default status pursuant to Algonquin's tariff;
- (2) Notice to the Replacement Customer regarding the Replacement Customer's suspension of service notice;
- (3) Notice to the Replacement Customer regarding the Replacement Customer's contract termination notice due to default or credit-related issues; and
- (4) Notice to the Replacement Customer that the Replacement Customer(s) is no longer creditworthy and has not provided credit alternative(s) pursuant to Algonquin's tariff.
- 14.14 Permanent Capacity Releases.

To the extent that any Customer desires to release all or any part of its firm rights under an Open-access Rate Schedule on a permanent basis, the procedures specified in this Section 14 shall apply. In addition, the Replacement Customer that will acquire the capacity from Customer must submit a request for service electronically via the LINK® System and provide the credit information as required by Section 3 of the General Terms and Conditions. For any permanent capacity release, the minimum bid acceptable to Algonquin shall be a bid for the remainder of the term of Customer's service agreement at the rate(s) Customer is obligated to pay Algonquin for the capacity to be permanently released. Algonquin may refuse to allow a permanent capacity release if it has a reasonable basis to conclude that it will not be financially indifferent to the release. If Customer's request to permanently release capacity is denied by Algonquin, Algonquin shall notify Customer via e-mail and shall include in the notification the reasons for such denial.

FERC Docket: RP10-77-000

Third Revised Sheet No. 551 Third Revised Sheet No. 551

Superseding: Second Revised Sheet No. 551

GENERAL TERMS AND CONDITIONS (continued)

15. TERMINATION FOR DEFAULT

- 15.1 Termination of Service. If either Algonquin or Customer shall fail to perform any of the covenants or obligations imposed upon it or them under and by virtue of an applicable rate schedule or effective Service Agreement, then the other party may at its option terminate such agreement by proceeding as follows. The party not in default shall provide written notice to the party in default stating specifically the cause for terminating the agreement and declaring it to be the intention of the party giving the notice to terminate the same; thereupon the party in default shall have thirty Days after receipt of such notice to remedy or remove the cause or causes stated in the notice, and if within the thirty-Day period the party in default does so remove and remedy said cause or causes and fully indemnifies the party not in default for any and all consequences of such breach, by a good and sufficient indemnity bond or otherwise, then such notice shall be withdrawn and the agreement shall continue in full force and effect. In case the party in default does not so remedy and remove the cause or causes or does not so indemnify the party giving the notice for any and all consequences of such breach, within the thirty-Day period, then, after any necessary authorization by regulatory bodies having jurisdiction, the agreement shall become void at the expiration of said period, provided that notice of termination has not been withdrawn prior thereto. Any cancellation of such agreement pursuant to the provisions of this paragraph shall be without prejudice to the right of Algonquin to collect any amounts then due to it for natural gas delivered prior to the time of cancellation, and shall be without prejudice to the right of Customer to receive any service for which it has paid but has not received, although entitled thereto, prior to the time of cancellation, and without waiver of any remedy to which the party not in default may be entitled for violations of such agreement.
 - 15.2Suspension of Service. Prior to a termination for default, Algonquin may suspend service to any Customer who fails to comply with the General Terms and Conditions, the applicable rate schedule, or Customer's service agreement. Algonquin's failure to invoke its right to terminate an agreement for default shall not be construed as a waiver of Algonquin's right to terminate service during any other period when Customer is in violation of the terms of this tariff or of Customer's service agreement.

FERC Docket: RP05-417-000

First Revised Sheet No. 552 First Revised Sheet No. 552: Effective

Superseding: Original Sheet No. 552

GENERAL TERMS AND CONDITIONS (continued)

16. FORCE MAJEURE

- 16.1Relief from Liability. Neither party shall be taken to have breached its obligations under the service agreement, the applicable rate schedule, or these General Terms and Conditions by reason of any act, omission or circumstance occasioned by or in consequence of any acts of God, strikes, lockouts, acts of the public enemy, wars, blockades, insurrections, riots, epidemics, landslides, lightning, earthquakes, fires, storms, floods, washouts, arrests and restraints of rulers and peoples, civil disturbances, explosions, damage, breakage or accident to machinery or lines of pipe, failure of third-party transportation, the binding order of any court or governmental authority which has been resisted in good faith by all reasonable legal means, and any other cause, whether of the kind herein enumerated or otherwise, not within the control of the party claiming suspension under this section, which act, omission or circumstance such party is unable by the exercise of due diligence to prevent or overcome.
- 16.2 Liabilities Not Relieved. Such causes or contingencies affecting performance by either party, however, shall not relieve such party of liability in the event of its concurring negligence or in the event of its failure to use due diligence to remedy the situation and remove the cause in an adequate manner and with all reasonable dispatch, nor shall such causes or contingencies affecting performance relieve either party from its obligations to make payments of amounts then due in respect of gas theretofore delivered.
- 16.3 Proration of Impaired Deliveries. The provisions of Section 24 of these General Terms and Conditions shall govern the apportionment of available capacity when such available capacity is impaired so that Algonquin is unable to deliver to Customer the daily or annual quantities of gas provided by effective service agreements.
- 16.4 Scheduling of Construction and Maintenance. Algonquin shall have the right to curtail, interrupt, or discontinue service in whole or in part on all or a portion of its system from time to time to perform repair, maintenance or improvements on Algonquin's system as necessary to maintain the operational capability of the system, or to comply with applicable regulatory requirements, or to perform construction pursuant to valid FERC authorization. Algonquin shall exercise due diligence to schedule repair, construction and maintenance so as to minimize disruptions of service to Customer and shall provide reasonable notice of the same to Customer.
- 16.5 Compliance with Directives of Governmental Agencies. Whenever in order to comply with orders, directives or regulations of duly constituted state, local or federal authorities, including, but not limited to, the Department of Transportation, the Federal Energy Regulatory Commission, and the Environmental Protection Agency, Algonquin must curtail deliveries to Customer and is unable to deliver to Customer the quantities of gas which Customer may then

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 553 Fourth Revised Sheet No. 553

Superseding: Third Revised Sheet No. 553

GENERAL TERMS AND CONDITIONS (continued)

16. FORCE MAJEURE (continued)

require up to the quantities of gas Algonquin is then obligated to deliver to Customer, Algonquin shall not be liable in damages or otherwise to Customer or any other person or entity for any such failure to deliver such quantities of gas to Customer except to the extent the orders or directives were issued as a result of imprudence or failure to exercise due diligence on the part of Algonquin.

- 16.6 Effect of Interconnecting Operations. In the event that any upstream entity involved in handling Customer's gas refuses or is unable to deliver gas to Algonquin, Algonquin shall not be required to continue deliveries of gas on behalf of Customer to the extent of such refusal or inability to deliver gas to Algonquin. In the event that any downstream entity involved in handling Customer's gas refuses or is unable to receive gas from Algonquin, Algonquin shall have the right to reduce deliveries of gas on behalf of Customer to the extent of such refusal or inability by the downstream entity to receive gas.
- 17. GOVERNMENTAL REGULATION

The service agreement, and all terms and provisions contained or incorporated therein, and the respective obligations of the parties thereunder, are subject to valid laws, orders, rules and regulations of duly constituted authorities having jurisdiction.

- 18. BILLING
- 18.1 Monthly Bill. Algonquin shall bill Customer on or before the seventh Day of each Month for charges relating to service rendered during the preceding Month. Such bill shall be delivered to Customer or its agent by posting Customer's final bill on Algonquin's LINK® System and posting a general notice of the availability of the final bills on Algonquin's Informational Postings Web Site. Algonquin will provide an e-mail notification, if an e-mail address has been designated by Customer, contemporaneously with the posting of the final bill on Algonquin's LINK® System. It is the Customer's responsibility to update e-mail address information provided to Algonquin as necessary. Customer may designate an agent to receive invoices and may designate such agent to receive the e-mail notifications of the availability of Customer's final bill on Algonquin's LINK® System.
- 18.2 Access to Billing Data. Both Algonquin and Customer shall have the right to examine at reasonable times the books, records, and charts of the other to the extent necessary to verify the accuracy of any statement, charge, or computation made under or pursuant to any of the provisions hereof.
- 18.3 Discounts. The rates will be discounted in the following order: (i) Gas Research Institute Surcharge, (ii) base rate reservation charge.

FERC Docket: RP10-77-000

First Revised Sheet No. 554 First Revised Sheet No. 554

Superseding: Original Sheet No. 554

GENERAL TERMS AND CONDITIONS (continued)

19. PAYMENT

19.1 Payment. On or before the nineteenth Day of each Month, Customer shall pay Algonquin, by wire transfer of federal funds to an address designated by Algonquin, the amounts billed by Algonquin in said Month pursuant to Section 18 of these General Terms and Conditions; provided, however, that when the nineteenth Day of the Month falls on a day that is not a Business Day, such payment is due on the first Business Day following the nineteenth Day of the Month. Any payment shall be considered to have been made as of the time that such payment is received by Algonquin. Payment will be applied in accordance with documentation provided by Customer. All payments shall be identified by invoice number(s) and if payment differs from invoiced amount, remittance detail shall be provided with the payment except when payment is made by electronic funds transfer (EFT), in which case, the remittance detail is due within two Business Days of the payment due date.

19.2 Late Payment.

- (a) Should Customer fail to make timely payment of any bill in whole or in part, payment of any unpaid amount, when made, shall be deemed a Late Payment and Algonquin shall charge interest from the due date of payment until the actual date of payment at an annual interest rate equal to the current national prime interest rate plus 1-1/2 percent prorated for the number of days the payment is late.
 - Algonquin shall bill Customer for any interest due under this section in its next regular billing, and Customer shall pay such amount in accordance with Section 19.1 above.
- (b) Notwithstanding the foregoing, where Customer fails to make timely payments three times during any twelve-month period, the applicable interest rate for subsequent Late Payments during such twelve-month period shall be twice the current national prime interest rate plus 1-1/2 percent, but in no event in excess of the maximum lawful rate.

FERC Docket: RP10-77-000

First Revised Sheet No. 555 First Revised Sheet No. 555

Superseding: Original Sheet No. 555

GENERAL TERMS AND CONDITIONS (continued)

19. PAYMENT (continued)

- (c) Algonquin may waive the interest on a Late Payment made within five days of the due date where Customer has diligently paid its prior bills on time, and lateness in payment is of a nonrecurring nature.
- (d) Subject to Section 19.3 below, should Customer fail to pay its bill for thirty days after payment is due and such failure continues beyond such thirty days, Algonquin, in addition to any other remedy it may have hereunder, may, after any necessary authorization by regulatory bodies having jurisdiction, suspend further delivery of gas to Customer until such amount is paid.
- 19.3 Disputed Bills. If Customer in good faith shall dispute the amount of any bill or parts thereof and shall pay to Algonquin such amounts as it concedes to be correct, and at any time thereafter within thirty days of a demand made by Algonquin shall furnish a good and sufficient surety bond guaranteeing payment of the amount ultimately found due upon such bills after a final determination which may be reached either by agreement or judgment of the courts as may be the case, then Algonquin shall not be entitled to suspend further delivery of gas unless and until default be made in the conditions of such bond. Customer shall pay the portion of the bill not in dispute and shall provide documentation identifying the basis for the dispute. No payment by Customer of the amount of a disputed bill shall prejudice the right of Customer to claim an adjustment of the disputed bill. Customer shall pay interest on that disputed portion of a bill for which Customer has withheld payment and which ultimately is found due, in accordance with the provisions of Section 19.2 (a) above.
- 19.4 Extension of Time. If presentation of bills by Algonquin is delayed after the seventh Day of the Month, then the time of payment shall be extended accordingly unless Customer is responsible for such delay.

FERC Docket: RP10-77-000

First Revised Sheet No. 556 First Revised Sheet No. 556

Superseding: Original Sheet No. 556

GENERAL TERMS AND CONDITIONS (continued)

- 19. PAYMENT (continued)
- 19.5 Billing Errors. In the event an error is discovered in the amount hereunder, such error shall be adjusted within thirty days of the determination thereof, provided that claim therefor shall have been made in writing.
 - (a) Measurement data corrections shall be processed within six (6) months of the production Month with a three-month rebuttal period.
 - (b) The time limitation for disputes of allocations shall be six (6) months from the date of the initial Month-end allocation with a three-month rebuttal period.
 - (c) Prior period adjustment time limits shall be six (6) months from the date of the initial transportation invoice with a three-month rebuttal period, excluding government-required rate changes.

These three standards shall not apply in the case of deliberate omission or misrepresentation or mutual mistake of fact. Parties' other statutory or contractual rights shall not otherwise be diminished by this standard.

FERC Docket: RP04-361-000

Original Sheet No. 557 Original Sheet No. 557: Effective

GENERAL TERMS AND CONDITIONS (continued)

20.POSSESSION OF GAS

- 20.1 Control. Subject to Section 20.2, as between Algonquin and Customer, Algonquin shall be deemed to be in control and possession of the gas hereunder upon receipt of such gas at Customer's Point(s) of Receipt until such gas shall have been delivered for Customer's account at Customer's Point(s) of Delivery, and Customer shall be deemed to be in control and possession of such gas prior to such receipt by Algonquin and after such delivery for Customer's account.
- 20.2 Responsibility. Algonquin shall have no responsibility with respect to the gas or on account of anything done, occurring or arising with respect to the gas before receipt of such gas by Algonquin or after delivery of the gas to Customer or for Customer's account, and Customer shall have no responsibility with respect to the gas or on account of anything done, occurring, or arising with respect to the gas while the gas is in Algonquin's possession; provided, however, in the event any gas flows through facilities of Customer prior to such delivery to Customer or for Customer's account, Algonquin shall have no responsibility with respect to such gas or on account of anything done, occurring or arising with respect to such gas while it is in the facilities of Customer.

21. WARRANTY OF TITLE TO GAS

Customer warrants that it will at the time of delivery of the gas to Algonquin have good and merchantable title to all of the gas subject only to security interests, liens and encumbrances that would not prevent Customer from tendering the gas for transportation hereunder or receiving redelivery of the gas from Algonquin. Title to the gas received by Algonquin for Customer's account hereunder shall remain with Customer during transportation of the gas by Algonquin. Customer agrees to indemnify and save Algonquin harmless from all suits, actions, debts, accounts, damages, costs, losses and expenses arising out of adverse claims of any and all persons to the gas other than claims arising solely through acts of Algonquin.

FERC Docket: RP10-77-000

Second Revised Sheet No. 558 Second Revised Sheet No. 558

Superseding: First Revised Sheet No. 558

GENERAL TERMS AND CONDITIONS (continued)

22. NOMINATIONS

- General. If Customer desires the delivery of gas on any Day under Algonquin's rate schedules, Customer shall give notice to Algonquin nominating the specific quantity of gas Customer requests under each of Customer's Service Agreements under each of Algonquin's rate schedules for each Day of the Nomination Period. Such nomination shall specify the begin and end dates upon which such deliveries are desired, provided the nomination begin and end dates are within the term of Customer's contract, the desired quantities at each Point of Receipt and each Point of Delivery under Customer's Service Agreements, any hourly change in Customer's rate of delivery, if deliveries are not projected to be made in equal hourly increments, and, for interruptible service for which Customer is paying less than the maximum rate, whether Customer elects to pay the applicable maximum rate in accordance with Sections 23.1(e)(2) and 24.5 of the General Terms and Conditions. To the extent Customer desires to change its nomination for any Day(s) within the Nomination Period, Customer must submit a new nomination for such Day(s). In no event shall Algonquin be required to accept a nomination for a quantity insufficient to be measured by the existing measurement facilities at the Point(s) of Receipt or the Point(s) of Delivery. When a nomination for a date range is received, each Day within that range is considered an original nomination. When a subsequent nomination is received for one or more Days within that range, the previous nomination is superseded by the subsequent nomination only to the extent of the Days specified. The Days of the previous nomination outside the range of the subsequent nomination are unaffected. Nominations have a prospective effect only.
- 22.2 LINK® or Electronic Data Interchange Nominations. Customer shall submit its nominations via the LINK® System or electronic data interchange, unless another means of submittal has been mutually agreed upon by Customer and Algonquin. A nomination made through electronic data interchange shall include all mandatory Commission-approved data elements.

FERC Docket: RP10-77-000

First Revised Sheet No. 559 First Revised Sheet No. 559

Superseding: Original Sheet No. 559

GENERAL TERMS AND CONDITIONS (continued)

22. NOMINATIONS (continued)

22.3Initial Nominations. The nomination of deliveries under all rate schedules to commence on the first Day of any Nomination Period shall be submitted to Algonquin no later than the time specified in the nomination timeline below, or such lesser period as is acceptable to Algonquin, prior to the first Day of the Nomination Period. Such nomination shall specify the quantity to be scheduled for each Day of the Nomination

The nomination timeline shall be Central Clock Time on the day prior to gas flow:

11:30 a.m. Nomination leaves control of the nominating party
11:45 a.m. Receipt of nomination by Algonquin (including from Title Transfer Tracking Service Providers (TTTSPs))
12:00 p.m. Quick response by Algonquin
3:30 p.m. Receipt of completed confirmations by Algonquin from upstream and downstream connected parties

4:30 p.m. Receipt of scheduled quantities by Customer and point operator

22.4Subsequent Nominations.

a) Prior notice nominations. Subsequent to the first Day of the Nomination Period, Customer may alter its nominations provided that the nomination given by Customer to Algonquin for deliveries on any Day shall be submitted to Algonquin no later than the time specified in the nomination timeline below, or such lesser period as is acceptable to Algonquin, prior to the commencement of such Day. Any scheduling nomination submitted after the initial scheduling nomination shall contain Customer's anticipated service requirements for the remainder of the Nomination Period.

The nomination timeline shall be Central Clock Time on the Day prior to gas flow:

11:30 a.m. Nomination leaves control of the nominating party
11:45 a.m. Receipt of nomination by Algonquin (including from Title Transfer Tracking Service Providers (TTTSPs))
12:00 p.m. Quick response by Algonquin
3:30 p.m. Receipt of completed confirmations by Algonquin from upstream and downstream connected parties
4:30 p.m. Receipt of scheduled quantities by Customer and point operator

FERC Docket: RP10-77-000

Second Revised Sheet No. 560 Second Revised Sheet No. 560

Superseding: First Revised Sheet No. 560

GENERAL TERMS AND CONDITIONS (continued)

Intra-day nominations under firm rate schedules. Algonquin shall allow Customer (b) to revise Customer's nomination under any firm rate schedule any time prior to the end of the Gas Day being scheduled provided such revision may be implemented, in Algonquin's reasonable judgment, by Algonquin without detriment to any other firm service Customer and provided such change can be confirmed in a timely manner with Customer's upstream operator and downstream operator and other operators of connecting facilities and supplies. There are two types of intraday nominations: (i) a nomination received during the Gas Day for the same Day of gas flow, and (ii) a nomination received after the nominations deadline for the following Gas Day. At least one (1) intra-day nomination can be submitted 4 hours prior to gas flow. Within four (4) hours of an intra-day nomination, Algonquin will notify affected Customer(s) and point operator(s) of changes in scheduled quantities. Such a change in nominated and scheduled quantities shall be processed after timely nominations have been scheduled and shall be made prospectively only. Algonquin shall be obligated to deliver the quantities of gas attributable to a revision of Customer's nomination at the same hourly rate as Customer's gas is delivered into Algonquin's system ("Upstream Delivery Hourly Rate") unless delivery at such hourly rate would impair firm service to another Customer. In the event delivery is not made at the Upstream Delivery Hourly Rate, Algonquin shall deliver the quantities at a uniform hourly rate or, in the case of Service Agreements resulting from the conversion pursuant to Order No. 636 of sales entitlements under former Rate Schedules F-1 and WS-1, at a rate equal to .06 multiplied by the scheduled daily quantity, or such greater hourly rate as may be consistent with Algonquin's system operational requirements to serve firm Customers. Any scheduling nomination submitted after the initial scheduling nomination shall contain Customer's anticipated service requirements for one Day only and shall include the effective date and time. Intra-day nominations do not replace the remainder of a standing nomination. There is no need to re-nominate if an intra-day nomination modifies an existing nomination. The interconnected parties shall agree to the hourly flows for the intra-day nomination.

If Customer submits an intra-day nomination on the same contract which already has been scheduled to flow through a Posted Point of Restriction, such intra-day nomination may be scheduled by Algonquin provided that the intra-day nomination has a scheduling sequence priority that is being scheduled and allowed to flow on the same Gas Day as the intra-day nomination pursuant to Sections 23.1 and 48.3 of the General Terms and Conditions and provided that the intra-day nomination does not result in a net increase in the total volume scheduled on the contract through the Posted Point of Restriction on that Gas Day.

(c) Intra-day nominations under interruptible rate schedules. Algonquin may allow, Customer to revise its nominations on a prospective basis under any interruptible rate schedule any time prior to the end of the Gas Day being scheduled, provided that, in Algonquin's reasonable judgment, Algonquin may implement such revision under existing system operating conditions and such revision shall cause no detriment to any other firm or interruptible service. There are two types of intra-day nominations: (i) a nomination received during the Gas Day for the same Day of

FERC Docket: RP10-77-000

Second Revised Sheet No. 561 Second Revised Sheet No. 561

Superseding: First Revised Sheet No. 561

GENERAL TERMS AND CONDITIONS (continued)

gas flow, and (ii) a nomination received after the nominations deadline for the following Gas Day. At least one (1) intra-day nomination can be submitted 4 hours prior to gas flow. Within four (4) hours of an intra-day nomination, Algonquin will notify affected Customer(s) and point operator(s) of changes in scheduled quantities. Any scheduling nomination submitted after the initial scheduling nomination shall contain Customer's anticipated service requirements for one Day only and shall include the effective date and time. Intra-day nominations do not replace the remainder of a standing nomination. There is no need to re-nominate if an intra-day nomination modifies an existing nomination. The interconnected parties shall agree to the hourly flows for the intra-day nomination.

If Customer submits an intra-day nomination on the same contract which already has been scheduled to flow through a Posted Point of Restriction, such intra-day nomination may be scheduled by Algonquin provided that the intra-day nomination has a scheduling sequence priority that is being scheduled and allowed to flow on the same Gas Day as the intra-day nomination pursuant to Sections 23.1 and 48.3 of the General Terms and Conditions and provided that the intra-day nomination does not result in a net increase in the total volume scheduled on the contract through the Posted Point of Restriction on that Gas Day.

- (d) Elapsed Prorated Scheduled Quantity. With respect to intraday nominations for reductions in previously scheduled quantities, Algonquin will accept, subject to the limitations set forth in Sections 22.4(b) and 22.4(c), any explicitly confirmed quantity, down to and including zero, for such intraday nomination; provided, however, if such intraday nomination requires confirmation from an upstream and/or downstream interconnected pipeline then any intraday nomination to reduce previously scheduled quantities will be subject to, and limited to, the reduced quantity confirmed by such upstream and/or downstream interconnected pipeline.
- 22.5 Minimum NAESB Nomination Standards. In the event the more flexible nomination procedures set forth in Section 22 above are inapplicable for any reason, nominations shall be submitted and processed in accordance with the minimum standards set forth in this Section 22.5. Algonquin shall support the following standard nomination cycles:
 - (a) The Timely Nomination Cycle: 11:30 am for nominations leaving control of the nominating party; 11:45 am for receipt of nominations by Algonquin (including from Title Transfer Tracking Service Providers (TTTSPs)); noon to send Quick Response; 3:30 pm for receipt of completed confirmations by Algonquin from upstream and downstream connected parties; 4:30 pm for receipt of scheduled quantities by Customer and point operator (central clock time on the Day prior to flow).
 - (b) The Evening Nomination Cycle: 6:00 pm for nominations leaving control of the nominating party; 6:15 pm for receipt of nominations by Algonquin (including from TTTSPs); 6:30 pm to send Quick Response; 9:00 pm for receipt of completed confirmations by Algonquin from upstream and downstream connected parties; 10:00 pm for Algonquin to provide scheduled quantities to affected Customers and point operators, and to provide scheduled quantities to bumped parties (notice to bumped parties), (central clock time on the Day prior to flow).

FERC Docket: RP10-77-000

First Revised Sheet No. 562 First Revised Sheet No. 562

Superseding: Original Sheet No. 562

GENERAL TERMS AND CONDITIONS (continued)

Scheduled quantities resulting from an Evening Nomination that does not cause another Service Requester on Algonquin to receive notice that it is being bumped should be effective at 9:00 am on the Gas Day; and when an Evening Nomination causes another Service Requester on Algonquin to receive notice that it is being bumped, the scheduled quantities should be effective at 9:00 am on Gas Day.

- (c) The Intraday 1 Nomination Cycle: 10:00 am for nominations leaving control of the nominating party; 10:15 am for receipt of nominations by Algonquin (including from TTTSPs); 10:30 am to send Quick Response; 1:00 pm for receipt of completed confirmations by Algonquin from upstream and downstream connected parties; 2:00 pm for Algonquin to provide scheduled quantities to affected Customers and point operators, and to provide scheduled quantities to bumped parties (notice to bumped parties), (central clock time on Gas Day). Scheduled quantities resulting from Intraday 1 Nominations should be effective at 5:00 pm on Gas Day.
- (d) The Intraday 2 Nomination Cycle: 5:00 pm for nominations leaving control of the nominating party; 5:15 pm for receipt of nominations by Algonquin (including from TTTSPs); 5:30 pm to send Quick Response; 8:00 pm for receipt of completed confirmations by Algonquin from upstream and downstream connected parties; 9:00 pm for Algonquin to provide scheduled quantities to affected Customers and point operators (central clock time on the Gas Day). Scheduled quantities resulting from Intraday 2 Nominations should be effective at 9:00 pm on the Gas Day. Bumping is not allowed during the Intraday 2 Nomination Cycle.
- 23. SCHEDULING OF SERVICE
- 23.1 Scheduling Sequence. Based upon Customer's nominations as confirmed with Customer's upstream and downstream transporters or operators, Algonquin shall schedule receipts and deliveries of gas in the following sequence:
 - (a) Firm services utilizing primary points: among Algonquin's firm service Customers according to the quantities of gas requested to be scheduled by such Customers from Primary Points of Receipt to Primary Points of Delivery up to but not in excess of the applicable contractual entitlement of such Customers;
 - (b) Firm service utilizing secondary points: among Algonquin's firm service Customers according to the quantities of gas requested to be scheduled by such Customers from Secondary Points of Receipt or to Secondary Points of Delivery up to but not in excess of the applicable contractual entitlement of such Customers, in accordance with Section 48.3(b) of the General Terms and Conditions;

FERC Docket: RP10-77-000

Third Revised Sheet No. 563 Third Revised Sheet No. 563

Superseding: Second Revised Sheet No. 563

GENERAL TERMS AND CONDITIONS (continued)

23.SCHEDULING OF SERVICE (continued)

- (c) Resolution of firm service imbalances: among Algonquin's firm service Customers scheduling excess receipts or deliveries for the purpose of resolving a prior imbalance in scheduled receipts or deliveries up to but not in excess of the applicable contractual entitlement of such Customers;
- (d) Old interruptible services: among Customers whose contractual entitlements resulted from the conversion of interruptible entitlements under Algonquin's former Rate Schedules T-LG, STB and SS-III.
- (e) New interruptible services: Among all other interruptible service Customers, excluding Park and Loan Customers, in the following order:
 - (1) among all Customers paying the maximum applicable rate, on a prorata basis based upon Customer's nominations received by Algonquin; and
 - (2) among all Customers paying a rate that is less than the maximum rate of Algonquin then in effect, in inverse order of the size of such discounts, and among Customers with the same size discount on a prorated basis based upon Customer's nominations received by Algonquin. In addition, Customer may elect to pay the applicable maximum rate at the time Customer submits its nomination in accordance with Section 22 of these General Terms and Conditions. In the event Customer's nomination is for service at a rate less than the applicable maximum rate and Algonquin determines that all nominations exceed available interruptible capacity, Customer making such election will be required to pay the applicable maximum rate in order to have its nomination accepted and scheduled by Algonquin to the extent interruptible capacity is available as determined by Algonquin. Such election shall be a one-time election effective for the remainder of the Nomination Period.
- (f) Resolution of interruptible service imbalances: among Algonquin's interruptible service Customers scheduling excess receipts or deliveries for the purpose of resolving a prior imbalance in scheduled receipts or deliveries up to but not in excess of the applicable contractual entitlement of such Customers;
- (g) Authorized overrun gas: among Customers requesting quantities of gas to be scheduled in excess of the applicable contractual entitlement of such Customers;
- (h) Park and loan service: among Customers requesting quantities of gas to be scheduled for park and loan service under Rate Schedule PAL based on net present value, with the transaction yielding the highest net present value receiving priority over transactions yielding a lower net present value. Net present value shall be calculated on the basis of the present value of the daily PAL charge per Dth to Algonquin for the term of the transaction as specified in Exhibit A of the PAL Service Agreement. PAL transactions yielding the same net present value shall be scheduled pro rata based on scheduled quantities. In making the determination of net present value, Algonquin shall apply the rate, as of the date of the review, stated in accordance with Section 154.501(d) of the Commission's regulations, to all bids.

FERC Docket: RP10-77-000

First Revised Sheet No. 564 First Revised Sheet No. 564

Superseding: Original Sheet No. 564

GENERAL TERMS AND CONDITIONS (continued)

23. SCHEDULING OF SERVICE (continued)

- 23.2 Notice of Amounts Scheduled. Upon completion of scheduling pursuant to Section 23.1, Algoriquin shall notify each Customer of the quantities scheduled for delivery to such Customer under each of Algoriquin's rate schedules if the amount scheduled is other than the amount nominated.
- 23.3 No Bump Policy. Once all or a portion of the nomination of a Customer who is paying the maximum rate is accepted and scheduled pursuant to Section 23.2 for any Gas Day, said daily scheduled service shall not be interrupted unless: (a) interruptible capacity is required to provide firm service prior to the Intraday 2 Nomination Cycle as described in Section 22; or (b) curtailment is necessary pursuant to the provisions of Section 24 of these General Terms and Conditions. Any Customer bumped pursuant to this Section 23.3 will be notified by telephone or facsimile and confirmed through the LINK® System and the Web Site.
- 23.4 Scheduling Penalty.

The following scheduling penalties shall apply, provided that if the unauthorized contract overrun penalty set forth in Section 31 of these General Terms and Conditions would also be applicable to a given quantity of gas, only the greater of either the scheduling penalty or the unauthorized contract overrun penalty shall apply.

(a) Daily. To the extent that deliveries to Customer on any Day vary from the quantity scheduled, for reasons not attributable to curtailment pursuant to Section 24 and system operating conditions require adherence to the Customer's daily flow obligation in order to enable deliveries of scheduled quantities to all firm Customers, Customer shall be subject to a charge equal to the maximum applicable rate under Rate Schedule AIT-1 per Dth by which such deliveries exceed 110 percent of scheduled deliveries.

The daily charge shall not be applicable unless Algonquin has first provided four Hours' notice on its Web Site. The notice shall specify the period during which such penalties shall be in effect, the Points of Delivery to which such penalties will be applicable and that the notice is being posted pursuant to the provisions of Section 23 of the General Terms and Conditions.

FERC Docket: RP10-77-000

First Revised Sheet No. 565 First Revised Sheet No. 565

Superseding: Original Sheet No. 565

GENERAL TERMS AND CONDITIONS (continued)

23. SCHEDULING OF SERVICE (continued)

(b) Hourly. To the extent that deliveries to Customer during any period of one Hour exceed 104 percent of 1/24th of the scheduled daily quantity (or .06 multiplied by the scheduled daily quantity in the case of Service Agreements resulting from the conversion pursuant to Order No. 636 of entitlements under former Rate Schedules F-1 and WS-1) without the consent of Algonquin, Customer shall be subject to a charge equal to the maximum applicable rate under Rate Schedule AIT-1 per Dth by which the deliveries to Customer exceed 104 percent of 1/24th of the scheduled daily quantity (or .06 multiplied by the scheduled daily quantity in the case of Service Agreements resulting from the conversion pursuant to Order No. 636 of entitlements under former Rate Schedules F-1 and WS-1).

Such charge shall be in addition to any daily scheduling charge assessed, provided that such hourly charge shall not be applicable unless Algonquin has first provided one Hour's notice on its Internet Web Site that system operating conditions require adherence to the Customer's hourly flow obligation in order to enable deliveries of scheduled quantities to all firm Customers. Any such notice shall specify the period during which such penalties shall be in effect, and the Points of Delivery to which such penalties will be applicable and that the notice is being posted pursuant to the provisions of Section 23 of the General Terms and Conditions.

- (c) Exception for No-Notice Service. The scheduling penalties described herein shall not apply to Customers receiving service under Rate Schedules AFT-E or AFT-ES, provided that all of the operational conditions specified in Section 5 of those rate schedules have been met.
- (d) Exception for Flow Control. The scheduling penalties described herein shall not apply to excess deliveries at Points of Delivery where Algonquin has installed remote electronic flow control equipment that is operational at the time of such excess deliveries.
- Credit of Penalties Received. Any penalty revenue collected by Algonquin pursuant to this Section 23.4 will be credited to those firm and interruptible Customers that did not incur penalties pursuant to this Section 23.4 in the Month for which penalty revenues were received ("Non-offending Customers"), based on the ratio of the Reservation and Usage Charges paid during that Month by each Non-offending Customer to the total amount of such charges paid during that Month by all Non-offending Customers. An operator that has executed an Operational Balancing Agreement and a transportation service agreement with Algonquin shall be eligible to receive penalty revenue credits as a Non-offending Customer for any Month in which the operator was not liable for penalties pursuant to this Section 23.4 under its own transportation service agreement(s). Such credits shall be calculated for each Month of the twelve (12) month period ending July 31 of each year and will be included on the Non-offending Customers' invoice for the Month following the date of the final Commission order approving Algonquin's penalty disbursement report; provided, however, that Algonquin will calculate and include such credits on Non-offending Customers' invoices for a period shorter than twelve months in the event and to the extent that the total accumulated amount of penalty revenue collected pursuant to this Section 23.4 by Algonquin as of the end of any Month exceeds \$1,000,000. Algonquin will file a penalty disbursement report within sixty days of July 31 or sixty days after the end of the Month in which the revenue collected exceeds \$1,000,000. Any penalty revenue credited to Non-offending Customers pursuant to this section shall include interest calculated in accordance with Section 154.501 of the Commission's regulations.

FERC Docket: RP04-361-000

Original Sheet No. 566 Original Sheet No. 566: Effective

GENERAL TERMS AND CONDITIONS (continued)

24. CURTAILMENT

- General. This Section 24 sets out the procedures whereby Algonquin will reduce or discontinue firm or interruptible services available under this tariff. If, due to any cause whatsoever, the capacity of Algonquin's transmission system, or any part thereof, is impaired so that Algonquin is unable to receive or to deliver the total quantity of gas which Algonquin is then obligated to receive or to deliver under agreements with Customer, then Algonquin shall allocate such reduced available capacity in accordance with this Section 24.
- 24.2 Causes of Curtailment. Algonquin shall have the right to curtail, interrupt or discontinue service in whole or in part on all or a portion of its system from time to time and at any time (a) for reasons of force majeure or when necessary, in Algonquin's reasonable judgment, to maintain the operational integrity of its system, (b) to perform routine maintenance, repairs, improvements and regulatory compliance activity, provided that Algonquin shall utilize due diligence to schedule such activity so as to minimize disruptions of service and shall provide reasonable notice of such interruptions to Customer, (c) when the service is interruptible to the extent, in Algonquin's reasonable judgment, such interruptible service would impair Algonquin's ability to meet its firm service obligations, (d) to the extent necessary in Algonquin's reasonable judgment to comply with applicable laws and regulations including, but not limited to, governmental permits governing the operation of Algonquin's facilities, and (e) to the extent necessary in Algonquin's reasonable judgment to protect people and property, including Algonquin's facilities, from harm or damage due to operational or safety conditions including but not limited to conditions such as excessive compressor discharge temperatures.

FERC Docket: RP10-77-000

First Revised Sheet No. 567 First Revised Sheet No. 567

Superseding: Original Sheet No. 567

GENERAL TERMS AND CONDITIONS (continued)

24.CURTAILMENT (continued)

- 24.3 Notice of Curtailment.
 - (a) Notice by Algonquin. Algonquin shall promptly provide Customer with notice of curtailment or interruption at a time and in a manner that is reasonable under then existing conditions. When possible, such notice shall be provided in advance. Algonquin shall confirm a telephonic notice of curtailment by posting the notice on the LINK® System and its Internet Web Site. Such notice shall specify that the notice is being posted pursuant to the provisions of Section 24 of the General Terms and Conditions. In the event of a curtailment of firm service, the notice shall include a concise statement regarding the cause of the curtailment. Algonquin shall provide a more detailed explanation of the cause of the curtailment upon request.
 - (b) Notice by Customer. Customer shall have the responsibility to inform its suppliers, other transporters and all others involved in the transaction as to any curtailment or interruptions.
 - (c) Notice in Emergency Situations. Algonquin shall provide notice to any Customer that is curtailed to a lower quantity as a result of a request submitted under the Emergency Situation relief provisions of subsection 24.6 below. Such notice shall be provided in a time and manner that is reasonable under then existing conditions but no later than within twenty-four hours of such curtailment. In the event Customer requesting Emergency Situation relief provides the sworn statement required by subsection 24.6, Algonquin shall provide a copy of the sworn statement at the request of any Customer curtailed to a lower quantity as a result of the anticipated sworn statement.
- 24.4 Order of Curtailment. In the event service is interrupted or curtailed then Algonquin shall reduce scheduled service beginning with category (1) below and continuing through the categories of service in the following order:
 - (1) Scheduled service for park and loan service under Rate Schedule PAL.
 - (2) Scheduled service for Authorized Overrun Quantities.
 - (3) Scheduled service for the resolution of imbalances under interruptible service agreements.
 - (4) Scheduled service for interruptible services not otherwise classified in paragraphs (1), (2), (3) and (5) herein.
 - (5) Scheduled service for interruptible services resulting from the conversion of contractual entitlements under Algonquin's former Rate Schedules T-LG, STB and SS-III.

FERC Docket: RP10-77-000

Third Revised Sheet No. 568 Third Revised Sheet No. 568

Superseding: Second Revised Sheet No. 568

GENERAL TERMS AND CONDITIONS (continued)

24.CURTAILMENT (continued)

- (6) Scheduled service under all firm service agreements.
- 24.5 Order of Curtailment Within Categories. Algonquin shall reduce service to those Customers affected by insufficient capacity beginning with requests within category (1) above, and continuing through the order listed above in Section 24.4. In the event that sufficient capacity is available to provide partial service in any of the categories defined in Section 24.4 above, then the available capacity shall be allocated within such category as follows:
 - (1) The capacity available for park and loan service under Rate Schedule PAL to the Customers affected within category (1) of Section 24.4 above shall be allocated among such Customers on the basis of net present value, with the transaction yielding the highest net present value receiving priority over transactions yielding a lower net present value. Net present value shall be calculated on the basis of the present value of the daily PAL charge per Dth to Algonquin for the term of the transaction as specified in Exhibit A of the PAL Service Agreement. PAL transactions yielding the same net present value shall be scheduled pro rata based on scheduled quantities. In making the determination of net present value, Algonquin shall apply the rate, as of the date of the review, stated in accordance with Section 154.501(d) of the Commission's regulations, to all bids.
 - (2) The capacity available for service to the Customers affected within category (2) of Section 24.4 above shall be allocated among such Customers as follows: first, pro rata among firm Customers paying the maximum rate; next, on a first-come/first-served basis among interruptible Customers paying the maximum rate; and next, among firm and interruptible Customers paying a discounted rate based upon the size of the discount, with the Customer paying the higher rate having the higher priority, subject only to the election of another interruptible Customer that is paying a discounted rate to pay the applicable maximum rate in order to afford its scheduled quantities a higher priority for the purposes of this subsection (2).
 - (3) The capacity available for service to the Customers affected within categories (3) and (4) of Section 24.4 above shall be allocated among such Customers, who have scheduled interruptible quantities pursuant to Section 23, on a prorata basis based on the quantities scheduled for such Customer(s), subject only to the election of another interruptible Customer that is paying a discounted rate to pay the applicable maximum rate in order to afford its scheduled quantities a higher priority for the purposes of this subsection (3).

FERC Docket: RP10-77-000

Second Revised Sheet No. 569 Second Revised Sheet No. 569

Superseding: First Revised Sheet No. 569

GENERAL TERMS AND CONDITIONS (continued)

24.CURTAILMENT (continued)

- (4) The capacity available for service to the Customers affected within category (5) of Section 24.4 above shall be prorated among the affected Customers based upon the quantity scheduled for each affected Customer under Algonquin's rate schedules subject to category (5) of Section 24.4 as compared with the total of the quantity scheduled for all affected Customers; provided, however, that no Customer shall be entitled to a capacity allocation in excess of the quantity for which service has been scheduled pursuant to Section 23 within the category on the particular Day involved.
- (5) The capacity available for service to Customers affected within category (6) of Section 24.4 above shall be prorated based on the quantity scheduled for each affected Customer at the affected Points of Receipt or Points of Delivery under the affected rate schedules, subject to adjustment pursuant to Section 24.6 below.
- 24.6 Emergency Situation. In the event Algonquin is advised by a Customer that, absent adjustment of the curtailment level contemplated by subsection 24.5, the Customer will be unable for the duration indicated by Customer to avoid irreparable injury to life or property (including environmental emergencies) or to provide for minimum plant protection ("Emergency Situation"), then Algonquin shall adjust its curtailment of all other Customers on a pro rata basis as necessary to deliver the quantities required to avoid or mitigate the threatened or existing Emergency Situation. While Algonquin will make adjustments in curtailment promptly upon notification by Customer, Customer must provide Algonquin within twenty-four hours of notification a sworn statement including:
 - (a) a detailed explanation of and an estimated duration for the Emergency Situation;

FERC Docket: RP10-77-000

Second Revised Sheet No. 570 Second Revised Sheet No. 570

Superseding: First Revised Sheet No. 570

GENERAL TERMS AND CONDITIONS (continued)

24. CURTAILMENT (continued)

- (b) affirmation that all sources of gas supply available to Customer, including peakshaving and storage, were and are being utilized to the maximum extent possible during the time period for which the exception to the curtailment provisions of this Section 24 is in effect;
- (c) affirmation that all interruptible services of Customer were and are being interrupted or curtailed during the time period for which the exception to the curtailment provisions of this Section 24 is in effect;
- (d) affirmation that no alternate fuel could be utilized or is available to be utilized to prevent the necessity for an exception to the curtailment provisions of this Section 24; and
- (e) affirmation that, if the Emergency Situation is an environmental emergency, the Customer has made and continues to make all feasible efforts to resolve the environmental emergency, including requests for waiver, exemption, and other relief from any regulation, directive, order or other requirement of a governmental body.

Customer shall provide Algonquin immediate notice of the cessation of the Emergency Situation

Algonquin shall not be liable for granting exceptions to the curtailment provisions of this Section 24 for any Customer based upon a request submitted by any such Customer to Algonquin under the Emergency Situation relief provisions of this subsection 24.6. In the event Customer does not provide the sworn statement as required by this subsection, then all quantities attributable to the adjustments made by Algonquin shall be billed, in addition to all other charges, at a rate equal to three times the daily Gas Daily posting for the High Common price for the "AGT Citygate" posting per Dth for the Day that Algonquin adjusts its curtailment levels to provide Emergency Situation relief pursuant to this Section 24.6. All revenues attributable to such charge shall be credited, pro rata on the basis of the increase in curtailment caused by this provision's having been invoked, to those Customers curtailed to a lower quantity as a result of the anticipated sworn statement. Notwithstanding any provision of this subsection, however, Customer shall not be entitled to relief under this subsection: (1) to the extent that an Emergency Situation is due to the Customer's failure to have adequate transportation arrangements in effect for the delivery of Customer's gas at the Point(s) of Delivery in effect hereunder during the relevant period, or (2) to the extent that the quantity of gas required to meet such Emergency Situation exceeds such Customer's firm contractual rights.

FERC Docket: RP10-77-000

Second Revised Sheet No. 571 Second Revised Sheet No. 571

Superseding: First Revised Sheet No. 571

GENERAL TERMS AND CONDITIONS (continued)

- 24. CURTAILMENT (continued)
- 24.7 Compensation. If Algonquin is requested to invoke Section 24.6, then the requesting Customer's bill for such Month shall be adjusted upward by an amount equal to the aggregate curtailment adjustment quantity requested by the Customer pursuant to Section 24.6 multiplied by the reservation charge for the applicable rate schedule multiplied by .0329. All revenues attributable to such adjustment shall be credited, on a pro rata basis, based on each Customer's additional curtailment quantity due to the Emergency Situation divided by the aggregate of all Customers' additional curtailment quantity due to the Emergency Situation, to those Customers curtailed to a lower quantity as a result of the Emergency Situation.
- Violation of Curtailment Orders. All quantities taken by Customer in violation of Algonquin's curtailment orders shall constitute unauthorized receipts or deliveries for which a curtailment penalty equal to three times the daily Gas Daily posting for the High Common price for the "AGT Citygate" posting per Dth on the Day the violation occurred shall be assessed. On any Day that Customer may be liable for scheduling penalties under Section 23 for the same quantities subject to a penalty charge under this Section 24.6, Algonquin shall impose on Customer only the greater of the two penalties. Algonquin will provide Customer notice of such curtailment orders and Customer shall be permitted one Hour, or such lesser time as is required to prevent the impairment of reliable service on the system, to reduce its tenders, takes or both in compliance with the curtailment orders. If Customer adjusts its tenders or takes within such notice period then no charge, as provided for herein, shall be assessed.

Any penalty revenue collected by Algonquin pursuant to this Section 24.8 will be credited to those firm and interruptible Customers that did not incur penalties pursuant to this Section 24.8 in the Month for which penalty revenues were received ("Non-offending Customers"), based on the ratio of the Reservation and Usage Charges paid during that Month by each Non-offending Customer to the total amount of such charges paid during that Month by all Non-offending Customers. An operator that has executed an Operational Balancing Agreement and a transportation service agreement with Algonquin shall be eligible to receive penalty revenue credits as a Non-offending Customer for any Month in which the operator was not liable for penalties pursuant to this Section 24.8 under its own transportation service agreement(s). Such credits shall be calculated for each Month of the twelve (12) month period ending July 31 of each year and will be included on the Non-offending Customers' invoice for the Month following the date of the Commission order approving Algonquin's penalty disbursement report; provided, however, that Algonquin will calculate and include such credits on Non-offending Customers' invoices for a period shorter than twelve months in the event and to the extent that the total accumulated amount of penalty revenue collected pursuant to this Section 24.8 by Algonquin as of the end of any Month exceeds \$1,000,000. Algonquin will file a penalty disbursement report within sixty days of July 31 or sixty days after the end of the Month in which the revenue collected exceeds \$1,000,000. Any penalty revenue credited to Non-offending Customers pursuant to this section shall include interest calculated in accordance with Section 154.501(d) of the Commission's regulations.

FERC Docket: RP10-77-000

Second Revised Sheet No. 572 Second Revised Sheet No. 572

Superseding: First Revised Sheet No. 572

GENERAL TERMS AND CONDITIONS (continued)

25. IMBALANCE RESOLUTION PROCEDURES

- 25.1 Customer's Responsibility. It shall be the responsibility of Customer to provide accurate and timely nominations of quantities proposed to be received and delivered by Algonquin under all of Customer's service agreements; to maintain as nearly as practicable equality between quantities actually taken by Customer under each service agreement and Customer's scheduled quantities under each service agreement; and to maintain a concurrent balance between receipts and deliveries under each of Customer's transportation service agreements. Except as otherwise expressly provided in Rate Schedules AFT-Es and AFT-Es, upon notice by Algonquin, Customer shall cause the rate of deliveries to Algonquin at the Point(s) of Receipt under a service agreement to equal as nearly as practicable the rate of deliveries by Algonquin at the Point(s) of Delivery under such service agreement.
- 25.2 Determination of Deliveries. The quantity delivered to Customer for purposes of this Section 25 shall be determined in accordance with Sections 5, 6, and 28 of these General Terms and Conditions.
- 25.3 Determination of Receipts. The quantity received by Customer at the Point(s) of Receipt under Customer's transportation service agreement(s) shall be determined in accordance with Section 27 of the General Terms and Conditions.
- 25.4 Allocation of Receipts Among Rate Schedules. Customer's receipts determined in accordance with Section 27 shall be apportioned among service agreements utilizing the same priority of rate schedules as is set forth in Section 28 of these General Terms and Conditions for the apportionment of deliveries among rate schedules.
- 25.5 Allocation of Receipts Among Service Agreements. In the event Customer has more than one service agreement under a single rate schedule, Customer's receipts under that rate schedule determined in accordance with Section 27 shall be apportioned among such service agreements according to a priority specified by Customer at the time of Customer's submission of nominations of receipts under such service agreements.
- 25.6 Resolution of Imbalances. Imbalances shall be determined separately by Customer, by contract and by individual pairs of Points of Receipt and Points of Delivery specified in Customer's nomination to Algonquin and Algonquin's notice of scheduled quantities provided to Customer by Algonquin pursuant to Sections 22.3, 22.4 and 22.5 of the General Terms and Conditions. During the Month, Cash-out Party may use any or all Imbalance Management Services, as defined in Section 1 of the General Terms and Conditions, to reduce or eliminate the cash-out liability for imbalances. In addition, Cash-out Party may trade an imbalance with another Cash-out Party, in accordance with Section 25.8 below, until the close of the seventeenth Business Day after the end of the Month in which the imbalances occurred. Any imbalance(s) not resolved via any of the Imbalance Management Services will be resolved as set forth in Section 25.10 herein.

FERC Docket: RP10-77-000

Second Revised Sheet No. 573 Second Revised Sheet No. 573

Superseding: First Revised Sheet No. 573

GENERAL TERMS AND CONDITIONS (continued)

- 25. IMBALANCE RESOLUTION PROCEDURES (continued)
 - 25.7 Imbalance Management Services. The Imbalance Management Services offered by Algonquin are identified in Section 1 of the General Terms and Conditions. Details of each service are included in the applicable Rate Schedule and Form of Service Agreement contained in this tariff, or in Sections 25.8 and 25.9 below.
 - 25.8 Imbalance Trading. Imbalance Trading is available for any imbalance that would otherwise be resolved pursuant to Sections 25.9 and 25.10 below. Algonquin shall allow Cash-out Parties to trade imbalances within the same Operational Impact Area, as defined in Section 1 of the General Terms and Conditions, if the two Cash-out Parties' imbalances are offsetting balances for the Month, such that the net imbalance after the completion of the trade for each Cash-out Party would be reduced to a quantity closer to zero. A Cash-out Party may trade any imbalance with another Cash-out Party, provided that any trades involving OBA imbalances shall not result in a transportation path which crosses a Posted Point of Restriction, as defined in Section 1 of the General Terms and Conditions; provided further that, to the extent that imbalances were incurred on Days when no Posted Point of Restriction was in effect, those imbalances are available for trading. An OBA Party that trades an imbalance resulting from actual deliveries by Algonquin in excess of scheduled deliveries ("due Algonquin") shall be assessed a transportation imbalance charge. OBA Party that trades an imbalance resulting from actual deliveries by Algonquin that are less than scheduled deliveries ("due OBA Party") shall be assessed a transportation imbalance credit. The transportation imbalance charge and the transportation imbalance credit shall be calculated by multiplying the traded quantity by the weighted average of the actual Commodity Charges owed on all quantities of gas delivered during the Month to that OBA Party.

FERC Docket: RP10-77-000

Second Revised Sheet No. 574 Second Revised Sheet No. 574

Superseding: First Revised Sheet No. 574

GENERAL TERMS AND CONDITIONS (continued)

25. IMBALANCE RESOLUTION PROCEDURES (continued)

Algonquin will provide the ability to post and trade imbalances at any time during the gas flow Month, and until the seventeenth Business Day after the end of the Month. To facilitate the trading process, Algonquin will, upon receipt of Cash-out Party's authorization, post a Cash-out Party's imbalance quantity on its Web Site. Authorizations to Post Imbalances that are received by Algonquin by 11:45 a.m. will be effective by 8:00 a.m. the next Business Day (central clock time). An Authorization to Post Imbalances will remain in effect until cancelled by the Cash-out Party. Imbalances previously authorized for posting will be posted as they become available, but no later than the ninth Business Day of the Month; however, Algonquin will not be required to post zero imbalances. The information posted will also identify the Cash-out Party, the contract, the Operational Impact Area and the gas flow Month applicable to the posted imbalance quantity. For purposes of determining the imbalance quantity that will be posted, all imbalances within an Operational Impact Area due Cash-out Party under all of Cash-out Party's contracts and all imbalances in that Operational Impact Area due Algonquin under all of Cash-out Party's contracts shall be summed together to yield a single net imbalance quantity for the Cash-out Party in that Operational Impact Area for the Month, unless otherwise agreed to pursuant to Section 25.9 below. Algonquin will provide to all Customers the ability to view, and upon request, download posted imbalance information.

Algonquin shall enable the imbalance trading process by (i) receiving the Request for Imbalance Trade, (ii) receiving the Imbalance Trade Confirmation, (iii) sending the Imbalance Trade Notification to all affected parties, and (iv) reflecting the trade prior to or on the next monthly Customer Imbalance or cashout. When trading imbalances, the quantity to be traded must be specified. After receipt of an Imbalance Trade Confirmation, Algonquin will send the Imbalance Trade Notification to the initiating trader and the confirming trader no later than noon (central clock time) on the next Business Day. Imbalance trades can only be withdrawn by the initiating trader and only prior to the confirming trader's confirmation of the trade. Imbalance trades are considered final when confirmed by the confirming trader and effectuated by Algonquin. Algonquin shall update the Cashout Party's imbalance data to reflect any final trades of imbalance quantities no later than 9:00 a.m. CT on the next Business Day after the trade is finalized.

FERC Docket: RP10-77-000

Third Revised Sheet No. 575 Third Revised Sheet No. 575

Superseding: Second Revised Sheet No. 575

GENERAL TERMS AND CONDITIONS (continued)

25.IMBALANCE RESOLUTION PROCEDURES (continued)

25.9 Imbalance Netting

For purposes of determining the imbalance quantity that will be subject to resolution pursuant to Section 25.10, all imbalances within an Operational Impact Area due Cashout Party under all of Cashout Party's contracts for a Month and all imbalances in that same Operational Impact Area due Algonquin under all of Cashout Party's contracts for that same Month shall be summed together to yield a single imbalance for that Operational Impact Area for the Month, unless otherwise requested in writing by Cashout Party. Any imbalance not resolved after Sections 25.7 and 25.8 above have been utilized will be subject to resolution in accordance with Section 25.10 below.

25.10 Cash-out Provision

Prior to or with the monthly transportation invoice, Algonquin will render each Cashout Party a statement detailing the unresolved imbalances. In the second Month after the imbalances occurred, a bill for the amount due Algonquin or a credit of the amount due Cash-out Party, as determined below, will be rendered with the monthly transportation invoice pursuant to Section 18.1 of these General Terms and Conditions.

- (a) Determination of Index Price. The Index Price for purposes of resolving imbalances shall be the average of the index cash out prices for the relevant Month established for purposes of resolving imbalances by the upstream pipelines that delivered gas to Algonquin for Customer's account at Customer's Points of Receipt, such average being weighted on the basis of Customer's receipts. In the event that none of the upstream facilities operators that delivered gas for Customer's account has established a cash out price for purposes of resolving imbalances, the cash out index price established by Texas Eastern Transmission, LP for deliveries at Lambertville, New Jersey during the Month in question shall be utilized as the Index Price. The upstream index price utilized in each case shall be inclusive of any zone, location or transportation charges assessed by the upstream pipeline in the course of cashing out imbalances on such upstream facilities that are attributable to points of interconnection with Algonquin's facilities.
- (b) Imbalance Due Algonquin. In the event of an imbalance caused when Cash-out Party's allocated deliveries as determined in accordance with Section 25.2 exceed Cash-out Party's allocated receipts as determined in accordance with Section 25.3, less an allowance for fuel determined in accordance with Section 32 of the General Terms and Conditions, Algonquin shall charge Cash-out Party for such excess delivery plus an allowance for fuel calculated by multiplying such excess delivery by the applicable fuel percentage established pursuant to Section 32 of the General Terms and Conditions. Such charge shall be based on the appropriate index price for the Month multiplied by one or more of the following factors:

Imbalance Level	Factor
Less than or equal to 5%	1.00
Greater than 5% but less than or equal to 10%	1.10
Greater than 10% but less than or equal to 15%	1.20
Greater than 15% but less than or equal to 20%	1.30
Greater than 20% but less	
than or equal to 25%	1.40 1.50
Greater than 25%	1.50

FERC Docket: RP10-77-000

Second Revised Sheet No. 576 Second Revised Sheet No. 576

Superseding: First Revised Sheet No. 576

GENERAL TERMS AND CONDITIONS (continued)

25. IMBALANCE RESOLUTION PROCEDURES (continued)

For purposes of determining the appropriate factor, Cash-Out Party's imbalance shall be determined by taking the lower of (a) the level of imbalance supplied pursuant to Section 40.2, or (b) the imbalance computed by comparing (i) the deliveries at the Point of Delivery as determined in accordance with Section 25.2 and (ii) the receipts at the Point of Receipt as determined in accordance with Section 25.3.

Provided, however, that in the case of an imbalance created by an action of Algonquin taken pursuant to Rate Schedule AFT-E or AFT-ES to decrease receipts by Algonquin for Cash-out Party's account the applicable factor shall be 1.0 to the extent Cash-out Party's receipts were in fact reduced pursuant to such action by Algonquin.

Provided further, for Rate Schedule AFT-1S or AFT-ES Customers, the level of imbalance shall not be determined until after Algonquin subtracts up to 4,828 Dth, as necessary, from each Rate Schedule AFT-1S or AFT-ES Customer's total monthly imbalance. Such amount subtracted up to 4,828 Dth will be cashed out at the appropriate index price for the Month.

The applicable imbalance percentage for purposes of determining the applicable cash out price factor shall be determined by dividing the amount of the excess delivery by the receipts (less fuel). The calculation of cash out charges relating to excess deliveries shall also include a transportation imbalance charge which shall be calculated by multiplying the excess delivery quantities by the weighted average of the actual Commodity Charges owed on all quantities of gas delivered during the Month to that Cash-out Party.

(c) Imbalance Due Cash-out Party. In the event of an imbalance caused when Cash-out Party's allocated deliveries as determined in accordance with Section 25.2 are less than Cash-out Party's allocated receipts as determined in accordance with

FERC Docket: RP10-77-000

Second Revised Sheet No. 577 Second Revised Sheet No. 577

Superseding: First Revised Sheet No. 577

GENERAL TERMS AND CONDITIONS (continued)

25.IMBALANCE RESOLUTION PROCEDURES (continued)

Section 25.3, less an allowance for fuel determined in accordance with Section 32 of the General Terms and Conditions, Algonquin shall make a cash out payment to Cash-out Party reflecting such excess receipts. Such payment will be based on the index price for the Month multiplied by one or more of the following factors:

Imbalance Level	Factor
Less than or equal to 5%	1.00
Greater than 5% but less	
than or equal to 10%	0.90
Greater than 10% but less	
than or equal to 15%	0.80
Greater than 15% but less	
than or equal to 20%	0.70
Greater than 20% but less	
than or equal to 25%	0.60
Greater than 25%	0.50

For purposes of determining the appropriate factor, Cash-out Party's imbalance shall be determined by taking the lower of (a) the level of imbalance supplied pursuant to Section 40.2, or (b) the imbalance computed by comparing (i) the deliveries at the Point of Delivery as determined in accordance with Section 25.2 and (ii) the receipts at the Point of Receipt as determined in accordance with Section 25.3.

Provided, however, that in the case of an imbalance created by an action of Algonquin taken pursuant to Rate Schedule AFT-E or AFT-ES to increase receipts by Algonquin for Cash-out Party's account, the applicable factor shall be 1.0 to the extent Cash-out Party's receipts were in fact increased pursuant to such action by Algonquin.

Provided further, for Rate Schedule AFT-1S or AFT-ES Customers, the level of imbalance shall be determined after Algonquin subtracts up to 4,828 Dth, from each Rate Schedule AFT-1S or AFT-ES Customer's total monthly imbalance. Such amount subtracted up to 4,828 Dth will be cashed out at the appropriate index price for the Month.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 578 Fourth Revised Sheet No. 578

Superseding: Third Revised Sheet No. 578

GENERAL TERMS AND CONDITIONS (continued)

25. IMBALANCE RESOLUTION PROCEDURES (continued)

The applicable imbalance percentage for purposes of determining the applicable cash out price shall be determined by dividing the excess receipts by the total receipts (less fuel). The calculation of the amount due Cash-out Party relating to excess receipts shall also include a transportation imbalance credit, which shall be calculated by multiplying the excess receipt quantities by the weighted average of the Commodity Charges owed on all quantities of gas delivered during the Month to that Cashout Party. Algonquin shall have no responsibility for the distribution of funds beyond the initial distribution to the Cash-out Party.

- (d) [Reserved for Future Use]
- (e) Credit. On a monthly basis Algonquin shall credit or debit, as appropriate, the Fuel Reimbursement Quantity Deferred Account as provided in Section 32 herein, the net proceeds from the operation of the imbalance resolution procedures contained in this Section 25.
- 26. ACTIONS REQUIRED TO PROTECT SYSTEM OPERATIONS
 - 26.1 If any of the following conditions occur, then Algonquin may take the actions described in Section 26.2 in addition to actions permitted elsewhere herein:
 - (a) Customer's actual average hourly deliveries over any period of time under all contracts under which gas is delivered for Customer's account at a given Point of Delivery exceed receipts of gas for Customer's account at relevant Point(s) of Receipt, or
 - (b) Actual average hourly deliveries under all contracts under which gas is delivered for Customer's account at a given Point of Delivery are less than receipts of gas for Customer's account at relevant Point(s) of Receipt, and
 - (c) Algonquin determines in its reasonable discretion that, if Algonquin does not take prompt action, Algonquin's ability to continue to deliver scheduled quantities to other Customers may be impaired by the further accumulation of such imbalances.

For the purposes of this Section 26, the term "Customer" refers to one or more of Customer, operator or agent, as applicable.

FERC Docket: RP10-308-000

Second Revised Sheet No. 579 Second Revised Sheet No. 579

Superseding: First Revised Sheet No. 579

GENERAL TERMS AND CONDITIONS (continued)

26.ACTIONS REQUIRED TO PROTECT SYSTEM OPERATIONS (continued)

- 26.2 Within a reasonable time after the occurrence of any of the conditions described in Section 26.1, Algonquin shall notify Customer. Following such notice, Algonquin and Customer shall agree upon the appropriate action that Algonquin should take to protect system operations.
- 26.3 In the event Customer and Algonquin cannot agree upon an appropriate action within a reasonable time, Algonquin may do any or all of the following:
 - (a) Adjust Customer's nominations under one or more contracts immediately and for the remainder of the Month as necessary to halt the further accumulation of imbalances on Customer's account,
 - (b) Revoke Algonquin's confirmation of receipts with Customer's upstream transporters, and revise any confirmed quantity as necessary to reflect adjustments of scheduled quantities on Algonquin's system,
 - (c) Take any actions that are within Algonquin's operational capabilities to reduce or eliminate Customer's excess receipts, and
 - (d) Take any actions that are within Algonquin's operational capabilities to reduce or eliminate Customer's excess deliveries.
- 26.4 In the event Algonquin takes action to alleviate excess imbalances it shall nonetheless remain the obligation of Customer to make such further adjustments to nominations during the remainder of the Month to resolve accumulated imbalances or to account for subsequent changes in actual deliveries.
- 26.5 Algonquin's ability to exercise, or Algonquin's actual exercise of its authority under this Section 26, shall have no effect upon Customer's liability for scheduling, overrun or imbalance penalties

FERC Docket: RP10-308-000

Third Revised Sheet No. 580 Third Revised Sheet No. 580

Superseding: Second Revised Sheet No. 580

GENERAL TERMS AND CONDITIONS (continued)

26.ACTIONS REQUIRED TO PROTECT SYSTEM OPERATIONS (continued)

otherwise applicable to Customer under other provisions of this tariff, provided that Customer's revised nominated and scheduled quantities shall provide the basis for the computation of such penalties.

Algonquin shall provide, via posting on the LINK® System and the Web Site, prior notice to all Customers of upcoming events that may necessitate the invocation of the provisions of this Section 26. At such time as Algonquin contacts a Customer pursuant to this Section 26, Algonquin shall post a notice to all Customers on its Web Site that it has contacted a Customer pursuant to this section. Such notice will specify that the posting is being made pursuant to the provisions of Section 26 of the General Terms and Conditions and will include a description of the circumstances that caused Algonquin to contact the Customer, but will not identify the specific Customer contacted. Algonquin shall rescind the invocation of the provisions of this Section 26 promptly upon the cessation of operating conditions that caused the relevant system problem, and shall post a notice on its Web Site specifying the factors that caused the provisions of this Section 26 to be invoked and then rescinded, to the extent such factors are known.

27.DETERMINATION OF RECEIPTS

The quantity received by Customer at the Point(s) of Receipt under Customer's transportation service agreement(s) shall be determined as follows:

- (a) At each Point of Receipt at which Customer's gas is the only gas entering Algonquin's facilities, Customer's receipts under all service agreements in aggregate shall equal the quantity in total measured by the upstream operator's meter at the relevant point;
- (b) At each Point of Receipt at which Customer's gas is not the only gas entering Algonquin's facilities, if Algonquin has executed an Operational Balancing Agreement under Section 30 herein with the operator of the upstream facilities that permits Algonquin to do so, Algonquin shall deem Customer's receipts to equal Customer's scheduled receipts;
- (c) At each Point of Receipt at which Customer's gas is not the only gas entering Algonquin's system, and if Algonquin has not executed an Operational Balancing Agreement pursuant to Section 30 herein with the operator of the upstream facilities, metered quantities shall be apportioned among all Customers pro rata according to scheduled quantities; provided however, before Algonquin may invoke this subsection Algonquin shall have provided nominations and shall have scheduled receipts of any gas received by Algonquin for Algonquin's account for Company Use Gas in a like manner to that required of Customers under Algonquin's firm open access transportation rate schedule; or
- (d) In accordance with such other schedule or methodology as to which Algonquin and Customer shall agree. Customer shall submit such schedule or methodology to Algonquin via the LINK® System, unless otherwise agreed to by Algonquin and Customer.

FERC Docket: RP07-557-000

Second Revised Sheet No. 581 second Revised Sheet No. 581 : Effective

Superseding: First Revised Sheet No. 581

GENERAL TERMS AND CONDITIONS (continued)

- 28. DETERMINATION OF DELIVERIES
- 28.1 Single Customer Delivery Points. At each Point of Delivery at which Customer's gas is the only gas being delivered by Algonquin, Customer's deliveries in aggregate under all Service Agreements shall equal the quantity in total measured at the Point of Delivery.
- 28.2 Multiple Customer Delivery Points. At each Point of Delivery at which Customer's gas is not the only gas being delivered by Algonquin, deliveries to each Customer will be determined in the following manner:
 - (a) If Algonquin has executed an agreement with the operator of the delivery point facilities that permits Algonquin to do so, Algonquin shall deem Customer's deliveries to equal Customer's scheduled deliveries;
 - (b) If Algonquin has not executed an agreement with the operator of the delivery point facilities as described in Section 28.2(a), metered quantities shall be apportioned among all Customers pro rata according to scheduled quantities; or
 - (c) In accordance with such other schedule or methodology as to which Algonquin and Customer shall agree. Customer shall submit such schedule or methodology to Algonquin via the LINK® System, unless otherwise agreed to by Algonquin and Customer.
- 28.3 Allocation of Deliveries Among Rate Schedules. Parties may agree upon one of the following allocation methodologies: ranked, pro rata, percentage, swing and operator provided value. In the event the parties cannot agree on a methodology, the methodology listed below shall be used as the default method. The party responsible for custody transfer (the party performing the measurement function) must provide the allocation. Algonquin will accept such allocation if operationally or administratively feasible.

Whenever Customer is receiving deliveries of gas at a Point of Delivery under more than one rate schedule, and unless Customer informs Algonquin otherwise via the LINK® System, unless otherwise agreed to by Algonquin and Customer, Customer's deliveries as determined in accordance with Section 28.2 will be apportioned among rate schedules as follows:

FERC Docket: RP10-77-000

First Revised Sheet No. 582 First Revised Sheet No. 582

Superseding: Original Sheet No. 582

GENERAL TERMS AND CONDITIONS (continued)

28. DETERMINATION OF DELIVERIES (continued)

- (a) Rate Schedule PAL. The quantity of gas scheduled for delivery under this rate schedule on such Day but not more than the total delivery. In the event that Customer has not scheduled quantities to be delivered under (b) through (j), the total quantity delivered by Algonquin to Customer on any Day shall constitute the deliveries on such Day under Rate Schedule PAL.
- (b) Rate Schedules AIT-1 and AIT-2. The quantity of gas scheduled for delivery under these rate schedules on such Day but not more than the total delivery less the deliveries under (a) above. In the event that Customer has not scheduled quantities to be delivered under (c) through (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) above shall constitute the deliveries on such Day under Rate Schedules AIT-1 and/or AIT-2, as applicable.
- (c) Rate Schedule AFT-4. The quantity of gas scheduled for delivery under this rate schedule on such Day but not more than the total delivery less the deliveries under (a) through (b) above. In the event that Customer has not scheduled quantities to be delivered under (d) through (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (b) above shall constitute the deliveries on such Day under Rate Schedule AFT-4.
- (d) Rate Schedule T-1. The quantity of gas scheduled for delivery under this rate schedule on such Day but not more than the total delivery less the deliveries under (a) through (c) above. In the event that Customer has not scheduled quantities to be delivered under (e) through (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (c) above shall constitute the deliveries on such Day under Rate Schedule

FERC Docket: RP10-77-000

First Revised Sheet No. 583 First Revised Sheet No. 583

Superseding: Original Sheet No. 583

GENERAL TERMS AND CONDITIONS (continued)

28. DETERMINATION OF DELIVERIES (continued)

- (e) Volume No. 2 Rate Schedules. The quantity of gas scheduled for delivery under any rate schedule contained in Volume No. 2 of Algonquin's FERC Gas Tariff, in the order specified in advance by Customer, on any Day but not more than the total delivery less the deliveries under (a) through (d) above. In the event that Customer has not scheduled quantities to be delivered under (f) through (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (d) above shall constitute the deliveries on such Day under Volume No. 2 Rate Schedules.
- (f) Rate Schedule AFT-CL. In the event deliveries are scheduled under Rate Schedule AFT-CL on any Day, the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (e) above. In the event that Customer has not scheduled quantities to be delivered under (g) through (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (e) above shall constitute the deliveries on such Day under Rate Schedule AFT-CL.
- (g) Rate Schedule AFT-1. In the event deliveries are scheduled under Rate Schedule AFT-1 on any Day, the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (f) above, but not in excess of the scheduled quantities if Customer has scheduled quantities to be delivered under the rate schedules listed in (h) through (j). In the event that Customer has not scheduled quantities to be delivered under (h) through (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (f) above shall constitute the deliveries on such Day under Rate Schedule AFT-1.
- (h) Rate Schedule AFT-1S. In the event deliveries are scheduled under Rate Schedule AFT-1S on any Day, the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (g) above, but not in excess of the scheduled quantities if Customer has scheduled quantities to be delivered under the rate schedules listed in (i) through (j). In the event that Customer has not scheduled quantities to be delivered under (i) through (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (g) above shall constitute the deliveries on such Day under Rate Schedule AFT-1S.

FERC Docket: RP10-77-000

Second Revised Sheet No. 584 Second Revised Sheet No. 584

Superseding: First Revised Sheet No. 584

GENERAL TERMS AND CONDITIONS (continued)

- 28. DETERMINATION OF DELIVERIES (continued)
 - (i) Rate Schedule AFT-E. In the event Customer has scheduled deliveries of gas under Rate Schedule AFT-E on any Day, the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (h) above but not in excess of the scheduled quantities if Customer has scheduled quantities to be delivered under the rate schedule listed in (j). In the event that Customer has not scheduled quantities to be delivered under (j), the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (h) above shall constitute the deliveries on such Day under Rate Schedule AFT-E.
 - (j) Rate Schedule AFT-ES. In the event Customer has scheduled deliveries of gas under Rate Schedule AFT-ES on any Day, the total quantity delivered by Algonquin to Customer on any Day, less the quantities determined pursuant to (a) through (i) above shall constitute the deliveries on such Day under Rate Schedule AFT-ES.
- Allocation of Deliveries Among Service Agreements. Whenever Customer is receiving deliveries of gas at a Point of Delivery under more than one Service Agreement under a particular rate schedule, and unless Customer specifies a priority for deliveries under such Service Agreements after or during confirmation and before the effective time of the gas flow, Customer's deliveries under that rate schedule, as determined in accordance with Section 28.3, will be apportioned among such Service Agreements pro rata according to scheduled quantities.
- 29. OPERATIONAL FLOW ORDERS
- 29.1 Purpose. In the event that Algonquin determines, in its reasonable discretion, that a quantity of gas is required for use as Company Use Gas, Algonquin shall issue a flow order pursuant to this section; provided, however, that this section may only be invoked if Algonquin determines that such quantity of gas is not obtainable

FERC Docket: RP10-77-000

First Revised Sheet No. 585 First Revised Sheet No. 585

Superseding: Original Sheet No. 585

GENERAL TERMS AND CONDITIONS (continued)

29.OPERATIONAL FLOW ORDERS (continued)

in the time frame required by utilizing off-system resources available to Algonquin, or pursuant to Section 32 of these General Terms and Conditions; and further provided that Algonquin shall have first taken any appropriate action pursuant to Section 26 of these General Terms and Conditions. Such flow orders shall be uniformly applicable to all volumes received at all Points of Receipt under all rate schedules.

- 29.2 Uses of Gas Obtained Pursuant to Flow Orders. Gas obtained pursuant to flow orders issued under authority of this section may be utilized only to provide Company Use Gas to the extent gas for such uses is not otherwise available to Algonquin from resources at its disposal.
- 29.3 Imposition of Flow Orders. Algonquin shall provide, via posting on the LINK® System and the Web Site, prior notice to all Customers of upcoming events such as operational problems that may necessitate the issuance of an operational flow order. If Algonquin invokes this section it shall confirm a telephonic or facsimile notice by posting the notice on the LINK® System and its Internet Web Site specifying:
 - (a) the increment to be added to the otherwise applicable Fuel Reimbursement Percentage provided for in Section 32 of these General Terms and Conditions,
 - (b) the total quantity of gas that Algonquin estimates it will require to meet immediate operational needs,
 - (c) the Hour at and Day on which the flow order will become effective, and
 - (d) the period of time during which Algonquin expects the flow order to remain in effect.

Such notice shall be posted twenty-four hours in advance of any flow order issued pursuant to this section or such lesser period of time as is practicable under the circumstances, shall specify that the flow order is being issued pursuant to the provisions of Section 29 of the General Terms and Conditions and shall specify the factors that caused the flow order to be issued, to the extent such factors are known. During the pendency of any such flow order Algonquin shall obtain the additional gas it requires by balancing receipts and deliveries, Algonquin being obligated to deliver the quantity of gas received for Customer's account reduced by the Fuel Reimbursement Quantity including quantities retained pursuant to this section.

After Algonquin has lifted the flow order, Algonquin shall post a notice on the LINK® System and the Web Site specifying the factors that caused the flow order to be issued and then lifted, to the extent such factors are known.

FERC Docket: RP10-77-000

First Revised Sheet No. 586 First Revised Sheet No. 586

Superseding: Original Sheet No. 586

GENERAL TERMS AND CONDITIONS (continued)

- 29. OPERATIONAL FLOW ORDERS (continued)
- 29.4 Compensation. On the invoice for the Month in which this Section 29 was invoked,
 Customer shall receive a credit equal to the product of the volume in Dth of Customer's
 gas retained by Algonquin for each Day pursuant to this Section 29 multiplied by the
 Midpoint price for the "AGT Citygate" Gas Daily posting on that Day.
- 30. OPERATIONAL BALANCING AGREEMENTS
- 30.1 General. Any imbalances arising under any transportation agreement at Points of Receipt into Algonquin's system, or at Points of Delivery from Algonquin's pipeline system, which Points of Receipt or Points of Delivery are subject to Operational Balancing Agreements as more fully described in Section 30.2 below, on the Day or Days such imbalances arise, will be resolved by Algonquin pursuant to the terms of the applicable Operational Balancing Agreement.
- 30.2 Terms Governing. For the purpose of minimizing operational conflicts between various natural gas facilities with respect to the delivery of gas to and from Algonquin's facilities, Algonquin is willing to negotiate and execute Operational Balancing Agreements with appropriate parties that operate natural gas facilities interconnecting with Algonquin's system (any such party referred to as "Operator" herein). Such Operational Balancing Agreements shall specify the gas custody transfer procedures to be followed by Algonquin and Operator for the confirmation of scheduled quantities to be received by Algonquin at Points of Receipt and delivered by Algonquin at Points of Delivery. Such Operational Balancing Agreements will provide that any variance between actual quantities and scheduled and confirmed quantities for any Day shall be resolved in-kind promptly, or pursuant to such cash out provisions as Algonquin and the Operator may agree upon. To facilitate such determination of variances on a timely basis,

FERC Docket: RP10-77-000

First Revised Sheet No. 587 First Revised Sheet No. 587

Superseding: Original Sheet No. 587

GENERAL TERMS AND CONDITIONS (continued)

30. OPERATIONAL BALANCING AGREEMENTS (continued)

Algonquin and the Operator will agree in the Operational Balancing Agreement on necessary measurement and accounting procedures. Algonquin shall post on its Internet Web Site those Points of Receipt and Points of Delivery which are subject to an Operational Balancing Agreement. Algonquin will also provide to any party upon request a copy of any executed Operational Balancing Agreement. A form of Operational Balancing Agreement is contained among the forms of service agreements contained in this tariff.

- 30.3 Prerequisites to Execution. It is Algonquin's intent to negotiate and execute Operational Balancing Agreements on a non-discriminatory basis with any Operator. However, Algonquin shall have no obligation to negotiate and execute Operational Balancing Agreements with any Operator that:
 - (a) is not creditworthy as determined pursuant to Section 3 of these General Terms and Conditions, substituting the term "Operator" for "Customer" for this purpose;
 - (b) does not maintain a dispatching operation which is staffed on a continuous, around-the-clock basis;
 - (c) would cause an increase in the level of regulation which Algonquin is subject to prior to the execution of the applicable Operational Balancing Agreement; or
 - (d) does not commit to timely determination of imbalances based on reasonable available measurement technology.
- 30.4 Right To Protect System Integrity. Nothing in this Section 30 nor any executed Operational Balancing Agreement shall limit Algonquin's rights to take action as may be required to adjust receipts and deliveries under any transportation agreement to reflect actual experience or to alleviate conditions which threaten the integrity of Algonquin's pipeline system, including maintenance of service to higher priority Customers or services.

FERC Docket: RP10-77-000

Second Revised Sheet No. 588 Second Revised Sheet No. 588

Superseding: First Revised Sheet No. 588

GENERAL TERMS AND CONDITIONS (continued)

- 30. OPERATIONAL BALANCING AGREEMENTS (continued)
- 30.5 Recordkeeping. Algonquin shall maintain records of volumes and amounts paid pursuant to Operational Balancing Agreements entered into under this Section 30. Such records shall be available for review.
- 30.6 Determination of Deliveries. Deliveries to shippers at points covered by Operational Balancing Agreements entered into under this Section 30 shall be determined in accordance with Section 28.2(a).
- 31. UNAUTHORIZED CONTRACT OVERRUN GAS
- 31.1 Unauthorized Contract Overrun Gas Defined.
 - (a) If, during any Hour, the total quantity of gas taken by Customer from Algonquin at all Point(s) of Delivery under any Service Agreement exceeds Customer's MHTQ under such Service Agreement during such Hour, and Algonquin has posted notice on the LINK® System and its Internet Web Site that system operating conditions require adherence to Customers' MHTQs in order to enable deliveries of scheduled quantities to firm Customers, then all such gas taken in excess of such entitlement shall constitute Unauthorized Contract Overrun Gas. Algonquin shall confirm an electronic notice by telephone or facsimile one hour prior to taking action pursuant to Section 31.1(a).
 - (b) If, on any Day, the total quantity of gas taken by Customer from Algonquin at all Point(s) of Delivery under any Service Agreement exceeds Customer's MDTQ under such Service Agreement on such Day, and Algonquin has not authorized such excess quantity, then all such gas taken in excess of such entitlement shall constitute Unauthorized Contract Overrun Gas. In the event Customer takes Unauthorized Contract Overrun Gas and such Unauthorized Contract Overrun Gas did not threaten system integrity or deliveries to firm Customers, such Unauthorized Contract Overrun Gas shall be deemed for purposes of the applicable rate schedules and these General Terms and Conditions to be "Authorized Overrun Gas".
- 31.2 Penalty Payment. In the event that Customer takes Unauthorized Contract Overrun Gas and such Unauthorized Contract Overrun Gas threatens system integrity or deliveries to firm shippers, Customer shall pay Algonquin an unauthorized contract overrun penalty equal to three times the daily Gas Daily posting for the High Common price for the "AGT Citygate" posting per Dth on the Day the violation occurred for all Unauthorized Contract Overrun Gas as defined in Section 31.1 above, subject only to the waiver provisions set forth in Section 31.3 below, in addition to the amounts otherwise payable by Customer to Algonquin under Algonquin's applicable rate schedules and these General Terms and Conditions.

FERC Docket: RP10-77-000

First Revised Sheet No. 589 First Revised Sheet No. 589

Superseding: Original Sheet No. 589

GENERAL TERMS AND CONDITIONS (continued)

31. UNAUTHORIZED CONTRACT OVERRUN GAS (continued)

Any penalty revenue collected by Algonquin pursuant to this Section 31.2 will be credited to those firm and interruptible Customers that did not incur penalties pursuant to this Section 31.2 in the Month for which penalty revenues were received ("Non-offending Customers"), based on the ratio of the Reservation and Usage Charges paid during that Month by each Non-offending Customer to the total amount of such charges paid during that Month by all Non-offending Customers. An operator that has executed an Operational Balancing Agreement and a transportation service agreement with Algonquin shall be eligible to receive penalty revenue credits as a Non-offending Customer for any Month in which the operator was not liable for penalties pursuant to this Section 31.2 under its own transportation service agreement(s). Such credits shall be calculated for each Month of the twelve (12) month period ending July 31 of each year and will be included on the Non-offending Customers' invoice for the Month following the date of the final Commission order approving Algonquin's penalty disbursement report; provided, however, that Algonquin will calculate and include such credits on Non-offending Customers' invoices for a period shorter than twelve months in the event and to the extent that the total accumulated amount of penalty revenue collected pursuant to this Section 31.2 by Algonquin as of the end of any Month exceeds \$1,000,000. Algonquin will file a penalty disbursement report within sixty days of July 31 or sixty days after the end of the Month in which the revenue collected exceeds \$1,000,000. Any penalty revenue credited to Non-offending Customers pursuant to this section shall include interest calculated in accordance with Section 154.501(d) of the Commission's regulations.

- 31.3 Waiver of Penalty. The penalty payment provided in Section 31.2 above shall not apply:
 - a) When the Unauthorized Contract Overrun Gas in any Hour under Section 31.1(a) above does not exceed 2 Dth or 2% of Customer's MHTQ, whichever is greater, and
 - b) When the Unauthorized Contract Overrun Gas on any Day under Section 31.1(b) above does not exceed 50 Dth or 2% of Customer's MDTQ, whichever is greater.
- 31.4 Reservation; Other Remedies. This Section 31, or the payment of a penalty for Unauthorized Contract Overrun Gas hereunder, shall under no circumstances be considered as giving Customer any right to take Unauthorized Contract Overrun Gas or be considered as a substitute for any other remedy available to Algonquin against the offending Customer for failure to respect its obligation to stay within its contract entitlements. Algonquin expressly reserves the right, when, in Algonquin's reasonable judgment, such remedy is necessary to protect service to Algonquin's other Customers, to shed the load of the offending Customer which is not staying within its contract entitlements by closing valves at Points of Delivery on a particular Day after such offending Customer has taken the total quantity of gas it is entitled to take on such Day from Algonquin, or to take such other action as Algonquin deems necessary or appropriate to protect deliveries to other Customers.
- 31.5 Notice of Unauthorized Overruns. Algonquin shall post notice on the LINK® System and its Internet Web Site when conditions triggering application of this Section 31 have occurred. Such notice shall specify that the posting is being made pursuant to the provisions of Section 31 of the General Terms and Conditions.

FERC Docket: RP10-117-001

First Revised Sheet No. 590 First Revised Sheet No. 590

Superseding: Original Sheet No. 590

GENERAL TERMS AND CONDITIONS (continued)

- 32. FUEL REIMBURSEMENT QUANTITY
- 32.1 General. The Fuel Reimbursement Quantity ("FRQ") shall be determined by multiplying Customer's receipts at the Point(s) of Receipt by the applicable Fuel Reimbursement Percentage ("FRP"), except in the case of Backhauls and/or Forwardhaul components of transportation on the HubLine Mainline facilities in which case the FRQ shall be zero. During the term of the Service Agreements executed hereunder, Algonquin will periodically track changes in its requirement to retain gas in-kind in compensation for the quantities of Company Use Gas used to provide service for Customers.
- 32.2 Fuel Reimbursement Percentage (FRP). The FRP shall be as quantified pursuant to this Section 32 and as set forth in the currently effective Sheet No. 40 of this tariff, shall be separately stated for system services and for each incremental service as required by Commission order. The FRP shall be subject to adjustment hereunder for service under all rate schedules unless otherwise explicitly provided in the rate schedule. The FRP shall include any increment added pursuant to a flow order issued pursuant to Section 29 of these General Terms and Conditions.
 - Specified Calendar Periods. FRPs shall be established distinctly for the duration of two calendar periods as follows: (i) winter period--December 1 through March 31 and (ii) spring, summer and fall period--April 1 through November 30.
- 32.3 Calculation of FRP Adjustments. FRPs shall be calculated and filed pursuant to this Section 32 on an annual basis for each calendar period specified in Section 32.2 above.
- 32.4 Calculations of FRP. For each filing hereunder for each specified calendar period Algonquin shall calculate the FRP, with separate calculations for system services and for each incremental service as required by Commission order, as the quotient obtained by dividing (a) the projected annual quantities of Company Use Gas for each specified calendar period by (b) the projected annual throughput for each specified calendar period.

FERC Docket: RP10-117-001

Second Revised Sheet No. 591 Second Revised Sheet No. 591

Superseding: First Revised Sheet No. 591

GENERAL TERMS AND CONDITIONS (continued)

32.FUEL REIMBURSEMENT QUANTITY (continued)

32.5 True Up.

- FRQ Deferred Account. Algonquin shall determine the monthly difference between (a) (i) the actual quantity of gas realized as a result of application of this Section 32 and (ii) the actual quantity of Company Use Gas for the Month less any quantity of gas obtained during the Month pursuant to Section 29. The under- or over-realization of in-kind compensation gas shall be recorded as a debit or a credit, as the case may be, each Month in the FRQ Deferred Account. A monetary value shall be assigned to the volume debited to the FRQ Deferred Account based on amounts paid by Algonquin in obtaining gas to support the under-realization of in-kind compensation gas. Furthermore, a monetary value shall be assigned to the volume credited to the FRQ Deferred Account utilizing the base index price in effect for purposes of resolving imbalances incurred during the Month at Lambertville, New Jersey, on the Texas Eastern system. The FRQ Deferred Account shall be debited for any amounts paid pursuant to Section 29.4. The FRQ Deferred Account shall also be debited or credited, as appropriate, for the net proceeds resulting from the resolution of imbalances under agreements cashed out in accordance with Section 25 of the General Terms and Conditions and Sections 7.6 and 8 of Rate Schedule PAL. The FRQ Deferred Account shall also be credited with the value of the gas retained by Algonquin pursuant to Section 7.5 of Rate Schedule PAL. The FRQ Deferred Account shall be accumulated in twelve-month accumulation periods, August to July. A separate sub-account shall be maintained in the FRQ Deferred Account for each incremental service as required by Commission order.
- (b) Carrying Charges. Each Month the FRQ Deferred Account shall be debited (if the balance in said account is a debit balance) or shall be credited (if the balance in said account is a credit balance) for a carrying charge, which shall be determined by the product of (1) and (2) below:

FERC Docket: RP10-117-001

Second Revised Sheet No. 592 Second Revised Sheet No. 592

Superseding: First Revised Sheet No. 592

GENERAL TERMS AND CONDITIONS (continued)

32.FUEL REIMBURSEMENT QUANTITY (continued)

- (1) The cash balance in the FRQ Deferred Account, exclusive of carrying charges accrued pursuant to this subsection (b), net of the related deferred tax amounts, if any, as of the end of the immediately preceding Month.
- (2) The annual interest rate as set forth from time to time in Section 154.501 of the Commission's regulations divided by 365, or 366 in a leap year, and then multiplied by the number of days in the applicable Month.
- (c) Calculation of Surcharge or Refund. With each annual filing hereunder Algonquin shall calculate surcharges or refunds, separately for system services and for each incremental service as required by Commission order, designed to amortize the net monetary value of the balance in each applicable sub-account of the FRQ Deferred Account at the end of the previous accumulation period. The surcharge or refund shall be based upon the allocation of the FRQ Deferred Account sub-account balance at the end of the twelve month accumulation period over the applicable actual throughput during the twelve month accumulation period, exclusive of Backhauls. A Customer's net debit or credit shall be due and payable sixty days after the Commission's acceptance of the annual filing.
- 32.6 Procedure for Filing. At least thirty days prior to the effective date of a change in the FRP hereunder, Algonquin shall file with the FERC revised tariff sheets setting out the proposed change and supporting workpapers showing the calculations developing such proposal. Algonquin shall file annually to revise the FRPs as provided in Section 32.3 herein and surcharge or refund as provided in Section 32.5 herein effective on December 1 of each year. Algonquin may file interim proposals between annual filings subject to approval by the Commission.

FERC Docket: RP06-519-000

First Revised Sheet No. 593 First Revised Sheet No. 593 : Effective

Superseding: Original Sheet No. 593

GENERAL TERMS AND CONDITIONS (continued)

- 33. GAS RESEARCH INSTITUTE CHARGE ADJUSTMENT PROVISION
 - 33.1 Purpose. Algonquin has joined with other gas enterprises in the formation of and participation in the activities and financing of Gas Research Institute ("GRI"), an Illinois not-for-profit corporation. GRI has been organized for the purpose of sponsoring Research, Development and Demonstration ("RD&D") programs in the field of natural and manufactured gas for the purpose of assisting all segments of the gas industry in providing adequate, reliable, safe, economic, and environmentally acceptable gas service for the benefit of gas consumers and the general public.
 - 33.2 [Reserved for Future Use]

FERC Docket: RP06-519-000

First Revised Sheet No. 594 First Revised Sheet No. 594 : Effective

Superseding: Original Sheet No. 594

GENERAL TERMS AND CONDITIONS (continued)

- 33.3 [Reserved for Future Use]
- 33.4 Remittance to GRI. Algonquin shall remit to GRI, not later than fifteen days after the receipt thereof, all monies actually collected and received by virtue of the voluntary contribution mechanism, less any amounts properly payable to a federal, state or local authority relating to the monies received hereunder. In no event will Algonquin be required to remit to GRI amounts not actually collected under this section. Algonquin shall indicate to GRI the amounts applicable to specific projects and project areas, if so indicated by Customer.
- 33.5 [Reserved for Future Use]

FERC Docket: RP10-77-000

Second Revised Sheet No. 595 Second Revised Sheet No. 595

Superseding: First Revised Sheet No. 595

GENERAL TERMS AND CONDITIONS (continued)

33.6 Voluntary Contribution Mechanism. Algonquin will include in invoices to Customers a voluntary contribution mechanism allowing Customers to make voluntary contributions to GRI in such amounts and for such GRI project(s) or project area(s) as Customers specify. Customers may, but are not required to, make such voluntary contributions. The voluntary contribution mechanism is not a pipeline rate, rate provision, or term or condition of service. It is included in the invoice in accordance with the January 21, 1998, Stipulation and Agreement Concerning GRI Funding approved by the Commission in Gas Research Institute, 83 F.E.R.C. Par. 61,093 (1998, order on Rehearing, 83 F.E.R.C. Par. 61,331 (1998). Any amounts remitted to Algonquin pursuant to the voluntary contribution mechanism shall be forwarded to GRI by Algonquin and are not refundable by Algonquin.

34.ANNUAL CHARGE ADJUSTMENT CLAUSE PROVISION

- Purpose. Annual charges are assessed against gas pipelines by the Federal Energy Regulatory Commission under Part 382 of the Commission's regulations prior to each fiscal year in order to cover the cost of the operation of the FERC. For the purpose of recovering annual charges assessed Algonquin by the FERC, this Section 34 establishes pursuant to Section 154.402 of the Commission's regulations an Annual Charge Adjustment ("ACA") Surcharge to be applicable to quantities sold or transported under all of Algonquin's rate schedules. This ACA Surcharge is in addition to any amounts otherwise payable to Algonquin under its rate schedules.
- 34.2 Rate Schedules Subject to ACA Surcharge. The ACA Surcharge shall be applicable to the quantities delivered for the account of Customer under all of Algonquin's rate schedules, except the AFT-CL Customer will only be assessed the ACA Surcharge for gas volumes transported under an AFT-CL Rate Schedule when the AFT-CL Customer does not pay an ACA Surcharge under another rate schedule for the same gas volumes transported..
- 34.3 Basis of the ACA Surcharge. The ACA Surcharge shall be that increment, adjusted to Algonquin's measurement basis (MMBtu), which has been approved by FERC orders approving an annual charge unit rate.
- 34.4 Filing Procedure. The initial ACA Surcharge or any subsequent changes in such surcharge shall be filed by Algonquin at least thirty days prior to the proposed effective date unless the establishment by the FERC of the annual charge unit rate for a particular fiscal year prohibits Algonquin from making a timely filing, whereupon the FERC's notice requirements under 18 CFR Section 154.1(c), 154.207 shall be deemed waived for purposes of this Section 34.

The proposed effective date of filings pursuant to this subsection shall be the effective date prescribed by the Federal Energy Regulatory Commission. Such filing shall become effective without suspension or refund obligation.

FERC Docket: RP05-208-000

First Revised Sheet No. 596 First Revised Sheet No. 596 : Effective Superseding: Original Sheet No. 596

GENERAL TERMS AND CONDITIONS (continued)

35.[Reserved for Future Use]

FERC Docket: RP05-208-000

First Revised Sheet No. 597 First Revised Sheet No. 597 : Effective Superseding: Original Sheet No. 597

Sheet No. 597 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 598 First Revised Sheet No. 598 : Effective Superseding: Original Sheet No. 598

Sheet No. 598 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 599 First Revised Sheet No. 599: Effective Superseding: Original Sheet No. 599

Sheet No. 599 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 600 First Revised Sheet No. 600: Effective Superseding: Original Sheet No. 600

Sheet No. 600 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 601 First Revised Sheet No. 601: Effective Superseding: Original Sheet No. 601

Sheet No. 601 was previously issued, but is now reserved for future use.

FERC Docket: RP10-77-000

Second Revised Sheet No. 602 Second Revised Sheet No. 602

Superseding: First Revised Sheet No. 602

GENERAL TERMS AND CONDITIONS (continued)

36.PERIODIC RATE ADJUSTMENTS

Algonquin and Customer recognize that Algonquin will from time to time experience changes in costs related to providing service under this tariff, including, but not limited to, changes in the cost of labor, benefits, materials and supplies, taxes, required rate of return, costs associated with the resolution of past disputes or outstanding uncertainties concerning amounts owed by Algonquin or Customer or attributable to Algonquin or Customer, and costs generated by decisions of the Commission, the courts or by an arbitration panel or other body having jurisdiction over Algonquin. The parties further recognize that it may be appropriate, equitable and consistent with cost responsibility to allocate such costs among Customers based on or taking into account past period factors, such as contract demand levels, throughput or other factors related to a prior period of time. Customer agrees that Algonquin shall have the right from time to time to make rate change filings which may include such costs and utilize an allocation methodology based in whole or in part on factors related to past periods. Customer shall have the right to intervene and protest any such filing.

37. [Reserved for Future Use]

FERC Docket: RP05-208-000

First Revised Sheet No. 603 First Revised Sheet No. 603: Effective Superseding: Original Sheet No. 603

Sheet No. 603 was previously issued, but is now reserved for future use.

FERC Docket: RP05-208-000

First Revised Sheet No. 604 First Revised Sheet No. 604: Effective Superseding: Original Sheet No. 604

Sheet No. 604 was previously issued, but is now reserved for future use.

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 605 Fourth Revised Sheet No. 605

Superseding: Third Revised Sheet No. 605

GENERAL TERMS AND CONDITIONS (continued)

- 38. STANDARDS OF CONDUCT COMPLIANCE PROCEDURES
 - 38.1 Uniform Application. All terms and conditions contained in this tariff shall be applied in a uniform and nondiscriminatory manner without regard to affiliation of any entity to Algonquin.
 - 38.2 Comparability of Service. Algonquin shall provide service under Rate Schedules AFT-1, AFT-E, AFT-1S, AFT-ES, AFT-CL, PAL, TTT and AIT-1 on a basis that is equal in quality for all gas supplies transported or otherwise nominated under that rate schedule.
 - 38.3 Informational Postings. All information required to be posted pursuant to the Commission's currently effective Standards of Conduct regulations will be provided on Algonquin's Internet Web Site under Informational Postings. Such information will be updated as required by applicable regulation(s) issued by the Commission.

FERC Docket: RP10-77-000

Third Revised Sheet No. 606 Third Revised Sheet No. 606

Superseding: Second Revised Sheet No. 606

GENERAL TERMS AND CONDITIONS (continued)

39. COMPLAINT PROCEDURES

Complaints concerning Algonquin's compliance with Algonquin's Standards of Conduct procedures should be communicated to Algonquin with a designation that it is a Standards of Conduct regulation complaint and shall contain a clear and complete statement of the nature and basis of the complaint, together with supporting documentation, if any. Information regarding the appropriate contact personnel shall be posted on Algonquin's Internet Web Site. Algonquin shall respond initially within forty-eight (48) hours and in writing within thirty (30) days to such complaints. In the event the required date of Algonquin's response falls on a Saturday, Sunday, or a holiday that affects Algonquin, Algonquin shall respond by the next Business Day.

FERC Docket: RP09-1063-000

Sixth Revised Sheet No. 607 Sixth Revised Sheet No. 607

Superseding: Fifth Revised Sheet No. 607

GENERAL TERMS AND CONDITIONS (continued)

- 40. ELECTRONIC COMMUNICATION
- 40.1 System Description
 - (a) Algonquin provides for interactive electronic communications with its Customers and other parties through the LINK® Customer Interface System (hereinafter called the "LINK® System"). The LINK® System shall be available on a nondiscriminatory basis to any party (such party is referred to herein as the "LINK® System Subscriber"), provided that such party (i) has a currently effective Valid Service Agreement, has executed a LINK® System Agreement prior to March 11, 2009, or has executed a LINK® System Agreement electronically via the LINK® System on or after March 11, 2009, (ii) has established its business entity in the LINK® System by submitting Contact Information pursuant to Section 40.4(a) below, and (iii) has designated a Local Security Administrator pursuant to Section 40.3 below. A party to a LINK® System Agreement is responsible for ensuring that the individual executing such agreement on its behalf has the appropriate authority. Use of the LINK® System by such individual is acknowledgement of that authority. Algonquin shall not be responsible for verifying the authority of an individual to execute a LINK® System Agreement on behalf of a party. For purposes of this Section 40 and the form of LINK® System Agreement only, a "Valid Service Agreement" includes:
 - (1) any service agreement pursuant to any of Algonquin's Rate Schedules;
 - (2) any service agreement pursuant to any of the Rate Schedules of East Tennessee Natural Gas, LLC;
 - (3) any service agreement pursuant to any of the Rate Schedules of Egan Hub Storage, LLC;
 - (4) any contract between Customer and Moss Bluff Hub, LLC;
 - (5) any service agreement pursuant to any of the Rate Schedules of Ozark Gas Transmission, L.L.C.;
 - (6) any service agreement pursuant to any of the Rate Schedules of Saltville Gas Storage Company L.L.C.;
 - (7) any service agreement pursuant to any of the Rate Schedules of Texas Eastern Transmission, LP;
 - (8) a capacity release umbrella agreement between Customer and any of the entities identified herein.

For parties with a LINK® System Agreement dated prior to September 22, 2004, such LINK® System Agreement is hereby superseded by the provisions of this Section 40.

By accessing the LINK® System, LINK® System Subscriber agrees to comply with the procedures for access to and use of the LINK® System as set forth in this Section 40.

Algonquin reserves the right to implement enhancements to the LINK® System at its sole discretion; provided however, all such enhancements when fully operational shall be available to all LINK® System Subscribers. Algonquin will exercise due diligence to ensure the LINK® System operates correctly and will provide timely and non-discriminatory access to on-line LINK® System help features and to any information available on the LINK® System that LINK® System Subscriber is entitled to access.

FERC Docket: RP10-77-000

Third Revised Sheet No. 608 Third Revised Sheet No. 608

Superseding: Second Revised Sheet No. 608

GENERAL TERMS AND CONDITIONS (continued)

- 40. ELECTRONIC COMMUNICATION
- 40.1 System Description (cont'd)
 - (b) The LINK® System provides on-line help, a search function that permits a LINK® Subscriber to locate information concerning a specific transaction, and menus that permit LINK® System Subscribers to separately access notices of available capacity, records in the transportation request log, and standards of conduct information. The LINK® System will permit a LINK® System Subscriber to electronically download information on transactions from the LINK $\scriptsize{\scriptsize{(B)}}$ System and to separate extremely large documents into smaller files prior to such download. Algonquin shall maintain and retain daily back-up records of the information displayed on the LINK® System and the Web Site and through electronic data interchange for three years and shall permit LINK® System Subscriber to review those records upon request. Completed transactions will remain on the LINK® System for at least ninety days after completion and will then be archived. Archived information will be made available by Algonquin if possible within two weeks after receipt of a Customer's request for such information. Information on the most recent entries will appear ahead of older information.
 - (c) Customers' Notices pursuant to Section 14 of the General Terms and Conditions shall be submitted electronically and, in addition, posted electronically by the Customer via the LINK® System. Electronic communications may also be transmitted, where applicable, via electronic data interchange, which will be available on a nondiscriminatory basis to any LINK® System Subscriber, provided such LINK® System Subscriber has entered into a trading partner agreement with Algonquin, in addition to the agreements specified in Section 40.1(a) above. Specifically, a LINK® System Subscriber has the option of utilizing the LINK® System for purposes of:
 - (a) requesting service under Algonquin's rate schedules set forth in Volume No. 1 of Algonquin's FERC Gas Tariff;
 - (b) executing, tracking and amending certain service agreements under Algonquin's rate schedules set forth in Volume No. 1 of Algonquin's FERC Gas Tariff;
 - (c) providing nominations and viewing allocations and operational imbalances under all rate schedules as a Customer of Algonquin pursuant to the applicable rate schedule and the General Terms and Conditions;
 - (d) exercising its rights as a Customer of Algonquin pursuant to Section 9 of the General Terms and Conditions or submitting a bid as a Replacement Customer of Algonquin under such section;
 - (e) exercising its rights as a Customer of Algonquin pursuant to Section 14 of the General Terms and Conditions (which if submitted utilizing the LINK® System will be posted at that time) or submitting a bid as a Replacement or Prearranged Customer of Algonquin pursuant to such section, or posting a Capacity Request for capacity release pursuant to such section;
 - (f) viewing those Point(s) of Receipt and Points(s) of Delivery that are subject to an Operational Balancing Agreement pursuant to Section 30 of the General Terms and Conditions;
 - (g) viewing and downloading operational data for any Gas Day on the second subsequent Gas Day;
 - (h) viewing Algonquin's notice of an OFO as contemplated by Section 29 of the General Terms and Conditions;
 - effectuating Imbalance Netting and Trading pursuant to Sections 25.8 and 25.9 of the General Terms and Conditions;
 - (j) requesting a discount of the maximum Recourse Rate(s) for service under Algonquin's Open-access Rate Schedules or viewing such discounts
 - previously granted; and
 (k) such other functions as may be available on the LINK® System from time to time.

FERC Docket: RP10-77-000

Fourth Revised Sheet No. 608A Fourth Revised Sheet No. 608A

Superseding: Third Revised Sheet No. 608A

GENERAL TERMS AND CONDITIONS (continued)

40. ELECTRONIC COMMUNICATION

40.2 Information. Algonquin shall post at least four times a day on the LINK® System and the Web Site information relevant to the availability of firm and interruptible capacity at Points of Receipt, on the mainline, and at Points of Delivery. The LINK® System and the Web Site will indicate whether the capacity is available from Algonquin directly or through Algonquin's capacity release mechanism as set forth in Section 14 of the General Terms and Conditions. The LINK® System and the Web Site shall provide the best available information about imbalances on an hourly and a daily basis. The LINK® System and the Web Site also include information allowed or required to be posted thereon by other provisions of the tariff including Section 14, information that Algonquin is required to post pursuant to the Commission's regulations, or other information Algonquin chooses to post in furtherance of the operation of its system.

40.3 Local Security Administrators

- (a) LINK® System Subscriber shall designate one or more persons to perform certain security functions on the LINK® System ("Local Security Administrator") by submitting for each such person the Local Security Administrator Designation information via the LINK® System using the applicable on-line form, as such form is amended from time to time in the LINK® System. LINK® System Subscriber shall update Local Security Administrator Designation information via the LINK® System as such information changes.
- (b) The Local Security Administrator shall, via the LINK® System, be responsible for (1) identifying those persons who are duly authorized by LINK® System Subscriber to use the LINK® System to perform one or more of the functions available on the LINK® System ("LINK® System User"); (2) providing LINK® System Users with individualized USERIDs and passwords; (3) maintaining LINK® System Users' account information; (4) adding and terminating LINK® System Users immediately upon a change in status requiring such addition or termination; (5) creating and modifying security rights for LINK® System Users; (6) approving or terminating Designation of Affiliated Companies information and Designation of Agency information pursuant to Sections 40.5 and 40.6, respectively; and (7) ensuring that USERIDs are used only as appropriate and as contemplated by these General Terms and Conditions and the LINK® System Agreement.
- (c) Algonquin shall be entitled to rely upon the representation of the LINK® System Subscriber's Local Security Administrator that the LINK® System User(s) identified by the Local Security Administrator may (i) transmit information to Algonquin; (ii) view information posted on the LINK® System; and/or (iii) perform the LINK® System contracting function in accordance with the security rights granted by Local Security Administrator.

FERC Docket: RP09-348-000

Fifth Revised Sheet No. 608B Fifth Revised Sheet No. 608B

Superseding: Fourth Revised Sheet No. 608B

GENERAL TERMS AND CONDITIONS (continued)

- 40. ELECTRONIC COMMUNICATION
- 40.4 Authorized Use of the LINK® System; Confidentiality (cont'd)
 - (c) LINK® System Subscriber shall be solely responsible for any unauthorized or otherwise improper use of USERIDs and passwords issued by or for its Local Security Administrator, including, but not limited to, the use of such USERIDs and passwords by LINK® System Users who are not within LINK® System Subscriber's employment or control.
 - (d) Algonquin reserves the right to disable for due cause any USERID issued to any LINK® System User. Algonquin shall provide notice to LINK® System Subscriber, LINK® System User and/or Local Security Administrator, as applicable, at the time that the USERID is disabled by Algonquin. In addition, upon thirty (30) days prior notice to the LINK® System User and the Local Security Administrator, Algonquin will disable any USERID that has not been used to access the LINK® System for fifteen (15) consecutive months.
 - (e) LINK® System Subscriber shall immediately notify Algonquin of the desire to delete a Local Security Administrator of LINK® System Subscriber by (i) e-mail to link-help@spectraenergy.com, or (ii) submission via the LINK® System using the applicable on-line form of revised Local Security Administrator Designation information for such Local Security Administrator indicating the desire for termination. Such revised information shall supersede in its entirety any Local Security Administrator Designation information previously submitted to Algonquin for such Local Security Administrator. LINK® System Subscriber shall be solely responsible for any unauthorized actions of Local Security Administrator due to LINK® System Subscriber's failure to so notify Algonquin of the need to delete such Local Security Administrator.
 - (f) Algonquin warrants that, without the express consent of LINK® System Subscriber or as otherwise provided in this FERC Gas Tariff, no Algonquin employee or agent will disclose to any third party any non-public information regarding research performed through the use of the LINK® System by LINK® System Subscriber.
 - (g) Any manual forms received by Algonquin prior to September 1, 2007 related to Contact Information, Designation of Agency, or Designation of Affiliated Companies, and any Local Security Administrator Designation Forms received by Algonquin prior to March 11, 2009, shall remain in full force and effect until the earlier of (i) termination by the LINK® System Subscriber or (ii) receipt of superseding information submitted pursuant to this Section 40.
- 40.5 LINK® System Subscriber; Affiliated Companies
 - (a) If LINK® System Subscriber belongs to a group of affiliated companies and requires LINK® System access on behalf of one or more of said affiliates, LINK® System Subscriber (i) shall, or shall cause one of the affiliates of LINK® System Subscriber to, submit to Algonquin via the LINK® System the Designation of Affiliated Companies information, and (ii) shall cause all other parties included in the affiliation to approve the Designation of Affiliated Companies information via the LINK® System. The Designation of Affiliated Companies information shall be submitted and approved via the applicable on-line form, as such form is updated from time to time in the LINK® System. The submission pursuant to item (i) herein shall be deemed to be the submitting party's approval of the Designation of Affiliated Companies information.
 - (b) When Designation of Affiliated Companies information changes, the LINK® System Subscriber shall cause revised Designation of Affiliated Companies information to be submitted and approved pursuant to Section 40.5(a) above. Such revised information shall supersede in its entirety any Designation of Affiliated Companies information previously submitted to Algonquin. LINK® System

FERC Docket: RP07-553-000

Second Revised Sheet No. 608C Second Revised Sheet No. 608C : Effective

Superseding: First Revised Sheet No. 608C

GENERAL TERMS AND CONDITIONS (continued)

40. ELECTRONIC COMMUNICATION

Subscriber warrants that access consistent with any Designation of Affiliated Companies information submitted and approved by LINK® System Subscriber and its affiliates in accordance with Section 40.5(a) above or manually submitted prior to September 1, 2007, pursuant to Section 40.4(g) above is appropriate and authorized. Determining the propriety of such access is the responsibility of LINK® System Subscriber and/or its affiliates, but Algonquin reserves the right to reject such Designation of Affiliated Companies information if it determines that granting such designation would violate any contractual, legal, or regulatory responsibility of Algonquin.

- (c) In order for LINK® System Users of LINK® System Subscriber to access the LINK® System on behalf of LINK® System Subscriber's affiliates designated pursuant Section 40.5(a) above, LINK® System Subscriber and each designated affiliate of LINK® System Subscriber must meet the requirements of a LINK® System Subscriber set forth in Section 40.1(a) of these General Terms and Conditions.
- (d) It is the obligation of the LINK® System Subscriber to notify Algonquin via the LINK® System when a company affiliation terminates, either by (i) submitting a request to terminate a company affiliation via the applicable on-line form, as such form is updated from time to time in the LINK® System, or (ii) submitting and approving superseding Designation of Affiliated Companies information in accordance with Section 40.5(a). An affiliate may request a termination of the company affiliation by submitting such request via the LINK® System. A request to terminate a company affiliation will be processed by Algonquin without consent from the non-requesting party.

40.6 LINK® System Subscriber; Agency

- (a) If LINK® System Subscriber desires to designate one or more persons or entities to act as an agent on behalf of LINK® System Subscriber ("Agent"), then for each such Agent, the LINK® System Subscriber (i) shall, or shall cause the Agent to, submit to Algonquin via the LINK® System the Designation of Agency information specifying the rights granted to the Agent and (ii) shall cause the other party to the agency relationship to approve the Designation of Agency information. The Designation of Agency information shall be submitted and approved via the applicable on-line form, as such form is updated from time to time in the LINK® System. The submission pursuant to item (i) herein shall be deemed to be the submitting party's approval of the information. Algonquin may require that LINK® System Subscriber provide additional documentation to confirm that LINK® System Subscriber desires Agent to act on its behalf.
- (b) In order for LINK® System Users of an Agent designated pursuant to Section 40.6(a) above to access the LINK® System on behalf of LINK® System Subscriber, such Agent must meet the requirements of a LINK® System Subscriber set forth in Section 40.1(a) of these General Terms and Conditions.
- (c) Algonquin may accept and fully rely upon Designation of Agency information submitted and approved in accordance with Section 40.6(a) above or manually submitted prior to September 1, 2007, in accordance with Section 40.4(g) above. Algonquin may fully rely upon all communications received from and direction given by Agent with respect to all actions indicated in the approved Designation of Agency information for which Agent is authorized to act on behalf of LINK® System Subscriber. Algonquin may grant Agent access to LINK® System Subscriber's data contained in the LINK® System as necessary to perform the functions identified in the approved Designation of Agency information. LINK® System Subscriber will defend, indemnify and hold harmless Algonquin from and against any and all claims, demands, liabilities and/or actions, and/or any

FERC Docket: RP07-553-000

First Revised Sheet No. 608D First Revised Sheet No. 608D: Effective

Superseding: Sub Original Sheet No. 608D

GENERAL TERMS AND CONDITIONS (continued)

40. ELECTRONIC COMMUNICATION

and all resulting loss, costs, damages, and/or expenses (including court costs and reasonable attorney's fees) of any nature whatsoever, that may be asserted against or imposed upon Algonquin by any party associated with Algonquin's reliance on Designation of Agency information provided pursuant to this Section 40.6.

(d) The rights specified in the approved Designation of Agency information having the latest commencement date shall supersede all prior rights granted by LINK® System Subscriber to Agent. In no event can an agency right granted to one Agent be simultaneously granted to another Agent.

It is the obligation of the LINK® System Subscriber to notify Algonquin when an agency relationship changes or terminates, either by (i) specifying a termination date in the approved Designation of Agency information, (ii) submitting a request to terminate an agency relationship via the LINK® System using the applicable on-line form, as such form is updated from time to time in the LINK® System, or (iii) submitting and approving superseding Designation of Agency information in accordance with Section 40.6(a). The Agent may request a termination of the agency relationship by submitting such request via the LINK® System. A request to terminate an agency relationship will be processed by Algonquin without consent from the non-requesting party.

LINK® System Subscriber and Agent must re-approve existing Designation of Agency information via the LINK® System using the applicable on-line form, as such form is updated from time to time in the LINK® System, on an annual basis. If, during this annual re-approval process, either the LINK® System Subscriber or the Agent desires a change to the Designation of Agency information, new Designation of Agency information must be submitted and approved in accordance with Section 40.6(a) above. Algonquin shall remove the security rights granted to all LINK® System Users of Agent pertaining to access granted by LINK® System Subscriber pursuant to the Designation of Agency information if LINK® System Subscriber and Agent do not re-approve the existing Designation of Agency information on an annual basis.

(e) Agent is authorized to act on behalf of LINK® System Subscriber under any or all of LINK® System Subscriber's Service Agreements with Algonquin as such Service Agreements are effective from time to time, or with respect to any or all meter locations as available from time to time, respectively, as specified in the Designation of Agency information, until LINK® System Subscriber properly notifies Algonquin that the agency relationship is terminated or superseded in accordance with Section 40.6(d). The designation of an Agent by a LINK® System Subscriber does not provide for an assignment of the rights and obligations of any Service Agreement between Algonquin and LINK® System Subscriber.

40.7 Liability

(a) Algonquin shall not be liable to LINK® System Subscriber nor any other party in damages for any act, omission or circumstance related to the LINK® System occasioned by or in consequence of an event of Force Majeure as defined in Section 16 of these General Terms and Conditions, that is not within the control of Algonquin and which by the exercise of due diligence Algonquin is unable to prevent or overcome. To the extent the information displayed on the LINK® System is originated solely by Algonquin and such information is subsequently determined to be inaccurate, LINK® System Subscriber shall not be subject to any penalties otherwise collectable by Algonquin based on Customer conduct attributable to such inaccuracy during the period the inaccurate information was displayed on the LINK® System.

FERC Docket: RP09-97-000

Second Revised Sheet No. 608E Second Revised Sheet No. 608E

Superseding: First Revised Sheet No. 608E

GENERAL TERMS AND CONDITIONS (continued)

40. ELECTRONIC COMMUNICATION

- (b) LINK® System Subscriber shall defend, indemnify and hold harmless Algonquin from and against any and all claims, demands and/or actions, and any and all resulting loss, costs, damages, and/or expenses (including court costs and reasonable attorney's fees) of any nature whatsoever, that may be asserted against or imposed upon Algonquin by any party as a result of the unauthorized or otherwise improper use of any USERID and/or password issued to or by LINK® System Subscriber and/or Local Security Administrator or any other unauthorized or improper use of the LINK® System by any LINK® System User or LINK® System Subscriber unless such improper use is the result of Algonquin's negligence or willful misconduct, including, but not limited to, distribution of USERIDs or passwords to persons that are not employed by, or agents or affiliates of, LINK® System Subscriber.
- 40.8 Electronic Mail (E-mail) Notification. For system-wide notices of general applicability, any provisions of this FERC Gas Tariff requiring that these matters be written or in writing are satisfied by Algonquin utilizing electronic transmission through the LINK® System in accordance with the procedures for utilization of the LINK® System or through electronic data interchange as provided for in Commission-approved or permitted data sets. Critical system-wide notices will be in a separate category from notices that are not critical. Algonquin will use electronic mail (e-mail) in order to facilitate certain notifications to Customers as required by this FERC Gas Tariff. Customer shall provide Algonquin with at least one e-mail address to which these notifications can be sent, and shall be responsible for updating such information as necessary. In addition to the requirement specified in Sections 24.3, 26.6 and 29.3 of these General Terms and Conditions to post notices on the LINK® System, Algonquin shall provide such notifications via email communication to those Customers that have provided such e-mail address information and have requested, via the LINK® System, e-mail notification of critical notices issued by Algonquin. Customer shall be responsible for providing accurate e-mail notification information to Algonquin, including timely updates to such information as necessary. All other provisions, including service agreement-specific notices, requiring items or information to be written or in writing remain unchanged unless otherwise agreed by Algonquin and Customer.
- Algonquin or an affiliate of Algonquin is the exclusive proprietor of the programming that generates the LINK® System and of all the copyrights and proprietary interests therein, except insofar as any third party (whose materials are made available in the files of the LINK® System under license to Algonquin or an affiliate of Algonquin) possesses a copyright or proprietary interest in such materials, but not of the files of and the information displayed on the LINK® System. A LINK® System Subscriber will not by virtue of this Section 40 or the executed LINK® System Agreement acquire any proprietary interests in the programming that generates the LINK® System. As part of Algonquin's continuing development of the LINK® System, third parties may desire to acquire rights to certain parts of the programming. Algonquin will attempt to accommodate, when appropriate, requests by LINK® System Subscribers to license these rights for use in the LINK® System.

FERC Docket: RP09-348-000

Original Sheet No. 608A.01 Original Sheet No. 608A.01

GENERAL TERMS AND CONDITIONS (continued)

40. ELECTRONIC COMMUNICATION

40.4 Authorized Use of LINK® System; Confidentiality

- (a) LINK® System Subscriber shall submit Contact Information to Algonquin via the LINK® System using the applicable on-line form, as such form is updated from time to time in the LINK® System. In addition, LINK® System Subscriber shall be required to submit updated Contact Information to Algonquin via the LINK® System as such information changes. Such revised information shall supersede in its entirety any Contact Information previously submitted to Algonquin.
- (b) LINK® System Subscriber shall not disclose to persons other than Local Security Administrator and LINK® System Users that are employed by LINK® System Subscriber, or properly designated affiliates or agents of LINK® System Subscriber, and shall otherwise keep confidential, all USERIDs and passwords issued by Local Security Administrator. In addition, LINK® System Subscriber shall cause Local Security Administrator and LINK® System User(s) to refrain from disclosing to any other person, whether or not employed by LINK® System Subscriber, and shall otherwise keep confidential, the individualized USERID and password issued to each such LINK® System User.

FERC Docket: RP10-146-000

Ninth Revised Sheet No. 609 Ninth Revised Sheet No. 609

Superseding: Eighth Revised Sheet No. 609

GENERAL TERMS AND CONDITIONS (continued)

- 41. REVENUE CREDITS
- 41.1 Interruptible Transportation and Park and Loan Service
 - Applicability. The credit available under this Section 41.1 for revenues under Rate Schedules AIT-1 and PAL shall apply to all Service Agreements under firm Part 284 transportation service rate schedules, except for Service Agreements for service on the Brayton Point Lateral, the Manchester Street Lateral, the Canal Lateral, the Cape Cod Lateral, the Northeast Gateway Lateral, the J-2Facility, and the Kleen Energy Lateral under Rate Schedule AFT-CL. The credit available under this Section 41 for revenues under Rate Schedule AIT-2 attributable to service on the Brayton Point Lateral shall apply only to Service Agreements for service on the Brayton Point Lateral under Rate Schedule AFT-CL. The credit available under this Section 41 for revenues under Rate Schedule AIT-2 attributable to service on the Manchester Street Lateral shall apply only to Service Agreements for service on the Manchester Street Lateral under Rate Schedule AFT-CL. The credit available under this Section 41 for revenues under Rate Schedule AIT-2 attributable to service on the Canal Lateral shall apply only to Service Agreements for service on the Canal Lateral under Rate Schedule AFT-CL. The credit available under this Section 41 for revenues under Rate Schedule AIT-2 attributable to service on the Cape Cod Lateral shall apply only to Service Agreements for service on the Cape Cod Lateral under Rate Schedule AFT-CL. The credit available under this Section 41 for revenues under Rate Schedule AIT-2 attributable to service on the Northeast Gateway Lateral shall apply only to Service Agreements for service on the Northeast Gateway Lateral under Rate Schedule AFT-CL. The credit available under this Section 41 for revenues under Rate Schedule AIT-2 attributable to service on the J-2 Facility shall apply only to Service Agreements for service on the J-2 Facility under Rate Schedule AFT-CL. The credit available under this Section 41 for revenues under Rate Schedule AIT-2 attributable to service on the Kleen Energy Lateral shall apply only to Service Agreements for service on the Kleen Energy Lateral under Rate Schedule AFT-CL.
 - (b) Basis of the Credit. Revenues to which the credit under this Section 41.1 shall apply ("Eligible Revenues") shall be the revenues actually received by Algonquin under Rate Schedules AIT-1 and PAL or Rate Schedule AIT-2 that are attributable to commodity charges and authorized overrun charges but not including imbalance resolution charges, scheduling penalties, unauthorized contract overrun penalties, GRI, ACA, revenues attributable to Section 8 of Rate Schedule PAL, or any other such charges or surcharges.
 - (c) Percentage of Eligible Revenues.
 - (i) Rate Schedules AIT-1 and PAL. Beginning with the Month following the Month in which Algonquin's cumulative annual revenues stemming from commodity charges and authorized overrun charges under Rate Schedules AIT-1 and PAL exceed the dollar amount allocated to Rate Schedules AIT-1 and PAL in Algonquin's currently effective rates ("Excess Eligible Revenues"), Algonquin shall credit to current Month invoices under the applicable rate schedules 90% of the Excess Eligible Revenues received during the prior Month. For purposes of determining cumulative annual revenues under Rate Schedules AIT-1 and PAL, the annual period shall commence on May 1 of each year. In the event that two or more maximum rates for service under Rate Schedules AIT-1 and PAL are applicable during the same twelve month

FERC Docket: RP10-77-000

Second Revised Sheet No. 610 Second Revised Sheet No. 610

Superseding: First Revised Sheet No. 610

GENERAL TERMS AND CONDITIONS (continued)

41. REVENUE CREDITS (continued)

period beginning on May 1, the threshold for revenue crediting hereunder shall consist of a time-weighted average of the amounts allocated to Rate Schedules AIT-1 and PAL during the periods when the various AIT-1 and PAL rates are in effect. Algonquin shall retain the remainder of the Eligible Revenues and Excess Eligible Revenues not required to be credited or refunded.

- (ii) Rate Schedule AIT-2. Beginning with the Month following the Month in which Algonquin's cumulative annual revenues stemming from commodity charges and authorized overrun charges under Rate Schedule AIT-2 exceed the dollar amount allocated to Rate Schedule AIT-2 in Algonquin's currently effective rates ("Excess Eligible Revenues"), if any, Algonquin shall credit to current Month invoices under the applicable rate schedules 50% of the Excess Eligible Revenues received during the prior Month. For purposes of determining cumulative annual revenues under Rate Schedule AIT-2, the annual period shall commence on May 1 of each year. In the event that two or more maximum rates for service under Rate Schedule AIT-2 are applicable during the same twelve month period beginning on May 1, the threshold for revenue crediting hereunder shall consist of a time-weighted average of the amounts allocated to Rate Schedule AIT-2 during the periods when the various AIT-2 rates are in effect. Algonquin shall retain the remainder of the Eligible Revenues and Excess Eligible Revenues not required to be credited or refunded.
- (d) Apportionment of Eligible Revenues. Eligible Revenues attributable to a Month shall be apportioned among all Customers under the applicable rate schedules by applying the following ratio for each Customer: (a) the Customer's total MDTQ in effect during that Month under the applicable rate schedules (b) divided by the summation of the total MDTQs in effect during that Month for all Customers under the applicable rate schedules; provided, however, that no Customer shall receive a credit under this Section 41.1 in excess of its reservation charges under the applicable rate schedules for that Month, with such excess being reallocated to the other Customers under the applicable rate schedule in accordance with the above ratios.
- (e) Credits Subject to Refund. In the event that any revenues credited pursuant to Section 41.1 are subject to refund, and are ultimately required to be refunded to Customers under Rate Schedules AIT-1 and PAL or Rate Schedule AIT-2, Algonquin shall recalculate the revenue credits that would have been due to Customer if the AIT-1 and PAL rates or AIT-2 rate, respectively, used for purposes of the refund computation

FERC Docket: RP10-77-000

First Revised Sheet No. 611 First Revised Sheet No. 611

Superseding: Original Sheet No. 611

GENERAL TERMS AND CONDITIONS (continued)

41. REVENUE CREDITS (continued)

had been in effect at the time of the required credit, and shall bill Customer for the differences between revenues actually credited, and the recalculated revenue credit, plus interest at the rate prescribed by the Commission's regulations.

- 41.2 Retained Upstream Capacity Release Credits.
 - (a) Applicability. The credit available under this Section 41.2 shall apply to all firm rate schedules except Rate Schedule AFT-CL.
 - (b) Basis of the Credit. Revenues to which the credit under this Section 41.2 shall apply ("Eligible Capacity Release Revenues") shall be the revenues actually received by Algonquin that are attributable to demand charges paid by parties to which Algonquin has released upstream capacity rights held by Algonquin, to the extent that the costs of such rights have been reflected in the rates paid by Customers under the rate schedules identified in Section 41.2(a).
 - (c) Apportionment of Eligible Capacity Release Revenues. Eligible Revenues attributable to a Month shall be apportioned among all Customers under the applicable rate schedules by applying the following ratio for each Customer: (i) the Customer's total MDTQ in effect during that Month under the applicable rate schedules, (ii) divided by the summation of the total MDTQs in effect during that Month for all Customers under the applicable rate schedules, provided that, in the case of a Customer paying less than the maximum reservation charge during any Month, the Customer's MDTQ for purposes of both (i) and (ii) above shall be reduced in the same proportion as the reservation charge paid bears to the maximum reservation charge.

FERC Docket: RP10-308-000

First Revised Sheet No. 612 First Revised Sheet No. 612

Superseding: Original Sheet No. 612

GENERAL TERMS AND CONDITIONS (continued)

- 42. FEES AND CONSTRUCTION OF NEW FACILITIES
 - 42.1 In addition to any other charges due under the applicable rate schedule, Customer shall reimburse Algonquin for the following:
 - (a) The costs of any facilities installed by Algonquin with Customer's consent to receive, measure, transport or deliver natural gas for the account of Customer; and
 - (b) Any and all filing and approval fees required in connection with Customer's service agreement that Algonquin is obligated to pay to the FERC or any other governmental authority having jurisdiction.

Any reimbursement due Algonquin by Customer pursuant to this Section 42.1 shall be due and payable to Algonquin within ten days of the date of Algonquin's invoice(s) for same.

FERC Docket: RP06-519-000

First Revised Sheet No. 613 First Revised Sheet No. 613: Effective

Superseding: Original Sheet No. 613

GENERAL TERMS AND CONDITIONS (continued)

43. REPORTS WITH RESPECT TO TARIFF

- 43.1 Fuel Reimbursement Quantity. On an annual basis, Algonquin files its Fuel Reimbursement Quantity ("FRQ") report in accordance with Section 32 of the General Terms and Conditions. The FRQ report contains: (1) the revised Fuel Reimbursement Percentages ("FRPs") for the calendar year beginning in December, (2) a workpaper supporting the revised fuel requirements, (3) the calculation of the FRQ deferral allocation, (4) workpapers showing the monthly accrual of the FRQ deferred balance for the period of August 1, through July 31 of each year, (5) the computation of each Customer's surcharge for the FRQ Deferred Account balance as of July 31, of each year plus carrying charges and (6) the calculation of the monthly carrying charges through November 30, of each year.
- Algonquin files its Penalty
 Revenue Disbursement report in accordance with Sections 23.4(e),
 24.8 and 31 of the General Terms and Conditions, on an annual
 basis or within sixty (60) days after the revenue collected
 pursuant to any of those sections reaches the dollar amount
 specified in those sections. The Penalty Revenue Disbursement
 report reflects the amount to be remitted to each Non-offending
 Customer pursuant to each of the referenced sections, as well as
 the applicable interest to be included in the remittance to each
 Non-offending Customer.
- 43.3 Contact Information.

All contact with respect to the reports submitted pursuant to this Section 43 should be made to:

David A. McCallum
Director, Rates and Tariffs
Algonquin Gas Transmission, LLC
5400 Westheimer Court
P. O. Box 1642
Houston, TX 77251-1642
Telephone: (713) 627-5340
Facsimile: (713) 627-5947

FERC Docket: RP09-1063-000

Fourth Revised Sheet No. 614 Fourth Revised Sheet No. 614

Superseding: Third Revised Sheet No. 614

GENERAL TERMS AND CONDITIONS (continued)

44. North American Energy Standards Board ("NAESB")

Algonquin has adopted all of the Business Practices and Electronic Communication Standards which were required by the Commission in 18 CFR ¶ 284.12(b) in accordance with Order Nos. 587, et al., in Docket Nos. RM96-1-000, et seq. In addition to the standards which are reflected in other provisions of this tariff, Algonquin is incorporating herein the following NAESB Wholesale Gas Quadrant ("WGQ") standards, definitions and data sets, Version 1.8, and

```
Recommendation WGQ 2008 Annual Plan Item 10 and Request No. R08026 , where applicable, by
reference:
General Standards:
      0.2.1,\ 0.2.2,\ 0.2.3,\ 0.3.1,\ 0.3.2,\ 0.3.3,\ 0.3.4,\ 0.3.5,\ 0.3.6,\ 0.3.7,\ 0.3.8,\ 0.3.9,
      0.3.10, 0.3.11, 0.3.12, 0.3.13, 0.3.14, 0.3.15, and 0.4.1.
Nominations Related Standards:
      1.2.1, 1.2.2, 1.2.3, 1.2.4, 1.2.5, 1.2.6, 1.2.8, 1.2.9, 1.2.10, 1.2.11, 1.2.12, 1.2.13,
      1.2.18, 1.2.19, 1.3.2(v), 1.3.2(vi), 1.3.3, 1.3.4, 1.3.11, 1.3.14, 1.3.15, 1.3.16, 1.3.17, 1.3.18, 1.3.19, 1.3.20, 1.3.21, 1.3.22, 1.3.23, 1.3.24, 1.3.25, 1.3.26, 1.3.27, 1.3.29, 1.3.30, 1.3.31, 1.3.32, 1.3.33, 1.3.34, 1.3.35, 1.3.36, 1.3.37, 1.3.38, 1.3.39, 1.3.40,
      1.3.41, 1.3.42, 1.3.43, 1.3.44, 1.3.45, 1.3.46, 1.3.47, 1.3.48, 1.3.49, 1.3.50, 1.3.51, 1.3.52, 1.3.53, 1.3.54, 1.3.55, 1.3.56, 1.3.57, 1.3.58, 1.3.59, 1.3.60, 1.3.61, 1.3.62, 1.3.63, 1.3.64, 1.3.65, 1.3.66, 1.3.67, 1.3.68, 1.3.69, 1.3.70, 1.3.71, 1.3.72, 1.3.73,
      1.3.74, 1.3.75, 1.3.76, 1.3.77, 1.3.79, 1.4.1, 1.4.2, 1.4.3, 1.4.4, 1.4.5, 1.4.6 and
      1.4.7.
Flowing Gas Related Standards:
      2.2.1, 2.2.4, 2.2.5, 2.3.1, 2.3.2, 2.3.3, 2.3.4, 2.3.6, 2.3.7, 2.3.8, 2.3.9, 2.3.10,
      2.3.11, 2.3.12, 2.3.13, 2.3.15, 2.3.17, 2.3.19, 2.3.20, 2.3.21, 2.3.22, 2.3.23, 2.3.25, 2.3.27, 2.3.28, 2.3.29, 2.3.30, 2.3.31, 2.3.32, 2.3.33, 2.3.34, 2.3.35, 2.3.48, 2.3.50,
      2.3.51, 2.3.52, 2.3.53, 2.3.54, 2.3.55, 2.3.56, 2.3.57, 2.3.58, 2.3.59, 2.3.60, 2.3.61, 2.3.62, 2.3.63, 2.3.64, 2.3.65, 2.4.1, 2.4.2, 2.4.3, 2.4.4, 2.4.5, 2.4.6, 2.4.7, 2.4.8, 2.4.9, 2.4.10, 2.4.11, 2.4.12, 2.4.13, 2.4.14, 2.4.15, 2.4.16, 2.4.17 and 2.4.18.
```

Invoicing Related Standards: 3.3.1, 3.3.2, 3.3.3, 3.3.4, 3.3.5, 3.3.6, 3.3.7, 3.3.8, 3.3.9, 3.3.10, 3.3.11, 3.3.12, 3.3.13, 3.3.14, 3.3.16, 3.3.20, 3.3.21, 3.3.22, 3.3.23, 3.3.24, 3.3.25, 3.3.26, 3.4.1, 3.4.2, 3.4.3 and 3.4.4.

Quadrant Electronic Delivery Mechanism Standards: 4.2.1, 4.2.2, 4.2.3, 4.2.4, 4.2.5, 4.2.6, 4.2.7, 4.2.8, 4.2.9, 4.2.10, 4.2.11, 4.2.12,

4.2.13, 4.2.14, 4.2.15, 4.2.16, 4.2.17, 4.2.18, 4.2.19, 4.2.20, 4.3.1, 4.3.2, 4.3.3, 4.3.5, 4.3.16, 4.3.17, 4.3.18, 4.3.20, 4.3.22, 4.3.23, 4.3.24, 4.3.25, 4.3.26, 4.3.27 4.3.28, 4.3.29, 4.3.30, 4.3.31, 4.3.32, 4.3.33, 4.3.34, 4.3.35, 4.3.36, 4.3.38, 4.3.39, 4.3.40, 4.3.41, 4.3.42, 4.3.43, 4.3.44, 4.3.45, 4.3.46, 4.3.47, 4.3.48, 4.3.49, 4.3.50, 4.3.51, 4.3.52, 4.3.53, 4.3.54, 4.3.55, 4.3.56, 4.3.57, 4.3.58, 4.3.59, 4.3.60, 4.3.61, 4.3.62, 4.3.65, 4.3.66, 4.3.67, 4.3.68, 4.3.69, 4.3.72, 4.3.73, 4.3.74, 4.3.75, 4.3.76, 4.3.78, 4.3.79, 4.3.80, 4.3.81, 4.3.82, 4.3.83, 4.3.84, 4.3.85, 4.3.86, 4.3.87, 4.3.89, 4.3.90, 4.3.91, 4.3.92, and 4.3.93.

Capacity Release Related Standards:

5.2.1, 5.2.2, 5.3.5, 5.3.10, 5.3.17, 5.3.19, 5.3.21, 5.3.22, 5.3.23, 5.3.29, 5.3.30, 5.3.31, 5.3.32, 5.3.33, 5.3.34, 5.3.35, 5.3.36, 5.3.37, 5.3.38, 5.3.39, 5.3.40, 5.3.41, 5.3.42, 5.3.43, 5.3.46, 5.3.47, 5.3.52, 5.4.1, 5.4.2, 5.4.3, 5.4.4, 5.4.5, 5.4.6, 5.4.7, 5.4.8, 5.4.9, 5.4.10, 5.4.11, 5.4.12, 5.4.13, 5.4.14, 5.4.15, 5.4.16, 5.4.17, 5.4.18, 5.4.19, 5.4.20, 5.4.21, 5.4.22, and 5.4.23.

FERC Docket: RP09-720-000

Original Sheet No. 614A Original Sheet No. 614A

GENERAL TERMS AND CONDITIONS (Continued)

44. NORTH AMERICAN ENERGY STANDARDS BOARD ("NAESB") (continued)

Contracts Related Standards:
 6.3.3 (EDI Trading Partner Agreement)

Internet Electronic Transfer Related Standards: 10.2.1, 10.2.2, 10.2.3, 10.2.4, 10.2.5, 10.2.6, 10.2.7, 10.2.8, 10.2.9, 10.2.10, 10.2.11, 10.2.12, 10.2.13, 10.2.14, 10.2.15, 10.2.16, 10.2.17, 10.2.18, 10.2.19, 10.2.20, 10.2.21, 10.2.22, 10.2.23, 10.2.24, 10.2.25, 10.2.26, 10.2.27, 10.2.28, 10.2.29, 10.2.30, 10.2.31, 10.2.32, 10.2.33, 10.2.34, 10.2.35, 10.2.36, 10.2.37, 10.2.38, 10.3.1, 10.3.3, 10.3.4, 10.3.5, 10.3.6, 10.3.7, 10.3.8, 10.3.9, 10.3.10, 10.3.11, 10.3.12, 10.3.13, 10.3.14, 10.3.15, 10.3.16, 10.3.17, 10.3.18, 10.3.19, 10.3.20, 10.3.21, 10.3.22, 10.3.23, 10.3.24, and 10.3.25.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 615 Fourth Revised Sheet No. 615

Superseding: Third Revised Sheet No. 615

GENERAL TERMS AND CONDITIONS (Continued)

45. DISCOUNT TERMS

- 45.1 Any Customer desiring a discount of the maximum Recourse Rates for service under Algonquin's Open-access Rate Schedules must submit a valid request for such discount pursuant to the procedures of this Section 45. To be considered a valid request, Customer must complete and submit a request for discount via the LINK® System, specifically including the information for all mandatory fields. Upon receipt of a valid request for a discount, Algonquin will log such request and either deny or grant such request.
- 45.2 In the event that Algonquin agrees to discount its maximum rates under any of its Open-access Rate Schedules, Algonquin and Customer may agree to the types of discounts specified herein without such discounts constituting a material deviation from Algonquin's pro forma service agreement. Algonquin and Customer may agree that a specified discounted rate will apply: (i) only to specified quantities under the service agreement; (ii) only if specified quantities are achieved or only with respect to quantities below a specified level; (iii) only during specified periods of the year or for a specifically defined period; (iv) only to specified points, combination of points, markets, transportation paths or other defined geographic area(s); (v) only in a specified relationship to the quantities actually delivered (i.e., that the reservation charge will be adjusted in a specified relationship to quantities actually delivered); (vi) so that the applicable rate may be adjusted in the following manner: when one rate component, which was equal to or within the applicable maximum and minimum Recourse Rates at the time Customer received the Discount Confirmation pursuant to Section 45.4 below specifying the terms of the discount, subsequently exceeds the applicable maximum Recourse Rate or is below the applicable minimum Recourse Rate, so that such rate component must be adjusted downward or upward to equal the new applicable maximum or minimum Recourse Rate, then other rate components may be adjusted upward or downward to achieve the agreed-upon overall rate, so long as none of the resulting rate components exceed the maximum Recourse Rate or are below the minimum Recourse Rate applicable to the rate component (such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised Statements of Rates; however, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable); and/or (vii) based upon published index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points for price determination (such discounted rate may be based upon a single published index price or the differential between published index prices or arrived at by formula; provided that the discounted rate shall not change the underlying rate design, shall not include any minimum bill or minimum take obligation, and shall define the rate component to be discounted). Notwithstanding the foregoing, no discounted rate shall be less than the applicable minimum rate.
- 45.3 In the event that Algonquin rejects Customer's request for a discounted rate, Algonquin shall notify Customer via e-mail of the reason for such rejection.

FERC Docket: RP10-77-000

Third Revised Sheet No. 615A Third Revised Sheet No. 615A

Superseding: Second Revised Sheet No. 615A

GENERAL TERMS AND CONDITIONS (Continued)

45. DISCOUNT TERMS (continued)

- The terms of any discount request granted by Algonquin pursuant to this Section 45 shall be transmitted by e-mail to Customer in the form of a Discount Confirmation. The Discount Confirmation shall identify the applicable Customer's name, contract number, rate schedule, term of the discount, discount rate, applicable quantities, point(s) of receipt and delivery, and/or the Algonquin path being discounted. The Discount Confirmation may also include other information required for posting under the Commission's regulations and other conditions consistent with Section 45.2. No particular discount transaction shall be contractually binding on either Algonquin or Customer until Algonquin has confirmed the terms of the discount upon Algonquin's e-mail to Customer of the Discount Confirmation for the transaction, subject to the underlying service agreement being fully executed. All discounts granted shall be effective no sooner than the beginning of the next Gas Day following the Gas Day on which the request is granted by Algonquin. Once the discount is contractually binding, the Discount Confirmation will constitute an addendum to the underlying service agreement. Each such addendum is an integral part of the underlying service agreement as if executed by both parties and fully copied and set forth at length therein.
- If Algonquin's Recourse Rates are subject to refund at any time during the 45.5 effectiveness of a Discount Confirmation, with respect to the applicable discounted rate, Customer shall be entitled to refunds of payments made by Customer only in the event that the final, non-appealable maximum Recourse Rate, whether usage-based or reservation-based, as determined by the Commission for a given time period is lower than the rate actually paid by Customer during such time period. Subject to the condition precedent set forth in the immediately preceding sentence, Customer's principal refund amount shall be equal to (i) with respective to usage-based rates, the product of (aa) the positive difference between the final, non-appealable maximum Recourse Rate and the discounted rate, and (bb) the quantities of gas delivered to Customer, or for Customer's account, during the refund period; and (ii) with respect to reservation-based rates, the product of (cc) the positive difference between the final, non-appealable maximum Recourse Rate and the discounted rate, (dd) the MDTQ covered by the discounted rate, and (ee) the number of months in the refund period (partial months shall be prorated for the number of days in the month that fall within the refund period and a discounted rate that is not a monthly rate shall be adjusted for purposes of this calculation to reflect the monthly equivalent of the rate).

FERC Docket: RP10-609-000

Fifth Revised Sheet No. 616 Fifth Revised Sheet No. 616

Superseding: Fourth Revised Sheet No. 616

GENERAL TERMS AND CONDITIONS (Continued)

46. NEGOTIATED RATES

Algonquin and Customer may mutually agree on a Negotiated Rate or rate formula with respect to rates, rate components, charges, or credits differing from the otherwise applicable tariff rate under Rate Schedules AFT-E, AFT-ES, AFT-1, AFT-1S, AIT-1, TTT, AFT-CL, PAL and/or AIT-2.

46.1 Definition

A Negotiated Rate may be less than, equal to, or greater than the maximum Recourse Rate and/or the minimum rate; may be a rate design other than the straight fixed-variable; and may include a minimum quantity. The Recourse Rates will be available to any Customer that does not wish to negotiate a rate.

46.2 Limitations

This Section 46 does not authorize the negotiation of terms and conditions of service.

46.3 Bidding for Capacity

The cap for bidding for capacity under the right of first refusal provisions of this tariff is the maximum Recourse Rate.

46.4 Capacity Release

- (A) Except as expressly provided for in Section 14 of the General Terms and Conditions, the release of capacity under a Negotiated Rate agreement is capped at the maximum Recourse Rate; provided, however, the Negotiated Rate Customer will continue to be obligated to pay Algonquin the difference by which the Negotiated Rate exceeds the rate paid by the Replacement Customer. Algonquin and a Negotiated Rate Customer may agree upon payment obligations and crediting mechanisms, which vary from or are different from those set forth in Algonquin's capacity release provisions.
- (B) To the extent that Algonquin agrees to a Negotiated Rate applicable to usage and/or fuel charges, Algonquin will consider, if requested by the Negotiated Rate Customer, and may agree with the Negotiated Rate Customer, on a not unduly discriminatory basis, to the terms and conditions pursuant to which Algonquin will offer such Negotiated Rate(s) to Replacement Customer(s). This agreement to flow through the Negotiated Rates for usage and/or fuel charges to a Replacement Customer will be documented as set forth in Section 46.7 below. A Replacement Customer to whom such offer is made may elect to pay the usage and/or fuel charges pursuant to Algonquin's recourse rates or pursuant to the offered Negotiated Rate.

46.5 Accounting Treatment

Algonquin will establish a new sub-account to record the revenues received from any Negotiated Rate transactions and shall maintain supporting information at a level of detail that would be sufficient for Natural Gas Act Section 4 rate change filing purposes. Algonquin will keep separate and identifiable each volume transported, billing determinant, rate component, surcharge, and revenue associated with a Negotiated Rate to permit filings in the form of Statements G, I, and J in future rate proceedings.

FERC Docket: RP10-609-000

Original Sheet No. 616A Original Sheet No. 616A

GENERAL TERMS AND CONDITIONS (continued)

46. NEGOTIATED RATES (continued)

46.6 Filing Requirement

Algonquin will file a Statement of Negotiated Rates prior to the commencement of service under a Negotiated Rate agreement or, for those Negotiated Rate agreements between Algonquin and a Replacement Customer that incorporate a Negotiated Rate for usage and/or fuel charges flowed through to the Replacement Customer pursuant to Section 46.4(B), as soon as reasonably practicable following the execution of such Negotiated Rate agreement. The Statement of Negotiated Rates will reflect the Customer's exact legal name, rate schedule, Negotiated Rate, the term of the Negotiated Rate, quantities, points of receipt and delivery to which the Negotiated Rate applies, the exact formula underlying a Negotiated Rate for any Negotiated Rate agreement, and any other rate-related terms that apply to the Negotiated Rate. Such Statement of Negotiated Rates also affirms that actual Negotiated Rate agreements do not deviate in any material respect from the form of service agreements.

FERC Docket: RP10-308-000

Third Revised Sheet No. 617 Third Revised Sheet No. 617

Superseding: Second Revised Sheet No. 617

GENERAL TERMS AND CONDITIONS (continued)

46. NEGOTIATED RATES (continued)

46.7 Documentation

Any Negotiated Rate agreed to by Algonquin and Customer pursuant to this Section 46 shall be implemented by Algonquin's completion of a pro forma Statement of Negotiated Rates with the applicable Negotiated Rate-related provisions as described in Section 46.7 herein. Algonquin shall tender such pro forma Statement of Negotiated Rates to Customer together with a transmittal letter for counter-execution by Customer, which transmittal letter shall have the sole purpose of memorializing Algonquin's and Customer's mutual agreement to the rate-related provisions reflected on such attached pro forma Statement of Negotiated Rates. After execution by both Algonquin and Customer, Algonquin shall file a Statement of Negotiated Rates with the Commission which shall contain raterelated provisions identical to the rate-related provisions reflected on the pro forma Statement of Negotiated Rates agreed to by Algonquin and Customer. Any Negotiated Rate agreed to pursuant to this Section 46 shall become effective only after acceptance by the Commission; prior to such date the rate applicable to any such service shall be the maximum Recourse Rate.

46.8 Effect of Negotiated Rate

Customer, by agreeing to a Negotiated Rate, acknowledges that the otherwise generally applicable maximum Recourse Rate(s) shall not apply or be available to Customer for service under the applicable Service Agreement during the period for which the Negotiated Rate is effective, notwithstanding any adjustment to such generally applicable maximum Recourse Rate(s) which may become effective during the period for which the Negotiated Rate is effective. If, at any time during the period for which the Negotiated Rate is effective, Algonquin is collecting its effective maximum Recourse Rate(s) subject to refund under Section 4 of the Natural Gas Act, Algonquin shall have no refund obligation to Customer even if the final maximum Recourse Rate(s) are reduced to a level below the Negotiated Rate. Customer's right to receive credits relating to Algonquin's penalty revenue or other similar revenue, if any, applicable to service on Algonquin's system shall be governed by Algonquin's FERC Gas Tariff and any applicable Commission orders and/or regulations.

47. Off-system Pipeline Capacity

From time to time, Algonquin may enter into transportation and/or storage agreements with other interstate or intrastate pipeline companies ("off-system pipeline"). In the event that Algonquin acquires capacity on an off-system pipeline, Algonquin will use such capacity for operational reasons and will only render service to Customers on the acquired capacity pursuant to Algonquin's FERC Gas Tariff and subject to Algonquin's approved rates, as such tariff and rates may change from time to time. For purposes of transactions entered into subject to this Section 47, the "shipper must have title" requirement is waived.

FERC Docket: RP10-308-000

Second Revised Sheet No. 618 Second Revised Sheet No. 618

Superseding: First Revised Sheet No. 618

GENERAL TERMS AND CONDITIONS (Continued)

48.SEGMENTATION, SECONDARY POINT RIGHTS AND RELATED SCHEDULING

48.1 Segmentation of Capacity.

- A. By Means of Nomination.
 - 1. In addition to any other transportation rights under the applicable rate schedule and related service agreements, Algonquin shall receive, transport and deliver to or for the account of a Customer under Rate Schedule AFT-1, AFT-E, AFT-1S, AFT-ES and AFT-CL on a segmented basis scheduled quantities of gas to or from (i) any point within Customer's Base Flow Path, and (ii) any secondary points located outside of the Base Flow Path of such service agreement for the Gas Day subject to the following requirements:
 - (a) aggregate quantities of gas received, transported or delivered under a Customer's Service Agreement, and, if such Service Agreement is the result of capacity release, the aggregate quantities of gas received, transported or delivered by all other Customers utilizing capacity rights derived from the original firm Service Agreement, shall not exceed the Customer's MDTQ or segment entitlements under such Service Agreement in any segment; and
 - (b) such service on a segmented basis shall not impair Algonquin's ability to render firm service and shall not adversely affect system operating conditions.
 - 2. For the purpose of determining whether any overlapping transactions exceed, in the aggregate (based on all relevant Customer utilization) the contractual entitlements of the original firm Service Agreement at any point (including, without limitation, the MDTQ or segment entitlements), a transaction that involves movement of gas in the same direction as that contemplated by the Primary Point(s) of Receipt and the Primary Point(s) of Delivery on the firm Service Agreement and a transaction that involves movement of gas that is counter to the direction contemplated by the Primary Point(s) of Receipt and the Primary Point(s) of Delivery on the firm Service Agreement that are nominated to the same delivery point for the same gas flow date and time shall not be deemed to be an overlap at that delivery point; provided, however, in no event shall Algonquin be obligated to deliver on a primary firm basis at that delivery point a quantity in excess of the MDDO applicable to that delivery point. For the purpose of determining whether any overlapping nominations in a segment exceed, in the aggregate (based on all relevant Customer utilization) the contractual entitlements of the original firm Service Agreement in any segment (including, without limitation, the MDTQ or segment entitlements), a transaction that involves movement of gas in the same direction as that contemplated by the Primary $\operatorname{Point}(s)$ of Receipt and the $\operatorname{Primary}$ $\operatorname{Point}(s)$ of $\operatorname{Delivery}$ on the firm Service Agreement and a transaction that involves movement of gas that is counter to the direction contemplated by the Primary Point(s) of Receipt and the Primary Point(s) of Delivery on the firm Service Agreement that are nominated on the same segment for the same gas flow date and time shall be deemed to be an overlap on that segment.

FERC Docket: RP10-308-000

First Revised Sheet No. 619 First Revised Sheet No. 619

Superseding: Original Sheet No. 619

GENERAL TERMS AND CONDITIONS (Continued)

48.SEGMENTATION, SECONDARY POINT RIGHTS AND RELATED SCHEDULING (continued)

48.1 Segmentation of Capacity. (continued)

- 3. In the event that a Releasing Customer and/or its Replacement Customer(s) submit overlapping nominations which in sum exceed in any segment or at any point the level of entitlement for which the Releasing Customer originally contracted, the relative priority of each nomination shall be determined first according to the scheduling procedures in Sections 22 and 23 of the General Terms and Conditions and the Secondary Point priorities specified in Sections 48.2 and 48.3 below, and then by applying the overlap priorities provided in the Releasing Customer's Notice to release capacity pursuant to Section 14.4(a)(15) of the General Terms and Conditions. If the Releasing Customer fails to provide overlap priorities in its notice, Algonquin shall allocate capacity to the overlapping nominations pro rata based on nominations.
- 4. Unless the point at which gas is received or to which gas is delivered is designated in Customer's service agreement as having an MDRO or MDDO, and the quantity received or delivered is within the applicable contractual limits, such point shall be a Secondary Point. The scheduling and curtailment priority of such receipts and deliveries shall be set forth in Section 48.3 below.

FERC Docket: RP10-308-000

Second Revised Sheet No. 620 Second Revised Sheet No. 620

Superseding: First Revised Sheet No. 620

GENERAL TERMS AND CONDITIONS (Continued)

48.SEGMENTATION, SECONDARY POINT RIGHTS AND RELATED SCHEDULING (continued)

- 48.1 Segmentation of Capacity (continued).
 - B. By Means of Capacity Release.

Releasing Customers can also segment firm capacity in their Base Flow Path through capacity release in accordance with the provisions of Section 14 of the General Terms and Conditions of Algonquin's FERC Gas Tariff, subject to the requirement that the release or multiple releases do not increase the total contractual entitlements (including, without limitation, the MDTQ, MDRO or MDDO) above the contractual entitlements of the original firm Service Agreement in any segment or at any point. The right to segment is subject to the requirement that aggregate quantities of gas received, transported or delivered under a Customer's Service Agreement, and, if such Service Agreement is the result of capacity release, the aggregate quantities of gas received, transported or delivered by all other Customers utilizing capacity rights derived from the original firm Service Agreement, shall not exceed the Customer's MDTQ under such Service Agreement in any segment.

Unless the point at which gas is received or to which gas is delivered is designated in Customer's Service Agreement as having an MDRO or MDDO, such point shall be a Secondary Point. The priority of deliveries to and from Secondary Points for scheduling, curtailment, and other tariff purposes shall be as set forth below in Section 48.3. As a general matter, Customers will have the right to utilize all Secondary Points so long as such use does not impair Algonquin's ability to render firm service, does not adversely affect Customers firm contractual rights, and/or does not adversely affect the safe and reliable operation of Algonquin's system.

FERC Docket: RP10-308-000

First Revised Sheet No. 621 First Revised Sheet No. 621

Superseding: Original Sheet No. 621

GENERAL TERMS AND CONDITIONS (Continued)

48.SEGMENTATION, SECONDARY POINT RIGHTS AND RELATED SCHEDULING (continued)

48.2 Secondary Points.

(a) In addition to the primary Point(s) of Receipt and/or Deliveryavailable under Customer's AFT-1, AFT-E, AFT-1S and AFT-ES service agreement(s), a Customer under Rate Schedules AFT-1, AFT-E, AFT-1S and AFT-ES may schedule and tender/deliver any quantity of gas at (i) any secondary Point(s) of Receipt on Algonquin's system or (ii) any secondary Point(s) of Delivery on Algonquin's system, including in excess of the MDRO and MDDO stated in the executed service agreement (but capped at the original firm contract holder's total MDTQ), with the exception that a transaction that involves movement of gas in the same direction as that contemplated by the Primary Point(s) of Receipt and the Primary Point(s) of Delivery on the firm service agreement and a transaction that involves movement of gas that is counter to the direction contemplated by the Primary Point(s) of Receipt and the Primary Point(s) of Delivery on the firm service agreement that are nominated to the same delivery point at the same time shall not be deemed to be an overlap of the MDTQ at that delivery point); provided, however, that Algonquin shall not be obligated to deliver a total quantity in excess of the total MDDO on a primary firm basis.

In addition to the primary Point(s) of Receipt and/or Delivery available under Customer's AFT-CL service agreement(s), a Customer under Rate Schedule AFT-CL may schedule and tender/deliver any quantity of gas at (i) any secondary Point(s) of Receipt on the applicable lateral or (ii) any secondary Point(s) of Delivery on the applicable lateral, including in excess of the MDRO and MDDO stated in the executed service agreement (but capped at the original firm contract holder's total MDTQ), with the exception that a transaction that involves movement of gas in the same direction as that contemplated by the Primary Point(s) of Receipt and the Primary Point(s) of Delivery on the firm service agreement and a transaction that involves movement of gas that is counter to the direction contemplated by the Primary Point(s) of Receipt and the Primary Point(s) of Delivery on the firm service agreement that are nominated to the same delivery point at the same time shall not be deemed to be an overlap of the MDTQ at that delivery point); provided, however, that Algonquin shall not be obligated to deliver a total quantity in excess of the total MDDO on a primary firm basis.

FERC Docket: RP10-308-000

Second Revised Sheet No. 622 Second Revised Sheet No. 622

Superseding: First Revised Sheet No. 622

GENERAL TERMS AND CONDITIONS (Continued)

48.SEGMENTATION, SECONDARY POINT RIGHTS AND RELATED SCHEDULING (continued)

48.2 Secondary Points. (continued)

These rights shall be subject to the following requirements:

- (1) quantities tendered must be within the accurate measurement capability and within the applicable operating restrictions imposed by the U.S. Department of Transportation with regard to the facilities installed at the Point(s) of Receipt and Point(s) of Delivery;
- (2) the requirement that the receipt and/or delivery of such quantity of gas will not impair the ability of Algonquin to satisfy Algonquin's firm obligations to other firm Customers and/or will not impair the safe operation of Algonquin's facilities; and/or
- (3) the requirement that Algonquin shall not receive, or deliver for the account of Customer in the aggregate a quantity of gas in excess of the Customer's rights in any portion of Algonquin's system.
- (b) Algonquin shall not be required to schedule any receipt at a Secondary Point of Receipt, nor shall Algonquin be required to schedule any delivery at a Secondary Point of Delivery if such receipt or delivery would impair deliveries to any firm service Customer at a Primary Point of Delivery or receipts at a Primary Point of Receipt.
- (c) Customer's receipts at Secondary Point(s) of Receipt may be reduced from time to time by Algonquin in order to permit Algonquin to purchase and receive into its system any quantities of gas that Algonquin desires to purchase and receive for purposes of satisfying Algonquin's Company Use Gas requirements or to satisfy Algonquin's obligations to receive gas pursuant to other firm service agreements under which such Point(s) of Receipt are Primary Points of Receipt. All such reductions in receipts will be allocated in accordance with Section 48.3 below.

FERC Docket: RP10-308-000

Third Revised Sheet No. 623 Third Revised Sheet No. 623

Superseding: Second Revised Sheet No. 623

GENERAL TERMS AND CONDITIONS (Continued)

48.SEGMENTATION, SECONDARY POINT RIGHTS AND RELATED SCHEDULING (continued)

- (d) In the event that Algonquin posts a notice, pursuant to Section 23.4(b) of the General Terms and Conditions, that hourly delivery rate restrictions are in force, deliveries by Customer to Secondary Points of Delivery must be made at a rate not to exceed 1/24 of Customer's daily scheduled quantity in any Hour. Customer's deliveries at such Secondary Point(s) of Delivery may be reduced from time to time to satisfy Algonquin's obligations to deliver gas pursuant to other firm service agreements under which such Point(s) of Delivery are Primary Points of Delivery. All such reductions in deliveries will be allocated in accordance with Section 48.3 below.
- 48.3 Status of Points and Related Scheduling and Curtailment.
 - (a) Scheduling Procedures for Secondary Points. With respect to scheduling at Secondary Points for timely nominations as defined in Sections 22.3 and 22.4(a) of these General Terms and Conditions, Algonquin will make available scheduled quantities for receipt at Customer's and operator's designated site by 4:30 p.m. CT on the Day prior to gas flow. Within four (4) hours of an intra-day nomination, Algonquin will notify affected Customer(s) and point operator(s) of changes in scheduled quantities. Based upon Customer's nominations, including rankings of individual Points of Receipt or Points of Delivery within a service agreement in the event the entirety of Customer's nomination cannot be scheduled, and subject to the provisions of Section 23 of the General Terms and Conditions, Algonquin shall schedule receipts and deliveries of gas at Secondary Points in the sequence listed below.
 - (1) among Customers under Algonquin's firm rate schedules according to quantities of gas requested to be scheduled by such Customers from (i) Secondary Points within the Base Flow Path to Primary Points, (ii) Secondary Points within the Base Flow Path to Secondary Points within the Base Flow Path, or (iii) Primary Points to Secondary Points within the Base Flow Path;
 - (2) among Customers under Algonquin's firm rate schedules according to quantities of gas requested to be scheduled by such Customers where either the Secondary Receipt Point, the Secondary Delivery Point, or both are outside the Base Flow Path.

Within each sequence listed in this Section 48.3(a), quantities of gas requested flowing through a Posted Point of Restriction within the Base Flow Path shall have priority over quantities of gas requested flowing through a Posted Point of Restriction outside the Base Flow Path.

FERC Docket: RP10-308-000

First Revised Sheet No. 624 First Revised Sheet No. 624

Superseding: Original Sheet No. 624

GENERAL TERMS AND CONDITIONS (Continued)

48.SEGMENTATION, SECONDARY POINT RIGHTS AND RELATED SCHEDULING (continued)

- (b) Curtailment at Secondary Points. Curtailment at Secondary Points shall be governed by the procedures set forth in Sections 24.4 and 24.5 of these General Terms and Conditions.
- (c) For the purposes of determining whether points are located within the Base Flow Path, Algonquin shall consider a movement of gas from a Point of Receipt to a Point of Delivery which is counter to the gas flow contemplated by the Primary Point(s) of Receipt and Primary Point(s) of Delivery specified in the service agreement as being outside of the Customer's Base Flow Path. In addition, for any movement of gas that traverses a segment(s) in which the total nominated quantity for that contract exceeds the firm contractual entitlement, that portion of the nominated quantity that exceeds the contractual entitlement shall be deemed to be outside of the Customer's Base Flow Path.

FERC Docket: RP10-515-000

Third Revised Sheet No. 625 Third Revised Sheet No. 625

Superseding: Second Revised Sheet No. 625

GENERAL TERMS AND CONDITIONS (Continued)

49. MATERIALLY NON-CONFORMING SERVICE AGREEMENTS

The following service agreements are being listed in accordance with Section 154.112(b) of the Commission's regulations. This list of agreements will be updated to reflect new agreements containing material, non-conforming provisions, with the exception of an extension in the term of one of the agreements identified below.

Customer Name	Contract Number	Rate Schedule	Primary Term Begin Date
ANP Bellingham Energy Company, LLC	99039	AFT-CL	4/01/2001
Bay State Gas Company	93001F	AFT-1	11/01/1993
Bay State Gas Company	93401	AFT-1	6/01/1993
Bay State Gas Company	94501	AFT-1	11/01/1994
Colonial Gas Company D/B/A National Grid	9227	AFT-1	11/01/1994
Colonial Gas Company D/B/A National Grid	510025	AFT-1	11/24/2003
Connecticut Light and Power Company	66668	AFT-CL	1/01/1998
Connecticut Light and Power Company	97130	AFT-1	1/01/1998
Connecticut Natural Gas Corporation	9₩005	AFT-1	4/01/2009
Consolidated Edison Company of New York, Inc.	510371	AFT-1	11/01/2008
Distrigas of Massachusetts LLC	92116	AFT-1	11/01/2002
Distrigas of Massachusetts LLC	510233	AFT-1	10/01/2004
Fore River Development, LLC.	510008	AFT-1	9/01/2004
Lake Road Generating Company, L.P.	98005LR	AFT-CL	8/01/2000
Mirant Energy Trading, LLC	510336	AFT-CL	8/09/2006
New England Gas Company - Fall River	510054	AFT-1	11/24/2003
Northern Utilities, Inc.	93002F	AFT-1	11/01/1993
NSTAR Gas Company	510482	AFT-CL	9/02/2009
NSTAR Gas Company	93004EC	AFT-E	4/01/1998
Orange and Rockland Utilities, Inc.	931003	T-1	8/23/1993
Taunton Municipal Lighting Plant	66667	AFT-CL	1/01/1998
The Boston Gas Company D/B/A National Grid	93002CR	AFT-1	11/01/1999
The Boston Gas Company D/B/A National Grid	93002EA	AFT-E	11/01/1999
The Boston Gas Company D/B/A National Grid	99012	AFT-1	11/01/1999
The Boston Gas Company D/B/A National Grid	9B100	AFT-1	11/01/1999
The Narragansett Electric Company			
D/B/A National Grid	510075	AFT-1	11/24/2003
The Narragansett Electric Company			
D/B/A National Grid	93011E	AFT-E	9/01/1994
Virginia Power Energy Marketing, Inc.	510294	AFT-CL	1/01/2005
Virginia Power Energy Marketing, Inc.	510295	AFT-CL	1/01/2005
Yankee Gas Services Company	93013EC	AFT-E	11/01/1997
Yankee Gas Services Company	9B106C	AFT-1	11/01/1997
Yankee Gas Services Company	93209C	AFT-1	11/01/1997
Yankee Gas Services Company	94000	AFT-1	11/01/1994
Yankee Gas Services Company	93009	AFT-1	9/01/1994
Yankee Gas Services Company	93409	AFT-1	6/01/1993

FERC Docket: RP04-361-000

Sheet Nos. 626 - 699 Sheet Nos. 626 - 699 : Effective

Sheet Nos. 626 - 699 are being reserved for future use

First Revised Sheet No. 700 First Revised Sheet No. 700 Superseding: Original Sheet No. 700

FORMS OF SERVICE AGREEMENTS

INDEX

Rate Scheo	ate Schedule	
AFT-1	Algonquin Firm Transportation Service	702
AFT-E	Algonquin No-Notice Firm Transportation Service	722
AFT-1S	Small Customer Firm Transportation Service	742
AFT-ES	Small Customer Firm No-Notice Transportation Service	762
AIT-1	Algonquin Interruptible Transportation Service	781
T-1	Transportation Service (New York and New Jersey)	791
AFT-4	Firm Transportation Service	811
	Capacity Release Umbrella Agreement	821
	Operational Balancing Agreement	831
	The LINK® System	841
AFT-CL	Firm Transportation Service	852
PAL	Park and Loan Service	861
TTT	Title Transfer Tracking Service	871
AIT-2	Interruptible Transportation Service	891

Third Revised Sheet No. 701 Third Revised Sheet No. 701 Superseding: Second Revised Sheet No. 701

This sheet was previously issued, but is now reserved for future use.

Third Revised Sheet No. 702 Third Revised Sheet No. 702 Superseding: Second Revised Sheet No. 702

FORM OF SERVICE AGREEMENT

(APPLICABLE TO RATE SCHEDULE AFT-1)			
Date:		Contract No	
	SERVICE AGR	EEMENT	
This AGREEMENT is entered is	nto by and between Algo ("Customer")	nquin Gas Transmission, LLC, ("Algonquin") and	
WHEREAS,			
NOW THEREFORE, in considerathe parties do agree as fol-		d of the mutual covenants herein contained,	
this Agreement and subject	ct to Algonquin's Rate	and pay for service pursuant to the terms of Schedule AFT-1 and the General Terms and orporated herein by reference and made a part	
(MATQ) for service under MATQ during the term of of Receipt and Point(s)	this Agreement and any this Agreement are list of Delivery, respective) and Maximum Annual Transportation Quantity right to increase or decrease the MDTQ or ed on Exhibit C attached hereto. The Point(s) ly, are listed on Exhibits A and B attached d herein by reference and made a part hereof.	
written notice of at leamore than 1 year; for see exactly one (1) year, the beginning of the primary subsequent notices for some after payment is due of such termination and of the date of termination, sufficient surety bond of guaranteeing payment to be entitled to terminate complies with the billing portions of this Agreement as required by the Gener parts of this Agreement of this Agr	Inot less that revice agreements with be notice must be submit term of the service agach service agach service agach service agach service agach service her; provided Algonquin gorovided further such to Customer either pays sor other form of security algonquin of such outst service pending the regardispute procedure curner and the customer either pays sor other form of security algonquin of such outst service pending the regardispute procedure curner and the customer either provisions and the service and Terms and Conditions until such time as such its as a "ROFR Agreemen is Tariff, the provision his article 3, a notice and Maximum Annual Tran	and shall continue for a term ending on and continue to be effective from erminated by Algonquin or Customer upon prior n 1 year for agreements with a primary term of oth a primary term and notice period of ted within ten (10) Business Days of the reement, and at least one (1) year for and otherwise mutually agreeable]. This nquin in the event Customer fails to pay part reunder and such failure continues for thirty ives ten days prior written notice to Customer ermination shall not be effective if, prior to uch outstanding bill or furnishes a good and y reasonably acceptable to Algonquin anding bill; provided that Algonquin shall not solution of a disputed bill if Customer rently on file in Algonquin's Tariff. Any or cash-out imbalances under this Agreement of Algonquin's Tariff shall survive the other balancing has been accomplished. t" as defined in the General Terms and of a termination notice by either Customer or of partial reduction in Maximum Daily sportation Quantity pursuant to Exhibit C or ms triggers Customer's right of first refusal	

FERC Docket: RP10-308-000

Fifth Revised Sheet No. 703 Fifth Revised Sheet No. 703

Superseding: Fourth Revised Sheet No. 703

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1)

- 4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted rate or a Negotiated Rate to which Customer and Algonquin have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Agreement. Provisions governing such Negotiated Rate and term shall be as specified on an appropriate Statement of Negotiated Rates filed, with the consent of Customer, as part of Algonquin's Tariff. It is further agreed that Algonquin may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Algonquin's Tariff, and Algonquin shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Customer any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
- 5. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

Algonquin:

Customer:

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable, except that in the case of conversions from former Rate Schedules F-2 and F-3, the parties' obligations under Article II of the service agreements pertaining to such rate schedules shall continue in effect:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN GAS TRANSMISSION, LLC
Ву:	Ву:
Title:	Title:

FERC Docket: RP08-268-000

First Revised Sheet No. 704 First Revised Sheet No. 704 : Effective Superseding: Original Sheet No. 704

Sheet No. 704 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 705 First Revised Sheet No. 705 : Effective Superseding: Original Sheet No. 705

Sheet No. 705 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 706 First Revised Sheet No. 706 : Effective Superseding: Original Sheet No. 706

Sheet No. 706 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 707 First Revised Sheet No. 707 : Effective Superseding: Original Sheet No. 707

Sheet No. 707 was previously issued, but is now reserved for future use.

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000 First Revised Sheet No. 708 First Revised Sheet No. 708 Superseding: Original Sheet No. 708FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1) Exhibit A Point(s) of Receipt Dated: To the service agreement under Rate Schedule AFT-1 between Algonquin Gas Transmission, LLC (Algonquin) and ______ (Customer) concerning Point(s) of Receipt. Primary Point of Maximum Maximum Daily Maximum Daily Maximum Receipt Obligation Receipt Pressure Receipt Signed for Identification

Algonquin: ____

Supersedes Exhibit A Dated _____

FERC Docket: RP09-97-000

First Revised Sheet No. 709 First Revised Sheet No. 709 Superseding: Original Sheet No. 709

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1)

	Exhibit B Point(s) of Deli	
Dated:		very
bacca.		
To the service agreement (Algonquin) and Delivery.	under Rate Schedule AFT-1	between Algonquin Gas Transmission, LLC (Customer) concerning Point(s) of
Primary Point of Delivery	Maximum Daily Delivery Obligation	Minimum Delivery Pressure

Signed for	Identifica	ation	
Algonquin:			
Customer:			
Supersedes	Exhibit B	Dated	

FERC Docket: RP09-97-000

First Revised Sheet No. 710 First Revised Sheet No. 710 Superseding: Original Sheet No. 710

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1)

Exhibit C

EXHIBIT
Transportation Quantities
Dated:
To the service agreement under Rate Schedule AFT-1 between Algonquin Gas Transmission, LLC (Algonquin) and (Customer) concerning transportation quantities.
MAXIMUM DAILY TRANSPORTATION QUANTITY (MDTQ): Dth Period
MAXIMUM ANNUAL TRANSPORTATION QUANTITY (MATQ): Dth
PARTIAL QUANTITY REDUCTION RIGHTS: Customer elects to partially reduce Customer's Maximum Daily Transportation Quantity by dth and Maximum Annual Transportation Quantity by dth, maintaining the existing Maximum Daily Transportation Quantity to Maximum Annual Transportation Quantity relationship, as of, or any subsequent anniversary date, upon providing [Notice period to be not less than the notice period required to terminate the entire contract] year(s) prior written notice to Algonquin.
Algonquin and Customer agree that, if this Agreement qualifies as a "ROFR Agreement", (i) the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is in addition to and not in lieu of any ROFR right to reduce Customer's Maximum Daily Transportation Quantity on a volumetric basis upon termination or expiration of this Agreement and (ii) only the partial reduction pursuant to the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is subject to the ROFR procedures specified in the General Terms and Conditions of Algonquin's Tariff and Customer may retain the balance of the Maximum Daily Transportation Quantity without being subject to the ROFR procedures.
Signed for Identification
Algonquin:
Customer:
Supersedes Exhibit C Dated

FERC Docket: RP08-268-000

Sheet Nos. 711 - 720 Sheet Nos. 711 - 720 : Effective

Sheet Nos. 711 - 720 are being reserved for future use

FERC Docket: RP09-97-000

Third Revised Sheet No. 721 Third Revised Sheet No. 721 Superseding: Second Revised Sheet No. 721

This sheet was previously issued, but is now reserved for future use.

FERC Docket: RP09-97-000

Third Revised Sheet No. 722 Third Revised Sheet No. 722 Superseding: Second Revised Sheet No. 722

FORM OF SERVICE AGREEMENT
(APPLICABLE TO RATE SCHEDULE AFT-E)

	(APPLICABLE TO RATE	SCHEDULE AFT-E)	
Date:		Contract No	
	SERVICE AG	REEMENT	
This AGREEMENT is entered into	• -	onquin Gas Transmission, LLC, ("Algo	onquin") and
WHEREAS,			
NOW THEREFORE, in consideration the parties do agree as followed		nd of the mutual covenants herein co	ontained,
this Agreement and subject	to Algonquin's Rate	ee and pay for service pursuant to the Schedule AFT-E and the General Terrocorporated herein by reference and management of the Schedule Sch	ms and
(MATQ) for service under the MATQ during the term of the of Receipt and Point(s) of	nis Agreement and an is Agreement are lis Delivery, respectiv	Q) and Maximum Annual Transportation y right to increase or decrease the ted on Exhibit C attached hereto. ' ely, are listed on Exhibits A and B ed herein by reference and made a pa	MDTQ or The Point(s) attached
including	imary Term") and shater unless and until[not less the less thate agreements with notice must be submited and service agreement; that any time by Algy bill for service he provided Algonquin ovided further such ustomer either pays other form of security gonquin of such outservice pending the raispute procedure cunecessary to correct Terms and Condition till such time as such as a "ROFR Agreeme Pariff, the provisions article 3, a notice Maximum Annual Tragement of its own te	and shall continue for a term end ll continue to be effective from terminated by Algonquin or Customer an 1 year for agreements with a print both a primary term and notice periotted within ten (10) Business Days of greement, and at least one (1) year and otherwise mutually agreeable]. Onquin in the event Customer fails ereunder and such failure continues gives ten days prior written notice termination shall not be effective such outstanding bill or furnishes at y reasonably acceptable to Algonquitanding bill; provided that Algonquitanding bill; provided that Algonquitanding bill; provided that Algonquitanding of a disputed bill if Customently on file in Algonquin's Tarist or cash-out imbalances under this sof Algonquin's Tariff shall survive balancing has been accomplished. nt" as defined in the General Terms of a termination notice by either e of partial reduction in Maximum Dinsportation Quantity pursuant to Eximate triggers Customer's right of finteriors of Algonquin's Tariff.	upon prior mary term of od of of the for This to pay part for thirty to Customer if, prior to a good and in in shall not tomer ff. Any Agreement ve the other and Customer or aily hibit C or

FERC Docket: RP10-308-000

Fifth Revised Sheet No. 723 Fifth Revised Sheet No. 723

Superseding: Fourth Revised Sheet No. 723

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-E)

- 4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted rate or a Negotiated Rate to which Customer and Algonquin have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Agreement. Provisions governing such Negotiated Rate and term shall be as specified on an appropriate Statement of Negotiated Rates filed, with the consent of Customer, as part of Algonquin's Tariff. It is further agreed that Algonquin may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Algonquin's Tariff, and Algonquin shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Customer any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
- 5. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

Algonquin:

Customer:

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN	GAS	TRANSMISSION,	LLC
Ву:	Ву:			
Title:	Title:			

FERC Docket: RP08-268-000

First Revised Sheet No. 724 First Revised Sheet No. 724 : Effective Superseding: Original Sheet No. 724

Sheet No. 724 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 725 First Revised Sheet No. 725 : Effective Superseding: Original Sheet No. 725 $\,$

Sheet No. 725 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 726 First Revised Sheet No. 726 : Effective Superseding: Original Sheet No. 726

Sheet No. 726 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 727 First Revised Sheet No. 727 : Effective Superseding: Original Sheet No. 727

Sheet No. 727 was previously issued, but is now reserved for future use.

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000 First Revised Sheet No. 728 First Revised Sheet No. 728 Superseding: Original Sheet No. 728

Supersedes Exhibit A Dated _____

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-E)

	Exhi	bit A			
	Point(s)	of Receipt			
Dated:					
To the service agreement (Algonquin) and	under Rate Schedule (C	AFT-E between A ustomer) concern	algonquin Gas ning Point(s)	Transmission, of Receipt.	LLC
Primary Point of Receipt	Maximum Daily Receipt Obligation		Maximum ipt Pressure		
Signed for Identification	n				
Algonquin:					
Customer:					

First Revised Sheet No. 729 First Revised Sheet No. 729 Superseding: Original Sheet No. 729

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-E)

Exhibit B			
	Point(s) of Delive	ry	
Dated	d:		
To the service agreemer (Algonquin) and	nt under Rate Schedule AFT-E b	etween Algonquin Gas Transmission, LLC concerning Point(s) of Delivery.	
Primary Point of Delivery	Maximum Daily Delivery Obligation	Minimum Delivery Pressure	
Signed for Identification			
Algonquin:			
Customer:			
Supersedes Exhibit B Dated			

First Revised Sheet No. 730 First Revised Sheet No. 730 Superseding: Original Sheet No. 730

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-E)

Exhibit C

Transportation Quantities
Dated:
To the service agreement under Rate Schedule AFT-E between Algonquin Gas Transmission, LLC (Algonquin) and (Customer) concerning transportation quantities.
MAXIMUM DAILY TRANSPORTATION QUANTITY (MDTQ): Dth Dth Period
MAXIMUM ANNUAL TRANSPORTATION QUANTITY (MATQ): Dth
PARTIAL QUANTITY REDUCTION RIGHTS: Customer elects to partially reduce Customer's Maximum Daily Transportation Quantity by dth and Maximum Annual Transportation Quantity by dth, maintaining the existing Maximum Daily Transportation Quantity to Maximum Annual Transportation Quantity relationship, as of, or any subsequent anniversary date, upon providing [Notice period to be not less than the notice period required to terminate the entire contract] year(s) prior written notice to Algonquin.
Algonquin and Customer agree that, if this Agreement qualifies as a "ROFR Agreement", (i) the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is in addition to and not in lieu of any ROFR right to reduce Customer's Maximum Daily Transportation Quantity on a volumetric basis upon termination or expiration of this Agreement and (ii) only the partial reduction pursuant to the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is subject to the ROFR procedures specified in the General Terms and Conditions of Algonquin's Tariff and Customer may retain the balance of the Maximum Daily Transportation Quantity without being subject to the ROFR procedures.
Signed for Identification
Algonquin:
Customer:
Supersedes Exhibit C Dated

FERC Docket: RP08-268-000

Sheet Nos. 731 - 740 Sheet Nos. 731 - 740 : Effective

Sheet Nos. 731 - 740 are being reserved for future use

FERC Docket: RP09-97-000

Third Revised Sheet No. 741 Third Revised Sheet No. 741 Superseding: Second Revised Sheet No. 741

This sheet was previously issued, but is now reserved for future use.

Third Revised Sheet No. 742 Third Revised Sheet No. 742 Superseding: Second Revised Sheet No. 742

FORM OF SERVICE AGREEMENT

(APPLICABLE TO RATE SCHEDULE AFT-1S)						
Date:		Contract No.				
	SERVICE A	GREEMENT				
This AGREEMENT is entered into	by and between Alc	gonquin Gas Transmission, LLC, ("Algonquin") and				
WHEREAS,						
NOW THEREFORE, in consideration the parties do agree as follow		and of the mutual covenants herein contained,				
this Agreement and subject	to Algonquin's Rate	se and pay for service pursuant to the terms of e Schedule AFT-1S and the General Terms and accorporated herein by reference and made a part				
(MATQ) for service under the MATQ during the term of this of Receipt and Point(s) of	is Agreement and and as as Agreement are list Delivery, respective	TQ) and Maximum Annual Transportation Quantity by right to increase or decrease the MDTQ or sted on Exhibit C attached hereto. The Point(s) wely, are listed on Exhibits A and B attached ted herein by reference and made a part hereof.				
written notice of at least more than 1 year; for servi exactly one (1) year, the results beginning of the primary tesubsequent notices for such Agreement may be terminated or all of the amount of any days after payment is due; of such termination and prothed the date of termination, Cu sufficient surety bond or consumption of the date of terminate secomplies with the billing of portions of this Agreement as required by the General parts of this Agreement untuit of this Agreement qualifies Conditions of Algonquin's Talgonquin, pursuant to this Transportation Quantity and the expiration of this Agreement	[not less the control of the service agreements with the service agreement, at any time by Alexander by Alexa	and shall continue for a term ending on and all continue to be effective from terminated by Algonquin or Customer upon prior nan 1 year for agreements with a primary term of both a primary term and notice period of atted within ten (10) Business Days of the agreement, and at least one (1) year for and otherwise mutually agreeable]. This gonquin in the event Customer fails to pay part hereunder and such failure continues for thirty gives ten days prior written notice to Customer termination shall not be effective if, prior to such outstanding bill or furnishes a good and ity reasonably acceptable to Algonquin standing bill; provided that Algonquin shall not resolution of a disputed bill if Customer arrently on file in Algonquin's Tariff. Any at or cash-out imbalances under this Agreement as of Algonquin's Tariff shall survive the other che balancing has been accomplished. The ent' as defined in the General Terms and on of a termination notice by either Customer or the of partial reduction in Maximum Daily ansportation Quantity pursuant to Exhibit C or earms triggers Customer's right of first refusal itions of Algonquin's Tariff.				

FERC Docket: RP10-308-000

Fifth Revised Sheet No. 743 Fifth Revised Sheet No. 743

Superseding: Fourth Revised Sheet No. 743

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1S)

- 4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted rate or a Negotiated Rate to which Customer and Algonquin have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Agreement. Provisions governing such Negotiated Rate and term shall be as specified on an appropriate Statement of Negotiated Rates filed, with the consent of Customer, as part of Algonquin's Tariff. It is further agreed that Algonquin may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Algonquin's Tariff, and Algonquin shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Customer any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
- 5. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

gon		
	qu_{\perp}	.11 .

Customer:

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable, except that in the case of conversions from former Rate Schedules F-2 and F-3, the parties' obligations under Article II of the service agreements pertaining to such rate schedules shall continue in effect:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN	GAS	TRANSMISSION,	LLC
Ву:	Ву:			
Title:	Title:			

FERC Docket: RP08-268-000

First Revised Sheet No. 744 First Revised Sheet No. 744 : Effective Superseding: Original Sheet No. 744

Sheet No. 744 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 745 First Revised Sheet No. 745 : Effective Superseding: Original Sheet No. 745

Sheet No. 745 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 746 First Revised Sheet No. 746 : Effective Superseding: Original Sheet No. 746

Sheet No. 746 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 747 First Revised Sheet No. 747 : Effective Superseding: Original Sheet No. 747

Sheet No. 747 was previously issued, but is now reserved for future use.

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000 First Revised Sheet No. 748 First Revised Sheet No. 748 Superseding: Original Sheet No. 748

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1S)

Exhibit A

Point(s) of Receipt

Dated:		
To the service agreement (Algonquin) and	under Rate Schedule AFT	-1S between Algonquin Gas Transmission, LLC (Customer) concerning Point(s) of Receipt.
Primary Point of Receipt	Maximum Daily Receipt Obligation	Maximum Receipt Pressure

Signed for	Identification
Algonquin:	
Customer:	
Supersedes	Exhibit A Dated

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000

First Revised Sheet No. 749 First Revised Sheet No. 749 Superseding: Original Sheet No. 749

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1S)

Exhibit B				
	Point(s) of Deli	very		
Dated:				
		S between Algonquin Gas Transmission, LLC (Customer) concerning Point(s) of		
Primary Point of Delivery	Maximum Daily Delivery Obligation	Minimum Delivery Pressure		
Signed for Identification				
Algonquin:				
Customer:				
Supersedes Exhibit B Dated				

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000

First Revised Sheet No. 750 First Revised Sheet No. 750 Superseding: Original Sheet No. 750

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-1S)

Exhibit C

Transportation Quantities		
Dated:		
To the service agreement under Rate Schedule AFT-1S between Algonquin Gas Transmission, LLC (Algonquin) and (Customer) concerning transportation quantities.		
MAXIMUM DAILY TRANSPORTATION QUANTITY (MDTQ): Dth Dth Period		
MAXIMUM ANNUAL TRANSPORTATION QUANTITY (MATQ): Dth		
PARTIAL QUANTITY REDUCTION RIGHTS: Customer elects to partially reduce Customer's Maximum Daily Transportation Quantity by dth and Maximum Annual Transportation Quantity to Maximum Annual Transportation Quantity relationship, as of, or any subsequent anniversary date, upon providing [Notice period to be not less than the notice period required to terminate the entire contract] year(s) prior written notice to Algonquin.		
Algonquin and Customer agree that, if this Agreement qualifies as a "ROFR Agreement", (i) the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is in addition to and not in lieu of any ROFR right to reduce Customer's Maximum Daily Transportation Quantity on a volumetric basis upon termination or expiration of this Agreement and (ii) only the partial reduction pursuant to the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is subject to the ROFR procedures specified in the General Terms and Conditions of Algonquin's Tariff and Customer may retain the balance of the Maximum Daily Transportation Quantity without being subject to the ROFR procedures.		
Signed for Identification		
Algonquin:		
Customer:		
Supersedes Exhibit C Dated		

FERC Docket: RP08-268-000

Sheet Nos. 751 - 760 Sheet Nos. 751 - 760 : Effective

Sheet Nos. 751 - 760 are being reserved for future use

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000

Third Revised Sheet No. 761 Third Revised Sheet No. 761 Superseding: Second Revised Sheet No. 761

This sheet was previously issued, but is now reserved for future use.

FERC Docket: RP09-97-000

Third Revised Sheet No. 762 Third Revised Sheet No. 762 Superseding: Second Revised Sheet No. 762

FORM OF SERVICE AGREEMENT
(APPLICABLE TO RATE SCHEDULE AFT-ES)

(APPLICABLE TO RATE SCHEDULE AFT-ES)		
Date:		Contract No
	SERVICE AGREEMENT	Г
This AGREEMENT is entered into by	and between Algonquin ("Customer").	Gas Transmission, LLC,("Algonquin") and
WHEREAS,		
NOW THEREFORE, in consideration o the parties do agree as follows:	f the premises and of t	the mutual covenants herein contained,
this Agreement and subject to	Algonquin's Rate Schedi	pay for service pursuant to the terms of ule AFT-ES and the General Terms and ated herein by reference and made a part
(MATQ) for service under this MATQ during the term of this A of Receipt and Point(s) of Del	Agreement and any right greement are listed on ivery, respectively, as	Maximum Annual Transportation Quantity t to increase or decrease the MDTQ or Exhibit C attached hereto. The Point(s) re listed on Exhibits A and B attached ein by reference and made a part hereof.
written notice of at least	nless and until termina [not less than 1 yeagreements with both a ce must be submitted worder the service agreement and of the service agreement; and of any time by Algonquin ll for service hereunder vided Algonquin gives the defurther such terminamer either pays such our form of security reasuin of such outstanding ce pending the resolutive procedure currently essary to correct or cams and Conditions of Alsuch time as such balar a "ROFR Agreement" as ff, the provision of a ticle 3, a notice of paximum Annual Transportant of its own terms transportant of its own terms transportant.	defined in the General Terms and termination notice by either Customer or artial reduction in Maximum Daily ation Quantity pursuant to Exhibit C or iggers Customer's right of first refusal

FERC Docket: RP10-308-000

Fifth Revised Sheet No. 763 Fifth Revised Sheet No. 763

Superseding: Fourth Revised Sheet No. 763

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-ES)

- 4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted rate or a Negotiated Rate to which Customer and Algonquin have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Agreement. Provisions governing such Negotiated Rate and term shall be as specified on an appropriate Statement of Negotiated Rates filed, with the consent of Customer, as part of Algonquin's Tariff. It is further agreed that Algonquin may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Algonquin's Tariff, and Algonquin shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Customer any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
- 5. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

$\Delta \pm \alpha$	OHIC	ruin	٠

Customer:

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN	GAS	TRANSMISSION,	LLC
Ву:	Ву:			
Title:	Title:			

FERC Docket: RP08-268-000

First Revised Sheet No. 764 First Revised Sheet No. 764 : Effective Superseding: Original Sheet No. 764

Sheet No. 764 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 765 First Revised Sheet No. 765 : Effective Superseding: Original Sheet No. 765

Sheet No. 765 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 766 First Revised Sheet No. 766 : Effective Superseding: Original Sheet No. 766

Sheet No. 766 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 767 First Revised Sheet No. 767 : Effective Superseding: Original Sheet No. 767

Sheet No. 767 was previously issued, but is now reserved for future use.

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000 First Revised Sheet No. 768 First Revised Sheet No. 768 Superseding: Original Sheet No. 768

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-ES)

Exhibit A

Point(s) of Receipt

Dated:			
To the service agreement (Algonquin) and	under Rate Schedule AFT-E	ES between Algonquin Gas Transmission, LI _ (Customer) concerning Point(s) of Recei	
Primary Point of Receipt	Maximum Daily Receipt Obligation	Maximum Receipt Pressure	

Signed for	Identification
Algonquin:	
Customer:	
Supersedes	Exhibit A Dated

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000

First Revised Sheet No. 769 First Revised Sheet No. 769 Superseding: Original Sheet No. 769

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-ES)

Exhibit B						
	Point(s) of Delivery					
Dated:						
To the service agreement (Algonquin) and	under Rate Schedule	AFT-ES between Algonquin Gas Transmission, LLC (Customer) concerning Point(s) of Delivery.				
Primary Point of Delivery	Maximum Daily Delivery Obligation	Minimum Delivery Pressure				
Signed for Identification						
Algonquin:						
Customer:						
Supersedes Exhibit B Dated						

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000

First Revised Sheet No. 770 First Revised Sheet No. 770 Superseding: Original Sheet No. 770

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-ES)

Exhibit C

Transportation Quantities
Dated:
To the service agreement under Rate Schedule AFT-ES between Algonquin Gas Transmission, LLC (Algonquin) and (Customer) concerning transportation quantities.
MAXIMUM DAILY TRANSPORTATION QUANTITY (MDTQ): Dth Dth Period
MAXIMUM ANNUAL TRANSPORTATION QUANTITY (MATQ): Dth
PARTIAL QUANTITY REDUCTION RIGHTS: Customer elects to partially reduce Customer's Maximum Daily Transportation Quantity by dth, maintaining the existing Maximum Daily Transportation Quantity to Maximum Annual Transportation Quantity relationship, as of, or any subsequent anniversary date, upon providing [Notice period to be not less than the notice period required to terminate the entire contract] year(s) prior written notice to Algonquin. Algonquin and Customer agree that, if this Agreement qualifies as a "ROFR Agreement", (i) the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is in addition to and not in lieu of any ROFR right to reduce Customer's Maximum Daily Transportation Quantity on a volumetric basis upon termination or expiration of this Agreement and (ii) only the partial reduction pursuant to the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is subject to the ROFR procedures specified in the General Terms and Conditions of Algonquin's Tariff and Customer may retain the balance of the Maximum Daily Transportation Quantity without being subject to the ROFR procedures.
Signed for Identification
Algonquin:
Customer:
Supersedes Exhibit C Dated

FERC Docket: RP08-268-000

Sheet Nos. 771 - 780 Sheet Nos. 771 - 780 : Effective

Sheet Nos. 771 - 780 are being reserved for future use

FERC Docket: RP10-308-000

Second Revised Sheet No. 781 Second Revised Sheet No. 781 Superseding: First Revised Sheet No. 781

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AIT-1)

	Date:		Contract No
		SERVICE	AGREEMENT
Th	his AGREEMENT is entered into h		Algonquin Gas Transmission, LLC,("Algonquin") and mer").
WH	HEREAS,		
	OW THEREFORE, in consideration he parties do agree as follows		es and of the mutual covenants herein contained,
1.	this Agreement and subject to	o Algonquin's F	take and pay for service pursuant to the terms of Rate Schedule AIT-1 and the General Terms and e incorporated herein by reference and made a part
2.	. Maximum Daily Transportation Maximum Annual Transportation	Quantity n Quantity	Dth Dth
3.	to thereafter written notice of at least Algonquin in the event Custom service hereunder and such far Algonquin gives ten days pringurther such termination shall customer either pays such out other form of security reason of such outstanding bill; propending the resolution of a correct or cash-out imbala	ary Term") and unless and unt . This A mer fails to pa ailure continue or written noti ll not be effectstanding bill nably acceptabl ovided that Alc disputed bill i in Algonquin's ances under thi riff shall surv	and shall continue for a term ending on and shall continue to be effective from a terminated by Algonquin or Customer upon prior agreement may be terminated at any time by any part or all of the amount of any bill for se for thirty days after payment is due; provided ace to Customer of such termination and provided attive if, prior to the date of termination, or furnishes a good and sufficient surety bond or a to Algonquin guaranteeing payment to Algonquin gonquin shall not be entitled to terminate service. If Customer complies with the billing dispute Tariff. Any portions of this Agreement necessary and a sufficient and the other parts of this Agreement until such ed.
4.	except during the specified and Algonquin have agreed. In the Discount Confirmation and term shall be as specific the consent of Customer, as may seek authorization from time to time to change any rand General Terms and Conditiplace such changes in effect herein shall be construed to	term of a disco Provisions gove to this Agreemed on an approp part of Algonqu the Commission ates, charges of ions of Algonqu in accordance deny Customer cipate fully in	applicable to service pursuant to this Agreement bunted rate or a Negotiated Rate to which Customer erning such discounted rate shall be as specified ment. Provisions governing such Negotiated Rate oriate Statement of Negotiated Rates filed, with min's Tariff. It is further agreed that Algonquin and/or other appropriate body at any time and from or other provisions in the applicable Rate Schedule min's Tariff, and Algonquin shall have the right to with the Natural Gas Act. Nothing contained any rights it may have under the Natural Gas Act, a rate or other proceedings by intervention or the or in part.
5.	considered duly delivered who facsimile. Customer or Algor	en mailed to th nquin may chang	In notices shall be in writing and shall be the applicable address below or transmitted via gethe addresses or other information below by the cessity of amending this Agreement:
	Algonquin:		
	Customer:		

FERC Docket: RP08-268-000

Second Revised Sheet No. 782 Second Revised Sheet No. 782 : Effective

Superseding: First Revised Sheet No. 782

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AIT-1)

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN	GAS	TRANSMISSION,	LLC
Ву:	Ву:			
Title:	Title:			

FERC Docket: RP08-268-000

Second Revised Sheet No. 783 second Revised Sheet No. 783 : Effective Superseding: First Revised Sheet No. 783

Sheet No. 783 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 784 First Revised Sheet No. 784 : Effective Superseding: Original Sheet No. 784

Sheet No. 784 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 785 First Revised Sheet No. 785 : Effective Superseding: Original Sheet No. 785

Sheet No. 785 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 786 First Revised Sheet No. 786 : Effective Superseding: Original Sheet No. 786

Sheet No. 786 was previously issued, but is now reserved for future use.

FERC Docket: RP04-361-000

Sheet Nos. 787 - 790 Sheet Nos. 787 - 790 : Effective

Sheet Nos. 787 - 790 are being reserved for future use

FERC Docket: RP10-77-000

First Revised Sheet No. 791 First Revised Sheet No. 791

Superseding: Original Sheet No. 791

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

This Agreement ("Agreement") is made and entered	d into this day of ,	, 19 , by
and between Algonquin Gas Transmission, LLC, a I	Delaware limited liability company	(herein
called "Algonquin"), and	(herein called "Customer" whether of	one or more
persons).		

In consideration of the premises and of the mutual covenants herein contained, the parties do agree as follows:

ARTICLE I SCOPE OF AGREEMENT

1.1 Subject to the terms, conditions and limitations hereof and of Algonquin's Rate Schedule T-1, Algonquin agrees to receive from or for the account of Customer for transportation on a firm basis quantities of natural gas tendered by Customer on any Day at the Point(s) of Receipt; provided, however, Customer shall not tender without the prior consent of Algonquin, at any Point of Receipt on any Day a quantity of natural gas in excess of the applicable Maximum Daily Receipt Obligation for such Point of Receipt plus the applicable Fuel Reimbursement Quantity; and provided further that Customer shall not tender at all Point(s) of Receipt on any Day or in any year a cumulative quantity of natural gas, without the prior consent of Algonquin, in excess of the following quantities of natural gas plus the applicable Fuel Reimbursement Quantities:

Maximum Daily Transportation Quantity ____ Dth
Maximum Annual Transportation Quantity ____ Dth

1.2 Algonquin agrees to transport and deliver to or for the account of Customer at the Point(s) of Delivery and Customer agrees to accept or cause acceptance of delivery of the quantity received by Algonquin on any Day, less the Fuel Reimbursement Quantities; provided, however, Algonquin shall not be obligated to deliver at any Point of Delivery on any Day a quantity of natural gas in excess of the applicable Maximum Daily Delivery Obligation.

FERC Docket: RP04-361-000

Original Sheet No. 792 Original Sheet No. 792: Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

ARTICLE II TERM OF AGREEMENT

- 2.1 This Agreement shall become effective as of the date set forth hereinabove and shall continue in effect for a term of _____ years ("Primary Term") and shall remain in force from year to year thereafter unless terminated by either party by written notice one year or more prior to the end of the Primary Term or any successive term thereafter. Algonquin's right to cancel this Agreement upon the expiration of the Primary Term hereof or any succeeding term shall be subject to Customer's rights pursuant to Section 8 of the General Terms and Conditions.
- 2.2 This Agreement may be terminated at any time by Algonquin in the event Customer fails to pay part or all of the amount of any bill for service hereunder and such failure continues for thirty days after payment is due; provided Algonquin gives ten days prior written notice to Customer of such termination and provided further such termination shall not be effective if, prior to the date of termination, Customer either pays such outstanding bill or furnishes a good and sufficient surety bond guaranteeing payment to Algonquin of such outstanding bill; provided that Algonquin shall not be entitled to terminate service pending the resolution of a disputed bill if Customer complies with the billing dispute procedure currently on file in Algonquin's tariff.

ARTICLE III RATE SCHEDULE

3.1 Customer shall pay Algonquin for all services rendered hereunder and for the availability of such service under Algonquin's Rate Schedule T-1 as filed with the Federal Energy Regulatory Commission and as the same may be hereafter revised or changed.

FERC Docket: RP04-361-000

Original Sheet No. 793 Original Sheet No. 793: Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

- 3.2 This Agreement and all terms and provisions contained or incorporated herein are subject to the provisions of Algonquin's applicable rate schedules and of Algonquin's General Terms and Conditions on file with the Federal Energy Regulatory Commission, or other duly constituted authorities having jurisdiction, and as the same may be legally amended or superseded, which rate schedules and General Terms and Conditions are by this reference made a part hereof.
- 3.3 Customer agrees that Algonquin shall have the unilateral right to file with the appropriate regulatory authority and make changes effective in (a) the rates and charges applicable to service pursuant to Algonquin's Rate Schedule T-1, (b) Algonquin's Rate Schedule T-1, pursuant to which service hereunder is rendered or (c) any provision of the General Terms and Conditions applicable to Rate Schedule T-1. Algonquin agrees that Customer may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Algonquin's existing FERC Gas Tariff as may be found necessary to assure that the provisions in (a), (b), or (c) above are just and reasonable.

ARTICLE IV POINT(S) OF RECEIPT

Natural gas to be received by Algonquin for the account of Customer hereunder shall be received at the outlet side of the measuring station(s) at or near the Point(s) of Receipt set forth in Exhibit A of the service agreement, with the Maximum Daily Receipt Obligation and the receipt pressure obligation indicated for each such Point of Receipt.

ARTICLE V POINT(S) OF DELIVERY

Natural gas to be delivered by Algonquin for the account of Customer hereunder shall be delivered on the outlet side of the measuring station(s) at or near the Point(s) of Delivery set forth in Exhibit B of the service agreement, with the Maximum Daily Delivery Obligation and the delivery pressure obligation indicated for each such Point of Delivery.

FERC Docket: RP04-361-000

Original Sheet No. 794 Original Sheet No. 794: Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

ARTICLE VI ADDRESSES

Except as herein otherwise provided or as provided in the General Terms and Conditions of Algonquin's FERC Gas Tariff, any notice, request, demand, statement, bill or payment provided for in this Agreement, or any notice which any party may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered, certified, or first class mail to the post office address of the parties hereto, as the case may be, as follows:

- (a) Algonquin:
- (b) Customer:

or such other address as either party shall designate by formal written notice.

FERC Docket: RP04-361-000

Original Sheet No. 795 Original Sheet No. 795 : Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

ARTICLE VII
INTERPRETATION

The interpretation and performance of the Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.

ARTICLE VIII AGREEMENTS BEING SUPERSEDED

When this Agreement becomes effective, it shall supersede the following agreements between the parties hereto.

FERC Docket: RP04-361-000

Original Sheet No. 796 original Sheet No. 796 : Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective agents thereunto duly authorized, the day and year first above written.

ALGONQUIN GAS TRANSMISSION, LLC

Зу:	
Title:	_
[CUSTOMER]	
3y:	
Pitle:	_

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000

Original Sheet No. 797 Original Sheet

 $\textbf{Original Sheet No. 797} \; \texttt{Original Sheet No. 797} \; : \quad \texttt{Effective}$

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

Exhibit A

Point(s) of Receipt

Dated:			
		between Algonquin Gas Transmission,(Customer) concerning Point(s) of	
	Maximum Daily Receipt Obligation	Maximum Receipt Pressure	
Signed for Identification	on		
Algonquin:			
Customer:			
Supersedes Exhibit A Dat	ced		

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000

Original Sheet No. 798 Original Sheet No. 798 : Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE T-1)

Exhibit B				
	Point(s) of De	livery		
Dated:				
To the service agreement (Algonquin) and Delivery.			nquin Gas Transmission, concerning Point(s) of	LLC
Point of Delivery	Maximum Daily Delivery Obligation		Minimum Delivery Pressure	
Signed for Identificatio	n			
Algonquin:				
Customer:				
Supersedes Exhibit B Dat	ed			

FERC Docket: RP04-361-000

Sheet Nos. 799 - 810 Sheet Nos. 799 - 810 : Effective

Sheet Nos. 799 - 810 are being reserved for future use

FERC Docket: RP08-268-000

First Revised Sheet No. 811 First Revised Sheet No. 811: Effective

Superseding: Original Sheet No. 811

FORM OF SERVICE AGREEMENT
(APPLICABLE TO RATE SCHEDULE AFT-4)

(APPLICABLE TO RATE SCHEDULE AFT-4)			
Date:		Contract No	
	SERVICE AGR	EEMENT	
This AGREEMENT is entered	into by and between Algo ("Customer")	onquin Gas Transmission, LLC,("Algonquin") and \cdot	
WHEREAS,			
NOW THEREFORE, in consider the parties do agree as for		d of the mutual covenants herein contained,	
this Agreement and subj	ect to Algonquin's Rate	e and pay for service pursuant to the terms of Schedule AFT-4 and the General Terms and corporated herein by reference and made a part	
(MATQ) for service unde MATQ during the term of of Receipt and Point(s)	er this Agreement and any this Agreement are list of Delivery, respective	e)) and Maximum Annual Transportation Quantity right to increase or decrease the MDTQ or ted on Exhibit C attached hereto. The Point(s) rely, are listed on Exhibits A and B attached ted herein by reference and made a part hereof.	
written notice of at lemore than 1 year; for sexactly one (1) year, the beginning of the primar subsequent notices for Agreement may be termined or all of the amount of days after payment is coff such termination and the date of termination sufficient surety bond guaranteeing payment to be entitled to termination complies with the billiportions of this Agreement of this Agreement of this Agreement of this Agreement.	eafter unless and until teast [not less that service agreements with be the notice must be submit by term of the service agreement; nated at any time by Algo any bill for service he due; provided Algonquin g diprovided further such the customer either pays sor other form of security algonquin of such outst be service pending the result of the customer either pays in the service pending the result of the customer either pays in the service pending the result of the service pending the service pending the result of the service pending the service pending the service pending the result of the service pending t	and shall continue for a term ending on and local continue to be effective from terminated by Algonquin or Customer upon prior in 1 year for agreements with a primary term or both a primary term and notice period of sted within ten (10) Business Days of the greement, and at least one (1) year for and otherwise mutually agreeable]. This enquin in the event Customer fails to pay part executed and such failure continues for thirty gives ten days prior written notice to Customer earmination shall not be effective if, prior to such outstanding bill or furnishes a good and expression years and the solution of a disputed bill if Customer executly on file in Algonquin's Tariff. Any cor cash-out imbalances under this Agreement of Algonquin's Tariff shall survive the other abalancing has been accomplished.	

If this Agreement qualifies as a "ROFR Agreement" as defined in the General Terms and Conditions of Algonquin's Tariff, the provision of a termination notice by either Customer or Algonquin, pursuant to this article 3, a notice of partial reduction in Maximum Daily Transportation Quantity and Maximum Annual Transportation Quantity pursuant to Exhibit C or the expiration of this Agreement of its own terms triggers Customer's right of first refusal under Section 9 of the General Terms and Conditions of Algonquin's Tariff.

FERC Docket: RP10-308-000

Fourth Revised Sheet No. 812 Fourth Revised Sheet No. 812

Superseding: Third Revised Sheet No. 812

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-4)

- 4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted rate or a Negotiated Rate to which Customer and Algonquin have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Agreement. Provisions governing such Negotiated Rate and term shall be as specified on an appropriate Statement of Negotiated Rates filed, with the consent of Customer, as part of Algonquin's Tariff. It is further agreed that Algonquin may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Algonquin's Tariff, and Algonquin shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Customer any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
- 5. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

	gon		
$_{\rm L}$	901	. y u -	L11 •

Customer:

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN	GAS	TRANSMISSION,	LLC
Ву:	Ву:			
Title:	Title:			

FERC Docket: RP08-268-000

First Revised Sheet No. 813 First Revised Sheet No. 813: Effective Superseding: Original Sheet No. 813

Sheet No. 813 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 814 First Revised Sheet No. 814: Effective Superseding: Original Sheet No. 814

Sheet No. 814 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 815 First Revised Sheet No. 815 : Effective Superseding: Original Sheet No. 815

Sheet No. 815 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 816 First Revised Sheet No. 816 : Effective Superseding: Original Sheet No. 816

Sheet No. 816 was previously issued, but is now reserved for future use.

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000

Original Sheet No. 817 original Sheet No. 817 : Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-4)

	Exhil	bit A		
	Point(s)	of Receipt		
Dated:				
To the service agreement (Algonquin) and				
Point of Receipt	Maximum Daily Receipt Obligation		Maximum Receipt Pressure	
Signed for Identification	on			
Supersedes Exhibit A Dat	ed			

Effective Date: 07/01/2004 Status: Effective FERC Docket: RP04-361-000 Original Sheet No. 818 original Sheet No. 818 : Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-4)

Exhibit B

Point(s) of Delivery

	POINT(S) O	I Delivery	
Dated:			
			7
Point of Delivery	Maximum Daily Delivery Obligation	Minimum Delivery Pressure	
	To the service agreement (Algonquin) and	Dated: To the service agreement under Rate Schedule (Algonquin) and Point of Maximum Daily	To the service agreement under Rate Schedule AFT-4 between Algonquin Gas Transmission, LLC (Algonquin) and (Customer) concerning Point(s) of Delivery. Point of Maximum Daily Minimum

Signed for	Identification
Algonquin:	
Customer:	
Supersedes	Exhibit B Dated

FERC Docket: RP08-268-000

Original Sheet No. 819 original Sheet No. 819 : Effective

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-4)

Exhibit C

Transportation Quantities

Dated:
To the service agreement under Rate Schedule AFT-4 between Algonquin Gas Transmission, LLC (Algonquin) and (Customer) concerning transportation quantities.
MAXIMUM DAILY TRANSPORTATION QUANTITY (MDTQ): Dth
MAXIMUM ANNUAL TRANSPORTATION QUANTITY (MATQ): Dth
PARTIAL QUANTITY REDUCTION RIGHTS: Customer elects to partially reduce Customer's Maximum Daily Transportation Quantity by dth and Maximum Annual Transportation Quantity by dth, maintaining the existing Maximum Daily Transportation Quantity to Maximum Annual Transportation Quantity relationship, as of, or any subsequent anniversary date, upon providing [Notice period to be not less than the notice period required to terminate the entire contract] year(s) prior written notice to Algonquin. Algonquin and Customer agree that, if this Agreement qualifies as a "ROFR Agreement", (i) the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is in addition to and not in lieu of any ROFR right to reduce Customer's Maximum Daily Transportation Quantity on a volumetric basis upon termination or expiration of this Agreement and (ii) only the partial reduction pursuant to the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is subject to the ROFR procedures specified in the General Terms and Conditions of Algonquin's Tariff and Customer may retain the balance of the Maximum Daily Transportation Quantity without being subject to the ROFR procedures.
Signed for Identification
Algonquin:
Customer:
Supersedes Exhibit C Dated

FERC Docket: RP08-268-000

Sheet No. 820 Sheet No. 820 : Effective

Sheet No. 820 is being reserved for future use

FERC Docket: RP06- 75-000

First Revised Sheet No. 821 First Revised Sheet No. 821: Effective

Superseding: Original Sheet No. 821

FORM OF SERVICE AGREEMENT FOR
CAPACITY RELEASE UMBRELLA AGREEMENT UNDER
RATE SCHEDULES AFT-1, AFT-1S, AFT-E, AFT-ES AND AFT-CL

This	Umbrella Ser	vice A	greement,	made a	and e	entered	into	this			
day	of	, by	and between	een				(herein	called	"Replacemen	it
Customer"),	and Algonqu	in Gas	Transmiss	sion, I	LLC,	a Delav	vare .	limited	liabili	ty company	(herein
called "Ald	gonguin").										

WHEREAS,

NOW, THEREFORE, for and in consideration of the mutual covenants and promises herein contained, the Replacement Customer and Algonquin hereby agree as follows:

ARTICLE I

SCOPE OF AGREEMENT

Subject to the terms, conditions and limitations hereof, so long as the financial evaluation and credit appraisal requirements are met in order for Replacement Customer to be on Algonquin's Approved Bidder list for capacity releases and execute this Umbrella Service Agreement pursuant to Section 14 of Algonquin's General Terms and Conditions, and this Umbrella Service Agreement is effective, Replacement Customer may bid from time to time on releases of capacity to be acquired under Rate Schedules AFT-1, AFT-1S, AFT-E, AFT-ES and AFT-CL pursuant to the procedures set forth in Section 14 of Algonquin's General Terms and Conditions. If at any time a bid submitted by Replacement Customer is accepted by Algonquin with respect to a given capacity release, Algonquin will promptly send to Replacement Customer an Addendum to this Umbrella Service Agreement, in the format attached hereto, depending upon the rate schedule under which the capacity is being acquired. An Addendum shall be deemed to be an executed Service Agreement under the rate schedule designated therein, subject to the terms and conditions of the rate schedule, the form of Service Agreement applicable to such rate schedule, and the General Terms and Conditions of Algonquin's tariff. The parties

FERC Docket: RP06- 75-000

First Revised Sheet No. 822 First Revised Sheet No. 822 : Effective

Superseding: Original Sheet No. 822

FORM OF SERVICE AGREEMENT FOR
CAPACITY RELEASE UMBRELLA AGREEMENT UNDER
RATE SCHEDULES AFT-1, AFT-1S, AFT-E, AFT-ES AND AFT-CL
(Continued)

agree that each Addendum is an integral part of this Umbrella Service Agreement as if executed by the parties hereto and fully copied and set forth herein at length and is binding on the parties hereto.

Subject to the terms, conditions and limitations hereof and of Algonquin's Rate Schedules AFT-1, AFT-1S, AFT-E, AFT-ES and AFT-CL, as applicable, and the forms of Service Agreement applicable thereto, Algonquin agrees to provide the applicable released service for Replacement Customer under the applicable rate schedule, provided that the Replacement Customer qualified under the financial evaluation and credit appraisal requirements set forth in Section 3 of Algonquin's General Terms and Conditions at the time it submitted the bid Algonquin accepted with respect to such release.

Replacement Customer hereby agrees to advise Algonquin of any material change in the information previously provided by the Replacement Customer to Algonquin pursuant to Section 3 of Algonquin's General Terms and Conditions.

ARTICLE II

TERM OF AGREEMENT

The term of this Agreement shall commence on _____ and shall continued in force and effect until ____ and ___ to ____ thereafter unless this Umbrella Service Agreement is terminated as hereinafter provided. If Algonquin determines at any time that Replacement Customer fails to meet the financial standards or credit criteria of Section 3 of the General Terms and Conditions, Algonquin may terminate this Agreement prospectively, without affecting Addenda previously executed but subject to the terms and conditions of Algonquin's tariff.

FERC Docket: RP06- 75-000

First Revised Sheet No. 823 First Revised Sheet No. 823 : Effective

Superseding: Original Sheet No. 823

FORM OF SERVICE AGREEMENT FOR
CAPACITY RELEASE UMBRELLA AGREEMENT UNDER
RATE SCHEDULES AFT-1, AFT-1S, AFT-E, AFT-ES AND AFT-CL
(Continued)

ARTICLE III

RATE SCHEDULE

This Umbrella Service Agreement does not have separate terms and conditions for particular services, but only provides a means for a Replacement Customer to utilize a service subject to the applicable provisions of the relevant Service Agreement and the terms and conditions for Rate Schedules AFT-1, AFT-1S, AFT-E, AFT-ES and AFT-CL, in the form attached hereto and fully incorporated herein as a part of this Umbrella Service Agreement.

Replacement Customer agrees that Algonquin shall have the unilateral right to file with the appropriate regulatory authority and make changes effective in (a) the rates and charges applicable to service pursuant to this Umbrella Service Agreement (b) the terms and conditions of this Umbrella Service Agreement, pursuant to which service hereunder is rendered or (c) any provision of the General Terms and Conditions, Rate Schedule applicable to this Umbrella Service Agreement, or Form of Service Agreement applicable to this Umbrella Service Agreement. Algonquin agrees that the Replacement Customer may protest or contest the aforementioned filings, and the Replacement Customer does not waive any rights it may have with respect to such filings.

FERC Docket: RP06- 75-000

First Revised Sheet No. 824 First Revised Sheet No. 824 : Effective

Superseding: Original Sheet No. 824

FORM OF SERVICE AGREEMENT FOR
CAPACITY RELEASE UMBRELLA AGREEMENT UNDER
RATE SCHEDULES AFT-1, AFT-1S, AFT-E, AFT-ES AND AFT-CL
(Continued)

ARTICLE IV

ADDRESSES

Except as herein otherwise provided or as provided in the General Terms and Conditions of Algonquin's FERC Gas Tariff, any notice, request, demand, or any notice which any party may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered, certified, or first class mail to the post office address of the parties hereto, as the case may be, as follows:

- (a) Algonquin:
- (b) Replacement Customer:

or such other address as either party shall designate by formal written notice.

ARTICLE V

INTERPRETATION

The interpretation and performance of this Umbrella Service Agreement shall be in accordance with the laws of $_$, without recourse to the law governing conflict of laws.

This Umbrella Service Agreement and the obligations of the parties are subject to all present and future valid laws with respect to the subject matter, either State or Federal, and to all valid present and future orders, rules, and regulations of duly constituted authorities having jurisdiction.

FERC Docket: RP06- 75-000

First Revised Sheet No. 825 First Revised Sheet No. 825 : Effective

Superseding: Original Sheet No. 825

FORM OF SERVICE AGREEMENT FOR
CAPACITY RELEASE UMBRELLA AGREEMENT UNDER
RATE SCHEDULES AFT-1, AFT-1S, AFT-E, AFT-ES AND AFT-CL
(Continued)

ALGONQUIN GAS TRANSMISSION, LLC

IN WITNESS WHEREOF, the parties hereto have caused this Umbrella Service Agreement to be signed by their respective agents thereunto duly authorized, the day and year first above written.

Ву:		
Title:	 	
[CUSTOMER]		
,		
Deva		
Ву:	 	
m! ± 1 = .		

FERC Docket: RP06- 75-000

First Revised Sheet No. 826 First Revised Sheet No. 826: Effective Superseding: Original Sheet No. 826

FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDER RATE SCHEDULES AFT-1, AFT-1S, AFT-E, AFT-ES AND AFT-CL (Continued)

Deal No.: _	
	dendum Contract No.:
Capacity Rel	ease Rate Schedule:
	Releasing Customer:
Releasing Cu	stomer's Contract No.:
Begin Date of	Dolongo.
End Date Of	Release:
Maximum Dail	y Transportation Quantity Dth
Maximum Annu	y Transportation Quantity Dth al Transportation Quantity Dth
Minimum Volu	me Commitment Dth/Monthly Billing Period
Is this capa	city subject to right of recall? Yes No
io oniio oupu	oro, basjess so right or resurr. res no
Recall Comme	nts:
Is this a pe	rmanent release? Yes No
011	
Other Commen	ts:
Rates:Check	all that apply:
	Maximum Rate Yes No
	Is this a release as a percent of Maximum Rate? Yes No
	If yes, what percent?
	Volumetric
	Reservation Charge

FERC Docket: RP06- 75-000

First Revised Sheet No. 827 First Revised Sheet No. 827 : Effective

Superseding: Original Sheet No. 827

FORM OF SERVICE AGREEMENT FOR
CAPACITY RELEASE UMBRELLA AGREEMENT UNDER
RATE SCHEDULES AFT-1, AFT-1S, AFT-E, AFT-ES AND AFT-CL
(Continued)

Capacity Release Rate Schedule:	
	aximum Daily eceipt Obligation

Primary
Point of Maximum Daily
Delivery Delivery Obligation

This addendum, entered into pursuant to Algonquin's capacity release program and to the executed Capacity Release Umbrella Agreement between Algonquin and the Replacement Customer, is heretofore made a part of and subject to the aforementioned Capacity Release Umbrella Agreement.

FERC Docket: RP04-361-000

Sheet Nos. 828 - 830 Sheet Nos. 828 - 830 : Effective

Sheet Nos. 828 - 830 are being reserved for future use

FERC Docket: RP10-77-000

First Revised Sheet No. 831 First Revised Sheet No. 831

Superseding: Original Sheet No. 831

OPERATIONAL BALANCING AGREEMENT
BETWEEN
ALGONQUIN GAS TRANSMISSION, LLC
AND

WHEREAS, the natural gas facilities operated by Operator and by Algonquin interconnect at the Interconnection Point specified on Exhibit 1 attached hereto;

WHEREAS, Algonquin and/or Operator have entered into one or more transportation agreements with various shippers ("Shippers") whereby Algonquin and/or Operator either receive natural gas ("gas") that Shippers cause to be delivered at the Interconnection Point or deliver gas that Shippers cause to be received at the Interconnection Point;

WHEREAS, from time to time, the quantity of gas delivered to or by Algonquin or Operator at the Interconnection Point is either greater or less than the quantity of gas nominated by Shippers and confirmed by Algonquin or Operator, resulting in inadvertent overor under-deliveries of the Shippers' nominated quantities;

WHEREAS, the Parties desire to provide for the elimination of such over- and underdeliveries by adjusting Shipper's nominations and/or deliveries and receipts of gas at the Interconnection Point;

NOW, THEREFORE, Algonquin and Operator agree that such over- or under-deliveries of gas at the Interconnection Point shall be eliminated in the following manner:

(1) Prior to the first Day of each Month, each Party shall notify the other Party in writing of the confirmed, scheduled nominations in Dth received by such Party from shippers for whom such Party will deliver or receive gas at the Interconnection Point during that Month. Notification of changes to such confirmed nominations shall be made orally as soon as reasonably possible, with written notice provided prior to the Day of the change in gas flow.

FERC Docket: RP10-77-000

First Revised Sheet No. 832 First Revised Sheet No. 832

Superseding: Original Sheet No. 832

OPERATIONAL BALANCING AGREEMENT (continued)

- (2) The Parties intend that the quantity of gas in Dth actually delivered and received each Day at the Interconnection Point will equal the confirmed, scheduled nominations of the Parties. Each Party will allocate the quantity that is to be delivered and received at the Interconnection Point among its Shippers pursuant to the confirmed, scheduled nominations. Any difference between the actual physical flow of gas at the Interconnection Point each Day and the confirmed, scheduled nominations of receipts and deliveries for such Day is the "Operational Imbalance," which is the responsibility of the Parties to eliminate pursuant to this Agreement.
- (3) Unless the Parties otherwise mutually agree in writing:
 - a. Each Party shall adjust the operations of its natural gas facilities as necessary in order to keep the measured flow of gas at the Interconnection Point as nearly as possible equal to the confirmed, scheduled nominations, but in any event to within [tolerance]. The measuring facilities owned and operated by a Party which measures receipts and deliveries at the Interconnection Point shall be a part of that Party's natural gas facilities for purposes of this Agreement.
 - b. Neither Party shall be expected to adjust the operation of its natural gas facilities below a minimum delivery pressure of _____ or above the maximum allowable operating pressures of its natural gas facilities at the Interconnection Point.
 - c. The Parties agree to cooperate with each other in the making of the adjustments required under subparagraph (3)a.

If it becomes apparent to either Party that the daily Operational Imbalance exceeds, or is projected to exceed, the [tolerance], such Party shall immediately orally notify the other Party thereof and the Parties shall immediately adjust the operations of their natural gas facilities pursuant to paragraph 3 in order to keep the confirmed, scheduled nominations and the Operational Imbalances within the applicable tolerances set forth above. In the event the Parties cannot, for any reason, adjust operations of their natural gas facilities as provided in this paragraph 3, the Parties shall adjust the allocation of gas to shippers at the Interconnection Point in

FERC Docket: RP10-77-000

First Revised Sheet No. 833 First Revised Sheet No. 833

Superseding: Original Sheet No. 833

OPERATIONAL BALANCING AGREEMENT (continued)

accordance with paragraph 5 in order to keep the confirmed nominations and the Operational Imbalances within the applicable tolerance. The Parties will orally notify each other of such adjustment prior to making the adjustment, with written notice given as soon as reasonably possible. Operator shall also provide notice to shippers when daily Operational Imbalances exceed the [tolerance].

- The actual measured quantity at the Interconnection Point each Month shall be determined and communicated by the measuring Party in writing to the other Party as soon as reasonably possible after the end of the Month, but in no case later than the Day of the Month following the last Day of the Month gas was delivered. The actual measured quantity shall be determined pursuant to the applicable provisions of the measuring Party's then-effective FERC Gas Tariff. Operational Imbalances shall be calculated initially by _____ and shall be agreed to in writing by the Parties prior to the _____ Day of the Month. The Parties shall correct any resulting Operational Imbalance for that Month by If a capacity constraint occurs on either Party's natural gas facilities which results in curtailment of quantities of gas through or after delivery and receipt of the gas at the Interconnection Point, the Party on whose natural gas facilities the constraint has occurred shall determine the reallocation of the quantity of gas to Shippers on its natural gas facilities in accordance with the provisions of its then-effective FERC Gas Tariff. Such Party shall promptly orally notify the other Party of the constraint and the reallocation of the quantity of gas to its Shippers, such notice to be confirmed in writing as soon as reasonably possible but not less than twenty-four hours after the occurrence of the constraint and the reallocation of the quantity of gas. If the constraint occurs at the measuring facilities at the Interconnection Point, the constraint shall be deemed to be on the natural gas facilities of the Party that operates those facilities. Measurement of gas for all purposes hereunder shall be in accordance with the (6) provisions set forth in the measuring Party's then-effective FERC Gas Tariff.
- (7) This agreement shall be effective as of _____ and shall continue until and from ____ to ___ thereafter until cancelled by either party upon not less than ____ prior written notice. Any remaining Operational

FERC Docket: RP04-361-000

Original Sheet No. 834 Original Sheet No. 834: Effective

OPERATIONAL BALANCING AGREEMENT (continued)

Imbalance will be corrected as provided in this agreement within _____ days after the final Operational Imbalance is calculated by the measuring Party or within such longer period of time as may be mutually agreed to by the Parties.

(8) The interpretation and performance of this agreement shall be in accordance with the laws of ______, without recourse to the law regarding the conflict of laws.

This agreement and the obligations of the parties are subject to all present and future valid laws with respect to the subject matter and to all valid present and future orders, rules and regulations of duly constituted authorities having jurisdiction.

- (9) No waiver by either Party of any one or more defaults by the other in the performance of any provision of this agreement shall operate or be construed as a waiver of any continuing or future default or defaults, whether of a like or different character, or a waiver of each of the Parties' obligations to eliminate Operational Imbalances by adjusting nominations and/or deliveries and receipts of gas at the Interconnection Point, as provided herein.
- (10) Notices shall be given to the following:
 - (a) Notices to Algonquin
 - (i) Oral notices:
 - (ii) Written notices:
 - (b) Notices to Operator
 - (i) Oral notices:
 - (ii) Written notices:

Notices required to be given in writing may be given by facsimile transmission.

FERC Docket: RP04-361-000

OPERATIONAL BALANCING AGREEMENT (continued)

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the day and year first above written.

ALGONQUIN GAS TRANSMISSION, LLC

By:	
Name: Title:	
iitie;	
_	
By:	
Name:	
Title.	

FERC Docket: RP04-361-000

Sheet Nos. 836 - 840 Sheet Nos. 836 - 840 : Effective

Sheet Nos. 836 - 840 are being reserved for future use

FERC Docket: RP09-1063-000

Fourth Revised Sheet No. 841 Fourth Revised Sheet No. 841

Superseding: Third Revised Sheet No. 841

FORM OF SERVICE AGREEMENT FOR THE LINK® SYSTEM

This LINK® System Agreement, executed this day of ,
, by and between, (Service Requester
Proprietary Number) (hereafter referred to as "LINK® System
Subscriber"), and ALGONQUIN GAS TRANSMISSION, LLC; EAST TENNESSEE NATURAL GAS
LLC; EGAN HUB STORAGE, LLC; MOSS BLUFF HUB, LLC; OZARK GAS TRANSMISSION,
L.L.C.; SALTVILLE GAS STORAGE COMPANY L.L.C.; and TEXAS EASTERN TRANSMISSION,
LP (whether one or more, hereafter referred to as "Pipeline"), witnesseth tha
for and in consideration of the mutual covenants and provisions herein
contained and subject to all of the terms, provisions and conditions herein
set forth, LINK® System Subscriber and Pipeline do hereby agree as follows:

ARTICLE I SCOPE OF AGREEMENT

- a. Pipeline shall make available for use by LINK® System Subscriber Pipeline's computerized electronic communication system, the LINK® Customer Interface System ("LINK® System"), to perform such functions as may be available on the LINK® System from time to time.
- b. Use of the LINK® System is subject to Pipeline's General Terms and Conditions or Statement of General Terms and Standard Operating Conditions, as applicable, as well as the provisions of any Rate Schedule, Contract and/or Service Agreement of the FERC Gas Tariff or Statement of General Terms and Standard Operating Conditions, as applicable, of the Pipeline with whom the LINK® System Subscriber is interacting to carry out the LINK® System transactions, as set forth in said Pipeline's currently effective FERC Gas Tariff or Statement of General Terms and Standard Operating Conditions, as applicable, as effective from time to time, and which are hereby incorporated by reference.
- c. LINK® System Subscriber agrees that Pipeline shall have the unilateral right to file with the appropriate regulatory authority and to make changes effective in (a) the rates and charges applicable to service pursuant to this LINK® System Agreement; and (b) any provision of Pipeline's FERC Gas Tariff or Statement of General Terms and Standard Operating Conditions, as applicable, related to this LINK® System Agreement. Pipeline agrees that LINK® System Subscriber may protest or contest the aforementioned filings, and LINK® System Subscriber does not waive any rights it may have with respect to such filings.

ARTICLE II TERM

The term of this LINK® System Agreement shall commence on the date of execution hereof and shall continue in full force and effect on a month to month basis until terminated by Pipeline or LINK® System Subscriber, within thirty days prior written notice of such termination.

FERC Docket: RP09-348-000

Third Revised Sheet No. 842 Third Revised Sheet No. 842

Superseding: Second Revised Sheet No. 842

FORM OF SERVICE AGREEMENT FOR THE LINK® SYSTEM

ARTICLE III
ADDRESSES

Except as provided in the General Terms and Conditions of Pipeline's FERC Gas Tariff or Pipeline's Statement of General Terms and Standard Operating Conditions, as applicable, any notice, request, demand, statement, bill or payment pursuant to this LINK® System Agreement shall be in writing and shall be considered as duly delivered when received online via the LINK® System, or when received as registered, certified, or regular mail at the address of the parties hereto, as the case may be, as follows:

(a) Pipeline: Spectra Energy Transmission

Attn: LINK® Services, Room WO 3I-32

5400 Westheimer Court P.O. Box 1642

Houston, TX 77251-1642

(b) LINK® System Subscriber:

[The address LINK® System Subscriber shall designate by submitting the on-line Contact Information as discussed in the Electronic Communications section of the General Terms and Conditions of the relevant Pipeline's FERC Gas Tariff or Statement of General Terms and Standard Operating Conditions, as applicable.]

ARTICLE IV
INTERPRETATION

The interpretation and performance of this LINK® System Agreement shall be in accordance with the laws of the State of Texas without recourse to the law governing conflicts of law.

This LINK® System Agreement and the obligations of the parties are subject to all present and future valid laws with respect to the subject matter hereof, either State or Federal, and to all valid present and future orders, rules, and regulations of duly constituted authorities having jurisdiction.

ARTICLE V AGREEMENTS BEING SUPERSEDED

When this LINK® System Agreement becomes effective, it shall supersede any LINK® System Agreements between the parties hereto with an earlier execution date.

FERC Docket: RP09-1063-000

Fifth Revised Sheet No. 843 Fifth Revised Sheet No. 843

Superseding: Fourth Revised Sheet No. 843

FORM OF SERVICE AGREEMENT FOR THE LINK® SYSTEM Continued

IN WITNESS WHEREOF, the parties hereto have caused this LINK® System Agreement to be signed by their respective agents thereunto duly authorized, the day and year first above written.

ALGONQUIN GAS TRANSMISSION, LLC
EAST TENNESSEE NATURAL GAS, LLC
EGAN HUB STORAGE, LLC
MOSS BLUFF HUB, LLC
OZARK GAS TRANSMISSION, L.L.C.
SALTVILLE GAS STORAGE COMPANY L.L.C.

TEXAS EASTERN TRANSMISSION, LP by its General Partner Spectra Energy Transmission Services, LLC

By:

Title:

Signature

LINK® System Subscriber

By:

Title:

Signature

FERC Docket: RP04-468-000

First Revised Sheet No. 844 First Revised Sheet No. 844 : Effective Superseding: Original Sheet No. 844

FERC Docket: RP04-468-000

First Revised Sheet No. 845 First Revised Sheet No. 845 : Effective Superseding: Original Sheet No. 845

FERC Docket: RP04-468-000

First Revised Sheet No. 846 First Revised Sheet No. 846 : Effective Superseding: Original Sheet No. 846

FERC Docket: RP04-361-000

Sheet Nos. 847 - 850 Sheet Nos. 847 - 850 : Effective

Sheet Nos. 847 - 850 are being reserved for future use

FERC Docket: RP09-97-000

Third Revised Sheet No. 851 Third Revised Sheet No. 851 Superseding: Second Revised Sheet No. 851

FERC Docket: RP10-146-000

Seventh Revised Sheet No. 852 Seventh Revised Sheet No. 852 Superseding: Sixth Revised Sheet No. 852

FORM OF SERVICE AGREEMENT

(APPLICABLE TO RATE SCHEDULE AFT-CL)					
Date:,					
SERVICE AGREEMENT					
This AGREEMENT is entered into by and between Algonquin Gas Transmission, LLC, ("Algonquin") and ("Customer").					
WHEREAS,					
NOW THEREFORE, in consideration of the premises and of the mutual covenants herein contained, the parties do agree as follows:					
1. Algonquin shall deliver and Customer shall take and pay for service on Algonquin's [Canal Lateral, Middletown Lateral, Cleary Lateral, Lake Road Lateral, Brayton Point Lateral, Bellingham Lateral, Phelps Dodge Lateral, Manchester Street Lateral, Cape Cod Lateral, Northeast Gateway Lateral, J-2 Facility, or Kleen Energy Lateral, as applicable,] pursuant to the terms of this Agreement and subject to Algonquin's Rate Schedule AFT-CL and the General Terms and Conditions of Algonquin's Tariff, which are incorporated herein by reference and made a part hereof.					
2. The Maximum Daily Transportation Quantity (MDTQ) and Maximum Annual Transportation Quantity (MATQ) for service under this Agreement and any right to increase or decrease the MDTQ or MATQ during the term of this Agreement are listed on Exhibit C attached hereto. The Point(s) of Receipt and Point(s) of Delivery, respectively, are listed on Exhibits A and B attached hereto. Exhibit(s) A, B, and C are incorporated herein by reference and made a part hereof.					
and shall continue for a term ending on and including ("Primary Term") and shall continue to be effective from to thereafter unless and until terminated by Algonquin or Customer upon prior written notice of at least [Incl less than 1 year for agreements with a primary term of more than 1 year; for service agreements with both a primary term and notice period of exactly one (1) year, the notice must be submitted within ten (10) Business Days of the beginning of the primary term of the service agreement, and at least one (1) year for subsequent notices for such service agreement; and otherwise mutually agreeable]. This Agreement may be terminated at any time by Algonquin in the event Customer fails to pay part or all of the amount of any bill for service hereunder and such failure continues for thirty days after payment is due; provided Algonquin gives ten days prior written notice to Customer of such termination and provided further such termination shall not be effective if, prior to the date of termination, Customer either pays such outstanding bill or furnishes a good and sufficient surety bond or other form of security reasonably acceptable to Algonquin guaranteeing payment to Algonquin of such outstanding bill; provided that Algonquin shall not be entitled to terminate service pending the resolution of a disputed bill if Customer complies with the billing dispute procedure currently on file in Algonquin's Tariff. Any portions of this Agreement necessary to correct or cash-out imbalances under this Agreement as required by the General Terms and Conditions of Algonquin's Tariff shall survive the other parts of this Agreement qualifies as a "ROFR Agreement" as defined in the General Terms and Conditions of Algonquin, pursuant to this article 3, a notice of partial reduction in Maximum Daily					
Transportation Quantity and Maximum Annual Transportation Quantity pursuant to Exhibit C or the expiration of this Agreement of its own terms triggers Customer's right of first refusal under Section 9 of the General Terms and Conditions of Algonquin's Tariff.					

FERC Docket: RP10-308-000

Fifth Revised Sheet No. 853 Fifth Revised Sheet No. 853

Superseding: Fourth Revised Sheet No. 853

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-CL)

- 4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted rate or a Negotiated Rate to which Customer and Algonquin have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Agreement. Provisions governing such Negotiated Rate and term shall be as specified on an appropriate Statement of Negotiated Rates filed, with the consent of Customer, as part of Algonquin's Tariff. It is further agreed that Algonquin may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Algonquin's Tariff, and Algonquin shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Customer any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
- 5. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

gon		
	qu_{\perp}	.11 .

Customer:

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN	GAS	TRANSMISSION,	LLC
Ву:	Ву:			
Title:	Title:			

FERC Docket: RP08-268-000

Third Revised Sheet No. 854 Third Revised Sheet No. 854 : Effective Superseding: Second Revised Sheet No. 854

Sheet No. 854 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 855 First Revised Sheet No. 855 : Effective Superseding: Original Sheet No. 855

Sheet No. 855 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 856 First Revised Sheet No. 856: Effective Superseding: Original Sheet No. 856

Sheet No. 856 was previously issued, but is now reserved for future use.

FERC Docket: RP09-97-000

First Revised Sheet No. 857 First Revised Sheet No. 857

Superseding: Original Sheet No. 857

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-CL)

Exhibit A

Point(s) of Receipt

Dated:

To the service agreement under Rate Schedule AFT-CL between Algonquin Gas Transmission, LLC (Algonquin) and _____ (Customer) concerning Point(s) of Receipt.

Primary Point(s)
of Receipt

Maximum Daily Receipt Obligation Maximum

Receipt Pressure

Algonquin's line pressure as may exist from time to time.

Signed for Identification

Algonquin:

Customer:

Supersedes Exhibit A Dated:

FERC Docket: RP09-97-000

First Revised Sheet No. 858 First Revised Sheet No. 858 Superseding: Original Sheet No. 858

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-CL)

Exhibit B

Point(s) of Delivery

Dated:

To the servi	ce agreement	under	Rate	Schedule	AFT-CL	between	Algonqu	in Gas	Transmiss	ion,	LLC
(Algonquin)	and					(Custor	ner) con	cernin	g Point(s)	of	
Delivery.											

Primary Point Maximum Daily Minimum of Delivery Obligation Delivery Pressure

Signed for Identification

Algonquin:

Customer:

Supersedes Exhibit B Dated:

Effective Date: 01/01/2009 Status: Effective FERC Docket: RP09-97-000

First Revised Sheet No. 859 First Revised Sheet No. 859 Superseding: Original Sheet No. 859

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AFT-CL)

Exhibit C

Transportation Quantities
Dated:
To the service agreement under Rate Schedule AFT-CL between Algonquin Gas Transmission, LLC (Algonquin) and (Customer) concerning transportation quantities.
MAXIMUM DAILY TRANSPORTATION QUANTITY (MDTQ): Dth
MAXIMUM ANNUAL TRANSPORTATION QUANTITY (MATQ): Dth
PARTIAL QUANTITY REDUCTION RIGHTS: Customer elects to partially reduce Customer's Maximum Daily Transportation Quantity by dth and Maximum Annual Transportation Quantity by dth, maintaining the existing Maximum Daily Transportation Quantity to Maximum Annual Transportation Quantity relationship, as of, or any subsequent anniversary date, upon providing [Notice period to be not less than the notice period required to terminate the entire contract] year(s) prior written notice to Algonquin.
Algonquin and Customer agree that, if this Agreement qualifies as a "ROFR Agreement", (i) the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is in addition to and not in lieu of any ROFR right to reduce Customer's Maximum Daily Transportation Quantity on a volumetric basis upon termination or expiration of this Agreement and (ii) only the partial reduction pursuant to the foregoing contractual right to partially reduce Customer's Maximum Daily Transportation Quantity is subject to the ROFR procedures specified in the General Terms and Conditions of Algonquin's Tariff and Customer may retain the balance of the Maximum Daily Transportation Quantity without being subject to the ROFR procedures.
Signed for Identification
Algonquin:
Customer:
Supersedes Exhibit C Dated

Effective Date: 05/01/2008 Status: Effective FERC Docket: RP08-268-000

Sheet No. 860 Sheet No. 860: Effective

Sheet No. 860 is being reserved for future use

Date:____

FERC Docket: RP10-308-000

Third Revised Sheet No. 861 Third Revised Sheet No. 861 Superseding: Second Revised Sheet No. 861

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE PAL)

Contract No.____

	SERVICE AGREEMENT
This	s AGREEMENT is entered into by and between Algonquin Gas Transmission, LLC,("Algonquin") and ("Customer").
WHEF	REAS,
	THEREFORE, in consideration of the premises and of the mutual covenants herein contained, parties do agree as follows:
t	Algonquin shall deliver and Customer shall take and pay for service pursuant to the terms of this Agreement and subject to Algonquin's Rate Schedule PAL and the General Terms and Conditions of Algonquin's Tariff, which are incorporated herein by reference and made a part mereof.
T r c	The Maximum Park Quantity or Maximum Loan Quantity, as appropriate, and the PAL Point(s) of Transaction are set forth in the Exhibit(s) A to this agreement. Customer shall initiate a request for each park or loan service transaction by executing and delivering to Algonquin one or more Exhibit(s) A. Upon execution by Algonquin, Customer's Exhibit(s) A shall be incorporated in and made a part hereof.
t W F F C C C C F F T F	This Agreement shall be effective on and shall continue for a term ending on and including ("Primary Term") and shall continue to be effective from thereafter unless and until terminated by Algonquin or Customer upon prior written notice of at least This Agreement may be terminated at any time by Algonquin in the event Customer fails to pay part or all of the amount of any bill for service hereunder and such failure continues for thirty days after payment is due; provided Algonquin gives thirty days prior written notice to Customer of such termination and provided further such termination shall not be effective if, prior to the date of termination, Customer either pays such outstanding bill or furnishes a good and sufficient surety bond or other form of security reasonably acceptable to Algonquin guaranteeing payment to Algonquin of such outstanding bill; provided that Algonquin shall not be entitled to terminate service pending the resolution of a disputed bill if Customer complies with the billing dispute procedure currently on file in Algonquin's Tariff. Any portions of this Agreement necessary to correct or resolve a Park Balance or a Loan Balance under this Agreement as required by Rate Schedule PAL and the General Terms and Conditions of Algonquin's Tariff shall survive the other parts of this Agreement until such time as such balancing has been accomplished.
e a i a t t t t t	Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted rate or a Negotiated Rate to which Customer and Algonquin have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Agreement. Provisions governing such Negotiated Rate and term shall be as specified on an appropriate Statement of Negotiated Rates filed, with the consent of Customer, as part of Algonquin's Tariff. It is further agreed that Algonquin may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Algonquin's Tariff, and Algonquin shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained therein shall be construed to deny Customer any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or other wise to contest increased rates in whole or in part.

FERC Docket: RP08-268-000

Second Revised Sheet No. 862 Second Revised Sheet No. 862: Effective

Superseding: First Revised Sheet No. 862

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE PAL)

5. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

Algonquin:

Customer:

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN GAS TRANSMISSION, LI
Ву:	Ву:
Title:	Title:

FERC Docket: RP08-268-000

First Revised Sheet No. 863 First Revised Sheet No. 863: Effective Superseding: Original Sheet No. 863

Sheet No. 863 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

Fourth Revised Sheet No. 864 Fourth Revised Sheet No. 864 : Effective Superseding: Third Revised Sheet No. 864

FORM OF SERVICE AGREEMENT FOR RATE SCHEDULE PAL (continued)

ALGONQUIN GAS TRANSMISSION, LLC PARK AND LOAN (PAL) AGREEMENT

	I	DATED	, 	
	EXHIBIT	A DATED		
ALGONQUIN:	Algonquin Gas Transmi 5400 Westheimer Court Houston, Texas 77056	;		
	Attention: Spectra F	Inergy Transmis	sion Marketing I	Department
CUSTOMER:				
	Commencement Service Date	Termination of Service Date	Maximum Park/Loan Quantity	Specific Points
Park Service				
Loan Service				
		ALGO	NQUIN GAS TRANSI	MISSION, LLC
		Ву		
		[CUST	OMER]	

FERC Docket: RP04-361-000

Sheet Nos. 865 - 870 Sheet Nos. 865 - 870 : Effective

Sheet Nos. 865 - 870 are being reserved for future use

FERC Docket: RP10-308-000

Second Revised Sheet No. 871 Second Revised Sheet No. 871 Superseding: First Revised Sheet No. 871

FORM OF SERVICE AGREEMENT
(APPLICABLE TO TITLE TRANSFER TRACKING SERVICE
UNDER RATE SCHEDULE TTT)

		UNDER RAIE SCHE	INCIE III)	
	Date:		Contract No	
		SERVICE AGR	REEMENT	
Thi		nto by and between Algo ("Customer")	onquin Gas Transmission, LLC,("Algonquin	") and
WHE	REAS,			
	THEREFORE, in considerat parties do agree as foll	-	nd of the mutual covenants herein contain	ned,
	pursuant to the terms of	this Agreement and sub ons of Algonquin's Tar	e and pay for Title Transfer Tracking se bject to Algonquin's Rate Schedule TTT a riff, which are incorporated herein by	
	including("P to thereaf written notice of at leas Algonquin in the event Cu service hereunder and suc Algonquin gives ten days further such termination Customer either pays such other form of security re of such outstanding bill;	rimary Term") and shall ter unless and until to the continues for	and shall continue for a term ending and continue to be effective from terminated by Algonquin or Customer upon ement may be terminated at any time by art or all of the amount of any bill for or thirty days after payment is due; prove thirty days after payment is due; prove if, prior to the date of termination, furnishes a good and sufficient surety be Algonquin guaranteeing payment to Algonal in shall not be entitled to terminate sustomer complies with the billing disputifif.	prior vided ided ond or nquin ervice
	except during the specifi and Algonquin have agreed in the Discount Confirmat and term shall be as spec the consent of Customer, may seek authorization fr time to time to change an and General Terms and Conplace such changes in eff herein shall be construed	ed term of a discounted. Provisions governintion to this Agreement. Sified on an appropriation as part of Algonquin's compart that the Commission and year the Commission and the Commis	cable to service pursuant to this Agree ed rate or a Negotiated Rate to which Cung such discounted rate shall be as spect Provisions governing such Negotiated Estatement of Negotiated Rates filed, as Tariff. It is further agreed that Algorian or other appropriate body at any time as ther provisions in the applicable Rate Statement of Algonquin shall have the rate Natural Gas Act. Nothing container rights it may have under the Natural Gas or other proceedings by intervention or in part.	stomer ified Rate with onquin nd from chedule ight to d s Act,
	considered duly delivered facsimile. Customer or A	d when mailed to the ap Algonquin may change th	otices shall be in writing and shall be oplicable address below or transmitted vote addresses or other information below into the state of amending this Agreement:	
	Algonquin:			
	Customer:			

FERC Docket: RP08-268-000

Second Revised Sheet No. 872 Second Revised Sheet No. 872: Effective

Superseding: First Revised Sheet No. 872

FORM OF SERVICE AGREEMENT
(APPLICABLE TO TITLE TRANSFER TRACKING SERVICE UNDER RATE SCHEDULE TTT)

- 5. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 6. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN GAS TRANSMISSION, LLC
Ву:	By:
Title:	Title:

FERC Docket: RP08-268-000

First Revised Sheet No. 873 First Revised Sheet No. 873 : Effective Superseding: Original Sheet No. 873

Sheet No. 873 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 874 First Revised Sheet No. 874 : Effective Superseding: Original Sheet No. 874

Sheet No. 874 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 875 First Revised Sheet No. 875 : Effective Superseding: Original Sheet No. 875

Sheet No. 875 was previously issued, but is now reserved for future use.

FERC Docket: RP08-268-000

First Revised Sheet No. 876 First Revised Sheet No. 876 : Effective Superseding: Original Sheet No. 876

Sheet No. 876 was previously issued, but is now reserved for future use.

FERC Docket: RP04-361-000

Sheet Nos. 877 - 890 Sheet Nos. 877 - 890 : Effective

Sheet Nos. 877 - 890 are being reserved for future use

FERC Docket: RP10-308-000

Eighth Revised Sheet No. 891 Eighth Revised Sheet No. 891 Superseding: Seventh Revised Sheet No. 891

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AIT-2)

	Date:		Contract No	_
		SERVICE AC	GREEMENT	
Th	is AGREEMENT is entered into by	and between Alc	gonquin Gas Transmission, LLC,("Algonquin ").	ı") and
WHI	EREAS,			
	W THEREFORE, in consideration of e parties do agree as follows:	f the premises a	and of the mutual covenants herein contai	.ned,
1.	this Agreement and subject to A	Algonquin's Rate	ke and pay for service pursuant to the te e Schedule AIT-2 and the General Terms an ncorporated herein by reference and made	nd
2.	Maximum Daily Transportation Qu Maximum Annual Transportation Q	uantity Quantity	Dth Dth	
3.			the utilization of capacity on the laterality is defined in Rate Schedule AFT-CL:	ıl
	Brayton Point Lateral Manchester Street Lateral Canal Lateral Cape Cod Lateral Northeast Gateway Lateral J-2 Facility Kleen Energy Lateral	Check Only One)		
4.	including ("Primary to thereafter up written notice of at least Algonquin in the event Custome: service hereunder and such fai. Algonquin gives ten days prior further such termination shall Customer either pays such outs: other form of security reasonal of such outstanding bill; provi- pending the resolution of a dis procedure currently on file in to correct or cash-out imbalance	y Term") and shanless and until This Agrace Tails to pay your continues of the written notice not be effective tanding bill or bly acceptable to ided that Algonguin's Tances under this Aff shall survive	and shall continue for a term ending all continue to be effective from terminated by Algonquin or Customer upon ement may be terminated at any time by part or all of the amount of any bill for for thirty days after payment is due; provered to Customer of such termination and provered if, prior to the date of termination, furnishes a good and sufficient surety has a Algonquin guaranteeing payment to Algonquin shall not be entitled to terminate such customer complies with the billing disputatiff. Any portions of this Agreement needs agreement as required by the General Terminate of the other parts of this Agreement until	prior prior prided pond or prided prided
5.	except during the specified terms and Algonquin have agreed. Proin the Discount Confirmation to and term shall be as specified the consent of Customer, as parmay seek authorization from the time to time to change any rate and General Terms and Condition place such changes in effect in herein shall be construed to de	rm of a discount ovisions governs of this Agreement on an appropriate of Algonquing ecommission and contains of Algonquing accordance with eny Customer any pate fully in resorts.	licable to service pursuant to this Agree ted rate or a Negotiated Rate to which Cuing such discounted rate shall be as spect. Provisions governing such Negotiated ate Statement of Negotiated Rates filed, 's Tariff. It is further agreed that Algdor other appropriate body at any time a other provisions in the applicable Rate Stariff, and Algonquin shall have the rath the Natural Gas Act. Nothing contained rights it may have under the Natural Gas ate or other proceedings by intervention or in part.	astomer cified Rate with gonquin and from Schedule right to ed as Act,

FERC Docket: RP08-268-000

Fourth Revised Sheet No. 892 Fourth Revised Sheet No. 892 : Effective

Superseding: Third Revised Sheet No. 892

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE AIT-2)

6. Unless otherwise required in the Tariff, all notices shall be in writing and shall be considered duly delivered when mailed to the applicable address below or transmitted via facsimile. Customer or Algonquin may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

Algonquin:

Customer:

- 7. The interpretation and performance of this Agreement shall be in accordance with the laws of the Commonwealth of Massachusetts, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- 8. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable:

[None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

CUSTOMER:	ALGONQUIN GAS TRANSMISSION, LLC
зу:	By:
Fitle:	Title:

FERC Docket: RP08-268-000

Second Revised Sheet No. 893 Second Revised Sheet No. 893: Effective Superseding: First Revised Sheet No. 893

Sheet No. 893 was previously issued, but is now reserved for future use.

FERC Docket: RP04-361-000

Sheet Nos. 894 - 1000 Sheet Nos. 894 - 1000: Effective

Sheet Nos. 894 - 1000 are being reserved for future use